# CAN INSTITUTIONS ADAPT TO NEW MODELS OF ECONOMIC GOVERNANCE?

## Case studies in North East England and Tees Valley.

## **Peter Morris**

June 2020

Submitted for the degree of Doctor of Philosophy

Centre for Urban and Regional Development Studies Newcastle University

#### Abstract

This thesis draws lessons for economic governance from two case studies of a model of local development introduced by the UK government in 2010. Its temporal context is the decision of the coalition government to abolish the regionalism introduced by the Labour government in 1997/98 and replace it with a model of localism based on functional economic areas (FEAs). Its spatial context is North East England, where the two case studies have been conducted to illustrate how localism is working in practice as a governance system. The thesis studies the governance of devolution in the North East Combined Authority (NECA) and in Tees Valley as a political, economic and institutional issue, analysing the responses of local leaderships to the new model and its formal institutional arrangements. It finds that informal institutions in the shape of identity, socio-cultural values and political traditions and practices can exert a powerful influence on local leadership groups and hence the establishment and operations of new formal institutions. The thesis discusses the different approaches taken by the new formal institutions to development, and the resulting economic outcomes. It concludes that governments intending to devolve powers for economic purposes must take full account of regional and/or local circumstances, including the political. Trying to impose an economic governance model on an unwilling locality does not work, and a bottom-up approach should be adopted. Economically, it finds that at this stage, and with so many spatially blind factors affecting the regional economy, it is impossible to say whether the transition from regionalism to localism is having any economic effect, though early signs are largely disappointing. The thesis also examines the accountability and inclusiveness of the new governance and finds problematic issues in relation to both. It concludes with recommendations for institutional reform. In what is the quite recent and continuously evolving institutional landscape of localism in England, the thesis makes a significant addition to a still under-researched field.

#### Acknowledgements

I must have been a difficult student, so I must first of all acknowledge the patience of my supervisors, Professor Danny MacKinnon and Dr Alistair Clark, in putting up with my over-confidence and occasional grumpiness and keeping me on the academic straight and narrow. I also thank staff in the Faculty of Humanities and Social Sciences for the training they provided in research methods during the early stages of my study.

This work could not have been carried out at all without the co-operation of the 43 people who agreed to be interviewed. All are busy people, and I am grateful for their time, but also what I believe was their frankness, without which the value of the work would have been very much diminished. They must remain anonymous, but they know who they are.

Finally, I thank my family and friends, who had to put up with long periods when I was sitting in front of my computer or had my head buried in a book instead of participating in sociable activities. I also thank my great niece, Jemima Woodhouse, for using her digital skills to replicate the diagram of Hambleton's framework for civic leadership, which is reproduced in Chapter 2.

This has been a work of close personal interest to me. I spent my entire career observing my native North East England struggling in vain to lift its economic performance from the lower reaches of the UK league tables where it has languished for decades. I have worked in a variety of positions in all the region's conurbations and want to see them and their hinterlands prosper. I would like to be able to offer a solution to their economic predicament but I can't, and this thesis does not propose a panacea, though it does suggest ways of building a sense of common purpose, which is a necessary first step.

This thesis is independent and funded by the author.

## Table of Contents

Abstracti
Acknowledgementsii
Table of Contentsiii
List of figures viii
List of tablesviii
Acronymsix
Chapter 1. Introduction and methodology1
1.1 The aims of the thesis 1
1.2 The wider devolution context: globalisation, regionalisation, localisation
1.3 Key concepts
1.3.1 Institutions
1.3.2 Place and identity, agglomeration and dominance
1.3.3 Leadership and accountability14
1.3.4 Path dependence17
1.4 Methodology 18
1.4.1 Philosophical approach 21
1.4.2 The case studies 23
1.5 Methods of data collection and analysis26
1.5.1 Documentary evidence

1.5.2 The interviews 2	29
1.5.3 Use of statistics	33
1.5.4 Ethical and reflexive considerations 3	33
1.6 Conclusion 3	35
Chapter 2. Literature Review: economic theories and political governance	37
2.1 Theories of economic geography3	38
2.1.1 New economic geography 3	38
2.1.2 Institutional economic geography 4	45
2.1.3 Evolutionary economic geography 4	46
2.2 Governance, the state and devolution 4	49
2.2.1 The emerging role of governance5	50
2.2.2 The continuing role of government5	52
2.2.3 Economic governance and devolution in England	53
2.2.4 Leadership, accountability and inclusion in local government6	50
2.3 Conclusion6	57
Chapter 3. The context of North East England: identity, culture and politics6	59
3.1 North East England's framework for place-based leadership	70
3.1.1 Socio-cultural and political framing7	70
3.1.2 Economic framing: the economic consequences of social values	31
3.1.3 Government framing 8	39
3.1.4 Environmental framing9	96
3.2 North East England's administrative geography9	<del>)</del> 9
3.2.1 North East and North of Tyne9	<del>9</del> 9

3.2.2 Tees Valley	
3.3 Conclusion	

Chapter 4. NECA: a failure of institutional adaptation	

4.1 From deal to no deal in 11 months	107
4.2 The role of the business leadership	115
4.3 The role of public professionals	120
4.4 The Campaign for Real Devolution	122
4.5 Political leadership – councils and cabinets	127
4.5.1 Gateshead leads the opposition	129
4.5.2 Durham shrugs its shoulders	131
4.5.3 Sunderland and South Tyneside – following the leader	134
4 C Destearing	177

4.6 Postscript	. 137
4.7 Conclusion	. 138

### Chapter 5. Turning Tees Valley blue......140

5.1 Tees Valley Unlimited: a basis for collaboration	. 141
5.2 Tees Valley Combined Authority – normal politics	. 143
5.3 Political pragmatism: no mayor, no deal; no deal, no money	. 145
5.4 Devolution: business builds on the past	. 147
5.5 Serious economics meets popular politics	. 150
5.6 Raised profiles – geographical and mayoral	. 156
5.7 The return of regionalism: a latent threat	. 160
5.8 Conclusion	. 164

Chapter 6. Contrasting approaches to governing economic geography	165
6.1 The mismatch of political and economic geography	165
6.2 Diverging approaches to economic development	174
6.2.1 North East – politics and business: uneasy partners	174
6.2.2 Tees Valley – size isn't everything	178
6.3 Economic strategy: the background	182
6.4 The Strategic Economic Plans	186

6.4.1 North East – closing the gap	186
6.4.2 Tees Valley – the UK's 'greatest single development opportunity'	189
6.5 The economic outcomes – still struggling, still lagging	191
6.5.1 North East	192
6.5.2 Tees Valley	193
6.6 Conclusion	194
Chapter 7. Holding power to account	197
7.1 Accountability	197
7.1.1 Tees Valley	198
7.1.2 North East Local Enterprise Partnership	203
7.1.3 North East and North of Tyne Combined Authorities	207
7.2 Public consultation	211
7.2.1 North East Combined Authority	212
7.2.2 North of Tyne Combined Authority	215
7.2.3 Tees Valley Combined Authority	217
7.3 Inclusion	221
7.4 Conclusion	230
Chapter 8. Conclusions and recommendations	233
8.1 Answers to the research questions	234
8.1.1 Research question 1	235
8.1.2 Research question 2	236
8.1.3 Research question 3	238
8.2 The contribution of the thesis	240
8.2.1 Empirical contribution	240

8.2.2 Methodological contribution	245
8.2.3 Conceptual contributions	246
8.3 Recommendations	252
8.4 Further research	255
Appendix A: List of events attended in the course of research	256
Appendix A: List of events attended in the course of research	
	258

## List of figures

Figure 2-1: Hambleton's model of place-based civic leadership
Figure 3-1: Change in national household final consumption per head by selected COICOP commodities, England and North East Region, 2009-2016
Figure 3-2: Workforce jobs by industry (selection) (SIC 2007)
Figure 3-3: Identifiable expenditure on economic affairs, per head, indexed
Figure 3-4: England's LEP areas, with the North East featured
Figure 3-5: Tees Valley identifying its five boroughs
Figure 6-1: Unemployment (%) July 2015-June 2016 by local authority
Figure 6-2: Unemployment (%) July 2015-June 2016 by NUTS3 area 171
Figure 6-3: Unemployment (%) July 2015-June 2016 by travel-to-work area
List of tables
Table 1-1: Selected statistics for North East and Tees Valley LEP areas (2016). 25
Table 1-2: Deprivation rankings (out of 39) for North East England LEP areas (2015).Source: DCLG25
Table 1-3: Meetings and reports taken into account
Table 1-4: Interviewees 29
Table 3-1: Council membership by party. March 2017. 77
Table 3-2: Overall effectiveness of schools inspected by region. As at March 2018 (newmethodology). Source: Ofsted
Table 3-3: Identifiable expenditure on selected services per head, indexed (UK=100). Source: Public Expenditure Statistical Analyses (PESA)
Table 4-1: Trade union representation in Labour groups (January/February 2018). Source: Council registers of interests
Table 6-1: Gross value added (balanced) per head by local authority. 172
Table 6-2: Population and employment change by local authority 173
Table 7-1: Attendance at NELEP board meetings, September 2016-March 2019 206
Table 7-2: Influence over local decisions. 224

### Acronyms

ASC	Audit and standards committee
BME	Black and minority ethnic
BIS	Department for Business, Innovation and Skills
BEIS	Department for Business, Energy and Industrial Strategy
СА	Combined authority
CARD	Campaign for Real Devolution
CBI	Confederation of British Industry
CURDS	Centre for Urban and Regional Development Studies
DCLG	Department for Communities and Local Government
DTVA	Durham Tees Valley Airport
DWP	Department for Work and Pensions
EDRAB	Economic Development and Regeneration Advisory Board
EEF	The Manufacturers' Organisation
EEG	Evolutionary economic geography
ERDF	European Regional Development Fund
ESF	European Social Fund
EU	European Union
FDI	Foreign direct investment
FEA	Functional economic area
FSB	Federation of Small Businesses
GOR	Government office for the region
GDP	Gross domestic product
GVA	Gross value added
IAMP	International Advanced Manufacturing Park
IPPR	Institute of Public Policy Research
loD	Institute of Directors
LEP	Local enterprise partnership
LGF	Local Growth Fund
LibDem	Liberal Democrat
MCA	Mayoral combined authority
MDC	Mayoral development corporation
MHCLG	Ministry of Housing, Communities and Local Government

NAO	National Audit Office
NECA	North East Combined Authority
NEECC	North East England Chamber of Commerce
NEG	New economic geography
NELEP	North East Local Enterprise Partnership
NEP	North East Party
NEPC	Northern Economic Planning Council
Nomis	Official labour market statistics
NP	Northern Powerhouse
NRST	Northern Region Strategy Team
NTCA	North of Tyne Combined Authority
NUTS	Nomenclature of Territorial Units for Statistics
OECD	Organisation for Economic Cooperation and Development
ONE	One North East (regional development agency)
ONS	Office for National Statistics
0&S	Overview and scrutiny
OSC	Overview and scrutiny committee
PDO	Protected designation of origin
PESA	Public Expenditure Statistical Analyses
PSA	Political Studies Association
PUA	Primary urban area
R&D	Research and development
RDA	Regional development agency
RES	Regional economic strategy
RGF	Regional Growth Fund
SEP	Strategic economic plan
SME	Small-to-medium-sized enterprise
STDC	South Tees Development Corporation
TEC	Training and enterprise council
TfN	Transport for the North
TTWA	Travel-to-work area
TUC	Trades Union Congress
TVCA	Tees Valley Combined Authority
	×

- TVU Tees Valley Unlimited
- UE Urban economics
- UK United Kingdom
- VONNE Voluntary Organisations Network North East

## Chapter 1. Introduction and methodology

## 1.1 The aims of the thesis

[T]oday I am putting on the table and starting the conversation about serious devolution of powers and budgets for any city that wants to move to a new model of city government - and have an elected mayor (Osborne, 2014).

[T]he current process is not real devolution – real power has remained with central government, while other important areas of policy have actually become centralised and local government capacity has been cut severely (Raikes, 2020).

These two statements, six years apart, crystallise the very different views of English devolution as conceptualized in the mind of a Conservative Chancellor of the Exchequer and perceived in that of a Labour councillor in a northern city.<sup>1</sup> This thesis describes how devolution has actually turned out in two areas of North East England in comparison with these two views, and explains how and why the outcomes in these two areas differ so radically from each other.

Since Chancellor George Osborne's *Northern Powerhouse* speech in 2014, proposing to devolve powers and resources to cities in the north of England in a bid to boost the region's economy, the UK government has been struggling with varying degrees of commitment to find willing partners among the local authorities. While councils have sometimes reacted with initial enthusiasm, they have also sometimes failed to follow through and sign the devolution deals on offer (Osborne, 2014; Sandford, 2016a; MacKinnon, forthcoming: 18). One such case was that of the North East Combined Authority (NECA).

To assess if institutions can adapt to meet the prerequisites for local and regional development, the aim of the thesis is, firstly, to explore in depth the reasons why NECA failed to reach a devolution deal with the government while neighbouring Tees Valley

<sup>&</sup>lt;sup>1</sup> Raikes is a Labour member of Manchester City Council (https://www.linkedin.com/in/luke-raikes-a7666023/).

succeeded when the two areas have so much in common, including Labour Party control of all their local authorities at the time. It also investigates what difference devolution has made to the approaches of the two areas to economic development and to the resulting outcomes, and what the consequences of the new economic governance have been for accountability and inclusion.

Building on its findings, the thesis further aims to contribute to an understanding of how central government and local governance can co-operate most successfully to bring effective devolution to England, with particular reference to the building of local partnerships and the reform of local institutions. Drawing on what is learned, the thesis concludes with recommendations designed to improve local economic governance in England by making it more accountable, inclusive, transparent and legitimate, engaging the public, strengthening scrutiny, motivating political leaders and facilitating a sense of common purpose among all leadership groups in society (8.3).

In exploring these questions, the thesis aims to make an original contribution to an understanding of the operation in practice of the local, as opposed to the previous regional, form of economic governance introduced to England by the UK government in 2010. It offers new insights into the factors that motivate local politicians and other elites, factors that have their roots deep in the history and socio-cultural environment of their individual geographies. It provides further research into the private meetings of the party political groups of councillors where these motivating factors exert their influence. These insights are based on a study not just of documentary evidence but a wide range of in-depth interviews with high-level regional leaders in the fields of politics, public administration, business, trade unions and the voluntary sector that can rarely have been matched in the current era of localism in England.

There are three research questions:

- Why did the local authorities in Tees Valley accept, and in the North East reject, UK government offers of devolution deals in 2016?
- 2. What is the effect of the resulting formal institutional arrangements in these areas on their approaches to economic development?
- 3. To what extent can the system of local enterprise partnerships (LEPs) and combined authorities (CAs) be described as accountable and inclusive?

Devolution is a continuing process already dating back several years, whereas much study of institutional change has traditionally considered abrupt, transformational change. The research therefore aims additionally to make a valuable addition to the body of literature which suggests important changes often take place incrementally, and analyses how and why that happens (Martin, 2010; Mahoney and Thelen, 2012: xi)

This chapter first sets out the global and UK national background to devolution and debates over the appropriate geography; defines the spatial and temporal scope of the study; expands on the research questions; and describes key concepts. Secondly, it states the methodology, including the philosophical approach and analytic framework; justifies the selection of case studies; describes the methods used; reflects on the situation of the researcher; and sets out the structure of the remaining chapters.

#### 1.2 The wider devolution context: globalisation, regionalisation,

#### localisation

The globalisation of recent decades has been accompanied by a movement towards the decentralization of governance and economic development to the sub-national level as governments have devolved powers and responsibilities to regional and local authorities (Rhodes, 1996; Jones, 2001; Jones *et al.*, 2004; Rodríguez-Pose and Sandall, 2008). Rodríguez-Pose and Gill (2003) note that sub-national units have been increasing their demands for power, but warn that the devolutionary process raises important issues concerning equity and welfare, public finance and territorial competition. They advise a cautious approach taking account of both the benefits and drawbacks: political support may be low, affecting legitimacy; responsibilities and funding may not match; and the ability of the centre to redistribute may be reduced, resulting in a less progressive system of transfers. Prud'homme (1995: 18) similarly argues that 'any reduction in the importance of national budgets relative to subnational budgets – a definition of decentralization - will increase inter-jurisdictional disparities'.

Nevertheless, the advocates of decentralization ranging from a strong form of federalism to local, community-based governance are numerous. In the context of wide regional economic disparities in the UK, Martin *et al.* (2016) argue for radical changes to the national political economy to decentralize many of its functions and

structures. Greater Manchester, England's devolution pioneer, has won independent support for more powers (Coyle *et al.*, 2019; Bounds, 2019b), and the independent UK2070 Commission, still at work when this thesis was being written, was moving towards recommending 'systematic and comprehensive devolution', backed by a £250bn budget over ten years (UK2070, 2019b: 10). Lord Heseltine called for more devolution in a report in 2019 (Walker, 2019g)., and the Prime Minister promised the Convention of the North that he would 'do devolution properly' (Pickard *et al.*, 2019).

Meanwhile, in the context of global capitalism and ecological concerns, Hambleton (2015b) makes the case for place-based leadership, focusing on cities, while in the context of small and medium-sized towns, the case is made for development at a still smaller geographic level by Kelly (2016a), in research building on the World Towns Leadership Summit in 2016.

Devolution in England in the 21<sup>st</sup> century has followed two distinct paths, regionalism followed by localism, with a radical shift between the two resulting from a change of government in 2010. Both have been accompanied by constant tinkering with both geography and governance.

Regionalism was introduced when the Labour Government established regional development agencies (RDAs) in England's nine regions, including London, in 1998. Legislation the following year gave London a mayor and assembly, with the first elections in 2000. Meanwhile, the regions outside London set up regional assemblies, also known and regional chambers, which were unelected voluntary bodies but had significant responsibilities including the production of regional spatial strategies (Sandford, 2013).

By 2004 new geographies were taking their place in Labour's policies. The government launched the *Northern Way*, a pan-northern, multi-level initiative to promote growth across three regions, including eight city-regions. Two years later the arrival of cityregional thinking was marked in a Local Government White Paper (Rees and Lord, 2013: 680).

Also in 2004 a referendum in North East England rejected by 78%-22% a government proposal to bring a form of democracy into regional governance in the shape of an

elected regional assembly (BBC, 2004). This failure was followed by the introduction of alternative forms of elective oversight. In 2007 nine regional ministers were appointed to scrutinise policy in their regions; in 2008 the House of Commons established eight regional select committees to examine regional policy; and in 2009 the Commons set up regional grand committees, open to all regional MPs. In another innovation in 2009, regional assemblies were replaced by Local Authority Leaders' Boards (Sandford, 2013)

None of these institutions survived the general election of May 2010. The change from regionalism to localism following the formation of a coalition government of Conservatives and Liberal Democrats was rapid and comprehensive. The abolition of RDAs was announced on 29 June, and accomplished in March 2012, and the revocation of all regional spatial strategies on 6 July (Sandford, 2013). Local authorities were invited to submit proposals for business-led LEPs, based on functional economic areas, and the first 24, including Tees Valley, were approved by the end of the year; the remaining 15 followed in 2011 (Ward, 2019b).

City-regionalism as a concept and pan-northern ambitions were, however, carried over from the Labour era. In 2011 Greater Manchester formed England's first combined authority, to be followed by four more areas, including NECA, in 2014 and another five by 2018, including Tees Valley and North of Tyne. Meanwhile the pan-northern concept of the *Northern Way* was replicated in another form in George Osborne's hopes of a *Northern Powerhouse* with its suggestion of more powers for cities willing to accept elected mayors (so-called metro mayors) *(Osborne, 2014; Sandford, 2016b; LGA, undated-d).* In two further pan-northern moves, Transport for the North was established in 2018 as England's first sub-national transport body and NP11 was launched as a government advisory body consisting of the chairs of the 11 northern LEPs (TfN, 2018; Sandford, 2019c).

From 2012 the government offered bespoke 'city deals', with funding and powers, to councils, LEPs and other local bodies, with the largest cities in the first wave and wider areas following. In all 38 deals have so far been confirmed, including Sunderland/South Tyneside and Tees Valley. Then in March 2014 all LEPs submitted strategic economic plans as the basis for funding growth deals. The first devolution deal was signed by Greater Manchester in 2014, followed by 11 more by 2016, of which three collapsed,

including that with NECA, but two were partially revived, including with North of Tyne (Ward, 2017d; Ward, 2019a; Sandford, 2019c).

As part of the national industrial strategy, LEPs and mayoral combined authorities are now preparing local industrial strategies to run alongside their strategic economic plans (BEIS, 2017: 216). Still more governance changes are in prospect following the Queen's Speech of December 2019 which promised to 'give communities more control over how investment is spent so that they can decide what is best for them' (Queen, 2019b).

This study commenced in autumn 2016 just as North East England's two CAs were making their devolution decisions – Tees Valley (TVCA) to accept a deal and the North East (NECA) to reject. Two years later NECA formally divided, with three councils breaking away to form their own CA, North of Tyne (NTCA), and sign their own devolution deal while four, all south of the Tyne, remained in a rump NECA without a deal. The North East LEP (NELEP) continues to work with all seven.

England's devolution agenda has been widely criticised as inadequate and problematic (Pike *et al.*, 2015; Martin *et al.*, 2016; Tomaney, 2016; Berry and Giovannini, 2018). It is not the purpose of this thesis to add a further assessment of the adequacies of devolution as currently being implemented in England. Rather, the thesis argues that the new localism, as a form of democratic governance, can only work for economies if the local politics, influenced by the socio-cultural environment, is amenable. Selecting areas for governance on economic grounds alone – functional economic areas (FEAs) - without taking account of the politics, does not always work, though economics may, anyway, sometimes only be an ostensible reason for the selection of geographies for devolution. Government may also have an eye to political advantage which, as we will see in Chapter 5, was the case in Tees Valley. The fact that one party sees possible advantage for itself does not however mean that local politics as a whole will necessarily be amenable - though in Tees Valley it was.

Through its case studies, this thesis firstly recounts the responses of local leadership groups to offers of devolution deals. It explains how and why the selection of FEAs as localities for political governance and hence for devolution deals can either succeed, with reservations, or fail. The evidence of this thesis shows that the coalition

government had introduced a policy of localism without always understanding - or, if it did, without taking due account of - the politics of the localities where it was to be implemented. It worked as a governance scheme in Tees Valley because the top-down policy was sufficiently consistent with the bottom-up wishes of local political and business leaderships. But that was not so in the North East, and the government's plans were confounded.

The explanation for the different decisions of the two areas and the response to research question 1, the study therefore finds, lies in the presence or absence of willingness and ability on the part of local politics to co-operate with the formal institutional arrangements demanded by central government if it is to support local governance with enhanced powers and resources. This willingness and ability depend, in turn, on a complex of informal institutions deeply embedded in local geography and history.

Secondly, in response to research question 2, the thesis asks what difference these decisions made to approaches to development in the areas concerned. It finds a sharp divergence between a North East business-led technocracy and the highly political approach which developed in Tees Valley after its election of a mayor.

Devolution has brought extra resources and a raised profile to Tees Valley, as well as a focus on the mayor personally as the centre of power and publicity. The process has become highly politicised and personalised. North of Tyne (NTCA), following its breakaway from NECA, is starting to implement the same institutional model, if not necessarily to follow the same political path or style, two years behind. Meanwhile the rump NECA faces an uncertain future, and economic development remains a largely technocratic exercise in the hands of the business-led LEP.

What effect, if any, the different approaches are having on economic outcomes is a separate question. Early indications are mixed-to-disappointing so far (6.5)<sup>2</sup>. This question, however, does turn attention to debate over the related issues of the respective merits of place-based and spatially blind development. Belief in the efficacy

<sup>&</sup>lt;sup>2</sup> Throughout this thesis numbers in parentheses are internal cross-references.

of place-based development is a *sine qua non* of both regionalism and localism, but contested, and the theoretical arguments are examined in section 2.1.1.

In response to research question 3, the thesis examines the extent to which the formal institutional arrangements of devolution, the CAs and LEPs, are accountable and inclusive. It finds a complex of problematic issues in both areas. Metro mayors are directly elected but were arguably never wanted by local people and certainly not by many of their councillors. CAs are elected, but indirectly, and the private sector majorities on LEPs not at all. Inclusion is regarded in different ways by different groups of people. To those inside the governance system it means access for all to the labour market; for those outside it means having a voice in decisions, which they feel they lack.

#### 1.3 Key concepts

This chapter now develops in more detail the above outline of the formal institutional context for devolved economic development in England, and explains what is meant by informal institutions. It describes other key concepts that will be discussed throughout the thesis.

#### 1.3.1 Institutions

Institutions have been defined as 'humanly devised constraints that structure political, economic and social interaction'. They can be both formal, such as constitutions, laws and property rights, or informal, such as sanctions, taboos, customs, traditions and codes of conduct. They evolve, and 'in consequence history is largely a story of institutional evolution in which the historical performance of economies can only be understood as part of a sequential story' (North, 1991: 97). A notable feature of this definition is that institutions are referred to as constraints, but this disguises the fact that in constraining behaviour they can simultaneously be enablers by creating the conditions in which economies can grow. As North says, political and economic institutions are essential parts of an effective institutional matrix. They may create an economic environment that induces increasing productivity (1991: 98). The extent to which institutions act as constraining and/or enabling factors in North East England is an important part of this study.

The formal institutions examined here are what Hodgson (2006: 18) calls organisations, which he defines as special institutions that involve criteria to establish their boundaries and to distinguish their members from non-members, principles of sovereignty concerning who is in charge, and chains of command delineating responsibilities. In this study – to be specific - CAs, local authorities, LEPs, governments and government departments, political parties and groups, business and voluntary sector organisations are all formal institutions. They need not be legally established.

Addressing the question of informal institutions 22 years after North (1991), Rodriguez-Pose (2013: 1037-1038) pronounces them 'notoriously difficult' to define. He lists them as norms, traditions, social conventions, values, interpersonal contacts, relationships and informal networks. Informal institutions, he says, facilitate economic benefits such as generating trust, promoting entrepreneurship, greasing labour markets and adapting to shocks. In North East England, this study finds, informal institutions, socially inherited and deeply embedded in the history and geography of the region and its localities, help determine the nature of its formal institutions.

Based on these findings, the significant informal institutions relevant to North East England's devolution story are sense of local identity, which trumps regional identity and gives rise to local rivalries and mistrust; political traditions and voting habits which have historically ensured power lies most of the time and in most of the region in the hands of a single party, Labour; and the widespread practice, not confined to North East England, of making important decisions in the privacy of party groups.

These informal institutions are common to the whole region, but there are differences between Tees Valley and the NECA area which moderate their effects in the former. Most significantly, rivalry has in the past been directed by Tees Valley's leaders outwards against their northern neighbours rather than inwards against each other, thus fostering internal trust and collaboration. This internal harmony may not survive the electoral upheaval which occurred in 2019 when, amid Brexit turmoil nationally, Labour lost overall majorities on all five Tees Valley councils and three parliamentary seats (3.1.1). But it had lasted long enough to bring devolution and a metro mayor.

The operations of these informal institutions are revealed through the case studies as they examine how politicians, their professional advisers, the business community,

trade unions, the voluntary sector and the general public responded to the governance structures, proposed and actualized, resulting from the transition from regionalism to localism and the government's offer of devolution deals.

The regional structure of formal institutions of economic governance inherited and immediately dismantled by the coalition had been established by the Labour government in 1998, as described above (1.2). The abolition of North East England's RDA, One North East, was more than an economic blow to the region, eradicating its principal driver of public development policy. It also meant the loss of the one organisation that held the region's 12 local authorities together in unstable cooperation. In a brief preview of the picture that will emerge during this thesis, a regional business leader said:

It's almost Balkanization. One thing One North East did was it totally united all 12 local authorities, but they were fighting against this other power that seemed to have all the money. As soon as that other power vanished, like Tito disappearing in Yugoslavia, they all fell at each other's throats (Business leader. August 2018).

The coalition's reasons for abolishing regions were spelled out in 2010. They were, it said, arbitrary and not efficient, effective or popular, and an artificial representation of functional economies. The regional approach missed the opportunities that came from local economic development activity and largely ignored the knowledge and expertise of the private sector, local authorities and local communities. The LEPs on which the new governance regime is based would, the government believed, give a clear vision and strategic leadership for sustainable private sector-led growth and job creation. The aim was to move power from Whitehall to localities, create partnerships between business communities and local authority leaders, focus on growing the private sector and increase local accountability, including, but not necessarily, by the creation of directly elected mayors. The government's reasons for abolishing regionalism were explained more simply by Elcock (2016: 11): 'The Conservative Party had generally regarded all forms of regional government as a waste of money' (BIS, 2010: 3, 7, 13; DCLG, 2010).

The coalition partners, Conservatives and Liberal Democrats, were not wholly united on this issue. A regional business leader, two national politicians of different parties and a political activist all told this study, independently, that the momentum to abolish RDAs came from the Tories, who were ideologically opposed to regionalism. LibDems on the other hand, according to two of these interviewees, argued unsuccessfully for at least the retention of One North East even if RDAs in other regions were abolished (Interviews November 2017, December 2017, January 2018, August 2018).

A first wave of 24 LEPs was approved in 2010 and the remainder the following year. As part of this process the North East region was divided between Tees Valley, part of the first wave, and the remainder of the region, which confusingly retained the name North East<sup>3</sup>. All LEPs submitted SEPs in 2014 as a basis for funding growth deals (Ward, 2019a). LEPs and CAs continued to operate under the Conservative governments of 2015, 2017 and 2019.

This new localism was said to recognise that places have specific geographic, historic, environmental and economic circumstances. Shifting power to the 'right' levels would increase democratic accountability and transparency, including through directly elected mayors (BIS, 2010: 3, 11). The local circumstances that were thus so important to the new governance regime, it is the argument of this thesis, were favourable to the devolution agenda in Tees Valley but not in the North East, where they were not 'right' but proved insuperably divisive when required to adapt to the new governance. It is an irony of the devolution process that a government which thus recognised the importance of local circumstances failed to understand, or perhaps simply ignored, what those circumstances actually were in the North East.

In June 2014, when Osborne set out his vision of a *Northern Powerhouse* which, using improved transport links, would create an agglomeration in the north of England to match the economic might of London, he made clear new formal institutions would be required. Osborne envisioned powerful city governments with enhanced powers and budgets devolved from the centre but, crucially, on condition that they accepted

<sup>&</sup>lt;sup>3</sup> Throughout this thesis 'North East England' refers to the region encompassing both the North East and Tees Valley LEP areas; 'North East' refers to the area of the North East LEP (NELEP). Further geographic detail, including maps, is in section 3.2.

elected mayors (Osborne, 2014). This led to the establishment later that year of four CAs, including NECA – but not yet with mayors - following the lead of Greater Manchester in 2011. TVCA was established in 2016.

CAs are combinations of two or more local authorities that may take on any functions their constituent authorities agree to share and statutory functions transferred to them by the government. The next step in the process saw the government negotiate devolution deals with a number of CAs to devolve additional powers, supported by annual grants for 30 years (subject to five-year reviews) and on condition of the election of a metro mayor (Sandford, 2016a; Sandford, 2016b).

#### 1.3.2 Place and identity, agglomeration and dominance

A place, according to Hambleton (2015a: 82-84) is somewhere somebody cares about. In the past, in his view, city politicians have often failed to appreciate the importance of the feelings people have for their home area and the social significance of neighbourhood life, based on personal experiences such as community celebrations and street festivals. Successful leaders work with the grain of these feelings, as Hambleton (2015a: xiv) illustrates through 17 case studies from around the world.

In England, councillors are elected to serve their municipality, which is what they do on a daily basis and what they are held accountable for by voters. These are their communities. Yet around once a month leaders attend CA or LEP meetings at which they are expected to put wider interests first and, it is argued here, they are not always able to do so. In North East England there are 12 municipalities but only two (now three) CAs and two LEPs which in theory correspond to FEAs. But FEA boundaries can be drawn and changed arbitrarily when politics demands, as demonstrated in the North East when one FEA became two after its politicians failed to agree on devolution. These two or three FEAs are themselves not the only economically possible geographies, as discussed in section 6.1.

Persuading councillors and voters to accept new political entities for the purpose of democratic local government which also coincide with economic areas suitable for place-based development can be difficult anywhere. Plans for CAs fell through in Great Lincolnshire and East Anglia, for example. In Yorkshire, the Sheffield mayor said he would not touch devolution 'with a barge pole' if it meant a Leeds takeover (Torr,

2018). Even in areas which have accepted devolution deals there have been difficulties associated with identity. A study of the Liverpool and West Midlands CAs found that local identity is a challenging concept for the public to pinpoint, and when defined tends to relate to a citizen's immediate locality rather than a wider sense of regional identity (PwC, 2016: full report, pp. 3, 11-13). The public's lack of propensity to identify with a wider region is matched by the inability of their leaderships to collaborate. According to Giovannini (2015), one of the most striking aspects in the current debate on devolution in the north of England is that the main actors seem to work in isolation, and a climate of mutual suspicion hinders decentralization from within.

It has been local authority elected representatives worldwide who have most strongly expressed reluctance to support mergers, according to the OECD, but local residents have also often declined to do so, including in the United States 'because they feared losing a certain quality of life which they ascribed to local identity'. Municipal mergers to form metropolitan governments have been rejected in Amsterdam and Rotterdam, and a merger that took place in Montreal was soon reversed. In fact: 'When the merger of municipalities is put to the popular vote it has to be said that the result is almost always a refusal' (OECD, 2006: 201). Metropolitan government was established in Auckland, New Zealand, in 2010 by amalgamating eight existing councils, but this was achieved by an act of parliament, not by a popular vote (Hambleton, 2015a: 189-192). McCann (2016: 492) reports that of 263 OECD metropolitan areas of over 500,000 population, 31% have no metropolitan governance body at all and 50% have a body without any regulatory powers.

In North East England a 2004 referendum vote to reject a regional assembly (1.2) ended the Labour government's attempt to introduce democratic accountability into the regional dimension on which it had based its economic development apparatus in the form of One North East, just as NECA's 2016 decision against devolution ended the coalition and Conservative governments' plans for mayoral democracy. While neither the regional assembly nor the devolution deal would have involved a merger of local authorities of the form described by the OECD, they did involve new arrangements for sharing political power over areas selected for economic reasons, and both were rejected.

The attachment that voters and their elected representatives feel to their local area may be further strengthened if they feel that a neighbouring municipality, typically a regional capital, is receiving preferential treatment due to its dominance. The role of cities or city-regions has been emphasized in England because they have often been the locations of CAs and devolution deals. The metro mayors of Manchester and Birmingham have become national figures. But the claim that cities act as growth hubs, creating jobs that benefit their hinterlands, is a controversial one which placed the role of Newcastle, North East England's regional capital, at the forefront of the devolution debate.

When the OECD reported on North East England it referred to the 'Newcastle city region' – in which it included all five Tyne & Wear metropolitan districts - as the growth centre (OECD, 2006: 40, 172). One North East also recognised the importance of urban cores as drivers of economic growth and identified the region as having two, centred on Tyne & Wear and Tees Valley (ONE, 2006: 14-16). These differences of view may have had a political rather than an economic rationale: One North East had to bear in mind the sensitivities of local politics; the OECD did not.

As will become clear in this study, Newcastle's perceived dominance is resented by some in the region, and the idea that it receives more than its fair share of development in a process of regional-level agglomeration is deeply felt in neighbouring municipalities. Some feel that development in Newcastle is at their expense. And it is not only the resources perceived to be going to Newcastle that is the problem, but the city's (represented by the city council's) assumption that they should, which is seen as arrogance: 'There is an acceptance that Newcastle is the regional capital; I don't think there is any question about that. But it's the way it behaves as the regional capital that is the problem' (Retired public official, April 2018). This suspicion was exploited during the North East devolution debate, as will be seen in the case study chapters.

#### 1.3.3 Leadership and accountability

The government's insistence on elected mayors to be accountable for the extra powers and resources provided as part of devolution deals has been one of the most contentious aspects of the process and was possibly a deal breaker in the North East. Leadership in English local government may be exercised in a variety of styles

depending on the formal system in place and the personalities of those involved, but councillors have traditionally favoured collective rather than individual leadership (2.2.4). Since local government reform in 2000 they have stuck as closely as possible to the collective model by having most council leaders elected by their fellow councillors, which in practice means by their party colleagues unless there is no overall political control. Only 16 councils in England, including one in the North East and one in Tees Valley, are led by directly-elected borough mayors, who are elected for four years (Sandford, 2017a). All are subject to political pressure within their party groups: one North East council leader who disagreed with his party group over devolution resigned (Milligan and O'Donoghue, 2016) and another is believed by some observers interviewed for this research to have voted against devolution because of group pressure (4.5.1, 4.6).

Council leaders and members of their cabinets, which fulfil the executive role, are thus still subject to party discipline. Some, however, are able to establish sufficient dominance over compliant party groups to be able to exercise considerable freedom over important decisions, as was the case over NECA's devolution deal (4.5.3). As to accountability to voters in council elections, it is dampened by voter loyalty and the link between local and national politics which often sees near-uniform swings across the country. The effects of this trend in North East England are discussed in section 3.1.1.

The election of metro mayors to lead CAs has raised the question of what style of leadership they should exercise. According to Hambleton (2015a: 11), what is required is a re-think of the role of the local leader, summed up in what he calls a caricature, but a helpful one, in the words: 'Out goes the notion of the "city boss"...in comes the "facilitative leader" orchestrating the efforts of multiple actors', not just collective but inclusive, creating a process of citizen empowerment. The extent to which this has happened in Tees Valley, where a metro mayor was elected in 2017, is examined in sections 5.5, 5.6 and 6.2.2.

Leadership is exercised by other groups in society too, alongside politicians, including through LEPs. According to Hambleton (2015b: 14, 17), local governance needs to respect and reflect that dispersal of leadership if decisions taken in the public interest

are to enjoy legitimacy. He looks for leaders who are empathetic, collaborative, imaginative and willing to take risks. The relative power of leadership groups varies by locality, as is examined in the case studies. With its discussion of place leadership at the sub-national level, the thesis thereby makes a contribution to what Horlings *et al.* (2018: 247) describe as the 'missing link' in our understanding of place-based development and local economies (Horlings *et al.*, 2018: 247). According to Quinn (2013: 749) leadership had been relatively under-researched in regional studies until recently and needed to be explored in greater detail. It is a subject of which the OECD (2015a: 57) said:

Leadership of local economies is a largely unexplored topic in academic literature even though leadership is widely referred to by employers and investors as one of the reasons they seek new opportunities in certain locations...[T]he critical work of local leaders in their local economies is worthy of further investigation... Avoiding leadership failure is a critical task in an arena of public policy where there are few clear scientific methods.

The establishment of CAs and LEPs raises new questions over accountability in local governance. In English local authorities, all councillors and the small number of borough mayors are directly elected by voters. Responsibilities are then shared between the executive, in the form of the leader (elected by the council) or mayor and cabinet, and backbench councillors, whose role is to hold the executive to account through their overview and scrutiny committees (OSCs) and to approve the policy framework and perform the key task of agreeing the annual budget (Hambleton, 2015a: 181-188). The scrutiny system can be problematic, as discussed in section 2.2.4, and the potential difficulties are compounded in the CAs, which have no councils of their own but are scrutinised by backbenchers of their constituent councils, who may feel somewhat remote from the work of the CA. These same backbenchers also scrutinise the LEPs, from which they may feel even more distanced.

There is a precedent for scrutiny and accountability in somewhat similar circumstances which, while not exact, is close enough to offer useful lessons. Under the Labour government's regional development system, scrutiny of RDAs was performed by regional chambers, made up of councillors and other civic leaders. The lessons of that

experience in North East England were studied at the time and are discussed in section 2.2.4. How accountability under the current governance of CAs and LEPs is working in the region is described in section 7.1.

#### 1.3.4 Path dependence

Path dependence is the idea that 'the current state of affairs cannot be derived from current conditions only, since the current state of affairs has emerged from and been constrained by previous states of affairs. Evolutionary geography deals with path dependent processes in which previous events affect the probability of future events to occur' (Boschma and Frenken, 2006: 280-281). Economically, path dependence is a consequence of the clustering of industries, with their sunk capital and skilled workforce. Once a region is on a particular path as the location of particular industries with their technologies and skills, it may become locked in.

The key defining characteristic of path dependence is an inability to shake free of history and create new paths, and this problem can be wide in scope, referring equally to material development and social dynamics, including the political, for path dependence is a concept that has come to be applied equally to institutions, which can also get locked in. However, both industries and institutions can avoid this situation by adapting in ways described in section 2.1.3 (Martin and Sunley, 2006: 399; Martin, 2010; Mahoney and Thelen, 2012).

The informal institutions of North East England form a socio-cultural environment which affects the ability of councillors to adapt to a new political situation and adopt new governance arrangements to facilitate devolution and its development opportunities. The evidence emerging from this study is that in the political sphere informal institutions are an important, sometimes decisive factor, facilitating change in Tees Valley but supporting the *status quo* in the North East.

In the economic sphere, informal institutional pressures might in similar fashion have affected the willingness of councillors to support business in taking new paths, but the evidence of this study is that this is not the case. Even in the North East, where councillors were resistant to institutional change, there is no evidence they opposed economic path creation.

As a purely economic factor the power of path dependence is gradually weakening as old industries are adapted to meet new demands, but as a political factor it remains important in the North East. This study uncovers powerful political and socio-cultural influences which, it argues, played an important role in the events examined, bringing deeply rooted pressures to bear on actors.

The extent to which the industries of North East England have avoided path dependence is discussed in Chapter 6 and the success of the region's governance institutions in shaking free in sections 4.7 and 5.8.

#### 1.4 Methodology

This thesis is structured in such a way that the concepts to be discussed, set out in the literature review (Chapter 2), lead naturally into the North East England context in which they are played out, but without reference to specific institutions or events (Chapter 3). This regional context leads in turn into the empirical findings of the case studies, which are specific to institutions and events (Chapter 4 and Chapter 5) and to the further empirical findings in relation to economic issues (Chapter 6) and accountability and inclusion (Chapter 7). This ordering allows the conceptual and contextual background to the research questions to be explored as preparation for the search for empirical answers, and it is in order not to break this progression from the conceptual to the contextual and thence to the empirical that the methodology is placed here rather than after the literature review, which is perhaps more usual.

The methodology is a comparative analysis of case studies designed to throw light on the responses of local actors to the institutional regime of localism introduced in England since 2010 and the extent to which they affect local approaches to economic development. A further objective is to assess the extent to which this regime is locally accountable and inclusive.

The mixed methods used are to a large extent qualitative, based on the researcher's perception and interpretation of official documents, interviews with key actors, and attendance at 29 relevant conferences and meetings (Appendix A). The devolution processes which are the focus of the study were public events, reported and commented on in mainstream media, and these media were in turn monitored and

interpreted by the researcher. Social media used by some actors was monitored, and contributed to the research findings. These qualitative methods were supported by quantitative methods using statistics relating to North East England's economic performance, and to public and private finances; and by electoral statistics to aid an understanding of the region's political situation.

Devolution forces its way to the forefront because of its prominence as a public controversy during the relevant period, because it forms a point around which issues of interest are crystallised, and because it provides a salient symbol of difference between the case study areas, which otherwise share many apparent similarities. Devolution helps throw light on both formal and informal institutions through the intensity of the regional debate around it. A specific and crucial objective is to determine what role informal institutions played in the decisions of five local authorities in Tees Valley, all controlled by the Labour Party at the time, to accept a UK government offer of devolution in 2016 and of seven local authorities in the North East, also all then controlled by Labour, simultaneously to reject a deal.

As a first step to answering the research questions, the results of the mixed methods used were combined to build a picture of the regional context against which the events concerned unfolded and thus to help explain the decisions and actions being studied. This contextual step was an essential prerequisite to answering all three questions. Within this overall methodology each of the three research questions required somewhat different approaches.

The first question, asking why the local authorities in Tees Valley and the North East took different decisions on devolution in 2016, is a question of the motivations of the 12 individuals who alone had the final deciding vote on those decisions - the seven council leaders constituting the North East Combined Authority (NECA) and their five Tees Valley counterparts. The two primary methods for ascertaining these motives were the official records of the CAs (agendas, reports and minutes) and interviews with the individuals, their colleagues and counterparts among other stakeholder groups. These primary sources were supported by data enabling an understanding to be formed of the political and socio-cultural background which exerts a powerful, if not always recognised, sway over regional governance. Explaining this background is the

main purpose of the context chapter (Chapter 3), and its consequences are seen in the case studies (Chapter 4 and Chapter 5), where they are presented in narrative form as the civic leadership groups identified in the analytic framework for the thesis (2.2.4) play out their roles in the devolution drama.

The answer to the second research question, addressing the effect of the resulting formal institutional arrangements in the case study areas on their approaches to economic development, is again addressed primarily through the official documents of the LEPs and CAs, in this case including their strategic economic plans, and the interviews. The context, particularly the socio-cultural and political background, is an important secondary factor. It suggests that the closer political balance in Tees Valley than in the NECA area, and the higher Tory hopes, encouraged a more high-profile, politically motivated approach. The Conservatives were encouraged to build proactively on their success.

The third research question, concerning the accountability and inclusiveness of the LEP/CA governance system, required a considerable methodological mix. The effective accountability of local government in general in the region to their voters is examined statistically in terms of entrenched voting patterns and low election turnouts, forming part of the North East context (Chapter 3). It is against this background that the discussion concerning the indirectly-elected nature of CAs and the unelected status of the private sector majority on LEPs (7.1) is to be seen; low voter engagement and lack of direct representation reinforce each other.

Accountability to the general public through the consultations carried out on devolution is considered qualitatively on the basis of the official reports of those consultations and of the researcher's attendance at three consultation meetings.

Internal accountability of LEPs is addressed through their official documents, supported by interviews, which examine the extent to which council leaders fulfil their role as the democratically, if indirectly, elected members of LEP boards and the nature of their engagement; this is to a significant extent a quantitative as well as qualitative investigation dealing with the frequency of inquorate meetings. A similar procedure is followed to examine the work of the overview and scrutiny committees (OSCs) of backbench councillors tasked with examining the work of their leaders and cabinets, and the LEPs; this is again partly quantitative research, dealing with numbers of inquorate meetings, and partly qualitative, discussing the nature of scrutiny.

Inclusiveness is considered in its economic and political aspects. Economic inclusiveness is examined qualitatively in terms of the work being done to enable marginalised groups to access the labour market, and quantitatively to report the success of these efforts as measured in official statistics. Political inclusiveness is treated as a LEP issue, as LEPs have considerable control over their own membership whereas CAs consist of the leaders chosen by their constituent local authorities and, indirectly, the voting public. LEP membership is examined quantitatively in terms of gender, ethnicity and the representation of the five civic leadership groups identified in the analytic framework (2.2.4) and different sectors of the economy. These primary data are supplemented qualitatively by interviews with representatives of those outside the LEP/CA system, such as the voluntary sector and those working in business development, to gauge their perceptions of the system's inclusiveness.

#### 1.4.1 Philosophical approach

This section outlines the philosophical approach of the thesis, which is critical realism supported by Aristotelian practical rationality, and justifies the accompanying narrative analytic method; discusses the comparative case study method and justifies the choice of the North East and Tees Valley as the two cases; and considers the question of the validity of studying a small number of cases.

The critical realist philosophical approach incorporates an ontology which claims that the world exists independently of our knowledge of it, and that social phenomena, such as the informal institutions of this study, by and large exist regardless of researchers' interpretations of them. Our knowledge of these phenomena is fallible, and epistemologically relativist, but not immune to empirical check, and not judgementally relativist (Sayer, 1992: 5-6; Jessop, 2005: 42-43).

Critical realism sees reality as stratified. At one level is the actual world of events as they occur (meetings take place, decisions are recorded etc.). At another is the empirical world of those events as they are experienced and interpreted by participants and observers. At the third level is a real world where are to be found the

causal mechanisms that underlie events. In this study, for example, exploration of the 'real' may lead to the informal institutions, political traditions and practices that cause local authority members to make the actual decisions they do, influenced by their own empirical perceptions and in turn empirically perceived and interpreted by observers.

In the world of local governance, actual events often take place in public while the real is confined to the private meetings of decision makers – what Ayres *et al.* (2017b: 861) call 'front' and 'back' stage. Front stage governance describes 'activities of visible and accountable office holders in elected bodies, constrained by bureaucratic rules', while back stage governance is 'the world of unseen decision-making where public officials are less constrained by formal rules and public scrutiny'. Critical realism requires the researcher to explore both these worlds.

This approach, applied to this thesis, enables the multiple factors leading to the different decisions in the North East and Tees Valley in relation to devolution to be explored in terms of the complex relationships between them and the ways they created the conditions, or causal mechanisms, which explain what happened. It calls for what Archer *et al.* (2004: cited in Easton (2010): 124) call 'judgemental rationality', meaning that the researcher's case can be publicly defended, with reasoned judgements about what reality is objectively like. Therefore, though critical realism leads to an interpretation of events, it is more than just one possible interpretation among many with no criteria for choosing between them; it is supported by defensible reasons established through research.

These strata of reality, the relationships that form and the causal mechanisms that are forged, lend themselves most readily to narrative analysis. While the North East England context is analysed thematically in Chapter 3, looking in turn at the four sides of the analytic framework (2.2.4), the events leading NECA and Tees Valley to their diverging decisions on devolution are subjected to a narrative analysis. It is the developing relationships between groups of actors, played out over time, that explain the devolution decisions, and the account of this drama is most clearly laid out in the narrative form utilised in Chapter 4 and Chapter 5.

These narratives recount events as they actually occurred, mainly on the basis of official documents. The empirical interpretations of these events by key actors are

ascertained during the course of the interviews. Documents and interviews, supplemented by the researcher's own empirical perceptions of the events he attended, are then triangulated to construct the narratives, which present the events at the critical realist level of the real. The rationale for the narrative method is summarised by Sayer (1992: 259):

By narrative I mean an account of some process or development in terms of a story, in which a series of events are depicted chronologically...Its power derives from the way in which putting things in chronological order, in a story, gives the appearance of a causal chain or logic in which each event leads to the conclusion.

Alongside its critical realist approach, this thesis is a work of what Flyvbjerg (2001) calls *'phronetic* social science' after the type of knowledge known to Aristotle as *phronesis*, or practical rationality. *Phronesis* is 'a rational faculty exercised for the attainment of truth in things that are humanly good and bad' (Aristotle, c350BC (1953)) and is akin to the judgemental rationality referred to above. It is, Flyvbjerg (2006: 221) argues, practical, concrete, context-dependent and non-predictive; it deals with real human situations such as those described in the case studies.

Aristotle's reference to 'things that are humanly good and bad' is a clear indication that *phronesis* takes account of values as well as epistemic, or scientific, facts. *Phronetic* judgements are ethical. As Flyvbjerg (2001: 57) says, *phronesis* is 'practical value-rationality'.

#### 1.4.2 The case studies

In comparative case studies, the cases are *what* is studied; comparison is *how* they are studied (Lijphart, 1971: 682). Case studies, according to Yin (2003: 1), are the preferred strategy when 'how' or 'why' questions are being posed, when the investigator has little control over events, and when the focus is on a contemporary phenomenon with some real-life context. They are thus appropriate here.

The suitability of the North East and Tees Valley as the choices for comparative case studies is enhanced by the fact that they share many characteristics, economically, socially and politically, which can be used as controls, allowing key differences to be identified. The two LEP areas fall into Flyvbjerg's category of cases with a significant variation which, in this study, is the acceptance in one case and rejection in the other of a devolution deal (Flyvbjerg, 2006: 230). Their shared economic situation, social conditions and politics offer the opportunity to observe and explain their divergence over devolution by understanding causal factors with their roots in informal institutions that are too complex for other research strategies (Yin, 2003: 15). **Table 1-1** shows how the North East and Tees Valley share some significant economic differences from the national situation.

**Table 1-2** shows how they share relatively high levels of multiple deprivation in some, though not all, domains. Of the 326 local authorities in England, Middlesbrough is the one with the largest proportion of highly deprived neighbourhoods, 48.8%, and Hartlepool follows in tenth place with 32.8%. Though North East England has no other local authority among the 20 most deprived, South Tyneside ranks as having the greatest increase in its proportion of deprived neighbourhoods between 2010 and 2015 (DCLG, 2015b: 10-12). Overall, Tees Valley is the second most deprived of the LEP areas and the North East is tenth (2015b: 19).

The extent to which the two LEP areas share a common party political balance is tabulated in section 3.1.1.

An advantage of the comparative case study, particularly at the sub-national level, is that it enables the researcher to dig deep into the detail. Snyder notes that focusing comparative method on sub-national units helps deal with the issue of having a small number of cases by making it possible to increase the number of observations. In this study, for example, it is possible to get down to a level of detail that might be impractical at national level or over a large number of cases in relation, for example, to economic strategies, professional advice, political decision-making and the motivations for it, political campaigning, and the opinions of business, the voluntary sector and the public. Snyder argues that sub-national units within one country can be matched on cultural, historical and socio-economic dimensions, making it easier to identify a small number of significant dissimilarities (Snyder, 2001: 95-96).
	North East	<b>Tees Valley</b>	Great Britain
Economically inactive	23.8%	26.3%	22.2%
Unemployed	6.7%	6.4%	4.8%
Workless households	19.7%	21.3%	15.1%
Top classification occupations <sup>4</sup>	38.7%	38.6%	45.4%
NVQ 4 and above	31.6%	30.8%	38.2%
Gross weekly pay – all full-time workers	£491.80	£495.70	£540.90
Jobs density <sup>5</sup>	0.71	0.69	0.85
Gross Value Added per head (indexed) <sup>6</sup>	73.1	72.6	100

 Table 1-1: Selected statistics for North East and Tees Valley LEP areas (2016).

 Source: Nomis

Table 1-2: Deprivation rankings (out of 39) for North East England LEP areas (2015). Source: DCLG

Nature of deprivation (domain)	North East	Tees Valley
Overall relative measure of deprivation	10 <sup>th</sup>	2 <sup>nd</sup>
Income	6 <sup>th</sup>	21 <sup>st</sup>
Employment	3 <sup>rd</sup>	2 <sup>nd</sup>
Education, training and skills	18 <sup>th</sup>	2 <sup>nd</sup>
Health and disability	4 <sup>th</sup>	2 <sup>nd</sup>
Crime	29 <sup>th</sup>	5 <sup>th</sup>
Barriers to housing and services	29 <sup>th</sup>	37 <sup>th</sup>
Living environment	33 <sup>rd</sup>	37 <sup>th</sup>

Though the two cases in this study were chosen largely on grounds of interest to the researcher, and convenience, any disadvantages of this method of selection are outweighed by important advantages. As Etikan (2016: 4) point out, the main disadvantage of convenience sampling is that participation is restricted to certain members of the target population and the results are therefore not necessarily generalizable. However, not only does this study enjoy the offsetting advantages already discussed of shared characteristics coupled with a significant variation, opportunity for context-dependent learning, and sub-national susceptibility to indepth study yielding richness of information; it also lends itself to the counsel of

<sup>&</sup>lt;sup>4</sup> Standard Occupational Classification 2010, Major Groups 1-3: managers, directors, senior officials, professionals, associate professionals, technical.

<sup>&</sup>lt;sup>5</sup> Ratio of total jobs to population aged 16-64.

<sup>&</sup>lt;sup>6</sup> 2016 figures are provisional.

Flyvbjerg (2006: 229) that it is often more important to clarify the deeper causes behind a given problem and its consequences than to describe the symptoms and how frequently they occur, as would be the aim of a random sample. 'Random samples emphasising representativeness will seldom be able to produce this kind of insight; it is more appropriate to select some few cases chosen for their validity'.

Lidstrom (1999: 98) argues that in local government studies it is the broad overviews that are problematic. Losing in depth what they gain in breadth, overviews can be superficial and unable to provide genuine understanding of the reality behind the formal structures; for example, how power is distributed and decisions are reached. Second, they tend to be mainly descriptive and only in exceptional cases do they explain why differences and similarities occur. Superficiality and lack of explanation are problems which this research aims to overcome, utilising what Stake (2000: 439) calls 'thick description'.

# 1.5 Methods of data collection and analysis

This section explains the temporal boundaries of the thesis; enumerates the official documents examined; and details the interviews conducted and the issues arising from them of credibility and interpretation, power relations and confidentiality. It describes the use made of statistics; and deals with ethical considerations. The researcher reflects on his own background and how it relates to the research.

The period covered by the case study research is the five years 2014-19. The 2014 starting point is chosen because both LEPs published the first versions of their SEPs that March, stating the economic targets they aimed to achieve in the coming decade. This was also the time that NECA, one of the main institutions studied, was established; its first meeting was in April 2014. TVCA followed two years later. The end of the study is March 2019 because that is the half-way point of the SEPs and an opportune moment to assess progress. It was also the (then) scheduled date of Brexit and thus of the anticipated start of a new economic era for the UK.

The research itself started in 2016 and, although systematic research stopped in March 2019, later significant relevant events occurring while the thesis was in preparation are noted. These include the 2019 council elections, scrutiny issues arising in Tees Valley,

the publication of Lord Heseltine's 2019 report on devolution (Heseltine, 2019), the first report of the UK2070 Commission on Rebalancing the UK Economy (UK2070, 2019b), the EU Commission for Economic Policy's report on place-based *Industrial Strategy* (Commission for Economic Policy, 2019), the Convention of the North in September 2019, a CBI report on devolution (CBI, 2019) and, at a very late stage, the 2019 general election.

#### 1.5.1 Documentary evidence

Answers to the research questions required first of all a detailed background knowledge of the North East England economy. The first research step was therefore to study economic reports and plans produced for and in the region in recent years. These include the LEPs' strategic economic plans (including updated versions in 2016, 2017 and 2019) and other reports from the recent past: One North East's regional economic strategies (ONE, 1999; ONE, 2006), an OECD territorial review (OECD, 2006), the 'North East Independent Economic Review' (Adonis, 2013) and 'Tees Valley: Opportunity Unlimited' (Heseltine, 2016). This selection made it possible to compare policies and priorities recommended by earlier and independent sources with those actually pursued by the LEPs and CAs and identify areas of agreement as to priorities, and any differences and the explanations for them, including the extent to which path dependence was a factor.

The research into devolution began at the critical realist level of the actual with reading and analysing the agendas, reports and minutes of the two LEPs and CAs. All agendas, reports and minutes of the main governance bodies were read - or at least noted if they appeared routine or irrelevant to the research questions - at an early stage and then periodically as new monthly papers appeared. The bodies covered in this way were the NECA leadership board, NELEP board, Tees Valley Unlimited (predecessor to Tees Valley LEP) and the TVCA cabinet. A similar same process was followed for the CAs' overview and scrutiny committees (OSCs), though at a later stage when it became clear that scrutiny was an issue for the research.

The meetings and documents taken into account are tabulated at **Table 1-3**. The numbers given take account of some subjective judgements: for example, appendixes

to reports, and reports not in the public domain, are not counted separately; inquorate meetings are included.

Analysis of these document started with the creation of a summary timeline of events and issues covered in the NELEP/NECA papers from March 2014 to January 2017. By this point the researcher had become sufficiently familiar with the issues to dispense with this procedure, and could readily tell the significant from the routine and feed the former directly into the analysis. In the case of Tees Valley it was possible to move directly to the second stage and focus largely on the principal controversy, which was over the mayor's plan to purchase Durham Tees Valley Airport, and on the mayor's leadership style. This familiarity with the main issues in both areas was achieved in time to inform the interviews, and the documents additionally provided a written record against which the memories and perceptions of interviewees could be independently checked.

Meeting and documents					
Body	Dates (inclusive)	Meetings	Reports		
NELEP	11/09/2014 to 21/03/2019	41	80		
NECA Leadership Board	15/04/2014 to 05/02/2019	41	258		
NECA OSC	14/06/2014 to 14/03/2019	30	105		
TVU	22/01/2014 to 27/01/2016	9	n/a		
TVCA Board	04/04/2016 to 22/03/2017	11	41		
TVCA Cabinet	31/05/2017 to 15/03/2019	18	118		
TVCA OSC	09/09/2016 to 01/03/2019	9	n/a		
NTCA Cabinet	04/11/2018 to 12 /03/2019	4	16		
NTCA OSC	17/12/2018 to 21/03/2019	3	9		
Totals		166	627		

These readings sometimes resulted in supporting or clarifying documents being sought from other sources such as government departments and individual councils. Government documents from several departments and agencies provided data, the most important being the Department for Business, Energy and *Industrial Strategy*, the Ministry of Housing, Communities and Local Government, and their predecessors, and the Cabinet Office. House of Commons Library briefings were also helpful. Regional mainstream media were monitored on a daily basis for reports of relevant events and opinions. Social media was discovered during the course of the study to have been in use for campaigning against NECA's devolution deal and the relevant site was thenceforward monitored. The Tees Valley mayor's highly active social media sites provided useful information.

The achievements of the CAs and LEPs in economic outcomes by March 2019 – the half-way point in the 10-year SEPs and the concluding date for this study - were compared quantitatively with the targets set out in the SEPs published in 2014, using UK official statistics and the LEPs own documents (6.5).

#### 1.5.2 The interviews

Interviews were the primary means for penetrating beneath the actual events recorded in official document to the underlying levels of the empirical and the real. While examination of the documents discussed above is an effective method of researching actual events, the interviews were essential for exploring the sociocultural environment that must be understood to answer the research questions. To get as wide a range of interpretations as possible, interviews were carried out with representatives of all five leadership groups identified in the analytic framework (2.2.4) Forty-two-interviews were conducted; in one case two colleagues chose to be interviewed together, making a total of 43 interviewees (**Table 1-4**).

Local politicians		
National politicians		
LEP members	4	
Business leaders	5	
Other private sector	2	
Voluntary and community sector		
Trade unions	2	
Political campaigner	1	
Public servants	3	
Miscellaneous (health, culture, ex-RDA)		

#### Table 1-4: Interviewees

The most important task was to secure interviews with those close to the devolution decision-making process. Interviews were secured with leading political figures from all seven NECA councils which collectively rejected devolution. Senior political

representatives of two of the five Tees Valley councils were interviewed, but as there was no evidence of disagreement between the councils it was considered superfluous to interview others. It was however essential to interview leading players in the devolution drama from the four other leadership groups identified in the analytic framework – public officials, business, trade unions and the voluntary sector. Account was taken of the fact that officials also had a voice through the documents studied.

An interview plan was prepared in October 2017 listing the position held by each interviewee justifying his/her selection, the relevant issues the interviewee was in a position to deal with and the research question/s to which s/he could be expected to help provide an answer. Interview request letters were sent by email in November 2017 specifying that the subject would be the role and impact of LEPs and CAs in the North East and Tees Valley. Within that broad framing, the semi-structured interviews enabled interviewees to raise any matters they considered relevant; some, for example, raised comparisons with the earlier regional development agencies. The researcher did, however, always steer discussion back to the issues of the research questions.

Some interviewees held region-wide responsibilities, some in one case study area or another. Many held leading positions in the region and some nationally, though all had backgrounds in the region. The national politicians and a civil servant interviewed were chosen for their combination of national and regional perspectives. Some interviewees filled dual roles (e.g. politician and LEP board member). Voluntary sector interviewees were selected on the basis of their involvement in economic development activities, such as business support, enterprise development, education and training, and employability. Some held senior positions, others closer to the grass roots. Eight interviewees were women; one was from an ethnic minority. More interviews were necessary in the North East than Tees Valley because the more turbulent events in the former called for a wider range of interpretations. At least one interview was conducted in each of the region's 12 municipalities. The researcher was fortunate in that very few of those asked to give interviews declined.

Interviews, conducted between November 2017 and December 2018, were carried out face-to-face and lasted 40 minutes on average. They were recorded, transcribed by the

author, digitally stored and coded in NVivo to aid analysis. Fifty-seven codes were used, with 1,932 references distributed among them (Appendix B). Quotations used were selected as those best suited to summarise views expressed by interviewees collectively and/or support points being made in the thesis.

Many of the interviews were with members of local elites, and scholars have identified a wide range of problems associated with this practice. How does one identify elites, asks Cochrane (1998) – a point discussed in relation to this research immediately above and in connection with the analytic framework (2.2.4). What is the relationship between researcher and interviewee? How can the researcher avoid being sucked into the interviewee's world and becoming its mouthpiece? Alternatively, is it legitimate to assume elites have something to hide which should be exposed? Cochrane (1998: 2030-2031) advocates responding to these questions with 'committed scepticism' – not simply believing elites, but still taking them seriously in their own right. He also advises that 'the key is not to be found in the elites themselves, but in the broader systems, processes, and structures of which they are a part' – in other words, the causal mechanisms of critical realism (1.4.1).

(Raco, 1999) argues that representations of the world arising from research, including interviews, are a fundamental part of political struggle and something which is the virtual monopoly of what he calls 'intellectuals'. This is a point addressed in relation to this research in the author's reflexive remarks in section 1.5.4. Raco says of the researcher's monopoly position:

This 'privilege' is one which can, therefore, be used in the pursuit of political agendas. On the one hand, critical research may challenge existing power relations through exposing and uncovering new forms of knowledge and helping to make organizations more accountable for their actions. On the other hand, intellectuals may help to reinforce existing power relations through exclusive processes of knowledge construction and distribution, which concentrate power further in the hands of the already powerful' (Raco, 1999).

Critical realism raises two practical issues for research based on interviews: how do we know interviewees are telling the truth, and how can we have confidence in our own explanation of what they say? On the first question, as Clark (1998: 82) says: 'I do not

believe that respondents tell us the truth if truth is defined as neutral, uncommitted observations about the given world'. Interviewees may have their own agendas. Clark's recommended response involves continuous checking against known facts, and fortunately checking was often possible in this case, not only against the views of other interviewees but also official documents. According to Stake (2000: 443-444), the solution involves using multiple perceptions to clarify meaning, verifying the repeatability of an observation or interpretation. It is a process that brings different methods – principally interviews and documents in this case – to bear on the same issues. In relation to one important area for this research, party political meetings in private, the task is particularly difficult, as it must rely on the memory and interpretations of individuals. But as Grix says: 'Research should be judged on how its constituent parts logically link together and not by the methods used' (Grix, 2002: 181).

Schoenberger recognises that interviewees themselves interpret the events they describe, and the researcher then has to interpret the interpretation. The best strategy, Schoenberger believes, starts with being well informed. The interviewee will be reassured by an informed researcher and is therefore likely to be more open, while the researcher will have an independent basis from which to assess what he or she is told. The researcher should constantly check validity by looking out for inconsistencies and asking for the same events or decisions to be described in different contexts (Schoenberger, 1992: 186-187).

Confidentiality is an issue experienced by Neal and McLaughlin (2009) in a way relevant to this research. They report that in their research anonymity was insufficient to avoid particular voices being recognised in a setting in which all the respondents were known to each other (2009: 694). In the research for this thesis interviewees were in some cases political colleagues or business associates of each other. The North East and Tees Valley political and business communities are relatively small and closely integrated and care had to be taken in the text to minimise the possibility that individuals would be recognisable.

Cochrane (1998) and Smith (2005) both call for reflexivity on the part of the researcher as a way of dealing with these issues arising from elite interviews. Personal reflexivity relating to this research is provided in section 1.5.4.

#### 1.5.3 Use of statistics

This is, for the most part, a qualitative study. When statistics are used, they are drawn from the UK Office for National Statistics (ONS), official labour market statistics (Nomis), HM Treasury, North East England's two LEPs, Eurostat and the OECD regional demography dataset. Where statistics for 2016 are used instead of more recent figures now available, it is because these are the statistics for the time councillors made their devolution decisions. While recognizing that statistics are open to interpretation and never beyond question, official statistics are accepted as valid for the purposes of this study.

Numbers relating to council elections, party political balance and trade union representation on local authorities are sourced from the council websites concerned; compilations are by the researcher.

Statistics are actual, but wield powerful empirical force. They influence perceptions which feed back to the level of the real and therefore drive actions. They contribute to the causal mechanisms that lead to decisions. How one views statistics depends on where one is coming from and in turn helps determine where one is going. We will see this process at work most clearly in the case studies in the attitudes of politicians and their officials to funding issues.

#### 1.5.4 Ethical and reflexive considerations

Practical steps were taken to ensure the confidentiality of interviews, anonymity of interviewees and security of data. Interviews were agreed and arranged by e-mail. Each interviewee was assured that his/her participation would remain confidential and any use of material from the interview would be anonymous. Interviewees were informed of the nature and aims of the research at the time the interview request was submitted. Interviews and all associated documents, including back-up copies, were stored anonymously on the researcher's laptop and an external hard drive. Identities were kept separately from the interviews. All this information is encrypted, in addition

to the researcher's routine password protection. The research posed and continues to pose no known risk to anyone.

Speaking reflexively, the researcher should clarify his own situation for the reader, expanding on the information in the *Acknowledgements*. A native of the North East and after a career in newspapers and then government, he feels committed to the region's prosperity and wellbeing. For professional reasons he has always attempted to maintain political impartiality and the only occasion he joined a party, when already in retirement, was on the day the 2017 general election was called. He joined the Liberal Democrats for the sole reason they were the only party with a Brexit policy he could support. He resigned again in under a year because he did not feel professionally comfortable placing his political impartiality in question. Brexit features in this thesis, but only incidentally.

His career experience placed the researcher in what is certainly an unusual relationship with his interviewees, and what he considers an advantageous one. It made him familiar with the regional political and economic context and comfortable dealing with local elites. However, his contextual knowledge, though still relevant, was somewhat out of date in detail at the time of the research, which he considers an added advantage as it meant he was not on such close personal terms with any interviewees as to be influenced by his relationship. He was previously professionally acquainted with two political interviewees of different parties, but had seldom or never been in touch with them for some years. He had worked with another interviewee more than a decade earlier and not seen him since. He was currently friendly with one other interviewee.

Finally, the researcher's experience reinforced, he believes, his ability to deal with the 'situated knowledge' issue – both his interviewees' and his own. Situated knowledge has been described as 'the idea that all forms of knowledge reflect the particular conditions in which they are produced, and at some level reflect the social identities and social locations of knowledge producers' (Rogers *et al.*, 2013). While the researcher believed that all his interviewees spoke sincerely, he recognised that they all reflected their own situated perspectives, as he does his own, and he was able, he

believes, to take that into account, applying the 'committed scepticism' recommended by Cochrane (1998) (1.5.2).

# 1.6 Conclusion

This chapter has stated the aims of the thesis and the research questions. It has introduced the reader to the key concepts being explored, to the critical realist and *phronetic* philosophical approach and to the analytic framework. The reader will now understand the basis on which the two case study areas were selected and the mixed methods used to study them, and why the interviewees were selected. The chapter has explained how the account of the region's contextual background given in Chapter 3 will contribute to the different approaches and methods required to address each of the three research questions. Ethical issues have been considered and the researcher has explained reflexively his own relationship to the research.

The strengths of the thesis are that the case studies facilitate an in-depth examination of the real-level causal factors that result in informal institutions influencing formal institutions. They provide early examples of the governance regime of LEPs and CAs, with particular reference to social values, political traditions and practices, including the private workings of party groups, and of democratic accountability, inclusiveness, leadership and the elected mayors issue. The researcher has had interview access to leading actors from all leadership groups involved. The aim is to come to a reasoned, plausible judgement on these issues which supports the recommendations with which the thesis concludes, aware however that alternative interpretations are possible. These issues are of wider interest than in North East England, not least to policy makers in the field of economic development, and the country's other LEP areas.

The rest of the thesis is structured as follows: Chapter 2, the literature review, covers theories of economic geography, including debates over agglomeration and placebased *versus* spatially blind development; and governance, including leadership and the mayoral question. Chapter 3 applies the analytic framework thematically to the context of North East England, placing the region within its socio-cultural and political, governmental, economic and environmental framings. Chapter 4 and Chapter 5 present the case studies covering the North East and Tees Valley respectively, in narrative form, taking a critical realist approach. They explain why the North East and

Tees Valley took different decisions on devolution (providing an answer to research question 1). Chapter 6, following the narration of events leading to the different decisions in the previous two chapters, outlines their consequences for the approaches of the case study areas to economic development (providing an answer to research question 2) and considers whether the different approaches are making any difference to economic outcomes. Chapter 7 addresses research question 3 by discussing what the case studies of Chapter 4 and Chapter 5 tell us about accountability and inclusion. Chapter 8 draws together the answers to the three research questions. It discusses what has been learned empirically, conceptually and methodologically. It makes recommendations for reform of the institutions of devolutionary governance. Finally, it sets out a further research agenda to support its suggestions.

# Chapter 2. Literature Review: economic theories and political governance

English devolution having been situated in its global, historical and political context (1.2), this chapter introduces the issues raised throughout the thesis through the medium of academic and some official, literature. Sub-national development faces issues in both economics and politics. Those discussed here include the key concepts introduced in section 1.3 and all have a significant place in the debates over devolution that emerge through the case studies described in Chapter 4 and Chapter 5. This chapter first discusses economic theories: new economic geography, focusing on agglomeration the debate over place-based or spatially blind policy; institutional economic geography, featuring place and identity, and path dependence.

Though these theories and concepts were almost certainly not in the minds of most North East England actors, in terms, during the events to be discussed in the case studies, and certainly did not feature explicitly in their debates, they form an essential background against which the events must be seen to be fully understood. In the context of new economic geography, a preference for place-based policies – though rarely if ever mentioned as such – is taken for granted by all concerned in the region; yet the literature makes it clear that there is an alternative, spatially blind policy, which governments must consider. Theories of institutional and evolutionary economic geography will be seen, as the argument of this thesis progresses through the following chapters, to underlie explanations of how the region's institutions, formal and informal, influenced the decisions on devolution being researched here.

Political issues examined in this chapter will also be found as the research progresses to play an important role in North East England: trust, legitimacy, the historical meaning of events, the continuing power and influence of government over governance, hopes and fears over fiscal devolution, leadership including the mayoral issue, accountability, inclusion, and the role of party groups and business leaders

All of these issues, economic and political, form essential background for understanding the events to be narrated in the case studies and for providing answers to the research questions.

# 2.1 Theories of economic geography

This section discusses the controversy between new economic geography, with its emphasis on spatially blind policies and agglomeration, and place-based strategies aimed at achieving even levels of development across all geographies. It then examines institutional and evolutionary economic geography, both of which are concerned with how the features of particular places, such as their politics and cultures, and their ability to adapt, affect their development.

### 2.1.1 New economic geography

The claim that cities act as growth hubs, creating jobs and benefiting their hinterlands, has long been controversial. Molotch (1976: 140) saw cities as growth machines driven by property interests keen to drive up land values and rents. He did not think they created jobs, merely distributed them as cities competed for their share of economic activity largely determined by spatially blind decisions outside their control - investment returns and the money supply (1976: 320).

The idea of the city as growth hub survives, if in different forms, and remains controversial. Moore-Cherry and Tomaney (2019: 1-2) note that 'metropolitan regions are now regarded as the drivers of economic growth' while using the term 'metrophobia' to describe a simultaneous reluctance to consider new metropolitan governance structures. Waite and Morgan (2019: 1-2) use the counter-term 'metrophilia' and note that while cities are seen by some policy-makers as 'engines of growth', empirical evidence casts doubt on this claim.

The role of cities is inextricably linked with the debate over place-based *versus* spatially blind development. Should cities be left free to agglomerate, with the resulting benefits spreading – it is hoped - to all, or should policy intervene to support growth in peripheral areas? New economic geography (NEG), made prominent by Krugman in the 1990s (Krugman, 2011: 2; Martin, 2011: 53), seeks to 'explain the riddle of uneven

spatial development' and why agglomerations form in a few places, leading to inequalities and tensions (Fujita and Thisse, 2009: 109).

NEG is controversial partly because it uses mathematical models to theorize the economic landscape, such generally valid abstract models paying attention to economies of scale, mobility of capital and labour, and transport costs, but not to contextual geographical differences, and thus producing, in the view of some, onesize-fits-all approaches to regional policy (Hassink and Gong, 2016: 3-4). The controversy was highlighted in an exchange between Krugman and the geographer Martin (1999: 80, 83), when the latter wrote that what he calls 'economic geography proper' involves not abstract models but 'a firm commitment to studying real places (a recognition that local specificity matters) and the role of historico-institutional factors in the development of those places'.

Economic geography proper, wrote Martin (2011: 54-56), is 'mainly narrative-based and discursive...typically emphasizing a range of institutional and cultural processes as well as – or instead of – economic mechanisms, where the interest is on "actual existing" spatial economic landscapes, not on hypothetical and idealized ones'. It is realist, pragmatic, empirically grounded and often case study orientated. In the terms of this thesis, Martin's approach is *phronetic*.

The controversy around NEG was heightened further when its influence was given expression in the World Bank's *World Development Report 2009* (Hassink and Gong, 2016: 4, 5). The Bank was opposed to the way the debate on development had become 'fixated on economic growth in lagging areas' (World Bank, 2009: 23). It argued that the drivers of economic transformation were agglomeration, bringing the benefits of the concentration of economic production; migration as people and businesses move closer to agglomerations; and specialisation (2009: xx-xxi, 34, 39). The bedrock of this process of integration was spatially blind institutions, such as laws, regulations and basic services, supported by connective infrastructure, and – but only as a last resort – targeted, place-based interventions (2009: 1, 22-23, 231). Economic growth would as a result, it agreed, be imbalanced. However:

To try to spread out economic activity is to discourage it. But development can still be inclusive, in that even people who start their lives far away from economic opportunity can benefit from the growing concentration of wealth in a few places. The way to get both the benefits of uneven growth and inclusive development is through economic integration (2009: xxi).

The Bank claimed not to be anti-equity, and argued progressive policies could ensure that living standards converge (2009: 2, 39). However, McCann (2016: 430) argues that often what are ostensibly space-blind policies in reality tend to be top-down, centrally driven policies which favour the interests of elites in capital or commercially dominant cities, and this is one reason for lack of trust leading some local actors to refuse to engage. Martin (2008: 3) argues that the claim associated with NEG that policies to reduce regional inequalities may be nationally inefficient is neither theoretically convincing nor empirically supported.

Urban Economics (UE), like NEG, focuses on cities, or city-regions, and their benefits of scale, population and density. UE incorporates assumptions of classical economics, such as that capital and labour are rational and mobile, but places more emphasis than NEG on human capital. Education, skills and entrepreneurship are seen as critical, and people move to cities for quality of life as well as jobs (Pike *et al.*, 2017: 129-131).

NEG and UE notwithstanding, advocates of decentralization and place-based interventions at various levels are numerous (1.2), and they found support in the *Barca Report 2009*, which argued for place-based development. The Barca Report was produced for the European Commission and was highly influential in the redesign of European cohesion policy for 2014-2020 (Hildreth and Bailey, 2013: 240), under which the North East LEP was awarded 540m euros and Tees Valley 202m (BIS, 2013b).

Yet the advocates of place-based policy can sometimes struggle to make the economic case. Empirical research by Pike *et al.* (2012) in the UK found limited evidence of any economic dividend of devolution, though this was hard to discern because its likely effects were overridden by national economic growth (2012: 25). According to Martin *et al.* (2016: 342), imbalance in the UK economy is a problem that has not been resolved by 90 years of regional policy.

The debate about the role of cities in the UK specifically is skewed by the fact that London is so much bigger than all other cities. McCann (2016) makes the case that the strong economic performance of London has not benefited the rest of the UK outside its southern hinterland. He therefore sees the UK's economic geography problem as between the London-based core and the peripheral regions, with varying urban performance elsewhere simply as a manifestation. According to Los *et al.* (2017: 790) the assumption that London would catalyse growth across the UK has simply not happened. Their analysis shows that for all other UK NUTS-2 regions<sup>7</sup> demand from London only accounts for between 0.7% and 4% of local GDP<sup>8</sup>.

Whether the same is true at a smaller scale of other UK cities in relation to their hinterlands has an important bearing on this study. Between 2000 and 2010 most cities (defined to include their wider urban areas) in the midlands and north of England grew more slowly than the national average, placing in question the suggestion that they should be allowed or encouraged to expand. Outside London, productivity barely increases with city size (OECD, 2015b: 48; McCann, 2016: 137, 140, 146-148, 150-151, 490). Lloyd (2015: 14-15) argues that the city growth model is inadequate because it rests on the notion of agglomeration economies and besides is not aimed at equity between regions, redistribution or social justice.

NEG thus poses a dilemma for the UK with its dominant capital city, raising the contested question of a trade-off between national growth and regional equality. Overman (2012) suggests building on success by prioritising investment in London and a few other cities – he names Birmingham, Manchester and Leeds – even if this means uneven spatial development. Leunig and Swaffield (2007: 5-6) agree, proposing expansion for the London economic powerhouse while accepting that not every town and city that has fallen behind can be regenerated. Some medium-sized cities – Manchester, Leeds and Newcastle are mentioned – can build on their strengths but, unlike London, are not successful enough to deliver prosperity to surrounding towns. Newcastle's place in this debate is an important feature of this study.

 <sup>&</sup>lt;sup>7</sup>NUTS: Nomenclature of Units for Territorial Statistics. NUTS level 2 regions are counties and/or groups of counties (Eurostat: https://www.ons.gov.uk/methodology/geography/ukgeographies/eurostat). North East England, including Tees Valley, is a NUTS2 region.
 <sup>8</sup>Gross Domestic Product (GDP) measures if and how much the economy is growing (https://www.gov.uk/government/news/gross-domestic-product-gdp-what-it-means-and-why-it-matters)

Pike *et al.* (2012: 11, 19-22, 25) find that when the economy is thriving London and the South East benefit most and regional disparities grow, but when boom turns to bust the economies of regions like the North East are less exposed to the market and disparities reduce. The implication, that reducing regional inequalities is linked to reducing national economic performance, is unlikely to please many. This outcome is linked, however, to the pursuit of growth-oriented national and regional economic policy, and other, more redistributive approaches may have different outcomes. An alternative view is that raising regional performance is an essential component of raising national performance. According to the government's *Industrial Strategy*, every region has a role to play in boosting the national economy, while the independent, though northern-based, Industrial Strategy Commission is still more explicit: 'The performance of the UK economy is held back by our high degree of regional imbalance' (BEIS, 2017: 216; Industrial Strategy Commission, 2017: 9).

This debate raises other important questions; how many agglomerations can a country have? Does the UK have only one, around London, or can there be miniagglomerations centred on second-tier cities – what Hildreth and Bailey (2013: 240) call 'mini spikes'? If so, what are the implications for the distribution of investment within as well as between regions? Should regional capitals be encouraged to grow to benefit their hinterlands (if that consequence would indeed follow) or should regional investment be evenly distributed, as pork-barrel politics demands? Besides, different definitions of a city may give different answers – is it the municipality, or a wider metropolitan area?<sup>9</sup> Former Chancellor of the Exchequer George Osborne's plan for a Northern Powerhouse suggests that England could have two agglomerations – London and the north, integrated by improved transport (Osborne, 2014). According to the independent UK2070 Commission, agglomeration works well in southern England and Scotland but plays a limited role in northern England, the midlands and Wales (UK2070, 2019b: 36). According to lammarino et al. (2019: 284), the jury is out on whether the benefits of agglomeration can be achieved through a more even distribution of medium-sized agglomerations.

<sup>&</sup>lt;sup>9</sup>See section 6.1 for a discussion of this question.

Present government policy is a nuanced mix of place-based policies aimed at boosting the national economy while working through LEPs and mayoral combined authorities (MCAs) to build on city, growth and devolution deals and introduce local industrial strategies with policies tailored to local circumstances (BEIS, 2017: 214-239).

Doubt is thus cast on both approaches: agglomeration around London has not produced much spillover benefit for other parts of the UK, but decades of regional policy have not worked either. This debate, however, reflects a relative view: agglomeration and regional policy have failed, true, if failure is seen as development which is not evenly spread across the country, or even more rapidly in lagging regions so they can catch up. But, as will be seen in section 3.1.2, the lagging North East of England has continued to get wealthier in absolute terms and, when lower housing costs are taken into account, people in the region are actually getting more prosperous more quickly than elsewhere, though they are still behind, and the fact that there remain neighbourhoods suffering multiple deprivation (1.4.2) draws attention to the fact that economics is only part of the argument for regional policy. There are issues of social justice too, and the question whether spatial disparities are necessary and, if so, politically tolerable as part of a trade-off to secure higher national growth.

Martin (2008) discusses justice as an issue alongside efficiency, asking: 'Are there equilibrium levels of spatial agglomeration and regional inequality that maximise national growth, and are these socially acceptable? Such questions remain largely unanswered yet are central to the trade-off issue'. Martin believes that even if there was a trade-off between national efficiency and regional equality, which he questions, 'policy surely should be concerned with...devising policies that both increase national growth and reduce regional inequality: that is, policies that increase both efficiency and equality' (2008: 3, 11).

The controversial implication of this debate is that maximising national growth is good for all in absolute terms but may actually increase disparities, while regional, placebased policy may be the only one acceptable to its beneficiaries on grounds of fairness but comes at the price of slower growth all round. Add the fact that regional policy under all governments this century has been spread across all regions, and it is hard to see who benefits, either absolutely (because growth is slower than would otherwise be

the case) or relatively (because leading regions are given the same assistance as lagging ones). When all parts of an unequal country are offered help equally, the result will be neither an economic dividend nor territorial justice (Morgan, 2006: 189, 201-203).

The Industrial Strategy Commission recognises both that there is a trade-off between efficiency and justice and that not everywhere can receive equal support. While there should be nowhere where the *Industrial Strategy* makes no impact at all, it says, the need to focus does mean some places will receive more attention than others: 'Policymakers must acknowledge a trade-off between economic efficiency and equitable treatment of communities. Sometimes, in these cases, it is right that decisions are made where the fairness objective predominates' (Industrial Strategy Commission, 2017: 9)

Martin *et al.* (2016) believe a radical, place-based approach is required. They diagnose the problem as caused by the concentration of economic, political and financial power in and around London and argue that northern areas have the potential to reverse their decline if the key institutional structures of the national political economy are devolved. They envisage devolution of large sections of public finance to a new federal system; regions or city-regions as the key units of economic governance, and locally accountable; a national investment bank; reform of the tax system; and a commission or similar body to oversee all this.

The UK2070 Commission, drawing on work by the IPPR think tank, is also moving towards a radical solution, provisionally calling in a first report in 2019 for a 'systematic and comprehensive' framework of political devolution and organisational decentralisation, and a budget for the regions of £250bn over 25 years. It has suggested a constitutional convention 'to achieve a stable devolution settlement' (IPPR, 2018; UK2070, 2019b: 10, 62, 64). Lloyd (2015: 23) is another calling for a constitutional convention to start the process of UK fiscal federalism. The case for 'fundamental reform', with the suggestion of a shift towards a federal system to rebalance the economy and reduce its concentration on London, is made as well by Martin and Gardiner (2018).

#### 2.1.2 Institutional economic geography

New approaches emerged in the 1990s, under the influence of postmodernism, emphasising the cultural, institutional and evolutionary foundations of economic processes. Postmodernism attracted the interest of geographers because of its embrace of difference and variety as characteristics of the world, including regionally (MacKinnon and Cumbers, 2011: 34). Institutional and evolutionary geography and new regionalism are of special relevance to this study and closely linked. Together they raise important questions about the significance of institutions, formal and informal, for regional and/or local growth and about the issue of path dependence. They support the *phronetic* element of the philosophical approach (1.4.1).

New regionalism examines the effects of social and cultural conditions within regions on economic growth and the way inherited institutional frameworks and routines influence how regions respond to economic challenges (2011: 37-38). Economic behaviour is embedded in regional cultures, according to Gertler (2003: 134): some 'hard luck cases' were once successful areas but fostered ties so strong, structures so rigid and attitudes so unbending that newcomers and new ways of doing things encountered insurmountable barriers. Gertler mentions parts of Germany, Switzerland and the US; MacKinnon and Cumbers (2011: 38) add North East England to the list.

Rodriguez-Pose (2013: 1037-1038) argues there is reason to believe that it is formal institutions, like the rule of law, that have most effect on economic development, while most analyses of the impact of informal institutions in advanced countries find their effects to be negligible. Storper *et al.* (2015: 111), in their comparative case studies of San Francisco and Los Angeles, found their interviewees and a large scholarly literature to be quite sceptical about the notion that even deliberate formal policies, if they were local, had significantly shaped divergence in development.

Economic geography distinguishes between the institutional environment of a territory and its institutional arrangements. The environment, explains Martin (2000: 79-80), encompasses both the informal institutions described in section 1.3.1, such as traditions, and formal ones like constitutions, while the arrangements are the organisations, like CAs and LEPs, referred to by Hodgson (2006) as special institutions. The environment and arrangements affect each other, and '[H]ow this interaction

varies across space, and how it shapes economic outcomes, are central issues of concern in institutionalist approaches to economic geography' (Martin, 2000: 80). Rodriguez-Pose (2013: 1043-1044) believes it would be futile and counter-productive to try to change the institutional environment, which forms the embedded culture of a place, but efforts should be made to overcome barriers to development created by the arrangements.

What Amin and Thrift (1994: 14-15, 258-259) call 'institutional thickness' can assist growth, such thickness consisting of the number of institutions, the degree of interinstitutional interaction, the formation of coalitions and the development of a common agenda. This provides a clue to the puzzle why some regions are institutionally thick and yet unsuccessful. Thickness alone is not enough; institutions need co-operation towards a common goal. This is in keeping with the critical realist view adopted in this study (1.4.1) that it is relations, not entities, which matter.

Tomaney (2014a: 134) notes that regional institutions tend to be viewed through separate lenses, one emphasising their contribution to economic performance and the other their expression of politically constructed spatial identity. The result is that economic governance is exercised by individuals who bring their embedded cultural norms to the task, blurring the distinction between formal and informal institutions – a point reinforced by this study. What those norms are in North East England and how they affect local economic governance will come to light as the narratives unfold.

#### 2.1.3 Evolutionary economic geography

Institutions are embedded in society not only spatially but temporally, in the history of their place. They are characterised by path dependence (1.3.4) and evolve incrementally, acting as 'carriers of history' and transmitting attitudes and values down the generations (Martin, 2000: 80). Though path dependence is often discussed as a feature of economies, for the purpose of this study its greater interest is its applicability to governance institutions.

Martin (2010: 13-15) is critical of equilibrium-based approaches to path dependence and draws attention to the idea that institutions, like economies, may change over time; he identifies three mechanisms to enable this to happen. Firstly, layering may

occur, in which new rules, procedures or structures are added to an existing institution; secondly, an institution may take on a new form or function, or both, or be reoriented to serve a new purpose, in a process of conversion; or thirdly, existing social-political-economic structures may recombine with new resources and properties to produce new structures. The abolition of RDAs and their replacement by CAs and LEPs, followed by the offers of devolution deals with their additional powers and resources as well as the creation of elected mayoralties, provide an opportunity to study - and an additional heuristic framework for analysing - how these processes of layering, conversion and recombination have been applied in actual cases of devolution in North East England.

This is the approach recommended by Mahoney and Thelen (2012: 31-32) in their study of gradual institutional change. Rather than promote abstract debate about metatheory or definitions, they say, their framework is intended to aid substantive analysis of change in individual cases. Ultimately, the arguments can be evaluated only through concrete cases and actual episodes – it is a *phronetic* process applied to the case studies of this thesis (4.7, 5.8).

Pike *et al.* (2010), whose purpose is to investigate the uneven resilience of different places, are also critical of equilibrium-based approaches, and concerned with renewal. They follow Grabher and Stark (1997) in distinguishing between adaptation and adaptability and arguing that while measures to implement adaptation in the face of exogenous shocks as quickly as possible – i.e. to return to the previous equilibrium – may be beneficial in the short run, they may hamper adaptability in the long term. For places where the previous equilibrium was at a low level, getting back to normal might be to the exclusion of seeking a better future. Adaptation and adaptability are thus in tension (Pike *et al.*, 2010: 62).

Sunley *et al.* (2017: 385) argue that we cannot assume the trade-off between adaptation and adaptability to be universal. In some instances incremental adaptation within a path allows learning and adaptation of dynamic capabilities that facilitate the creation of a new trajectory so that it generates adaptability. This is a process seen in North East England as, for example, shipbuilding has transformed itself to serve

various offshore industries, and the Tees Valley chemicals industry is adapting to meet the need for low-carbon energy (Chapter 6).

Keating remarks that failure to appreciate the value framework of a territory underlies many of the failures of regional development policies, such as the assumptions that peasant economies could painlessly move into tourism or that mining and steelmaking communities could become nurseries of entrepreneurs (1998: 5-6). It is an important issue for, as Garcia-Rodriguez *et al.* (2016: 2039-2043) report, empirical studies have demonstrated how the cultural aspects of a region can affect entrepreneurial intention even more than economic variables because they tend to present a more permanent character associated with fundamental values. According to Angula-Guerrero *et al.* (2017: 682-683) there is wide consensus that individual entrepreneurial activity and attitude cannot be explained without considering contextual factors, so that personal attributes and local conditions, including cultural and social norms, should be considered jointly. This has actually been done by Huggins *et al.* (2018) among others and the results relevant to North East England are reported in section 3.1.2.

A counterpart informal institution to lack of entrepreneurialism in old industrial areas is found by Robinson (2002: 326-333) in North East England in the form of community spirit. Robinson had doubts about policy direction, with its emphasis on trying to catch up with the south of England, wondering if this was possible and in any case whether it was what the people of the region wanted; might they not prefer to keep some aspects of regional life that risked being undermined by economic growth, such as affordable housing, reasonable public transport, a generally good environment and the ability of public services to attract and retain staff? Above all, perhaps, there were still elements of collectivism and community, born of generations of struggle in the tough industries of the past, not yet destroyed by individualism and consumerism. Community he saw as the region's main strength and foundation for the construction of a better, socially inclusive future. Robinson's concerns have been echoed more recently in the question discussed by Pike et al. (2017: 55): 'What kind of local and regional development and for whom?' and their conclusion that: 'Definitions...are broadening to include economic and social, environmental, political and cultural concerns'. These issues as they apply to North East England are discussed in section 3.1.2.

Taken together, the economic theories outlined above encompass a range of issues that extend beyond the purely economic to the political and moral, some of which may be addressed at regional and/or local level while others are matters of national policy. The fundamental questions are those associated with the debate about place-based and spatially blind policy: should place-based policy be pursued at all; if so, does it have trade-off implications, positive or negative, for national growth; if negative, is this morally justified on grounds of justice for lagging areas; should the aim be a relative one, to close gaps, or an absolute one, to increase prosperity for all but not necessarily to the same extent? The debate about the trade-off, if it exists, between maximising growth and reducing disparities, arises at every level – over London agglomeration and regional imbalance, over regional capitals as growth hubs, and between clusters and deprived communities as discussed below (2.2.4). If the answers to these questions result in place-based policy, as they have in one form or another for more than two decades, what form of decentralised governance is most appropriate?

Governance issues are discussed in the following sections of this chapter and their application in North East England in the case study chapters. Institutional adaptation in the socio-cultural-political field, its connection with community values and the consequences for entrepreneurialism are discussed in sections 3.1.1 and 3.1.2.

## 2.2 Governance, the state and devolution

This section traces the emergence of governance in England and the continuing role of government in overseeing governance. It describes the history of devolution and questions whether either Labour's RDA-based model or the current system have legitimacy. It discusses issues of leadership, including the debate over elected mayors and accountability; and inclusion. It discusses issues raised in the literature that will be found to trouble North East England leaders in their devolution debates – their relationship with government, particularly over finance; the conflict between regional and local, municipal interests; and leadership and the mayoral controversy. Scrutiny, inclusion, and the role of party groups and business leaders, all discussed here, will also be found to be problem areas in North East England.

#### 2.2.1 The emerging role of governance

If development policy is determined in the UK at national level, it is implemented partly through regional and/or local institutions, which are subject to frequent reorganisation. Since about 1980 scholars have been paying attention to the fact that states are no longer subject simply to government but to 'governance', which is a popular but imprecise term but can nevertheless safely be said to apply to the local and regional development studied here. The interest in governance came to prominence because a shift occurred in state-market-society relations brought about in part by the internationalisation of the economy, reinforced by a complex of reasons including, in the UK, public sector reforms during the 1980s rise of neoliberalism which moved the emphasis from a hierarchic bureaucracy towards greater use of markets, quasi-markets and networks, especially for the delivery of public services. These reforms intensified the effects of globalisation and the rise of transnational institutions such as the EU. This complexity and fragmentation meant the state increasingly depended on other organisations to deliver its policies (Rhodes, 1996: 652; Jessop, 1998: 29; Bevir, 2011: 459).

New economic and social conditions and attendant problems emerged which could not be resolved readily, if at all, through either top-down state planning or the market (Jessop, 1998: 32). This was a challenge to what Gamble (1990) calls the Westminster model based on parliamentary sovereignty, strong cabinet government, ministerial accountability through elections and the hierarchical bureaucracy of Whitehall. Supplementing this form of government came new networks of organisations involved in managing the economy and delivering public services: Jessop (1998: 31) mentions public-private partnerships, industrial districts and trade associations, among others, while Stoker (1998: 19) lists agencies, direct service organisations, opted-out hospitals and schools, and the voluntary sector. Rhodes (1996: 653-660) eventually settles on a definition of governance as self-organising, inter-organisational networks with characteristics including interdependence, continuing interaction, a basis of mutually agreed rules and a significant degree of autonomy from the state.

Governance has the same objectives as government – described by Stoker (1998: 17) as 'creating the conditions for ordered rule and collective action' – but by different means. The institutions of governance have no recourse to the authority and sanctions

of government (Stoker, 1998: 17), so they rely on trust (Rhodes, 1996: 652). According to Jessop (1998: 35-37), as well as trust, the key to success is continued commitment to dialogue, exchange of information, reduction of opportunism, interdependent decisions and solidarity. 'Shared values and norms are the glue which holds the complex set of relationships together' (Rhodes, 2007: 1245-1246). The extent to which these qualities are prevalent or absent in the institutions of North East England will become evident as the case studies in this thesis are developed. While CAs, as local government bodies, have authority in some areas, they and the LEPs still depend on trust and the other desirable qualities mentioned, such as in their dealings with each other and the business community.

The institutions of governance are, in one view, characterised by autonomy, both from each other and – with qualifications discussed below (2.2.2) – from central government. Those involved, according to Rhodes (1996: 658), have to exchange resources like money, information and expertise to achieve their objectives, maximise their influence and avoid becoming dependent on others. Stoker (1998: 18) likewise states that 'governance is about autonomous self-governing networks of actors'. Torfing (2007) finds on the basis of 12 case studies that governance networks have significant advantages; they are conducive to knowledge and understanding, flexible, good at mobilising private actors, reduce resistance, help to build consensus and cope with political conflicts, and close the gap between decision-makers and citizens. On the downside, good networks are hard to form due to political apathy and low expectations of the joint benefits, a lack of capable actors, reliance on precarious social and political processes, lack of leadership and frustration caused by absence of clear results (2007: 12-13).

Clarke and Cochrane (2013: 15) worry that governance can be technocratic and antipolitical in the sense of favouring apparently effective solutions and freezing out alternative approaches. They argue that participation has been partial and unequal, favouring the wealthy, the educated and the 'responsible' and excluding the inarticulate, the poor and 'extremists'. Elites, often unelected, set the agendas and position themselves as experts, with others as amateurs. Alternatively, consensus is achieved between stakeholders, glossing over issues and narrowing political debate. As Tomaney (2014a: 134) says: '[T]he sense in which economic development is a technical

exercise which is disturbed by an excess of politics is a theme of the literature on regional institutions', such an exercise being an attempt to avoid failures like corruption and inefficiency by offering 'hygienic' or technical solutions to what are essentially contentious political questions. This issue is discussed in a North East England context in section 6.2.1.

Where the literature of governance focuses on institutions and policies, that on governmentality – a term discussed by Bevir (2011) - concentrates on meanings and discourses. Policy actors are not treated straightforwardly as rational pursuers of power or cogs in institutional wheels but as drawing on historically contingent webs of meaning (2011: 457). This is an extremely promising approach for exploring the informal institutions featuring in this study, with many actors aware of the historical meanings of the events in which they were involved, as the interviews make clear. According to Bevir: 'In this view, people act for reasons that they form against the backcloth of inherited traditions that influence them' (2011: 463).

#### 2.2.2 The continuing role of government

Government forms one aspect of the state, an aspect seen by Jessop (2016: 72-73) as politically organised, coercive and administrative, with general and specific powers. It sits alongside the other elements of the state, which are clearly demarcated territory, a stable population and the 'idea of the state' giving it legitimacy.

The autonomous nature of governance institutions does not mean that central government is without influence, or even power. For one thing, it controls funding, which it uses to steer governance the way it wants. As Rhodes (1996: 662) says: 'So far, British government has compensated for its loss of hands-on controls by reinforcing its control over resources. Decentralising service delivery has gone along with centralising financial control'. The government also has a role in what Jessop (1998: 42) calls 'meta-governance': while remaining hands-off as far as strategies and initiatives are concerned, the government still has a major role in shaping the context, organising the dialogue, ensuring coherence and acting as a source of regulatory order and a last resort in case of failure. Rhodes (1996: 660) agrees that, although the state does not occupy a privileged, sovereign position, 'it can indirectly and imperfectly steer networks'.

Jones (2001) sees regional policy as a way for the government to shift problems from itself to another level. Labour's RDAs, he thinks, formed part of a political strategy aimed at rescaling, instead of resolving, an economic and democratic deficit, and 'ended up enacting a less-than-glamorous role as subcontractor to central government'. The same might be said of today's LEPs, at a different spatial level, as they technocratically administer growth funds handed down from the centre. CAs, however, being led by elected politicians, are liable to be less compliant and not always willing to accept responsibility for what they see as the government's problems, at least without what they consider the requisite funding. This was the view taken by the four south of Tyne NECA councils when they rejected a devolution deal (Chapter 4).

More fundamentally, Jones (2001) casts doubt on the economic necessity for regionalism by drawing attention to research which he says has done much to highlight that many successful regions are located within dynamic national political economies and innovation systems - a finding already noted in the discussion of NEG (2.1.1), and subversive of the economic (though not necessarily the political) case for regionalism (2001; 1185, 1198, 1202).

#### 2.2.3 Economic governance and devolution in England

'There has not been a long history of regionalism in England' according to a House of Commons briefing paper, which traced the story back no further than campaigns for devolution in some regions during the 1990s, leading to the establishment of RDAs in 1998 (Leeke *et al.*, 2003: 30). Keating (1998: 10) agrees, noting that: 'In England, the regions are mere administrative divisions with no institutions of their own'. Jones (2001: 1191) says that English regionalism has been 'somewhat inert'. By many measures, the UK is one of the most centralised countries in the developed world (Slack and Cote, 2014: 30).

This absence of robust regionalism in England may seem odd in the context of the simultaneous drive to decentralisation globally noted in section 1.2. It is explained by the fact that of the discourses that drove global devolution, in England identity (in this context) was not a strong factor while the case for regionalism as a route to democracy and/or economic improvement was unenthusiastically and weakly developed by

government, as evidenced by its failure to persuade the voters of North East England to support an elected regional assembly in 2004 (Rodríguez-Pose and Sandall, 2008: 62).

The weakness of the public debate over English devolution is a factor underlying the differences that will be revealed in this thesis's empirical research over the right geography for devolution, exemplified by the splitting of the North East Combined Authority (NECA) in 2018 (4.1) and continuing speculation in some quarters of the possible reintegration of Tees Valley with the rest of the region (5.7). Radical federalism is one suggested form of devolution, as we have seen (2.1.1), but there are others. An alternative approach is to build from the bottom, offering the advantage, or at least the hope, of creating a governance unit of whatever size can gain local political and popular support - what Rodríguez-Pose and Gill (2003) call legitimacy accompanied by commensurate responsibilities and resources. Government policy offers two of these requirements for devolution – responsibilities and resources - up to a point. Legitimacy, however, is something top-down government cannot confer. This is an important consideration: what will win the support of the public, stakeholdes and those who have to make it work. According to McCann (2016: 497-498), one of the challenges involved in identifying the most appropriate governance is to be perceived as legitimate and to develop trust and credibility with citizens, the private sector, civil society and other levels of government. Legitimacy, judged in this way, will be found in this thesis to have been achieved in Tees Valley, but not in the North East.

Economically, the bottom-up approach may not result in what Hildreth and Bailey (2013: 245) call the 'right geography' – at least initially. But it would, as Martin (2015: 261-262) suggests, enable people with local understanding to take account of the economic, social and, most importantly, political and institutional circumstances. Kinossian (2018: 372) agrees:

[I]n order to explain development patterns and to address uneven development more effectively, research and policy should pay more attention to the politicoinstitutional, social and cultural evolution of regions and localities in broader geographical and political contexts rather than generalising from the experiences of the few most successful.

This would likely mean, at least to start with, different-sized areas and populations and different policy solutions in different places. Support for this type of approach emerged from discussions between five of the fastest-growing urban areas in England – all in the south: 'When it comes to urban development, one size rarely fits all, and policy makers must recognize this' (Ramuni, 2019).

The point that regional devolution needs to be at the most appropriate democratic, not just economic, level is argued by Giovannini (2015). For the most part, she says, the underlying message in the current regional and city-regional agenda seems to be that devolution will lead to economic renewal for the regions lagging behind. Yet this is only one side of the coin, because to really flourish regional economies need to be nurtured from the bottom, through a system of governance which is ultimately accountable to the people, and not only to Westminster. McCann (2019: 11) argues the need for bottom-up institution-building to support place-based economic development involving all sectors of the community: '[M]odern place-based approaches are fundamentally about building the local institutional capacity necessary to ensure that genuinely locally-tailored policies are designed with the explicit involvement of local communities'.

Research by the Association for Public Service Excellence reached a similar conclusion 'The success of devolution, as with public services, will be measured not by the purity of the boundary or the structure but by the desire of those within them to make it work' (Rogers, 2018).

Besides, small-scale geographies need not necessarily mean lack of economic development. Towns can be the foci, as noted in Kelly (2016a) (1.2). Pike *et al.* (2017: 132-133) have reported that geographical units such as medium and smaller-sized cities/city regions and larger and medium-sized towns have been strong growth performers over the longer term. Research by Leach (2013) can also provide a model. She found

significant evidence that local economies with higher levels of small businesses and local ownership perform better in terms of economic success, job creation (especially in disadvantaged and peripheral areas) local multiplier effect, social

inclusion, income redistribution, health, wellbeing and civic engagement, than economies more dependent on centralised economic actors.

This could be an alternative route for those local authorities that choose not to go down the combined authority/devolution deal route. However, while Leach argues that much good could be done locally with relatively small changes using mechanisms already in place, this approach has its limits. Scaling up, she says, would still require decentralization of policy, support services, subsidies, tax, competition, banking, infrastructure and measures of success (Leach, 2013: 929-931). This is as radical a proposal in its way as the new policy models recommended by Martin *et al.* (2016) and others (2.1.1) and potentially as difficult to win government support for.

Evidence for the success of a more local approach of the type advocated by Hambleton (2015a), based on cohesive communities with inclusive leadership, is accumulating, both internationally and in the UK, and a common feature of case studies is that successful communities share interests that extend beyond the economic functionality that dominates the current English model (Kelly, 2016a: 29-33). It is an understanding of this community spirit, supporters argue, that is needed if successful development locations are to be shaped. One American community leader said expecting solutions to come from the government was rarely successful: 'What I tell communities is, if something gets done it is going to be because you do it. No one cares that much about the community except those people who live there or used to live there' (Grisham, V. quoted in Gilette, 2004).

Some disagree, though. According to Shutt and Liddle (2019: 196-197), commentators argue there are grave dangers in bottom-up approaches that seek to achieve consensus only at the local level, and the limits of localism are exposed when agreement cannot be reached.

The absence of meaningful devolution in England should not be taken as meaning there have been no regional policies. Quite the contrary; there has been a succession of regional policies since the 1930s (Tomaney, 2006: 3), culminating in Labour's RDAs, established in 1998. They were non-departmental public bodies, appointed by and accountable to ministers, though there were 'some arrangements in place to ensure

that the RDAs are responsive to regional views' including four local councillors on every RDA board (Allen, 2002: 13).

The Labour government also encouraged the development of non-elected regional chambers (1.3.3), also known as regional assemblies. These were voluntary bodies with no legal powers which scrutinised the work of the RDAs. They were made up mainly of local authority councillors, alongside 'social and economic partners' from civil society (Leeke *et al.*, 2003: 30). These institutional arrangements for England came at the same time as more significant powers were being devolved to Scotland, Wales and Northern Ireland. RDAs' accountability to regional chambers is further discussed in section 2.2.4.

RDAs and regional chambers, alongside the government offices for the regions (GORs) inherited from the Conservatives, formed a troika which worked with a fragmented array of predominantly single-purpose, quasi-autonomous bodies with overlapping responsibilities and strategies (Pearce, 2008: 109). The Audit Commission, quoted in Pearce (2008: 109), reported that although 'working relationships between the troika tend to be good, and there is evidence of them working together to set priorities and develop a shared agenda for the region, few working at the local or regional level feel that this equates to regional leadership in an overarching sense'. One strength of the chambers was their 'social and economic partner' members, who were praised by senior RDA and GOR officials as less parochial and more visionary than their local authority counterparts (2008: 108).

The chambers were eventually phased out between 2007 and 2010 (Sandford, 2013: 4) in favour of local authority leaders' boards, which formed part of what Tomaney (2014b: 32) calls a twin-pole structure with regional ministers and House of Commons regional committees at the national pole alongside leaders' boards at the local pole.

Though the coalition government that came to power in 2010 immediately abolished the RDAs and GORs and introduced business-led LEPs, followed by CAs alongside them, the establishment of local authority leaders' boards, however short-lived, had activated 'dormant or latent institutional resources, with local government leaders discovering their potential central role in regional governance' (2014b: 32).

CAs, consisting of two or more local government areas, with or without a directlyelected mayor, enable councils to take decisions across boundaries on issues which extend beyond the interests of any one council (Morse, 2017: 5). Whether the introduction of CAs, which are indirectly elected, has dealt satisfactorily with concerns about the democratic accountability of local and regional governance is a question for debate, and is discussed in section 7.1. According to Tomaney (2014b: 30) the establishment of NECA was at least an attempt to introduce some form of regional democratic structure to countervail the business-led LEP. Sandford (2017b) believes devolution has little to do with territorial governance at all but rather resembles a contract under which the government determines the terms on which it will outsource specific responsibilities. Accountability, governance and even geography take second place. Sandford (2019a) also argues that the system incentivises CAs to trade policy priorities in exchange for discretionary central funding.

Martin *et al.* (2016) believe regional authorities need greater powers and resources to stimulate growth, as noted above (2.1.1). So do Filipetti and Sacchi (2016: 1793), who found from a study of 21 OECD countries that the growth effects of fiscal decentralization depended critically on the authority of sub-national governments: tax decentralization led to higher or lower rates of economic growth when coupled with high or low administrative and political decentralization. Tax decentralization was more conducive to growth if sub-national taxes accrued mostly from autonomous revenues such as property taxes, they found. This is step that the UK government is planning to take through the business rates retention scheme (MHCLG, 2018a; Sandford, 2019b).

However, fiscal devolution is not without its concerns for local government because of fears that poorer areas with smaller tax bases will lose out ((Prud'homme, 1995; Rodríguez-Pose and Gill, 2003) (1.2), and this is a concern in the North East as we will see (4.5). Fiscal devolution would still leave relatively poor areas dependent on some form of redistribution from the centre unless uneven development was to be allowed to be accompanied by uneven wellbeing.

Ayres *et al.* (2017a) take a critical look at the devolution process yet hold out hope that deals, and particularly mayors may, in accordance with a 'wedge and crack' theory

developed by Wright (2004), enable minor initial concessions to be expanded into more significant changes. However, English devolution, so far, conclude Ayres *et al.* (2017a), has been a revolution that has not occurred, and has been marked by a 'rhetoric-reality gap'. England has seen little of the bottom-up public pressure associated with identity and democratic representation that characterised devolution in Scotland. Instead, the process has been driven by economic motives while paradoxically having to face the *realpolitik* of finding politically workable governance arrangements.

The result is a new form of top-down control, backed by financial incentives, assimilating local elites but with weak citizen mobilisation. Why, ask the authors, would local authorities risk becoming a lightning rod for centrally imposed cuts in return for only modest financial and policy incentives. The answer: 'Local authority leaders have consistently suggested that the importance of the deals lies not in their initial content but in their potential to evolve into a quite different relationship in the future' (2017a: 1). This was a hope expressed with confidence by the Newcastle Council leader during consultation on the establishment of North of Tyne CA<sup>10</sup>.

Could it happen? The government does not hold all the cards, and some council leaders have described it as 'desperate' to do deals. So: '[L]ooking to the future, there is clearly a strong chance that if the economic, social and democratic benefits of "devolution" deals begin to be realised then the centre-periphery model will shift, implying that the government has ceded some control' (2017a: 1, 2, 7-10). According to Magrini (2018), mayoral combined authorities (MCAs) are already better placed than their counterparts without deals to manage the challenges of an increasingly automated labour market. They can make their local economies more attractive by improving transport and addressing housing needs, and use their control of adult education to support people in an evolving labour market, as well as being better placed to secure more devolved powers in future.

A question worth asking here in the political context, as it was in effect in the economic by Martin *et al.* (2016) (2.1.1), is whether what the regions of England have been granted over the past two decades – whether Labour's troika or the current LEPs

<sup>&</sup>lt;sup>10</sup> Personal observation

and CAs - is actually devolution at all. Just what devolution is, is a 'complex and heterogeneous process' according to Rodríguez-Pose and Gill (2003: 334). They specify three factors – legitimacy, the decentralisation of resources and the decentralisation of authority (or responsibilities). While details of the last two of these are specified in the devolution deals agreed between government and CAs, legitimacy is a more subtle concept. Rodríguez-Pose and Gill (2003: 335) argue that the legitimacy of sub-national and national governments is determined for the most part by processes of history and political support, taking account of factors such as culture, language, religion, ethnicity and economic development which combine to create regional identity and legitimise regional government. Much of this, it might be thought, is not applicable in England which, though more diverse than in the past, is not differentiated by language, religion or ethnicity on a regional basis and only superficially on a cultural basis. But as we will see in Chapters 3-5, very localised identities and historic rivalries can pose continuing problems even in what appears a homogenous region like North East England.

Legitimacy in England traditionally comes through the ballot box, which is where regionalism was rejected in North East England in 2004. Under Labour, coalition and Conservative governments, regionalism and localism have focused on economic development, with political support and legitimacy as an afterthought. Labour's attempt at a North East elected regional assembly came six years after the establishment of RDAs, and the coalition's and Conservatives' CAs followed at various periods after their respective LEPs. The result has been fraught with difficulties.

#### 2.2.4 Leadership, accountability and inclusion in local government

Labour's English regionalism coincided with a reform of local government introduced in 2000. Reform and attempts to strengthen local leadership have been driven in several European countries by two concerns, according to Wollmann (2008: 279-280) – a democratic deficit reflecting a lack of transparency and accountability, and a performance deficit as local government appears less able to cope with social, economic and environmental challenges. Wollmann identifies varying degrees of movement towards individualist solutions such as mayors, and collectivist ones like committee systems. In England, he argues, self-government came historically to be seen as the collective action of local citizens according to an essentially voluntary
formula covering both decision-making and the execution of tasks. From this premise, local government by executive committee followed (2008: 292).

Not surprisingly, then, when Labour's Local Government Act 2000 obliged councils to choose between three constitutional models, 81 per cent rejected the two options involving mayors and preferred the alternative cabinet-and-leader system, which came closest to the collectivist tradition (2008: 284). The leader may approximate to a local 'prime minister', but he or she is at least elected by his/her peers and can be removed by a majority of the council, unlike directly elected mayors, whose position is thus more presidential.

Nevertheless, some believe the choice between the English tradition of collective responsibility and the mayoral model of strong and accountable individual leadership, need not be a dichotomy. The skill to bring actors from diverse areas of public life together in a common cause is recognised as especially important when governance is fragmented. Borraz and John (2004: 112-114) see leadership in these conditions as 'crucial to the functioning and success of local governance' and believe that 'leaders can make the shifting framework of individuals and organisations work together'. The OECD (2015a: 11) sees local leaders are being adept at influencing and persuading other stakeholders: 'They need to become expert in sharing, listening and networking'.

Hambleton (2015a: 187) lists the arguments for and against directly elected mayors. Their potential advantages are visibility, legitimacy and accountability, strategic focus and authority to decide, stable leadership, the attraction of new people into politics, partnership working and the building of coalitions. Disadvantages are celebrity posturing, the wrong geographic area, scope for corruption, weak power of recall, cost, and the continuing over-centralisation of the state. However, he backs away from prescribing a particular model of governance, saying merely that in democratic societies the citizens should decide.

The overview and scrutiny (O&S) system, through which backbench councillors hold a council's executive to account and can call in decisions for review, was found by Stoker *et al.* (2004: 9-10) to be 'problematic'. In some cases call-in was hardly used at all and in others frequently. The determining factor was political relations on the council. If competition was fierce and the parties closely matched, call-in was used to create

publicity and score points. Where there was a safe majority, party loyalties seemed to make challenges to the executive problematic and difficult to sustain. Wollmann (2008: 285) argues that the influence of backbench councillors has been diminished as their role has been confined to scrutiny, and a mayor would potentially distance backbenchers still further from power. Leach (2010: 336-337) argues that the executive-scrutiny division has demotivated backbenchers, perhaps explaining the resistance of dominant party groups to strong council leadership. O&S was carried forward from councils into CAs, and its operations in NECA and Tees Valley are discussed in section 7.1.

Scrutiny of CAs and LEPs bears some resemblance to that of RDAs by regional chambers, made up of councillors and other civic leaders, as noted in section 1.3.3. Blackman and Ormston (2005), who noted that accountability is very open to interpretation and the focus of much analysis, conducted a study of scrutiny of One North East by its regional chamber, which consisted of 72 members of whom 47 were council representatives and 21 'economic and social partners', alongside four MPs and MEPs.

The researchers found a tension between the RDA's focus on measurable economic targets, characterised as objective facts , which it was held responsible for achieving by the government which appointed and funded it, and the assembly's view that it was there to reflect the subjective views and experiences of stakeholders, which were regarded as just as valid. An example of this difference of approach was the RDA's strategy of clustering certain industries, as required by the government in the belief that all areas would benefit as a result of 'trickle down', and the assembly's scepticism that this would benefit deprived communities (Blackman and Ormston, 2005: 383-384).

Economic inclusion is a task falling largely to LEPs, which work hard to help more people access jobs, as we will see (7.3). Their approach has been questioned, however, for prioritising growth over social issues (Sissons *et al.*, 2018). The debate resembles that over place-based and spatially blind development, and regionally over clusters, as we have just seen, in the sense that once again there is concern over an implied and supposed trade-off between efficiency and justice. The authors find from a study of

devolution in six areas, including Tees Valley, that the focus is on creating more good jobs rather than improving low-pay sectors and challenging the existing development model and the inequalities it produces: 'Inclusive growth concerns appear to be largely side-lined in the devolution processes as a narrative of growth (and assumption of trickle-down) prevails' (2018: 10).

Inclusion also means participating in political decision-making. Young (2000: 6-14) draws a clear distinction between economic and political exclusion and argues that problems such as economic exploitation or a refusal to help needy people should be so named, and not confused with political exclusion. Nevertheless, she adds, social and economic inequalities such as deprivation do help to account for political exclusion. The membership of LEP boards is an important factor in connection with inclusive decision-making and is discussed in section 7.3.

Concerns over political inclusion are closely connected with a lack of public engagement. In the case of devolution, the Political Studies Association (PSA) found that decision-making had involved a small number of people at a high level in central and local government who had believed that a streamlined and closed process offered the best opportunity to broker a good deal for their area. The PSA believes there is potential to reinvigorate civic participation in communities in England, though it does not explain how this is to be done (PSA, 2016: 10). One possibility is participatory or deliberative democracy like the citizens assemblies that have been held in relation to devolution in some areas, in which a representative group of citizens are selected at random to learn about, deliberate upon, and make recommendations in relation to a particular issue or set of issue (Giovannini, 2017). Public consultation is another form of engagement and is discussed in connection with devolution in North East England in section 7.2.

Copus and Erlingsson (2012: 238, 242-243) found in a study of England and the Nordic countries that in spite of declining membership political parties continued to dominate local government. The party group and its private meetings were a forum for councillors to question, challenge, criticise and influence the party leadership, but in public, the system privileged unity, loyalty and discipline over debate, responsiveness and the articulation of citizen views, with councillors expected to defend group

decisions even in the face of public opposition. Ordinary party members and citizens were excluded from effective input into local politics. The operation of party groups in the North East is described in Chapter 4.

Councillors often regard themselves as having been elected because voters identify with their party, from which they believe it follows their duty to the electorate is to implement party policy rather than defer to whatever the wishes of local people may happen to be on any particular topic (Copus, 2004: 198). This view is deeply embedded within Labour in particular. Jones (1969: 163-164), in his study of borough politics in Wolverhampton, records that at the turn of the 20th century, while Conservative and Liberal councillors formed fluctuating coalitions on different issues, Labour from its earliest days tried to act as a bloc.

Even if strong, individual leadership of the facilitative, networking type (Hambleton, 2015a; OECD, 2015a) is able to overcome all the obstacles in its path to becoming established, party politics is still likely to prove a continuing complicating factor. Borraz and John (2004: 116, 117) argue: 'Much depends on the leaders' ability to wield influence above party politics and to have a direct relationship to the citizens', and leaders must 'free themselves from their party's grasp and find new political resources to sustain them in office'; they have to build networks. One explanation for the difficulty in doing this is provided by Stoker and Mossberger (1995: 820) who argue in relation to Europe generally that partisan politics is more important in local politics [than in the US] and sometimes leads to ideological conflict between central and local government. Copus (2004: 1) and Newton (1976: 13-30) see municipal and parliamentary battles as part of the same political war.

Just where power lies today between mayors or leaders and their cabinets, backbench councillors in their scrutiny committees, and party groups is understandably an underresearched question in relation to the recently-established CAs. Analysis of the party group is one area where this study makes a valuable new contribution. As Copus says:

The party group has not been granted the attention it deserves in reviews of the organisation and management of local government. Part of the reason for overlooking the group system is that the group is not part of the formal political

decision-making process of local government, but it is widely recognised by members and officers as the place where council policy is set. (Copus, 2004: 92).

Mossberger and Stoker (2001: 812) argue that both local government and business possess resources needed to govern, and this is a reminder that leadership is needed in business as well as politics. In a study focusing on a business context at times of economic change, featuring the English West Midlands and the Italian region of Tuscany, Bailey *et al.* (2010: 462) define place-renewing leadership as 'a form of public-private strategic leadership that empowers institutional or social forms of decision-making to absorb and adjust...to path-breaking economic change'. But Tomaney (2016: 4-6) takes a more circumspect view of links between political leaders and business – at least, if the formal institutional framework is not right, which he believes is the case in England. He worries that contacts could come to resemble a typical American model, lacking transparency and scrutiny and favouring urban property interests. He regards vehicles for distributing resources from central to regional and/or local government, such as devolution deals and city deals, as 'secret deals' between business and political elites, and the whole system as regressive.

Business and political elites, and other groups, are brought together in a framework of 'new civic leadership' developed by Hambleton (2015a: 109-137) that will be used in this thesis to aid analysis of both the context and actors of regional/local governance. Hambleton sees local leadership as not confined to elected politicians but as involving four other groups as well – business, public administration professionals, trade unions and the voluntary sector.

These five leadership groups operate within the fourfold constraints of government, economics, socio-cultural influences and the environment. In this model **(Figure 2-1)**, government provides the national context and determines the degree of autonomy, including fiscal powers, available to regional/local leadership and the legal and policy framework in which they must operate. Economic logic dictates that areas must compete in the market to promote growth. Socio-cultural influences comprise the mix of people in the area and the values they hold. Hambleton identifies a tension between the economic and political objectives of governance resulting from these

influences, and also argues that environmental factors, the fourth side of the framework, have been neglected.



### Figure 2-1: Hambleton's model of place-based civic leadership

These forces do not disable local leadership but they do combine to limit it, and it is within these constraints that new civic leadership must operate. How it does so depends on personal qualities, institutional design, the task at hand and the context. Hambleton argues for transformational leadership in which leaders have the self-confidence to empower others – to exercise facilitative leadership, which he defines as 'shaping emotions and behaviours to achieve common goals' (2015a: 124). Such emotional engagement is aligned to place and identity which, like the environment, Hambleton believes has been seriously neglected. It is within this place-based context that the five groups of the new civic leadership operate.

Hambleton's model of new civic leadership is used in this study to aid thematic analysis of the context of North East England in Chapter 3 and narrative analysis of the roles played by each leadership group in the region's devolution story in Chapter 4 and Chapter 5. Additionally, the study examines the role played in the devolution debate by public opinion which, though not a leadership group or a feature of Hambleton's model, has a part in the story. Public consultation has a statutory role in the establishment of combined authorities, mayoral and non-mayoral (7.2). Public opinion will be seen in Chapter 5 as intrinsic to the Tees Valley mayor's leadership style as he shapes it, appeals to it, responds to it and uses it in support and justification of his policies and actions and as a political weapon against his opponents.

Hambleton's model is not a perfect fit for analysing North East England. Firstly, while Hambleton regards the environment as a non-negotiable 'good', the governance institutions of North East England treat it as an economic opportunity for the region's energy-related industries and as a challenge in making the region an attractive place to live and invest. Secondly, economics is not only one side of the framework within which activity takes place but also, in the shape of growth, the objective of that activity. Thirdly, Hambleton's discussion is in the context of local authorities rather than CAs and LEPs. Finally, while Hambleton's focus is public service innovation, this study deals with economic development, which involves just one aspect of such innovation. With these *caveats*, the model's focus on collaboration between civic leadership groups within a relevant framework of constraints makes it appropriate and helpful for this study.

The governance issues outlined above constitute a formidable list of issues to be dealt with by local authorities seeking devolution: the need for trust to be built and networks to be created; historical awareness and sense of community to be taken into account; technocracy to be avoided; suspicion of central government's continuing involvement, of directly elected mayors and of the risks of fiscal devolution to be overcome; the opaqueness of party group decision-making; and the 'problematic' nature of the scrutiny system. All these obstacles have to be tackled, and the ways this was done in the North East and Tees Valley are described in the case studies.

## 2.3 Conclusion

The literature reviewed in this chapter discusses and explains a wide range of issues, economic and political, which will be found in the course of this study to raise questions and play roles, often disruptive, in the governance of North East England. It provides the conceptual background an awareness of which is the first step towards

fully understanding events to be narrated in the case studies (Chapter 4 and Chapter 5) and the issues of economic development, accountability and inclusion discussed in Chapter 6 and Chapter 7, and thus towards answering the research questions.

The concept of the city as growth hub, or city-regional agglomeration, will be found to be a divisive issue, linked to concerns over fair shares of development for all and tension between place-based and spatially blind development within the region as well as between regions. This in turn will be discovered to lead to underlying strains between the regional and municipal responsibilities of council leaders with their roots in local socio-cultural identities and finding expression through formal institutions of governance. Identity will be found to be a feature of the region's public life which can either unite, as in Tees Valley, or divide, as in the North East. Problems will be found with accountability, scrutiny, inclusion, and the roles of party groups and business leaders, all discussed in the literature.

The next chapter will take the issues raised here one step further by describing the North East regional context in which these questions form and essential, if often unarticulated and perhaps even unrecognised, background.

# Chapter 3. The context of North East England: identity, culture and politics

Following the exposition of theoretical issues underlying England's devolution policies of the past decade in Chapter 2, this chapter takes the investigation of this thesis to its next stage by describing the regional context against which the events of the case studies were played out. It forms a bridge between the world of theories and ideas in Chapter 2 and their realisation in the actual events of North East England to be discussed in the following chapters.

North East England is analysed here thematically in terms of Hambleton's four-sided framework for civic leadership – its socio-cultural and political, economic, governmental and environmental framing. In doing so, the chapter places key concepts introduced in section 1.3 in their regional context – socio-cultural and political factors which form North East England's informal, and help shape its formal, institutions; and identity, which helps unite Tees Valley but divides the rest of the North East. It contextualises, too, issues raised in the literature, including entrepreneurialism and community values (section 2.1.3); and relations between regional/local governance and central government (2.2.1, 2.2.2 and 2.2.3).

In the socio-political context, this chapter discusses the meaning of identity in the region and finds a contrast between local rivalries at the municipal level and solidarity in the community. Economically, it discusses regional attitudes towards entrepreneurialism, aspiration, wealth and wellbeing, education and enterprise, and whether the region's economic performance is affected by the importance local people place on values other than economic development, like community and public services. In its examination of governmental framing, the chapter describes how the powers devolved to North East England's local government, as well as other regions, have changed over time and discusses what will emerge in Chapter 4 as the vexed question of funding. Environmentally, the chapter finds that in contrast to the approach of Hambleton (2015a) to the environment as a non-negotiable good (2.2.4), North East governance regards it as an opportunity to adapt traditional industries to new 'green' purposes.

This chapter finally places the region in its geographical and administrative context, with a description and history of its two LEPs and two (now three) combined authorities (CAs). This geography with its LEPs was created by the coalition government in 2010-2011, some five-to-six years before the devolution decisions which are a central focus of this thesis, and should therefore be seen as part of the established regional context by that time, albeit not one with which all regional actors were content (5.7).

# 3.1 North East England's framework for place-based leadership

This section, placing North East England in Hambleton's four-sided framework for civic leadership, deals first with the region's socio-cultural and political character, which together constitute one side, followed by its economic, governmental and environmental framing.

## 3.1.1 Socio-cultural and political framing

It is an easy mistake to think of North East England as homogenous. This was an underlying assumption of the Labour Government when it held a referendum in 2004 on the establishment of a regional assembly. How wrong they were. The region rejected the proposal by 78%-22% (Rallings and Thrasher, 2006: 928-930). The fact that even Labour, so closely identified with the voters of North East England, could be wrong about such a matter indicates how easy it can be to misjudge. According to the OECD, there is lack of common purpose in the region; a belief persists there is a strong regional identity representing a vital asset for the region as a whole, but: 'While there is a strong element of truth in this, the sense of identity is not mobilised for political objectives...Nor does the sense of common culture and history prevent serious rivalry between localities and between organisations' (OECD, 2006: 204).

So in North East England, what is identity, and what are its implications? Governments over the past 60 years have constantly amended the region's boundaries and even its name – sometimes the North East, sometimes the North (Board of Trade, 1963; NEPC, 1966; NRST, 1977). The separation of Tees Valley from the wider region in 2010 was just the latest change. A survey of North East voters after the 2004 referendum found regional and local layers of identity, with the latter undermining the case for a regional body:

There can be no doubting the degree to which respondents identified with their region – 87% said they felt 'very' or 'fairly closely' attached to the North East - but for many this was an overlap on top of their continuing, equally fierce loyalty to their own city, town or village. This was often articulated in concern that a regional tier of government would weaken rather than strengthen the influence of their own area. (Rallings and Thrasher, 2006: 934)

Interview evidence for this study confirms that however strongly attached people in North East England may feel to their region, local identity remains important too. A NELEP board member, talking of identity in the North East, said:

Traditional family structures and traditional communities have...everybody wedded to their town and having pride in their town and having a born hatred of the town three miles down the road, and if there's a job there you are still not going to move. (NELEP board member, private sector. January 2018)

In 2019, as the Prime Minister was promising the Convention of the North to 'do devolution properly' (1.2) a parliamentary petition calling for English devolution was amassing only 275 signatures nationally, including ten in North East England (Unboxed, 2019).

Sense of local identity can be an asset, supporting strong communities and effective civic leadership, as Hambleton (2015a) argues. But it can also present difficulties if it hinders the establishment of democratic governance over geographic areas that are simultaneously appropriate for place-based economic development. In North East England, it is argued here, there is a mismatch in one case (the North East), but not the other (Tees Valley), between the functional economic areas (FEAs) upon which the LEPs are based and the areas which form people's sense of local identity and which are therefore appropriate for political governance.

Following case studies that included NECA and Tees Valley, the National Audit Office (NAO) concluded that key factors for progress were a common sense of purpose, a history of joint working leading to sound working relationships, and clear and aligned geographical areas. TVCA, it found, built on established joint working arrangements. In contrast the NECA council leaders, in addition to funding issues, disagreed over the

requirement to have an elected mayor, and following the failed devolution process of 2016 relations between them deteriorated (Morse, 2017: 23-25).

In the North East, in contrast to Tees Valley, the pragmatism needed to share power on a consistent basis is lacking. Cooperation is intermittent and prone to break down, as when NECA split. True, rivalry does not always prevent co-operation when it is to mutual advantage, as the Newcastle Gateshead Initiative for place marketing exemplifies (NewcastleGateshead Initiative, undated). Nevertheless, that there are still contrasting identities between two such closely connected municipalities as Newcastle and Gateshead, their centres facing each other over the Tyne and linked by seven bridges, seems odd to outsiders. A former Newcastle Council leader told the House of Lords of incidents of non-co-operation by Gateshead and other local councils dating back to the 1960s. Actually, they can be traced to the middle ages (Page, 1891; Fraser, 1966: 257; Beecham, 2018: cols. 110-112GC).

So in spite of a shared regional identity, it is competing local identities that are stronger in the North East's town halls. Tales of rivalry are rife, and though they may appear petty to outsiders, the resulting divisions have serious consequences. The rivalry is most commonly seen in terms of Newcastle *versus* the rest, though with varying degrees of animosity, felt most strongly south of the Tyne due to the perception that Newcastle is the main beneficiary of development, with little spillover benefit for others. According to one interview account, the Newcastle leader spent one NECA meeting being shouted at for most of the two hours by a majority of his fellow leaders 'about how Newcastle always got everything and how their boroughs were being left behind and therefore they weren't prepared to countenance any money going to Newcastle through these arrangements' (North East councillor. March 2018).

Whether this resentment is justified is contested, and both views were expressed by interviewees. A study of one measure of investment, foreign direct investment (FDI), found that between 1997 and 2017 in the UK the benefits flowed overwhelmingly to core cities<sup>11</sup>, with Newcastle among those more than doubling their number of

<sup>&</sup>lt;sup>11</sup> Birmingham, Bristol, Cardiff, Glasgow, Leeds, Liverpool, Manchester, Newcastle, Nottingham and Sheffield.

projects, and large towns<sup>12</sup>, which include all the other boroughs in Tyne & Wear and Tees Valley, also being successful. But there was almost no growth for smaller places. North East England did, though, see FDI shared more equally between its core city and large towns than the UK average, with 113 projects to Newcastle and 256 shared between the large towns, a phenomenon accounted for by the region's relative strength in manufacturing (EY and Centre for Towns, 2018: 1-3, 16, 23, 24). An independent report on the North East's inward investment effort found that while it 'may be the case' that a city-region branding would help compared with a regional brand, it played down the significance. Anyway, it remarked without further explanation but implying the local rivalry issue, a city-region brand was 'undeliverable in the region' (NELEP, 2018e: item 6, p. 43).

Newcastle has so far received the largest share of the £120m North East Fund, which provides finance for small and medium-sized enterprises (SMEs). Figures in January 2019 showed that of 80 investments made, 39.5% by value and 47.7% by number were Newcastle-based 'but there are increasing numbers of investments in most local authority areas indicating uptake across the region' (NELEP, 2019b: item 4).

Rivalries between the North East's authorities were denied by hardly a single interviewee for this thesis, though a minority played them down, arguing that the region should not be seen as unique in this respect, or that some political horsetrading was to be expected. A NELEP official suggested that rivalry was sometimes played up by politicians in public, as they wanted to be seen to be fighting for their area, while in private they were supportive (Interview. January 2018). This is a point relevant to the debate about path dependence at the socio-political level. It suggests councillors are willing to support the economic measures necessary for renewal but wary of being seen to sacrifice local political interests.

The business community is well aware of the political rivalry and impatient with it, accepting pragmatically that Newcastle as regional capital is the brand to attract investment. One business leader said: 'There's no doubt that Newcastle's the regional

<sup>&</sup>lt;sup>12</sup> Population over 75,000 but not a core city.

capital. I think the quicker the region gets over that the better, to be honest, and not all parts of the region are equal' (Regional business leader. August 2018).

Media are aware of the tensions too. A North East newspaper, featuring the development of a tech hub in Gateshead, quoted an executive as saying they branded themselves as Newcastle, but as it expanded 'it is our responsibility to start saying Gateshead'. The paper commented: 'Newcastle may be well known for its tech scene but if everything goes to plan, Gateshead will surpass it as an international hub for emerging technology' (Manning, 2018) - as if the global tech world knew or cared on which side of the river a project was located.

Behind the pettiness are serious issues; the distribution of resources is an important matter, and there is a feeling in some quarters that their areas get less benefit from regional or sub-regional institutions than others. This is why it was so important to Sunderland when it got its new *Northern Spire* bridge, a highly visible £117m structure. The bridge exemplifies many of the issues which prove so troublesome in the North East: local identity, rivalry, jealousy, the 'fair-shares-for-all' question, prestige and perception. It brought civic pride and a royal visit (Sunderland Council, 2018a), but a former NELEP board member (interview. January 2018) doubted its justification on economic grounds and a regional business leader agreed:

Sunderland got the bridge to nowhere... because it was their turn, in inverted commas. Now what difference does that really make to the economy? (Regional business leader. August 2018)

Two politicians of different parties who both had a hand in approving the bridge when in government justified it because it opened up land alongside the A19 corridor, where an international advanced manufacturing park (IAMP) is now being developed (interviews. December 2017 and January 2018). It also contributes to a link between the A19 and Sunderland docks. But in the mistrustful atmosphere of North East politics, the industrial park too, with its position straddling the border between Sunderland and South Tyneside, raised suspicion; it was there so both councils could lay claim to it, commented a close observer (interview. April 2018). The industrial park and the bridge seem, intuitively, to make sense, located as they are close to the A19 and the Nissan car factory. But in the North East such suspicions are no surprise.

The Tyne & Wear Metro light railway is another grievance for Sunderland, where they still remember paying towards its construction in the 1970s and 1980s even though it did not initially run there. The Metro was extended to Sunderland in 2002, but the memory remains and makes Sunderland wary of any joint governance scheme that does not give it a veto. A neighbouring councillor said:

The thing that was said in Sunderland for years and years and years and years was we pay more for the Metro than anyone else...and it doesn't even come here. You could sit down and you could argue the massive economic benefits of the Metro system and the benefits that Sunderland derives from that...but the prevailing view was 'We pay for it but we don't get it'. (NECA councillor).

This is not denied in Sunderland, as confirmed in interview (Sunderland councillor. March 2018).

Such dissatisfactions can stem from perception. A concert hall and a footbridge across the Tyne were given by an interviewee as examples of investment in Newcastle, as well as sometimes in Gateshead, which supports impressive public buildings while equal sums invested in industrial sites in Sunderland or Tees Valley, while just as economically important, do not result in equally edifying, visible projects (Former RDA member. December 2017). Still today, more prestige developments for shopping, entertainment, conferences and science continue to come to the regional capital and its near neighbour (Whitfield, 2019; Wayman, 2019b).

This reflects another aspect of local rivalry: its ubiquity. Newcastle may be the main target of envy, but some in the region see Newcastle and Gateshead as joint chief beneficiaries of development, while in Tees Valley it is Tyne and Wear, which includes Sunderland, that is seen as dominating attention. Complaints arise in other geographic areas as well. One interviewee spoke out for rural areas on the region's western fringes which suspect they are losing out to the eastern coastal strip (Labour councillor. March 2018). Local jealousies are everywhere.

Rivalry and jealousy over resources are not confined to the North East. In Yorkshire, Gray *et al.* (2018: 152-153) found 'jam spreading' to appease all local authorities and, in the *Northern Powerhouse*, concern that other cities and non-metropolitan areas would lose out to Manchester. Amid such competition it is not surprising there is unease over any new institution that will have power over investment decisions, and even where it will be based, with an assumption during the devolution debate that the proposed metro mayor would be located in Newcastle, as the RDA had been.

In Tees Valley, Newcastle is regarded by some as a domineering member of the family, by others as a stranger who has little to do with them at all. Some south of the Tees identify with Yorkshire. Resentment against Newcastle is not universal in Tees Valley but it is widespread, and its roots lie in the fact that the former RDA was headquartered there. The rivalry surfaces at popular level, with Teessiders keen to distinguish themselves from 'Geordies' (Robson, 2018; Price, 2018a). Now that Tees Valley has its own institutions, and with its neighbours to the north divided, it sees itself as a flagship for the region. Its internal cohesion is a reflection of that quality which Rhodes (1996) identified as essential to good governance, built up over two decades: trust (2.2.1).

However, the division manifest within the region's town halls is only one aspect of political identity. Solidarity has its place too. On 8 July 2017, more than a decade after the closure of North East England's last deep colliery, an estimated 200,000 people crowded into the medieval centre of Durham City to greet Jeremy Corbyn, Labour Party leader, at the Miners' Gala. The annual gala, first held in 1871, has been dubbed 'the biggest demonstration of working class culture in Europe' and is emblematic of the trade union solidarity and community spirit which people in the region exhibited most conspicuously during the miners' strike of 1984-85 (Chronicle, 2013; Halliday, 2017; Welford, 2017). Their strength came from unity. 'Unity is strength' is a famous union slogan, and it is the tradition of union-based solidarity that was the foundation of the politics that has given the Labour Party unbroken or virtually unbroken control of most of the region for around a century. In March 2017, towards the end of the municipal year in which NECA and TVCA decided to accept or reject devolution deals, the political composition of the councils in North East England was as set out in **Table 3-1.** 

Council elections that May saw the Conservatives take control of Northumberland County Council for the first time since 1974, with 33 seats (one short of an absolute

majority), while Labour fell from 31 seats to 24. Labour lost 20 seats on Durham County Council but retained control with 74. A Conservative won the mayoralty of Tees Valley by a narrow majority.

These Labour setbacks came on a day when the party lost 382 council seats nationally and the Conservatives gained 563 (BBC, 2017; Northumberland County Council, 2017; Seddon, 2017; TVCA, undated-a), suggesting a contemporary reflection of the conflated parliamentary and municipal battles long a feature of English politics (Newton, 1976; Copus, 2004) (2.2.4). Elections for six councils in the region in 2018 saw virtually no change and the devolution issue had no evident electoral effect; Labour won 306 seats, suffering a net loss of two (BBC, 2018). Its dominance was reflected in parliamentary elections too; Labour won 26 of the region's 29 constituencies in the 2017 general election, with Conservatives confined to the rural and semi-rural fringes.

Council	Con	Lab	LibDem	Other	Vac.	Total
			NORTH EAST	· ·		
Durham	4	94	9	17	2	126
Gateshead		54	12			66
Newcastle		55	20	3		78
N.Tyneside	7	51		2		60
North'land	20	31	10	5	1	67
S.Tyneside		53		1		54
Sunderland	6	66	2	1		75
Total	37	404	53	29	3	526
			TEES VALLEY			
Darlington	17	29	3	1		50
Hartlepool	3	19		11		33
M'brough	4	31		10		45
Redcar &	9	28	11	11		59
Cleveland						
Stockton	13	30	1	12		56
Total	46	137	14	45		243

Table 3-1: Council membership by party. March 2017. Source: council websites

In 2019 however, with Brexit controversy at its height, there was a significant shift away from Labour. First, in council elections in May, the party's overall strength in Tees Valley was cut from 137 councillors in 2017 to 92 and it was left without an overall majority on any of the five councils. In the NECA area, however, in spite of losing 20 seats overall, it retained control of all five contested councils by majorities of between 27 and 42<sup>13</sup>.

Then, in the 2019 general election, significant Labour losses spread throughout the region. Of seven parliamentary seats in Tees Valley, the Conservatives added three to the one already held, in Northumberland they gained one in a former mining area, and in semi-rural parts of County Durham took three. But Labour retained all 12 seats in its Tyne and Wear metropolitan heartland, just as it continued to control all five councils there (Kelly, 2019; BBC, 2019d).

Low turnout may be a factor reinforcing Labour's hold. In the 2017 elections Northumberland had the highest turnout nationally, at 40.7%, and it was in Northumberland, alone in the NECA area, that voters enforced change, bringing the Conservatives to office. Turnout in County Durham by contrast was the third lowest of 36 local elections nationally, at just 31.3% compared with a national average of 35.1% (Electoral Commission, 2017a; Electoral Commission, 2017c: 5-6).

Dominance by a single party could, warned a specialist in regional culture interviewed for this study - a man with roots in the region and behind him a professional career in cultural administration which brought him into regular, close contact with councillors lead to complacency and perpetuate in-fighting: 'I think it certainly breeds a lack of challenge. It forces factionalism within the Labour Party, because that is where your fights are' (Retired public official. April 2018).

An alternative interpretation could be that voters' apparent lack of desire for political change in most parts of the North East is partly a reflection of a belief that their local government is well run. North East England performs well on the European Quality of Government Index, which ranks 202 EU regions according to public perceptions of institutional quality at regional level measured by quality of public services, impartiality of delivery and low corruption. North East England ranks second in the UK, one place behind the West Midlands, and 28<sup>th</sup> overall (European Commission, 2018).

<sup>&</sup>lt;sup>13</sup> Source: council websites

A business interviewee saw mayors as a solution to the turnout problem by offering the prospect of change (Former NELEP board member. January 2018). Unfortunately for this argument, elections for six metro mayors nationwide in 2017 brought turnouts even lower than for local councils, averaging just 27.8%. The Tees Valley mayoral turnout was lowest of all at just 21.3% (Electoral Commission, 2017b: 5). Nor was this simply due to the novelty of the metro mayoral office; the election for the wellestablished position of borough mayor of North Tyneside attracted a turnout of 34.3% - no higher than usual for council elections (Electoral Commission, 2017c: 6).

The outcome of political, economic and social attitudes in the North East has been that regional governance has in some respects stood still since the millennium. Robinson *et al.* (2017),following up a study of 2000, found that 67% of North East council seats were held by Labour compared with 66% in 2000, and councillors were even older than in 2000, with an average age of 60.3; over half were retired. Where there have been changes are in education, with over half now having university degrees, and the representation of women, who account for 43% of councillors, though still only two out of 12 council leaders/mayors. But black and ethnic minority (BME) communities and the disabled are under-represented and as a result 'the region's councillors still do not properly reflect the range and diversity of their communities'.

This socially, if not politically, unrepresentative group retains further influence via membership of a range of organisations that make up the 'extended world of local governance'. The authors conclude that 'local democracy continues to be at a low ebb, with disappointing turnouts at elections, a dominant political party subject to little challenge, and widespread public indifference'. A follow-up paper refers to a 'substantial democratic deficit' in the region (Robinson *et al.*, 2017: 6-7, 39, 142; Shaw and Robinson, 2018: 849).

The age of councillors is not irrelevant, for many of them remember the deindustrialisation of the 1980s, and – in powerful examples of the informal institutions at work in the North East and discussed in terms of institutional and evolutionary geography (2.1.2, 2.1.3) - this can make them reluctant still to do business with a Conservative government. Emotions aroused by those memories and expressed in interviews range from anger at police actions during the miners' strike to nostalgia for

the values and trade union solidarity of working communities. The regional cultural specialist quoted above reflected on the events of the 1980s and their lasting legacy, and warned against underestimating, still, 30 years later

the absolute loyalty to your marras [workmates] in the mine and on the shipbuilding site and in heavy industry...There is the sense that to do anything other than vote for the party that we created in the 20<sup>th</sup> century, that we created, I mean we the class, created... And then you get Thatcher. The thought that you would actually vote for them bastards who did this to us and our people, who never cared, who stripped the money from the land and took it to the south, the thought that you would vote for them is viscerally unthinkable still to a large number of people (Retired public official. April 2018).

A Labour MP has memories of shipyard communities of the same era. Reflecting on what comes across almost as a golden age, he remembers trade union leaders who not only looked after the material welfare of their members at work but set standards of behaviour. Mrs Thatcher had destroyed that sense of community and undermined the Labour vote by dismantling the large unionised workplaces and selling off the council houses (Labour MP. January 2018).

Nothing similar was said by any interviewee in Tees Valley. In fact they barely looked backwards at all beyond the SSI steelworks closure at Redcar in 2015 (5.1) and the era of RDA domination, as they saw it, from Newcastle.

In the North East, until the 2019 general election, opportunities to improve relations with Conservatives, at least to the pragmatic level seen in Tees Valley (Chapter 5), were severely limited by the Tories' electoral failure in most of the area. The two Conservative MPs were far away on the rural fringes, out of sight and mind of Labour councillors in their heartlands. Conservative control of Northumberland did not occur until after NECA had rejected devolution. Unmoderated by meaningful intercourse with their opponents, NECA had no friends looking after their interests in high places. Labour campaigners against the NECA devolution deal could see nothing at all in its favour and regarded the process as a cynical Tory plot to undermine real devolution. A campaigner said:

The Conservative government's policy, right across the UK where these mayoral combined authorities have been created and these metro mayors have been established - what you can see is an attempt to prevent genuine regional devolution from taking place by breaking up the identity of the region. (Labour activist. November 2017)

The view of the Labour MP quoted above was that the government had two political motives for devolution. One was to muddy the waters over local government funding by offering small amounts for devolution while taking much greater sums away from core services. The other in the mind of Conservative strategists was:

Let's offer them something they can shape themselves and let them quarrel over what the boundaries should be, how the money should be applied, what the functions should be. Leave the quarrelling to them and let's see what happens (Labour MP. January 2018).

The informal institutions discussed in this chapter take their place, tacitly but potently, in the consciousness or perhaps sub-consciousness of decision-makers, alongside formal agendas and reports, whenever important decisions are to be taken. But where the informal institutions are most influential of all is another institution, in this case formal, if not official. Decisions about important matters of public policy within the competence of local government are debated and determined not at council or cabinet meetings, where the naïve might assume power is exercised, but in the privacy of the party political groups on local authorities. Only afterwards are they exhibited with unanimous acquiescence, if not necessarily whole-hearted support, to the public gaze. So it was in the case of North East devolution, as we will see in 0. The process was different in Tees Valley where, true, the decision was taken in the Labour groups of four of the five councils - though in Redcar & Cleveland, where there was a minority administration, others had to be persuaded too – but the issue was not as contentious. The Tees Valley process is narrated in Chapter 5.

3.1.2 Economic framing: the economic consequences of social values This section describes North East England's economic situation, which forms a second side of Hambleton's framework, and discusses entrepreneurialism and the role of education. The emphasis here is on social factors affecting the economy rather than the economy itself, which is dealt with in Chapter 6.

Popular attitudes in North East England towards economics, and especially to the important question of entrepreneurialism, are difficult to gauge. Low levels of entrepreneurialism and accompanying, potentially countervailing respect for community values were noted by Hudson (2000b) and Robinson (2002) and remain relevant to the debate.

Huggins and Thompson (2017) and Huggins *et al.* (2018) have created 'psychocultural' profiles of regions, which combine community values with individual personality traits. They then compare the results with economic performance. Of 11 regions and nations in Great Britain, North East England comes top for social cohesion, and psychoculturally is the third most 'inclusively amenable', but ranks tenth for economic competitiveness. Being socially 'nicer' is thus likely to come at the cost of economic rewards, the researchers conclude (2017: 68, 69). Their work supports the finding of Angula-Guerrero *et al.* (2017) that individual entrepreneurial activity and attitude should be seen in the context of cultural and social norms (2.1.3).

Interviews for this study suggest the lifestyle aspirations of people in the region are often modest, once a satisfactory standard of living is achieved – defined by one interviewee as 'getting by' (Professional man, private sector, Newcastle. November 2017) or more precisely by the specialist in regional culture quoted above as 'as long as we can have our two weeks wherever, in Torremolinos, as long as we can have a car, as long as we can manage, that's fine' (Retired public official. April 2018). Young to middle-aged professionals told stories of coming from families where the height of ambition was to have a secure low-to-mid-level job with a large employer, and they believed such attitudes were still common (Professional man, private sector, Newcastle. November 2017; business support executive, County Durham. October 2018).

While the region has some of the greatest concentrations of deprivation in England, as we have seen in section 1.4.2, other statistics suggest a mixed picture for the region's prosperity. Overall, households in the region are continuing to get wealthier, albeit not as fast as the national average. While households in the region have less total wealth

than any other region of England, with a median figure of £165,200 in 2014/2016 compared with £268,100 for England, this was still an increase of £17,000 over 2012/14 - a rise of 11.4% compared with 18.5% for England - and reflects the fact that people have less of their money in property than elsewhere (26% compared with 37% in England) (ONS, 2018a: figs 11, 13, 22, 23).

North East regional households have proportionately more of their wealth in private pensions – the largest and fastest-growing component in national aggregate household wealth, following the introduction of automatic enrolment in workplace pension schemes in 2012 - than anywhere else in England (52% compared with 40%). And they suffer less under the burden of debt: the proportion of households in the region with problem debt fell from 7% to 4%, equal lowest in England with the South East (ONS, 2018a: figs 1, 11, 39).

Young workers are actually catching up with their counterparts elsewhere. Research published in 2019 by the Resolution Foundation found that millennials born in the North East region in 1986-1990 are progressing faster in the labour market than their counterparts in other regions when compared with their elders born in 1971-1975. The intergenerational comparison found that while the 1986-1990 cohort in the North East still has earnings, employment rates and degree attainment levels below national average, it is catching up. Pay is up 13% over the earlier generation compared with 3% nationally, employment rate up 4% compared with 1% and degree attainment up 99% compared with 62%. What is more, the region is benefiting from among the highest growth rates in both earnings and employment for all age groups since the early 2000s. And at the same time housing costs are up only 2% compared with 20% nationally (Gustafsson, 2019).

This picture of the region's growing wealth and prosperity is mirrored in its spending habits (ONS, 2018c). Though annual spending by people in the region is the second lowest at £15,727 per head, or £3,347 below average, this is substantially offset by the lowest housing costs, £2,077 below average. The region saw a larger than average increase in total spending (28% compared with England average 22%) between 2009 and 2016; in 2015-16 total spending per head in the region increased by 8.1%, the highest in the UK. Measured by spending rather than income, these figures suggest the

region's relatively subdued housing costs have enabled people to devote more to items like household goods and services, clothing and footwear, recreation and culture, and restaurants and hotels (Figure 3-1).



Figure 3-1: Change in national household final consumption per head by selected COICOP<sup>14</sup> commodities, England and North East Region, 2009-2016. Source: ONS

If lack of aspiration above a certain level may be a problem at the bottom-to-middle of the jobs ladder, modesty of ambition can be inhibiting higher up too, with entrepreneurs starting businesses then easing off when they achieve what they regard as a good standard of living, perhaps again reflecting the value attached to other aspects of life and a satisfaction with 'getting by', albeit at a higher level:

You can get to a level of business and have a really, really good lifestyle in the North East and not have to continue to grow your business; there isn't the same drive. We have a lot of what I would style as lifestyle businesses...Therefore even the businesses that do get started and do manage to scale to a level then tend to cap out...Is that lack of entrepreneurial spirit? I don't know. There's certainly a feel that there's a lack of drive to take some of these businesses to where they can be. (NELEP board member. January 2018)

<sup>&</sup>lt;sup>14</sup> Classification of individual consumption by purpose.

Another business leader agreed:

More and more people are wanting to start up their own. It's the level of growth; it's whether we have enough businesses that really want to take on the world... There are a lot of people who just want to have a lifestyle, and who is anybody to disagree with the intentions of whoever wants to run a business? (Regional business leader. August 2018).

North East England (including Tees Valley) does have the lowest number of businesses in England per head of population, but new business creation as a proportion of existing businesses is running consistently very close to or even above the national average. In 2017 the region's business birth rate as a proportion of existing businesses was fifth out of 12 UK nations and regions, at 12.6% compared with a national average of 13.1%. In the seven years 2011-17 it was above the UK average twice, equal to it twice and below three times, and of the eight English regions outside London (which is a top-end outlier), the region was in top place twice, top equal twice, third once and fifth twice (ONS, 2018d).

The problem for North East England is that these figures build on a low baseline. The number of businesses in the region rose during 2011-17 by 18%, equal second highest in England outside London, but its total number of enterprises was still considerably the lowest, with just 287 businesses per 10,000 population compared with an England average of 406 (excluding London) (ONS, 2018b; ONS, 2018d).<sup>15</sup> The region was running a little faster than comparable regions but was still a long way from catching up.

NELEP reports that in its area in 2018 there were 318 private sector enterprises per 10,000 population compared with 474 in England excluding London. If North East performance matched England's excluding London the area would have 25,500 more businesses. In 2017 there were 44 business births and 40 business deaths per 10,000 adult population in the North East compared with 65 births and 62 deaths in England excluding London. NELEP comments: 'Whilst the North East underperforms on each of

<sup>&</sup>lt;sup>15</sup> Author's calculation.

these measures, it has made progress on them since 2014 – suggesting that the business base is growing and becoming more dynamic' (NELEP, 2019e: 10).

This discussion suggest that the region's problem may indeed be a legacy of its history of large employers, reflected partly in a low baseline of private enterprises and partly in a comparative weakness in entrepreneurship. However, perplexity over the source of the problem is reflected in NELEP's answers to its own question: 'What next?' Its list of replies is headed by; '[I]mproving our understanding of the factors underpinning our low business birth rate' (NELEP, 2019e: 11).

There may still be a link to the number of people in North East England with large employers, but often now in the public sector. More people in North East England than nationally do work in the public sector – 29.2% compared with 25.1% in the UK in the public administration and defence; education; and human health and social work sectors. More also still work in manufacturing - 10.4% compared to 7.7% (Figure 3-2).



Figure 3-2: Workforce jobs by industry (selection) (SIC 2007). Source: Nomis

Perceptions of North East England's spirit of enterprise differ widely. Of 19 interviewees for this study who discussed entrepreneurialism, six stated that problems associated with the region's industrial heritage were history: 'Our industrial past is just a cheap excuse that's used to demonstrate why we haven't got more businesses' (Regional business leader. August 2018); 'a bit of an out-of-date caricature' (Public official, Tees Valley. January 2018). Another business leader was insistent that entrepreneurial spirit and community values were anyway far from mutually exclusive; many of his members running regionally-based businesses were intent on contributing to their communities by providing good-quality, well-paid jobs (Regional business leader. September 2018). Others – though fewer, on the evidence of these interviews – think the problem lingers, partly now centred on the public sector rather than industry. A digital entrepreneur said well paid jobs in government and academia were dampening entrepreneurial spirit (Professional man, Newcastle. November 2017).

Some look at the schools when seeking responsibility for the region's economic problems. Few interviewees blamed them directly, though a politician with national experience said: 'The North East needs to look at itself more closely than it does. Why are our secondary schools performing so much worse than London secondary schools?' (Liberal Democrat. December 2017). Others were more nuanced, concerned principally about lack of contact between schools and the world of work, with a consequent lack of preparation, expectation and aspiration among school leavers. While several saw the universities as a source of enterprise, no matching ambition was seen coming direct from secondary education.

Higher educational attainment is explicitly targeted in the Tees Valley SEP (TVU, 2014g: 6), and NELEP prioritises it too: its Skills Board is chaired by the NELEP chair personally. According to a NELEP interviewee, the best thing anyone could do for the North East economy would be to improve his or her skill level (NELEP board member, private sector. January 2018). A council leader said it was just not good enough that skills locally were a grade or two lower than in the South East (NECA council leader. May 2018).

These perceptions must be seen in the context of actuality. North East England schools rank only just below average for England, measured by the proportion of schools judged outstanding or good; it has 85% in these categories, ranking fifth equal of nine regions including London. The England average is 86% (Ofsted, 2018) (**Table 3-2**). There are some notable inspection results at local level: Gateshead has 38% outstanding schools, second highest anywhere in England outside London. A year earlier, however, before the introduction of a new methodology, the North East had had the lowest proportion of outstanding or good schools, at 67% (BEIS, 2017: 98, 230).

In spite of school performance, however, many parts of the region are dogged by poor levels of educational attainment. Pupil attainment measured by average grade per GCSE subject in secondary school is in the bottom 25% in England in all 12 local education authorities in the region except Gateshead and North Tyneside, and of those only North Tyneside is above the national average. The Education Policy Institute (EPI) notes that levels of educational disadvantage have become 'firmly entrenched' in some regions, of which it highlights the North East. Low performance, the EPI finds, is sometimes correlated with levels of persistent disadvantage, but not always. In London, for example, Tower Hamlets has many disadvantaged pupils in secondary schools but a relatively low attainment gap (Hutchinson *et al.*, 2018).

 Table 3-2: Overall effectiveness of schools inspected by region. As at March 2018 (new methodology). Source: Ofsted

 %
 Outstanding
 Good
 Requires improvement
 Inadequate

%	Outstanding	Good	Requires improvement	Inadequate
England	21	65	10	4
London	32	60	6	2
South East	21	68	9	2
East	18	68	10	4
West Midlands	19	66	11	5
South West	18	66	12	4
East Midlands	17	67	12	5
North West	23	65	10	3
Yorks & Humber	17	63	14	5
North East	23	62	12	3

Nor is the issue simply one of funding. When the Prime Minister announced in August 2019 that cash would be provided to raise funding per pupil to a minimum of £4,000 a year in primary schools and £5,000 in secondary schools, only three councils in North East England (Darlington, North Tyneside and Stockton) were not already above that floor for primaries and none were below for secondary schools though, as the EPI pointed out, these were averages for each local authority, so individual schools may have fallen below the baselines (Department for Education, 2018; Education Policy Institute, 2019).

According to NELEP, performance in its area's secondary schools is mixed, with poor outcomes in many and no clear pattern about what underpins this under-performance (NELEP, 2019a: 42). Research in 2019 provides a possible answer to the puzzle why pupil attainment in North East England should be relatively low when school performance is more or less up to standard. A report for the Northern Powerhouse Partnership shows that differences in average school performance scores between regions change dramatically once adjustments are made for pupil background, with North East England showing the biggest improvement. Schools in the North East are doubly disadvantaged under current performance measures by teaching not just relatively poor intakes [eligible for free school meals] but also disproportionately white British pupils, both of which characteristics are associated with below average progress (Leckie *et al.*, 2019: 1, 8).

This suggests, as the report says, that responsibility for improving performance should be rebalanced towards society and government rather than schools (2019: 2). As education and skills are such an important part of economic development, this supports the case for a broad, multi-agency approach to the economy, with a focus on motivating disadvantaged pupils to learn through support in their communities as well as their schools.

#### 3.1.3 Government framing

This section discusses North East England's governmental framing, including the hotly contested issue of funding, which forms a third side of Hambleton's framework.

It is central government that decides the size, shape, responsibilities, powers and functions of councils; it is central government which can, and does, abolish individual councils, or entire layers of local government, or create new types of councils when it deems that circumstances, or politics, or policy requires such change (Copus et al., 2017: 17).

These authors see local government in England as a 'constant tussle' between the pressures of centralisation and localism in the context of globalisation, urbanisation, Europeanisation, austerity, increased public demand and economic downturn. In spite of localism and devolution, they say, the direction of travel has been clearly top-down (2017: 2, 5-6). The constancy of this tussle is reflected by Ayres *et al.* (2017a: 2), who note that 'England has been a landscape of permanent administrative reconfiguration during the second half of the 20<sup>th</sup> century'.

North East England has experienced the effects of this central power on several significant occasions in the past half century. The Local Government Act 1972

established Tyne and Wear Metropolitan County Council, bringing together all or parts of 24 existing councils in five new metropolitan districts (National Archives, undated: Schedule 1). The Local Government Act 1985 abolished the metropolitan county council, though the five districts remain as unitary authorities (Copus *et al.*, 2017: 89). Teesside County Borough, bringing together areas of Middlesbrough, Stockton and Redcar & Cleveland, was in existence in 1968-74, and Cleveland County Council, covering the same area plus Hartlepool, from 1974 to 1996. Thirteen district councils in Northumberland and Durham were abolished in 2009.

Which powers are under local control is also largely in the hands of the centre, as remains the case with devolution to CAs. As Ayres *et al.* (2017a: 3) remark: 'Research has indicated that the scope of devolution was firmly limited in practice by government priorities' (2017a: 9): largely, but not wholly, for they also note, as we have seen (2.2.3) that some council representatives report the government was 'desperate' to do deals and did not hold all the cards. This is borne out by a government offer of more powers made to a NECA leader (4.5.1). Devolution deals have generally consisted of what Sandford (2016a) calls a 'menu with specials', with the core powers being further education, business support, the Work and Health Programme, EU structural funds, Transforming Cities funding, fiscal powers and planning and land use.

As well as determining their very existence and their powers/responsibilities, the government controls the funding of local authorities. As already discussed (2.2.2), it has retained control of resources even as it has decentralised responsibilities. The consequences of this in the North East will become clear when we see that 'unfair funding' was an important factor in NECA's decision to reject its devolution offer (4.1).

When the government established the Local Growth Fund (LGF) in 2013, it pledged to maintain it at £2bn a year to the end of the next parliament (Treasury, 2013: 57). The White Paper *Investing in Britain's Future* in 2013 put 'resources under the strategic influence of LEPs' at as much as £20bn in the years to 2021, including £5.3bn EU funding (Treasury, 2013: 57). The £20bn overall figure was described by Martin *et al.* (2016: 353) as 'a significant move towards fiscal devolution' but still 'far short of the scale required'.

LGF resulted from a report commissioned by the government from Lord Heseltine. Money was to be allocated to LEPs on the basis of local growth deals and covering the six years 2015-2021. The North East region fared relatively well; NELEP received £379.6m and Tees Valley £126m in tranches. On a per capita basis both were in the second highest tier of LEPs (Ward, 2019a). However, these sums included some previously-committed projects, and NELEP had £270m actually available while Tees Valley's £90m first tranche included £18m previously committed (Cabinet Office *et al.*, 2014; NELEP, undated-f).

For North East England, EU funding was another important element of the package. The North East was allocated £539.6m for 2014-2020 and Tees Valley £202.6m (BIS, 2013b). The awarding of EU money to programmes and projects is controlled by the government, with advice from local committees in each LEP area (TVU, 2016b: Annex A).

It is a common perception that LEPs and CAs are poorly funded compared with RDAs. Not one interviewee for this study said otherwise, and RDA finances are discussed briefly here because a tendency to compare LEP funding unfavourably with RDA funding contributed to the background of dissatisfaction with funding generally against which NECA decided to reject its devolution offer.

One North East invested £3bn over 13 years (ONE, 2012: 6), an annual average of £230m a year, and in addition managed the European Regional Development Fund (ERDF), which brought £318m to North East England in 2007-13 (ONE, 2011: 27), an average of £45m a year, bringing the average annual total under the RDA's management in that period to £275m. However, following the global financial crisis of 2008 RDA finances were already being reduced under the Labour Government, collectively down from £2,263m in 2009/2010 to £1,178m in 2010/11 (Ward, 2019b: 19).

If the ambitions of the North East and Tees Valley strategic economic plans (SEPs), published in 2014, had been met, the region as a whole would have exceeded investment by the RDA. NELEP's SEP alone envisaged total spending of up to £1,592m over six years to 2020/21 (NELEP, 2014a: 87), an average of £265m annually. But the figures were only best estimates and dependent on the LEP's partners' budgets, as the

SEP acknowledged (2014a: 86). Actual spending turned out much less, and future plans are commensurately modest.

The North East invested £127m in 2015/16, £116m in 2016/17 and £111m in 2017/18. However, over the three years these sums included £39m, £37m and £26m for renewing the Tyne and Wear Metro, a responsibility which did not fall to the RDA (NECA, 2016c; NECA, 2017a; NECA, 2018l). Tees Valley Invested £44m in 2016/17 and £51m in 2017/18 (TVCA, 2017f; TVCA, 2018a). Total investment in the region therefore peaked at £162m in 2017/18, though without Metro renewal this was £135m – almost exactly half the annual average under the RDA.

Going forward, in early 2019 the rump NECA was anticipating capital spending of £134m in 2019/20, £114m in 1920/21 and £255m in 1921/22. However, these comparatively large sums are heavily weighted towards transport, and particularly the Metro, reflecting the fact that following its split from North of Tyne, NECA was tasked as the accountable body for transport covering both CAs. Furthermore, to add to the ongoing Metro asset renewal programme, a new fleet of trains was due to start being paid for from 2019/20, rising to a peak of £184m two years later. Excluding these Metro-related sums, and with LGF funding due to end in 2020/21, NECA's capital budget was set to fall to £70m in 2019/20, £46m in 2020/21 and £36m in 2021/22 (NECA, 2019b: 3) unless new funding was announced.

Meanwhile, in North of Tyne it was possible by early 2019 to begin to see the funding effects of its devolution deal. NTCA would have its £20m annual devolution grant coming in but as yet few projects ready to spend it on, with planned investment of £4m in 2019/20, £9m in 2020/21 and £15m in 2021/22. The result, even without anticipated funding from other sources, was expected to be an investment reserve of £50m by 2021/22 (NTCA, 2019a: 11).

In Tees Valley, its £15m annual devolution grant was being supplemented by a series of other government grants, most significantly by £71.5m over five years from the Transforming Cities Fund (TVCA, 2018i: 4-6). Its ten-year investment plan for 2019-29 envisaged spending of £588m, but this was heavily front-loaded and reduced sharply as LGF and EU funding ended and some grants tailed off. Investment in the three years 2019-22 was planned at £107m, £69m and £84m, though by 2025/26 – with new

funding sources including the proposed UK Shared Prosperity Fund (EU funding replacement) still undetailed - investment of only around £31m a year could be confidently anticipated (TVCA, 2019a: 29).

In spite of the uncertainties and the effects of factors such as front loading in Tees Valley, the lack of ready-to-go projects in North of Tyne and its resulting £50m accumulating reserve, lack of information about future funding including the UK Shared Prosperity Fund, and the Metro effect, it is clear that in 2019-22 Tees Valley, the smallest of the now three parts of North East England, is investing more in its economy than NECA and NTCA combined - £260m compared with £180m (Metro excluded). The *Northern Powerhouse* Minister was correct when he said that: 'It's clear that mayoral combined authorities, with that single, accountable, electable individual, are drawing more money and more power down from government than other areas' (Walker, 2018ak).

The sums discussed above are not definitive. While based on official documents, and thus not mere estimates, they are not always the only documents available. As noted above, for example, there are different versions of the amounts of LGF allocated to the North East and Tees Valley LEPs. NELEP's AGM in September 2019 was told growth deal investment for its area, including public and private match funding, was £672m and enterprise zone investment was £167m (Woods, 2019). Several scholars have reported that certainty on funding for regional development is elusive and subject to interpretation (Pike *et al.*, 2015: 12, 15; Pike *et al.*, 2017: 267; Sandford, 2019a: 113-115). Similar problems have been found in the US by Storper *et al.* (2015: 114-116). Some documents, such as RDA reports, have been archived and those with direct knowledge of them have dispersed.

The sums allocated to LEPs for regional development form only a small proportion of total public spending in the regions<sup>16</sup>. Total public expenditure on services in North East England was over £24bn in 2014-15<sup>17</sup>. Total public spending in the region has been consistently above the national average for many years, though there has been a

<sup>&</sup>lt;sup>16</sup> All public spending figures in the remainder of this section are taken from HM Treasury's Public Expenditure Statistical Analyses, Chapter 9, published annually and available at https://www.gov.uk/government/collections/public-expenditure-statistical-analyses-pesa
<sup>17</sup> Where 2014/15 statistics are used it is because they were the most recent available when councillors were debating their decisions on devolution.

slight relative decline. When One North East was producing its final regional economic strategy (RES) in 2005/06 indexed expenditure per head (i.e. UK=100) was 108 in the North East region. This declined to 107 in 2010/11, when the Labour Government gave way to the coalition and austerity took hold, and to 105 in 2014/15, though in 2016/17, after the devolution decisions, it rose again to 106. **Table 3-3** shows areas where spending in the North East region was above the level for England during at least part of the decade preceding devolution<sup>18</sup>.

SPENDING INDEX							
	2005/06		2010/11		2014/15		
	N.E.	England	N.E.	England	N.E.	England	
Total economic affairs	106	93	90	90	84	90	
Public order and safety	102	99	103	98	105	97	
Housing etc.	105	87	123	91	128	85	
Health	108	98	107	99	107	99	
Recreation etc.	120	89	110	87	89	86	
Education	108	98	103	100	98	99	
Social protection	111	97	112	98	112	98	
All spending	108	97	107	97	105	97	

Table 3-3: Identifiable expenditure on selected services per head, indexed (UK=100). Source: Public Expenditure Statistical Analyses (PESA)

The region's slowly declining share of spending was in the context of a declining total under austerity. Spending per head in real terms<sup>19</sup> in England as a whole rose to £9,116 in 2010/11 but then fell steadily to £8,645 in 2014/15, a reduction of 5.1%. For the North East region, affected by the double decline of total and share, the fall was from £9,979 to £9,355, a 6.2% cut (Treasury, 2011: table 9.4). The region thus suffered a bigger-than-average cut but was still receiving a larger-than-average total by almost  $£1.2bn^{20}$ .

However, the extra money was not going to activities to boost economic performance but rather, largely, to mitigate the effects of under-performance. In 2014/15 North East England's indexed figure was 112 for social protection, including pensions,

<sup>&</sup>lt;sup>18</sup> International services and defence are omitted.

<sup>&</sup>lt;sup>19</sup> Real terms figures are the nominal figures adjusted to then current price levels.

<sup>&</sup>lt;sup>20</sup> Author's calculation.

benefits and personal social services, the highest proportion in England, and by 2016/17 it had risen to 114.

**Figure 3-3** shows how even the economic affairs budget was being used to offset failure rather than promote success. The overall economic affairs budget declined from £1.544bn in 2005/06 to £1.529bn in 2010/11 and more sharply to £1.356bn in 2014/15. Within that total, spending on employment policies was consistently high compared with the UK average, reflecting not so much a welcome relatively high level of activity as an unwelcome need for it due to relatively high unemployment; indexed *per capita* spending on employment policies was comfortably the highest in England every year between 2005/06 and 2014/15. Meanwhile spending on enterprise and economic development, needed to boost the economy, plummeted following the 2008 financial crash; capital investment fell by more than half between 2006/07 and 2010/11. Other spending that would have helped the region, on science & technology and transport, remained consistently below average. Education, another budget essential to the economy, also suffered a decline, down from 108% of UK average in 2005/06 to 105% in 2010/11 and 98% in 2014/15. In 2016/17 it was down to 96%.

For council leaders whose main job is to lead their individual councils, local government spending is especially important, as will become clear when the question of 'fair funding' emerges as a fatal stumbling block for NECA on the road to devolution (Chapter 4). As with some other budgets, noted above, North East England consistently received more than its equal, if not its fair, share for local government, on a fluctuating but generally downward trend from 108% to 105% of UK average over the decade to 2014/15. It remained at 105% it 2016/17. An above-average starting point, however, does not make the pain of above-average cuts seem less. Northern primary urban areas<sup>21</sup> in general, including in the North East, suffered cuts up to almost twice the Great Britain average between 2009/10 and 2017/18 – down 26.6% in Newcastle, 21.1% in Sunderland and 17.8% in Middlesbrough, compared with 14.3% for Great Britain (Centre for Cities, 2019a: fig. 5). Gray and Barford (2018: 553-554) find that the biggest spending cuts have been in cities most dependent on government grants including Newcastle and Middlesbrough.

<sup>&</sup>lt;sup>21</sup> For a discussion of economic geography see section 6.1.



Figure 3-3: Identifiable expenditure on economic affairs, per head, indexed. Source: Public Expenditure Statistical Analyses (PESA)

The way local politicians think about figures like these is illustrated in election literature. A Labour Party newssheet delivered to homes in Newcastle referred to reductions in grants to the city council like this: 'The Tories and Lib Dems have stolen £2,481 from every household in Newcastle since 2010' (Labour Party, 2019a).

The statistics enable some trends to be discerned. Overall, the North East region receives a larger than equal share of public spending, mainly due to its need for social protection. Investment in success as opposed to ameliorating failure took a hit after the financial crisis. Local government funding, starting from a relatively high level, has been cut more than average but remains above average. The region has fared relatively well in winning slices of UK growth funding and EU funds but the cake is only about half what was available to One North East. Finally, following its devolution deal, Tees Valley has proved more able to win government support than the North East, but even there future funding is uncertain following the ending of LGF and EU funding.

## 3.1.4 Environmental framing

This section discusses how North East England deals with environmental issues constituting the fourth side of Hambleton's framework. This is the only side which Hambleton regards as non-negotiable because to ignore it would be irresponsible (Hambleton, 2015a: 113). North East England's approach to environmental protection
is different, however, reflecting its industrial base. Sustainability and the environment have their places in the North East and Tees Valley SEPs, but are regarded as opportunities rather than limitations, in the form of developing renewable energy technologies, the offshore, sub-sea and low carbon sectors, and the circular economy (TVCA, 2016a: 16-17; NELEP, 2017c: 20). NECA is engaged in work on sustainability in both transport and urban development (NECA, 2016q: 1-8; NECA, 2017c: 131-136). For NELEP, sustainability is a means of supporting the economy; its vision is stated as 'to rebalance the economy and create Europe's premier location for low carbon, sustainable, knowledge-based private sector-led growth and jobs' (NELEP, undated-a: 1). It sees potential to gain advantage from new investments in offshore wind, electric vehicles, micro generation and research and innovation on low carbon technologies (NELEP, 2014a: 29). According to the 2014 version of the North East SEP, the investment focus was on 'creating a distinctive area that supports sustainable growth, environmental sustainability and resilience'.

NELEP also recognised the need to strengthen the cultural and visitor offer with projects such as a convention centre, museum improvements and investment in coastal towns and rural 'assets', but again the purpose was not so much environmental as economic: 'There is real potential through projects like these to increase the number of visitors to the area, specifically international and businesses (sic) visitors, and to increase significantly the levels of visit spend'. Good housing was 'essential for creating sustainable communities which will attract and retain economic investment and skilled workers' (NELEP, 2014a: 12, 13).

North East England contains a national park, nine areas of outstanding natural beauty and two world heritage sites. One North East made significant use of these environmental assets to promote the region, featuring a full-page map in its first regional economic strategy, which was littered with photographs of regional attractions (ONE, 1999: 15). The North East SEP 2014, focused on its economic task, in contrast, made barely any mention of the natural environment, and when it did so it was in terms of the rural economy (NELEP, 2014a: 28). Tables and figures took the place of the RDA's photographs. When photos made an appearance in later documents, they showed the workplace (NELEP, 2018b).

The revised SEP published in 2019 gives greater prominence to the area's environmental assets, and without always linking them so closely to economic objectives. Environmental assets take the lead position in the SEP's section on 'Our offer', pointing to the 'high quality living environment with historic towns and cities, a varied landscape and coastal environment and strong cultural and leisure offer', accompanied by references to attractions including the world heritage sites and national park. The SEP features photographs of local attractions, both natural and cultural. Nevertheless, the economic message is still present: 'The North East is a great place to live, learn, work and do business' (NELEP, 2019a: 4, 12-13).

North of Tyne (NTCA), home to the region's national park, makes somewhat more of its environment, referring to its 'stunning natural landscapes' (NTCA, 2017e). It is possible that green issues will receive more focus as a result of the NTCA devolution deal, which assigns it the role of 'national rural exemplar', though the emphasis is on rural business growth and productivity (MHCLG, 2018c). A rural business growth investment fund was one of the first items on NCTA's agenda (NTCA, 2019b: 2). Meanwhile, more ambitiously, a proposal was submitted to the government for a Borderlands Inclusive Growth Deal to build on connections between Northumberland, Cumbria, Carlisle and southern Scotland, which won a £260m UK government grant in March 2019 towards total funding of £394.5m (Borderlands Partnership, 2018; NELEP, 2019a: 11; Walker, 2019e). A councillor told this study: 'It's up to us from Northumberland to come up with those rural programmes that will make a real difference to our county and our rural areas' (Northumberland councillor. May 2018).

Tees Valley is conscious of its environment, and focused on its quality. 'Place' is one of its six strategic priorities. But again there is an economic purpose, and it is the manmade environment that is the main concern of its SEP. Place is said to be 'central to the creation of a diversified and inclusive economy', with vibrant town centres, housing offering affordable and aspirational choices and a commercial property market to support new investment. Culture too is to contribute to diversifying the economy. The aim is an environment that offers a comfortable work-life balance, where health and wellbeing flourish and businesses want to invest. Clean technology at work, better housing and dealing with deprivation and social exclusion are the objectives (TVCA, 2016a: 31).

TVCA's 2019 Ten-Year Investment Plan includes both place and culture & tourism among six priority growth-generating themes. Of £588m to be spent by 2029, place is allocated £50m and culture & tourism £60m. The area is bidding to be City of Culture 2025. Up to £20m is allocated for Hartlepool waterfront and the same for Darlington and Stockton railway heritage (TVCA, 2019a).

# 3.2 North East England's administrative geography

North East England is the smallest of the nine English regions which formed the basis for economic development until the Labour government gave way to the coalition in 2010. It has a population of 2,644,000 (Nomis). Localism saw the region replaced by two functional economic areas (FEAs), the North East and Tees Valley, each with a LEP and later a CA. In 2018 the North East split again as a result of local political disagreement into two areas – North of Tyne, and the south of Tyne authorities which retained the name North East. Each has its own CA but both continue to be served by a single LEP (NELEP).

### 3.2.1 North East and North of Tyne

The North East **(Figure 3-4)** was the larger of the original two CA areas, with a population of 1,972,000 (Nomis). Until it divided in 2018, it covered 3,200 sq. miles and encompassed in Newcastle one of Great Britain's ten core cities, the mid-sized city of Sunderland (though actually almost the size of Newcastle in population) and the smaller city of Durham, plus a network of towns and smaller settlements including post-industrial, coastal and remote rural communities in the counties of Durham and Northumberland. There are large and international university populations in the main cities. The economy is underpinned by natural, cultural, sporting and heritage assets (Bolton and Hildreth, 2013; NELEP, 2014a; Core Cities UK, 2017). Administratively it consisted of seven local authorities – the five metropolitan councils in Tyne & Wear (Gateshead, Newcastle, North Tyneside, South Tyneside and Sunderland) and the counties of Durham and Northumberland.

#### Local Enterprise Partnership (LEP) Areas



Figure 3-4: England's LEP areas, with the North East featured. Source: Northumberland County Council

The North East Combined Authority (NECA) was established in April 2014 as a legal body, headed by a leadership board consisting of the leaders (including one elected borough mayor) of the seven councils who also sat on the board of the North East Local Enterprise Partnership (NELEP). All were Labour Party representatives until 2017, when Conservatives took control of Northumberland County Council. One, the borough mayor, is a woman. The chair of NELEP sits on the NECA leadership board, without a vote. NECA holds leadership responsibility for three of the six themes in the area's strategic economic plan (SEP) – economic development and regeneration; employability and inclusion; and transport. Partnership working was affirmed as an integral part of NECA's formal arrangements, and representatives from NELEP, North East England Chamber of Commerce (NEECC), CBI, Federation of Small Business and TUC, alongside councillors, form the membership of NECA's economic development and regeneration advisory board (EDRAB) (NECA, undated-c).

In 2016 NECA voted 4-3 in effect to reject a devolution deal offered by the government, a decision that is a central focus of this study (Chapter 4). This led during 2017-18 to a process which resulted in the three councils which had supported the deal, all located north of the Tyne – Newcastle, North Tyneside and Northumberland – going through a process of secession from NECA in order to establish the separate North of Tyne CA (NTCA) and sign their own devolution deal. NTCA came formally into existence on 2 November 2018.

The rump NECA, consisting of the four south of Tyne councils of Durham, Gateshead, South Tyneside and Sunderland, remains the largest of the region's three CAs in population with 1,152,000 residents (Nomis). It has similar internal governance arrangements to its predecessor, including a leadership board consisting of its four council leaders, all Labour, with the NELEP chair as a non-voting member. All are men. It retains an economic development board (EDRAB) with the same external representatives as previously (NECA, 2019a).

When North of Tyne split from the North East it took 819,000 of the population with it (Nomis). Containing the large rural county of Northumberland, it covers 2,014 of NECA's original 3,200 sq. miles. NTCA is governed by a cabinet headed since May 2019 by a metro mayor and consisting of the leaders and deputy leaders of Newcastle and Northumberland and the borough mayor and deputy mayor of North Tyneside, plus a non-voting representative of NELEP. Five of the politicians, including the metro mayor, are Labour and two Conservative. Two are women. The members' portfolios reflect the CA's responsibilities: business competitiveness, employability and inclusion, housing

and land, economic growth, place and productivity, and education improvement (NTCA, 2018c). NTCA has an inclusive economy board and a housing and land board (NTCA, 2018b).

NELEP dates from 2011. By its constitution it is led by a board of 18 of whom nine, including the chair, are from the private sector, seven from the local authorities and one each from higher and further education. The local authority representatives are the leaders of the seven councils. In 2019, five of the board were women (three from the private sector, one from further education and the borough mayor) and two, both private sector, were from ethnic minorities. NELEP's private sector members are recruited to reflect the geography of the NELEP area, key sectors and different sizes of business operation; they may serve two three-year terms and are unpaid. Unlike the RDA before it, NELEP has no guaranteed place for the trade union movement and had until early 2020 none for the voluntary sector (NELEP, 2018e: item 9, p. 1; NELEP, undated-a: 1-2; NELEP, undated-d).

#### 3.2.2 Tees Valley

Tees Valley (Figure 3-5) is the smaller of the LEP areas, with a population of 671,000 and covering 307 square miles (Demographia, undated; Nomis). It is polycentric, with small towns in Darlington, Hartlepool, Middlesbrough and Stockton, while Redcar & Cleveland is semi-rural. The area lies between County Durham and North Yorkshire and bears the visible signs of industrialisation in the 19th and 20th centuries by the steel and bulk chemicals industries; but it benefits from the North Yorks Moors National Park on its doorstep.

Tees Valley's institutional arrangements are significantly different from those of the North East and North of Tyne. Tees Valley's LEP, originally known as Tees Valley Unlimited (TVU), was among the first wave of 24 LEPs announced in 2010, and had pre-existing institutional roots in collaboration between the area's local authorities and business community dating to 1996 (TVCA, undated-c). It was, a member told this study, 'a LEP before LEPs were ever invented' (TVLEP board member, private sector. March 2018). It was an example of a business-based, cross-sector collaborative

institution of the type identified by Burcher and Mayer (2017) as contributing to social capital and regional economic dynamism.



Figure 3-5: Tees Valley identifying its five boroughs Source: Tees Valley Combined Authority

TVU held its last separate meeting on 27 January 2016, before effectively merging with Tees Valley Combined Authority (TVCA). The intention, members agreed, was that the two bodies would work together, providing a strong voice for the area. TVCA would include the five local authority leaders (including one borough mayor) plus the LEP chair. Other LEP members would be in attendance. The TVCA Board and LEP would be one meeting, with TVCA-specific and LEP-specific responsibilities – one meeting but divided into two. Later that year TVCA accepted a devolution deal, and its first metro mayor was elected on 4 May 2017 (TVU, 2016a: 4) – a Conservative to head a cabinet consisting of five Labour council leaders. The metro mayor and four of the cabinet were men. In autumn 2018 three women from the private sector were appointed to the LEP board to increase diversity, following which the board had nine men and five women; none of the TVCA or LEP members was from an ethnic minority (TVCA, 2018h; TVCA, undated-d). Since elections in 2019 the cabinet has consisted (in addition to the

mayor) of one Conservative, one Labour member and three Independents. Two are women (TVCA, 2019e).

# 3.3 Conclusion

This chapter has taken an essential step towards explaining why NECA and Tees Valley took the contrasting decisions they did on devolution, in answer to research question 1, by placing key concepts such as formal institutions and the informal socio-cultural and political institutions which shape them, and issues from the literature such as governmental relations and entrepreneurialism, in their regional context, where they exert profound influence on decisions-makers.

Placing North East England within Hambleton's four-sided analytic framework for place-based leadership, the chapter has found that socio-culturally, while the region shares community values based on its legacy of heavy industry and trade unionism, at the political level local identity trumps regional identity, leading to rivalry and mistrust. Tees Valley, however, is internally cohesive, though political tensions are mounting following the election of a Conservative mayor and Labour losses in the 2019 council elections.

Economically, the region presents a confusing picture in relation to its spirit of enterprise. Some worry that it lacks entrepreneurial spirit and there is a willingness to settle for 'getting by', based on secure employment or a lifestyle business. Others, however, think the region is as enterprising as elsewhere. These findings suggest that more research is needed into factors affecting entrepreneurialism in the region, as NELEP recognises (3.1.2).

Governmental control over most public funding available to the region has resulted in its receiving more than its equal share of the national cake, but only because it needs it to soften the effects of its economically lagging position. Under austerity, the region's declining share of a declining total has resulted in less being available for development. Environmentally, the region's SEPs see the demand for sustainable technologies as opportunities for local industry, while the purpose of maintaining and improving the natural and built environment is to contribute to growing the economy.

The following two chapters will bring this contextual investigation to its fulfilment by analysing in narrative form why, against this background, the two areas took the decisions they did, assessed at the critical realist levels of the actual and the real. In Chapter 4 we will see how socio-cultural-political factors, resistance to the concept of an elected mayor and discontent over funding combined to bring about NECA's rejection of the government's devolution offer and the CA's consequent split into two.

# Chapter 4. NECA: a failure of institutional adaptation

This case study develops the key concepts of section 1.3 and builds on the economic theories and political issues raised in Chapter 2 and contextualised regionally in Chapter 3 to provide a narrative answer to the first research question as it relates to the North East (as opposed to Tees Valley), asking why the area rejected a devolution deal. It describes the roles of four of the five groups identified in Hambleton's analytic model of civic leadership – political, business, public service and trade union leaderships – in NECA's decision first to seek, then to reject, a devolution deal with the government. The fifth leadership group in Hambleton's model, the voluntary sector, had a marginal role. The outcome was that the NECA area, already separated by government decision from Tees Valley, was split again along the Tyne.

The chapter's narrative accords with critical realism's levels of the actual, recorded largely in official documents; the empirical, as those events were interpreted through interviews; and the real, which analyses how actors, perceptions and circumstances interacted to determine events.

The events are first described at the critical realist level of the actual, as they were enacted largely at what Ayres *et al.* (2017b) call 'front stage' and recorded in official documents, though supplemented by interviews describing 'back stage' events. The following two sections describe the roles of the business and public service leaderships. There follows a section on a campaign against the devolution deal, carried on predominantly, perhaps even wholly, within the Labour and trade union movement. The focus then turns to the political leadership, now at the 'back stage' level of the real, enacted mainly in the privacy of party groups where the pressures exerted by the other leadership groups and campaigners were resolved into decisions.

The narrative sees the formal institutions introduced in section 1.3.1, NECA and NELEP, fulfilling their roles as decision-makers and would-be influencers respectively. Informal institutions, the region's socio-cultural and political traditions, also introduced conceptually in section 1.3.1 and further discussed in the regional context in section 3.1.1, are seen playing their full part in the real-world decision to reject devolution.

Political party groups, discussed theoretically in the literature (2.2.4) are found to have a crucial role.

### 4.1 From deal to no deal in 11 months

Ostensibly, NECA valued unity above all else: 'United in purpose, boldness of vision and determination' headlined NECA's first constitution. NECA, it said, builds on a strong track record of joint working, and provides an accountable and stable platform to take on more powers, responsibility and funding from the government (NECA, 2017d). Yet when faced with its most important institutional decision, it split in two in an extended crisis played out in low-key public leadership board meetings and heated private arguments over much of 2016. Now that NECA has split, the constitutional vision statement of its early days has been dropped (NECA, 2017d; NECA, 2018c).

NECA's early existence was marked by optimism, notably expressed by its thematic lead for transport. He and leading officials told NECA in July 2014 that the new body offered 'the opportunity to influence our connectivity to the rest of the UK and the world, to attract new investment, and to ensure that transport makes a strong contribution to sustainable and inclusive economic growth – helping us to realise our aspirations for communities, the environment and the economy'. In a reference to Chancellor George Osborne's Northern Powerhouse speech, they remarked that 'the inception of NECA, combined with the government's acknowledgement of the limitations of our current transport network, offers a unique opportunity to make the case for a step change in our transport connectivity'. NECA would have a 'substantial' budget for public transport, with the prospect of further devolved funding. A government announcement of Local Growth Fund (LGF) allocations added to the enthusiasm: it was 'gratifying' that the North East had been one of the big winners, and in terms of transport had received almost everything it had bid for (NECA, 2014c: 2). NECA's overview and scrutiny committee (OSC) congratulated officials on securing the third highest LGF award (NECA, 2015f: item 3, p.3). A NECA council leader told this study:

Looking back for the first year or so there was quite a lot of optimism about what we might achieve. But then, as always with these things, it came down to relationships (Council leader. March 2018). In June 2014 NECA wrote to the government 'to encourage further and greater devolution'. That October it responded positively to '*Northern Futures'*, a government initiative to devolve more decisions and 'create an economic core in the heart of the region [northern England] that can compete with the biggest cities and regions in the world' (NECA, 2014a: 13-24; NECA, 2014b: 45; NECA, 2015b: 6).

In December, as debates were stoked over devolution in the rest of the UK following Scotland's decision in a referendum to reject independence (Cameron, 2014; Colomb and Tomaney, 2016: 1), and in response to a challenge by Osborne to city regions to come forward with proposals, NECA's chair wrote to the Chancellor requesting an urgent meeting. NECA wanted 'a substantial devolution of power, funding and responsibilities' including an investment fund, enhanced responsibility for around £500m EU funding, a guarantee that the government would deliver commitments under existing city deals, investment in housing and transport, employability support, devolution of skills funding and business support, creation of Transport North East and an integrated approach to public service delivery (NECA, 2015c: 23-28).

The following months saw consultation on devolution by NECA with its stakeholders, including a meeting with MPs and members of the House of Lords, and in June 2015 NECA's leadership board heard that feedback had shown clear support from the public and the business sector (7.2.1). In reference to elected mayors, the board, rather cryptically, noted 'the importance of being mindful about concentrating heavily on the elected mayor issue, including by the media' (NECA, 2015a: 2-3; NECA, 2015d: 8).

Following 'positive' discussions with the Communities Secretary, NECA wrote to the government on 17 July 2015 seeking to open detailed negotiations. Less than two months later it submitted a statement of devolution intent, in which it expressed 'the ambition for the North East to be at the forefront of an ambitious programme of real devolution of powers, funding and responsibilities from Whitehall to combined authorities'; there was 'a once-in-a-generation opportunity to achieve a real devolution deal for the people of the North East. We are determined to seize that

opportunity as an ambitious region with strong public support for a significant shift of power and responsibility from Whitehall to the North East... accompanied by a strengthening of local and community leadership'. It pledged to consider 'with an open mind' the most appropriate governance structures, including an elected mayor, and expected the same from government. Finally, with a message often to be repeated, it reminded the government that 'devolution does not sit in isolation from the wider impact of cuts to public services, and to local government in particular', and called for 'fairer funding' (NECA, 2015d: item 4).

By the time NECA met again in November its devolution deal, offering 'significant powers for employment and skills, transport, housing, planning, business support and investment', and £30m a year for 30 years (subject to five-year reviews) (Treasury, 2015), had been signed and, superficially, all was well. And yet, with the ink hardly dry, there was a reminder that hurdles remained. Final agreement was conditional on factors including the legislative process, the government spending review, further public consultation, agreement by the constituent councils, and formal endorsement by the leadership board and ministers (NECA, 2015e: 1-3). Nevertheless, there was no indication at that stage that elected members dissented from the deal, though they did see many elements which still required attention and stressed the importance of consultation and fiscal devolution (NECA, 2016a: 2-3). It is clear with hindsight, however, that the process through which the agreement would eventually be rejected was already beginning.

When NECA discussed the deal in January 2016, the mood had already become cautious; members expressed concerns about 'risks', listed as the fact that some crucial detail had not yet been discussed; the timescales; and that the current status of the deal was not final (NECA, 2016f: 5-6). Two months later the list of issues had grown to 11, and some had negative implications for the deal, including reduced and reducing local authority budgets, the importance of fair funding and the need for further negotiations with the government to achieve clarification, assurances and better conditions (NECA, 2016f: 12-13).

The deepening negativity was crystallised at NECA's March 2016 meeting with news (already revealed in the media) that Gateshead's cabinet had rejected the deal. Whilst supporting the principle of devolution, Gateshead did not endorse the deal and did not consent to becoming part of a mayoral CA (MCA). Meanwhile, Newcastle had agreed the deal and the other five councils wanted clarifications on a range of mainly financial issues. They were emphatic that the government 'must fully deliver on the commitments made in the proposed agreement', indicating that a vital ingredient, trust, was lacking (NECA, 2016e: 2). Outstanding financial issues related to LGF, the promised £30m annual investment fund, 'rural proofing', transport funds, fair funding, and the devolution of air passenger duty to Scotland (NECA, 2016d: 17-18; NECA, 2016e: 2).

In May 2016, NECA heard from its officials that substantial progress had been made in discussions with government. Ministers had made clear throughout, however, that devolution was conditional on having a mayor. NECA had kept an open position on this and had sought to include appropriate checks and balances on mayoral power. NECA members recognised good progress but thought work was still needed. Apart from money for highways, their concerns at this stage were no longer financial but related less specifically to getting a deal that was best for the region, with the right powers and responsibilities. In spite of these doubts, six leadership board members voted to go ahead. But Gateshead voted against, making itself liable to be expelled from NECA. One of the implications was that Gateshead would become a local transport authority, separate from the integrated transport authority covering all areas surrounding it geographically (NECA, 2016g: 2; NECA, 2016h: 4, 12; NECA, 2016k: 9).

If May seemed to mark a step forward (albeit without Gateshead), July saw devolution stopped in its tracks. A legal requirement at this stage was for NECA to publish for consultation a governance scheme covering its new responsibilities. But it did not do so. Instead, in the wake of the Brexit referendum, at a meeting on 4 July 2016, members 'expressed concern about the current position and emphasised the need for long-term assurances over the terms of the devolution deal, particularly in relation to funding'. Nine days later Sajid Javid, Communities Secretary in the post-Brexit government, replied. Devolution, he said, was more important than ever. He committed to allowing councils to retain 100% of taxes raised locally to support

economic growth, referred to a fair funding review, said work was ongoing on the question of EU funds and that an update on future funding would be published shortly. Six more days later NECA met again and considered whether Javid's letter should be considered as the firm assurances they had sought. 'A number of views were expressed', the minutes record (NECA, 2016k: 1; NECA, 2016l: 1-2; NECA, 2016m: 2; NECA, 2016n: 2).

In spite of the opaque nature of this record, it is clear from the minutes that those who wished to proceed argued their case. Points raised during discussion are recorded as having included the importance of progressing devolution, access to funding and having a strong voice at the negotiating table, the risks of delay, strong business support for devolution and the promises made during the Brexit campaign about honouring funding commitments. On the other hand, there was mention of the importance of full assurances from the new government on the terms of the agreement signed the previous October, and the position of Gateshead. NECA was split, and in spite of adjourning for private discussions was unable to resolve its differences. All voted to move to the next stage of devolution – publication of and consultation on the governance scheme – but a majority of 5-2 added the proviso 'subject to receipt and acceptance...of assurances from the government, there was according to one account confusion and a shouting match lasting over two hours split between the pre-meeting and the adjournment:

We got to a position which I didn't like where I thought I'd lost; in fact I definitely had lost, four votes to three. Went into the meeting, and [name of council leader] got up to moved what I thought had been agreed at the meeting beforehand, and we were an hour later. He started to move it, and [name of another leader] jumps up and says 'That's not what we agreed'. There was then a commotion and a shouting match, so [chair] says 'We're going to have to adjourn, we'll go back to a private discussion'. So we went back for a discussion and had another hour's argument and eventually had another vote and this time I won 4-3. That's exactly what happened. (Council leader. February 2018).

It was not unusual for the leaders' private meetings to be difficult. One said they were always difficult and understandably so because everyone was passionate about getting the best for the region as they saw it (Council leader. March 2018)

NECA received another letter from Javid on 15 August in which he confirmed full funding for EU projects signed off before the forthcoming autumn budget statement and that there would be new funding arrangements for later projects while the UK remained in the EU. He confirmed the requirement for an elected mayor. Eight days later NECA members met Javid for talks, when he confirmed the funding arrangements in his letter, and said the government supported devolution to NECA and that it was a stepping stone to further devolution. He remained unequivocal that an elected mayor was required (NECA, 20160: 3). A civil servant involved in the negotiations told this study:

They [NECA] asked for some reasonable things which he [Javid] gave them and then they asked for some unreasonable things. But what they kept doing was delaying and delaying and delaying. Their tactic was not to do anything, not force anything, I don't know if they were trying to call anyone's bluff. Eventually we said unless you make this progress then the deal will lapse. (Civil servant. January 2018)

The NECA requests that the civil servant criticised as unreasonable related particularly to European funding. The councillors wanted guarantees that after Brexit they would still receive the EU funds they were expecting, guarantees which the government felt unable to give at that early stage of the Brexit process. Council chief executives knew that the demands of their political masters were unreasonable, the civil servant claimed, though in the end, he said, most of what they wanted was secured by negotiation (Civil servant. January 2018).

Control over EU funding was a critical issue and a symbol of the distrust between NECA and the government. One of those who voted against the deal complained they had been led to believe that after devolution the funds, amounting to nearly half a billion pounds, would be controlled by the elected mayor, but later this pledge was weakened so that the mayor would only have influence over the funds (NECA council leader. March 2018).

By late summer there were signs that Gateshead Council might be getting cold feet about being isolated. Its cabinet met at 9.00am on 6 September 2016 – the day NECA was to meet at 2.00pm to decide whether to proceed with the deal. Councillors received a report from officials detailing the implications of being left out but also warning of the danger of opting back in only to find others opting out. Gateshead officials recommended the cabinet give its consent after all but with the proviso that it could withdraw again if one or more of the other councils pulled out. In the event the cabinet made no decision at all, deferring the matter to take account of the outcome of NECA that afternoon (Gateshead Council, 2016a; Gateshead Council, 2016d: 10).

The final act was played out at a special leadership board that afternoon, delayed for about 15 minutes while the leaders held a private pre-meeting<sup>22</sup>. Members expressed again their concerns about funding issues and also the view that the deal was not a good one; on the other hand, some noted the importance of going to the next stage and being at the negotiating table, the risks of not going ahead including to transport projects, and the unlikelihood of a better deal. Members commented on the importance of devolution to the North East and their full commitment to 'real' devolution, but in the end the board voted 4-3 not to proceed to the next stage (NECA, 2016p: 7-8). Newcastle, North Tyneside and Northumberland backed the deal; Durham, Gateshead, South Tyneside and Sunderland did not. One person who was present laid bare the hidden tensions when the final vote was taken:

[Name of leader] went from abstaining to - God, this is really awful! [His] group would not have accepted him being the determining vote...[He] would have been happy to abstain...and for it to happen, but not for it to be the determining vote, because he had difficulty inside his group (Council leader. February 2018).

The deal's rejection was followed by more than a year of apparent stagnation. But behind the scenes the three north of Tyne authorities were involved in intense negotiations to set up their own MCA. Their efforts, never secret but never transparent either, broke into the light of day when the government's backing for the new North of Tyne Combined Authority (NTCA) was announced in the November 2017 Budget. Local

<sup>&</sup>lt;sup>22</sup> Personal observation.

media were enthusiastic, strap-lining stories: 'North of Tyne devolution: why you should be excited' (Hill *et al.*, 2018). In an 'Invest in the North East' campaign in a regional paper, devolution figured in fifth place in its list of 50 reasons for businesses to do so (Whitfield, 2018). Speculation started about who might be the first metro mayor.

In early 2018 an assurance came from the government that the North East would be compensated for EU funding lost as a result of Brexit. The news came in a letter to Northumberland County Council's new leader, enabling a Conservative politician to announce the resolution of one of the issues that had proved a deal breaker when NECA rejected its agreement 16 months earlier (Kelly, 2018). It was a reflection of the party political game which we will see in Chapter 5 being played much more extensively in Tees Valley. Meanwhile, pressure from Labour opponents continued with a newspaper report, based on quotes from a Newcastle MP, that a North of Tyne mayor could mean higher council tax bills (Walker, 2018a).

When Newcastle and North Tyneside Councils approved the NTCA plans through their cabinets in December 2017, they made a point of not breaking completely with their south of Tyne counterparts. They would continue to collaborate with the rump NECA through a proposed transport committee (inevitably, in view of the integrated nature of local public transport) and with NELEP (again, as they had to), their cabinets were told, and 25% of the 10,000 new jobs expected to be created by the deal would, it was estimated, be taken by south of Tyne residents (Graham, 2017).

NECA gave its consent to the departure of the North of Tyne councils in April 2018 by 6-0 votes. Gateshead abstained, with its leader commenting: 'At some stage some future government...will have to do something to pull this governance together. It's dysfunctional to say the least' (Muncaster, 2018; NECA, 2018a: 13-15). Theresa May brought her cabinet to Gateshead and confirmed the deal (Graham and Eden, 2018; Seddon, 2018c; Walker, 2018c; Walker, 2018d). The legislation passed that October, leaving just enough time for a mayoral election in May 2019 (Walker, 2018e). Three Labour and two Conservative MPs were content to be photographed with the *Northern Powerhouse* Minister at the signing ceremony (Holland, 2018a). Even while the legislation was in Parliament came an example of the harmful effects of splitting

the Tyne and Wear conurbation, with its Metro running on both sides of the river. The decision on whether the first Metro extension should be north or south had been 'parked', the House of Lords was told (Shipley, 2018). A business leader said:

It's been the most frustrating process in the North East. We have had four years now of talk...and it just felt as though the civic leadership wasn't ready for a devolution deal and they would find any reason to reject it...There was a whole range of different excuses as to why they didn't want to sign up to a deal. To their credit the North of Tyne authorities have not buckled under that...It just means that we are one step behind other areas like the Manchesters, like the Liverpools (Business leader. August 2018).

Within days the new NTCA held its first meeting and the political parties started their searches for a mayoral candidate. Labour's process was at one stage reported to be 'in disarray' over the lack of women candidates. Its choice eventually came down to a straight fight between two men from Newcastle, the city council leader and a Momentum activist first elected to the council the previous May. The Momentum candidate won selection by 2,514-1,913 votes (Seddon, 2018a; Seddon, 2018b; Seddon, 2019b; Tighe, 2019b; Seddon, 2019c) and went on to win the election with 56% of the vote, including second preferences, on a 32% turnout against Conservative, LibDem, UKIP and Independent opponents, all men (Holland, 2018b; BBC, 2019a; Seddon, 2019d; NTCA, 2019f).

### 4.2 The role of the business leadership

NELEP first received a presentation on devolution in March 2015 and was kept updated regularly. The minutes of its meetings record little or nothing more than that reports were noted (NELEP, 2015a: 2; NELEP, 2015b: 5-6; NELEP, 2015c: 3; NELEP, 2016b: 7). Then, in April 2016, when it was already public knowledge that controversy within NECA was threatening the deal (Graham, 2016b), NELEP's chair reacted with a statement that the LEP was publicly on board with devolution and would be happy to assist in any way (NELEP, 2016c: 6). This offer was repeated a month later, with NELEP noting that it had a responsibility for the economic wellbeing of the North East (NELEP, 2016d: 6). Nevertheless, devolution still appeared likely enough for one NELEP board member to resign from the Conservative Party with a view to standing for the

mayoralty as an independent (such was the poverty of Tory expectation in the North East) the following year (Hill, 2016a).

Business as usual ended in summer 2016 with the dual shocks of the Brexit referendum and NECA's rejection of its devolution deal. Following the Brexit vote NELEP played an active role alongside other business organisations in pressing for their voice to be heard in the debates over both devolution and Brexit. Business leaders met informally on 8 July 2016 to formulate a response to Brexit, co-ordinated by a working group representing NELEP, the CBI, EEF, FSB, North East England Chamber of Commerce (NEECC), Entrepreneurs Forum and NECA represented by the economic directors of the local authorities (NELEP, 2017e: item 5 (iii)).

In July NELEP was still working on the assumption that devolution would proceed (NELEP, 2016e: 5), and NECA's decision on 6 September not to go ahead came as a shock. Three weeks later NELEP's chair wrote to the government to reaffirm its commitment to leading strategic economic growth and to place NELEP at the head of the process. It had been an unsettling time for many North East businesses, he said, and it was now even more important, post-Brexit and with uncertainty around devolution, that business leadership through NELEP was 'front and centre' of the agenda (Hodgson, 2016). The business leader who had earlier declared his intention to stand for the mayoralty resigned from NELEP and set up the North East Devolution Commission 'to explore ideas for the region's future', backed by 100 local business people (O'Donoghue, 2016; Rowell, 2016; NELEP, 2016a: 3). It launched a youth employment policy the following month (Ord, 2016), though has since had a profile low to the point of invisibility.

Just before Christmas the Chamber of Commerce wrote to NECA criticising its rejection of the devolution agreement in strong terms (NECA, 2017b: 5-6):

While business across the UK are faced with the uncertainty of Brexit, our members also remain frustrated by unfulfilled devolution and are calling on NECA to set out very clearly how it intends to contribute to the realisation of the ambitions originally set out in the SEP... Ongoing uncertainty on both fronts puts the North East at a disadvantageous position compared with other regions of the UK.

On 5 January 2017 the NELEP chair and vice-chair, accompanied by representatives of the other business organisations, met the seven NECA council leaders for a discussion that was 'open, honest and conducted in a positive manner'. Afterwards the business leaders agreed key principles on which to seek government support and move the devolution agenda forward, including action in time for a mayoral election in May 2018 and involving as many of the local authorities as were willing and did not frustrate the process. The business community wanted to be acknowledged as a senior partner, the new devolution deal to be at least as good as the previous one and to have more business-focused outcomes, the SEP to continue as the principal economic policy document, and NELEP and business leaders to be part of any negotiating team (NELEP, 2017b: item 8).

Following news that the three north of Tyne councils were in talks with the government about their own deal, the NELEP chair spoke out in a newspaper column. It was clear that despite a huge effort by NELEP and the business community, he wrote, the seven councils were not in a position to progress devolution. It was important for the North East not to be left behind, so they would work with the three councils and any others wishing to move to a deal with business-focused outcomes (Hodgson, 2017).

In April, when NELEP (with NECA) responded to the green paper *Building our Industrial Strategy,* it took the opportunity to air its views on governance issues and, though diplomatic, hinted at frustration. After outlining its work to implement the SEP, it noted that underlying it 'should be [implying 'is not yet'] a framework of economic governance offering long term and stable institutional support for the economy through properly empowered local institutions'. Regular changes in institutional structure and capacity had not served regions like the North East well, it added; partners across the region were said to support significant devolution to strengthen the local institutions to make an economic impact (2017d: 19).

Then, in July 2017, as NECA appeared still paralysed, NELEP stepped forward again, in an apparent attempt to fill a gap in local leadership (NELEP, 2017e: item 5). It approved a series of measures to communicate and promote the SEP, strengthen the economic evidence base, shape the economic policy environment and take control of the

region's response to Brexit. The informal group of business leaders that had formulated the North East's response to Brexit the previous summer was put on a firmer footing, reporting to the NELEP board and with its membership extended to include the TUC and universities. Over the following 18 months the Brexit group published reports on migration, the impact of Brexit on the area's economy and key messages, and prepared a Brexit toolkit for businesses. In December 2018, as MPs debated the Brexit Withdrawal Agreement, the group issued a special bulletin to its stakeholders detailing its work (NELEP, 2018d; NELEP, 2018e: item 7).

Members of NELEP's team were trained and tasked with spreading its message at dozens of presentations. NELEP planned a programme of research, an 'evidence forum' of key partners, regular public commentary on economic data and an annual 'state of the region' report. It planned to fulfil its role of boosting the regional economy by operating as 'a collaborative organisation working across sectors', influencing, convening and supporting. It contemplated, too, controversially, stepping beyond the bounds of economic and industrial policy to address linked social issues like inequality, suggesting it might use SEP funding to tackle them. A NELEP report stated:

The [2017 general] election campaign and the result itself exposed key issues of inequality. All the main parties expressed concern in the run up to the vote, but the outcome has been catalytic. A particular focus is on the issue of intergenerational division and inequality which found particular expression in both the referendum and the June election. Narratives around 'equitable' and 'inclusive' growth are now at the heart of the national debate, with debate around tax, the quality of employment, and access to housing for young people and families. These are aligned to key programmes in the SEP and provide opportunities to elaborate proposals through policy and investment programmes (NELEP, 2017e: item 5 (ii) p.3).

In Hambleton's model of civic leadership, co-ordinating the response to such issues is a role an elected mayor would have played. A regional business leader told this study: 'I think there is a definite feeling around the whole patch that some proper leadership is required' (Interview. August 2018). In the absence of a mayor, the task was thus potentially being taken on by a body of unelected business leaders (though with a

minority of councillors on their board). This caused unease in both political and business circles, and NELEP backed off. A NELEP board member from the private sector said later that one of the NECA leaders warned that social policy had to be either done or not, but not half done (NELEP board member, private sector. January 2018).

In spite of the Brexit and devolution traumas, NELEP asserted in its 2016/17 annual review that 'we are in an extremely strong position to continue driving the SEP forward'. One advance NELEP had made during 2016/17 was to 'significantly improve collaboration with the North East business community, our membership organisations and key sector groups'. Though it attributed these closer links to work on refreshing the SEP, the joint reaction of business organisations to Brexit and NECA's rejection of the devolution deal must also have helped bring them together (NELEP, 2017f: 1, 8).

Relationships between the NELEP business sector and the NECA politicians remain sensitive. There is jealousy, for example, over access to ministers. A civil servant involved in devolution negotiations said: 'The LEP in particular felt excluded from a lot of those [devolution] discussions, led by a small group of elites within the local authorities within NECA' (Interview. January 2018). From the other side, a council leader complained: 'I can't pick up a 'phone to Sajid Javid. I don't speak to ministers and secretaries of state. But [LEP chair] does. Daily' (NECA council leader. February 2018).

Through all the political machinations, business generally takes the pragmatic view that it must work with whatever is in place: 'It is what it is', sighed one regional business leader (Interview. September 2018). 'There is a sense that the LEP will do what the LEP will do' said another (Interview. September 2018). NELEP too has to take as it finds. A NELEP board member who deals with the NECA leaders regularly, reflecting on relationships, said:

I think the frustration is between the political leaders locally. It's a very complex situation. You learn a lot doing roles like I do and one of the things that I think I have learned is that the politics within a party are much more complex than the politics between parties, and a lot of the leaders here have power bases built upon very, very localised politics. And it took me some time to get to grips with

that, and some of the sub-optimal decisions that become apparent through that. (NELEP board member, private sector. January 2018)

# 4.3 The role of public professionals

While the North East's political and business leaders struggled to build a working partnership, another of the five civic leadership groups in Hambleton's framework, the professional public servants were getting on with their jobs, but in the view of some were failing to maintain the impartiality which was intrinsic to their role. A Labour Party activist told this study:

Don't forget that behind the scenes the chief executives and other senior officers in the councils were working closely with the DCLG [Department for Communities and Local Government] to press the government line, and there was a lot of pressure, really a great deal of pressure, political pressure, on council leaders not just by Osborne but also by Lord O'Neill, who came to the region numerous times and tried to lean on those who were not happy to try to persuade them it was a good idea. So in a way I think the local authorities were brave, actually, and in my view were outstanding in terms of the democratic decision making they took (Anti-devolution deal campaigner. November 2017).

The officials who advised NECA were certainly upbeat about the deal from the start, consistently recommended moving forward, and maintained that position for as long as possible without seeming to interfere in what had clearly become a controversy capable of resolution only by their political masters. The official tone was set in a report to NECA at its first meeting after signing the provisional deal. It detailed the 'significant' powers to be devolved, and said the deal paved the way for further devolution and would 'bring considerable additional resources' (NECA, 2015e: 2). As late as 4 July 2016 officials were recommending the leadership board to move to the next stage (NECA, 2016k) and even on 19 July, although the official recommendation was simply to consider the next steps, there was advice that devolution 'will have a positive impact on the objectives of NECA' (NECA, 2016m).

Gateshead's cabinet was recommended by its chief executive at a crucial meeting on 22 March 2016 to approve the deal subject to clarification on outstanding issues. Though risks were noted, the overall message was positive:

In its totality the agreement will support increasing the region's ability to support all in society, through new responsibilities for skills training for young people, support for those struggling to find employment, improving transport services and infrastructure and helping the region build more homes (Robinson, 2016a: 13-15).

But it was too late: Gateshead's Labour group had met privately and overwhelmingly rejected the deal (Seddon, 2016). The vote was 43-7, according to one informed source<sup>23</sup>. On 12 July the chief executive tried again, recommending that the cabinet reconsider, 'having regard to the interests of unity across the NECA area' (Robinson, 2016b). As late as 6 September, the day NECA finally rejected the deal, Gateshead's acting chief executive was recommending the cabinet to agree unless at least one other council backed out. He warned of the costs of being the only council to hold out against the deal (Barker, 2016: 6, 8). A Gateshead councillor recalled being warned that the consequences would be dire:

We would have to set up our own transport authority, we would have to have our own concessionary fares scheme, we'd have to employ 50, 60, 100 staff to monitor and administer that. When they broke up the debts we would be disadvantaged by it. All the pension liabilities would tie up our financial departments and our legal department. So they were beating us over the head to get back in line. (Councillor. February 2018)

Newcastle officials, with top-level political support, backed the deal, while the approach of officials in the other councils was generally cautious, pointing out the risks and potential disadvantages but either avoiding a specific recommendation or recommending taking the next step forward while continuing to talk to the government about outstanding issues (Forbes and Lewis, 2016; Mason and Davey,

<sup>&</sup>lt;sup>23</sup> Personal information

2016; Melia, 2016; O'Donnell and Henig, 2016; Swales, 2016; Sunderland Council, 2016b).

# 4.4 The Campaign for Real Devolution

A campaign against NECA's provisional deal emerged almost as soon as it had been signed in October 2015. It shows trade unionists, one of five civic leadership groups in Hambleton's framework, playing a dual role. Many councillors, sometimes in leading positions, were union members. They sat in the Labour groups which had the ultimate say over the devolution deal while simultaneously being members of unions campaigning against the deal.

Within a month of the provisional deal being signed the Labour MP for Newcastle East, Nick Brown, made his opposition known through his webpage (Brown, 2015). Brown is one of the region's most influential figures, though most of his time is now spent on the national stage. A former Newcastle councillor, he served in the Blair and Brown governments, was Minister for the Region, and survived into the Corbyn era as Chief Whip. His posting marked (though did not cause) the start of a 10-month struggle, and by the following month a campaign had formed.

Though the Campaign for Real Devolution (CARD) was not branded a Labour campaign, it was conducted almost wholly within the Labour and trade union movement. It was in effect the work of one long-standing, well-connected Labour Party member, not holding elective office, who organised the support of its political and union wings to target the Labour groups of councillors. One Labour MP who supported CARD's campaign said of him: 'The prime mover is [name]..a very senior figure, academic man, highly respected...I think he did very well. He is well thought of and that gained support', though he added: 'The opinions of the rank and file councillors were shaped to some extent by [name's] campaign, but also by exposure to the arguments'.

This campaign organiser understood the vital fact that in the local government of North East England, power is exercised through the Labour groups who determine policy in the privacy of their meetings and control most of the area's councils most of the time. Party political groups wield a dual power, organisational and normative, as

discussed by Elder-Vass (2010: 7): they control votes in council and exert moral pressure.

By early December 2015 CARD<sup>24</sup> had written to every Labour councillor in the area spelling out its reasons for opposing the deal, which it described as simply a form of 'enhanced localism'. This letter set the tone for the entire campaign. It went into considerable detail but the basic messages remained the same: the net financial benefits were marginal; council services would remain starved of funds; the deal was undemocratic because of the imposition of a mayor; it cemented the division of the region; regional control of EU funding was likely to prove illusory; it was not a starting point for further genuine devolution; and voters should be given a say. The letter also introduced a phrase which, with some variations in wording, would become familiar: 'It is based on a flawed city-region economic growth model which is analytically and empirically unsound'.

On 18 March 2016 ten Labour MPs and three trade union leaders wrote an open letter to NECA drawing attention to 'increasing strength of feeling' among MPs and unions against the deal. Four days later CARD was celebrating news that Gateshead's cabinet had rejected the deal. The decision was reported in local media as having stunned many. A LibDem councillor said devolution had been 'sacrificed by those in the Labour party who view elected mayors as a threat to their own establishment and control of the region' (Kelly, 2016b; Seddon, 2016). Two months later the Gateshead Council leader resigned, described by the same opponent as 'collateral damage in the battle against the regional elected mayor' (Milligan and O'Donoghue, 2016).

Trade unions played a big part in the CARD campaign. Regional leaders of the GMB, Unison and Unite wrote jointly to all Labour councillors on 8 April 2016 to ensure they were aware of union policy on devolution:

With this decision being one for Labour groups, we urge you to ensure that the proposals are brought under full scrutiny within your Labour group in the course of the next few weeks. We feel it is vitally important that all Labour groups give full consideration to the latest information before making an informed decision

<sup>&</sup>lt;sup>24</sup> CARD's campaign is documented on its Facebook page at <u>https://www.facebook.com/Campaign-for-Real-Devolution-CARD-944653188965455/</u>

on whether to commit their council to the MCA in its current form. We ask you to support the policy of your trade union by rejecting the current deal, and would be pleased to offer you any support you feel would be helpful to get this debate raised in your Labour group.

The use of the phrase 'your trade union' is a reminder that in many cases the union leaders were writing to their own members, and in some cases these members constituted a majority within their Labour group. Research in January/February 2018 revealed how many councillors were members of the three unions active in the campaign<sup>25</sup> (Table 4-1).

The GMB, Unison and Unite between them accounted for a majority of the Labour group in Gateshead, which was the first to reject the deal, with Unison as the largest union. This is consistent with the opinion of a civil servant close to the devolution negotiations, who told this study that the unions were particularly strong movers in the CARD campaign and that: 'Unison were leading the charge' (January 2018).

Council	Labour group total	GMB	Unison	Unite	Three unions total
Durham	74	19	7	10	36
Gateshead	54	7	16	14	37
Newcastle	53	10	12	6	28
N. Tyneside	52	12	12	7	31
North'land	24	4	9	2	15

Table 4-1: Trade union representation in Labour groups (January/February 2018). Source: Council registers of interests

Any argument, however, that councillors automatically follow the line of their trade union faces the problem that the three unions also had majorities in the Labour groups in Newcastle and North Tyneside, and perhaps Northumberland, which backed the devolution deal, and the two council leaders known to have been members of one of

<sup>&</sup>lt;sup>25</sup> The Gateshead, Newcastle and North Tyneside council memberships at that time were the same as in 2016, when the devolution deal was rejected, apart from by-elections. The Northumberland and Durham memberships were significantly different following all-out elections in 2017. The 2018 figures give only an indication of union strengths in these local authorities. Union membership for South Tyneside and Sunderland could not be established.

the three campaigning unions (Gateshead and Northumberland) took opposite sides when the final NECA vote came. A CARD campaigner made the controversial claim that in Labour groups which failed to support the CARD line, the decision was due to a failure of group democracy:

Unfortunately not all the Labour groups were enabled to have a proper discussion and debate about the issues, and that for me was very sad and very undemocratic. The three in particular who did not have full debates were North Tyneside, Northumberland and Newcastle. The other authorities, Durham and Gateshead in particular, had quite detailed discussions within the Labour groups, and the Labour groups took the decision...It is interesting that the three authorities where there was no proper democratic full discussion are the ones who actually supported the idea of a mayoral combined authority (CARD campaigner. November 2017).

This campaigner, echoing Wollmann (2008) (2.2.4), ascribed failure to debate devolution critically within some Labour groups to the cabinet system introduced in 2000, giving leaders, as he saw it, considerable patronage in appointing the cabinet.

This version is vigorously contested, however. One leader of a deal-supporting council said devolution had been discussed at virtually every Labour group meeting for three years, and anyone saying otherwise was either inadvertently or deliberately trying to mislead people. Devolution had won widespread group support and on several occasions the group had voted unanimously to press ahead (NECA council leader. March 2018).

Political group meetings are private and only those present know for sure what happened. The CARD campaigner quoted here is not a councillor and did not attend the meetings concerned. According to his interpretation of events, the Labour groups that supported his position had proper democratic debates while none of those which opposed his position did so. This view carries a tinge of self-justification. The interview evidence, however, suggests he is correct to say that debates were particularly vigorous in Gateshead and Durham.

By 15 February 2016 three mailings had been made by CARD to all Labour councillors in all seven NECA authorities urging them to oppose the deal. A public meeting was held on 5 March addressed by councillors, trade unionists, Labour activists and at least one shadow cabinet member, to discuss how rejection of the deal could be justified and implemented in a democratic manner. A second public meeting was held on 7 May. Perhaps the only voice at these meetings to remind campaigners that not everyone in Labour was opposed to devolution deals came from the party's mayoral candidate for Tees Valley, who explained why colleagues there had decided to accept a deal. Another mailing followed ten days later arguing that a powerful executive mayor without an elected assembly was undemocratic and saying the money on offer amounted to only 4% extra for investment. The campaign thereafter petered out, but it had done enough.

The CARD campaign was not secret; it was on social media, and public meetings were held and reported in mainstream media. Nevertheless, in a process with so little public engagement and a region with such marginalised and ineffectual political opposition, it was in effect a Labour Party and trade union in-house affair, and all that was required for victory was to persuade around 400 Labour councillors, meeting in private in their party groups. A civil servant involved in the negotiations speculated later about the motives behind the campaign, and referring to the self-declared independent candidate for mayor referred to above (4.2) - businessman, former Conservative and NELEP board member - added:

There was certainly a rhetoric of 'you're passing on Tory cuts'. I think people were scared about a particular candidate who had made a powerful bid [name]; they were scared he would be successful. I think there was also a little bit of protectionism among council leaders that having a mayor would force them not to have their own priorities first in their cosy little cabal (Civil servant. January 2018).

Exactly how influential CARD was in swinging the vote in the Labour groups is hard to say for sure. A leading councillor in Sunderland said he had never heard of CARD (March 2018). Another in South Tyneside said: 'I think I had one conversation with [campaign organiser] and that was pretty much it. So it had no effect down here. I

wasn't in any way politically pressured' (March 2018). One Gateshead councillor remembered receiving CARD's correspondence but claimed not to have been influenced by CARD at all (backbench Labour councillor. March 2018). Another said he was aware of very few councillors who had attended CARD events. A retired member of one of the three campaigning unions, he added: 'I do not feel under any pressure whatsoever from [the union] to vote in a particular way on any given issue...You are under the impression that the trade unions have sufficient influence to direct councillors' vote. That is not the way it operates' (Gateshead backbench Labour councillor. February 2018). A Durham councillor had heard of a CARD meeting but not attended. He was aware though of the trade union letters it issued: 'They were also taken into consideration. A lot of councillors are members of those trade unions ...and as part of the Labour movement trade unions are fundamental to that. That all comes into the mix in terms of how this plays out' (Durham backbench councillor. April 2018).

The CARD campaign displays both the influence of the trade union movement in local government and its limitations. Union leaders are well connected, have members sitting on councils in significant numbers and holding important leadership and cabinet roles, and are able to ensure their voice is heard. On the other hand they cannot rely on those members to automatically follow the union line, as demonstrated by the decisions of at least two and perhaps three councils with union majorities in their Labour groups to back devolution. They lose other battles too, such as over redundancies (4.5.1). Close connections between the Labour Party and the trade unions date back more than a century and are an established part of the UK political system. In most, though not all, councils checked for this research councillors' membership of trade unions is public knowledge through their register of interests.

# 4.5 Political leadership – councils and cabinets

This section focuses on why and how four councils responded to the pressures described above and rejected NECA's devolution offer. The interest centres on these four because, after NECA agreed a provisional deal with the government in October 2015, acceptance was the default position. Why, this thesis asks, did four councils turn down an offer of extra powers and funding less than a year after accepting it?

A NECA council leader told this study:

I ended up from March to September [2016] sitting in meetings...sitting talking, sandwiches and crisps, and I cannot believe this is how democracy works. I cannot believe the future of this region actually revolves round me having some sandwiches over lunch with [another council leader]; it cannot be right. It didn't involve trade unions, stakeholders, academic institutions, constituencies, members of parliament - and members of parliament were furious: 'What the [expletive deleted]'s going on here?' This is having massive political consequences but it was being determined by half a dozen people in private rooms (Council leader. February 2018).

That is one version, from the inside, of how it happened. The reality was not always quite like that, for decisions are sometimes taken by rather more than half a dozen people in private rooms; they are taken by a few dozen people in private rooms when party groups hold their meetings – though in some councils, as we will see (4.5.3), leaders are given a very free hand.

Another version, looking in from the outside – but in close contact with the process – came from a business leader:

At the core of it was some quite left-wing politics [in Gateshead] that fundamentally was going to fight anything that was a Tory dogma, that was going to fight anything about mayors because they don't believe in that particular form of governance, and that they also, a number of the individual leaders, feel threatened by joining something that was bigger, and seeing it although it was very much a matter of trying to bring devolved powers from Westminster back to the North East - actually lessening the authority of their own little fiefdoms. So there were a number of both personal objectives and political objectives coming together (Business leader. August 2018).

Full councils and their cabinets did actually debate devolution in public, and the minutes and reports which contributed to decisions can be viewed online. Members of the public may, however, be unaware of the private decisions already taken that constrained their elected representatives in public. Cabinets made the decisions, though full councils usually met first to make their views known. Six of the seven

councils discussed the devolution deal between 24 February and 21 March 2016 (the exception being South Tyneside), followed by decisions by their cabinets. NECA was told on 24 March that the decisions were: Newcastle to support the deal, Gateshead to endorse the principle of devolution but not the deal or an MCA, and the remainder to support subject to clarifications (NECA, 2016e) (4.1).

### 4.5.1 Gateshead leads the opposition

Gateshead has a history when it comes to resisting government initiatives, and is proud of its ability to stand on its own feet. Its attitude might be characterized as bravado: 'We are confident about our own capacities', said a councillor (February 2018). Another recalled with pride that Gateshead refused to join the Tyne and Wear Urban Development Corporation (UDC) of the 1980s but still managed to achieve some regeneration (February 2018). An official referred cockily to Newcastle, the regional capital across the Tyne, as 'Gateshead North'<sup>26</sup>.

Gateshead is key because it led the way in rejecting the deal, and understanding why is challenging. 'Momentum Gateshead and Blaydon' is linked as a related page on the CARD Facebook page (Momentum, undated). Nevertheless, a backbench councillor discounted any suggestion that Momentum had dictated the decision; some councillors were supporters and others not (Gateshead councillor. April 218). Another Gateshead backbencher was actually unsure altogether why Labour councillors voted against the deal:

It's hard to say to what extent something actually influenced different people...We debated as Labour councillors this issue at length and came to a decision (Gateshead councillor. February 2018).

The limitations of union influence on councils in general, discussed above in relation to the CARD campaign (4.4), were made clear by a union leader. The union's strongest representation, he said, was in Durham and Sunderland, but that did not stop those councils making the union's members redundant (Union official. October 2018).

<sup>&</sup>lt;sup>26</sup> Personal observation.

Nevertheless, the CARD campaign message did get through to some. When Gateshead Council debated the deal on 25 February 2016 one member, a Unison official, referred to it as 'an enhanced form of localism' and 'based on a city region economic growth model that does not suit our region' – both forms of words reminiscent of CARD literature (Gateshead Council, 2016b) (4.4). When the cabinet rejected the deal on 22 March, of nine councillors round the table seven were members of one (in one case two) of the campaigning unions, and one has since become a Unison-sponsored MP (Gateshead Council, 2016c; Parliament, 2018a). The council leader was one of the two not in these unions, and two months later was replaced by a GMB member. A suggestion that some councillors are under union influence without necessarily understanding the reasons came from a leading member of another council: 'Sometimes I think it was as basic as "we've got a message from the union and we're against it". They might not have put it in those terms – don't get me wrong' (Labour council member, NECA. March 2018).

Links with Momentum notwithstanding, Gateshead cannot be said with confidence to be dominated by the Labour left. It is not even clear how 'left' would be defined. However, given its consistent resistance to Tory initiatives, a different adjective is suggested by a trade union banner that hangs in a Civic Centre interview room<sup>27</sup>. Displaying a portrait of Lenin, the word for Gateshead Council it brings to mind is 'bolshie'.

For all the internal politics, Gateshead had serious concerns about being asked to take on devolved responsibilities with inadequate pre-determined budgets to fulfil them. One councillor recalls being told by a minister: 'You can have any function you want, except education. You can have DWP if you want (laughs). There's power for you. There's a function for you'. He added:

Even if we got some kind of fiscal - and we are miles and generations away from fiscal devolution - the North East of England needs to be very, very, very careful about fiscal devolution because on which tax would we benefit? Business rate? Put up a local tourist tax? There isn't a single form of taxation where we don't

<sup>&</sup>lt;sup>27</sup> Personal observation.

heavily depend on redistribution from central government. So that was our objection to devolution (Gateshead councillor. February 2018).

Durham is also aware of this danger (4.5.2), as was made clear in interview (April 2018). Bailey (2018) has argued that tax reforms ostensibly intended to incentivize councils to boost economic growth, including business rates retention, will concentrate capital for investment in areas already affluent and growing and probably further disadvantage northern regions. A mismatch between responsibilities and funding and a weakening of the ability of the centre to redistribute were among the reasons for caution over devolution given by Rodríguez-Pose and Gill (2003) and Prud'homme (1995) noted in section 1.2.

### 4.5.2 Durham shrugs its shoulders

Durham's decision to vote against the deal came as a surprise to some. A business leader from outside Durham, asked why he thought the four south of Tyne councils had rejected the deal, simply replied: 'If anyone could explain to me why Durham did I'd be grateful' (Business leader, North Tyneside. December 2018). Outside observers believe the decision within Durham Labour group must have been close, and difficult for the Durham leader. A fellow NECA council leader said:

It was very finely hung. Leaders have a huge amount of influence inside of group and council; you would expect to support a leader; you look for leadership; you expect them to have spent the time and effort to research issues and to persuade people and take people with them. So when it starts to get towards 50:50 votes in a group meeting leaders' positions are becoming very strained; you are stretching every ounce of loyalty out of people (Council leader. February 2018).

Durham was the only council to poll all residents on the deal. Almost 82,000 responded and there was general support for the concept of devolution (Henig, 2016)<sup>28</sup>. After receiving the results the cabinet deferred a decision on 23 March, the day after Gateshead had rejected the deal. By 24 April all Labour group members had

<sup>&</sup>lt;sup>28</sup> For further discussion of public consultation see section 7.2.

received a letter from six county Labour MPs urging them to do likewise: 'It is economically dysfunctional, unequal in its application and reduces democratic control at local level', they wrote, repeating the wording of a CARD union letter of 8 April (Hill, 2016b). On 11 May the cabinet met again and was torn. On the one hand, said the leader, 'moving forward was reflective of the results of the poll'. On the other, 'fair funding was integral to making devolution work and without it there were no prospects to proceed'. Faced with a dilemma, the cabinet agreed to move to the next stage, but only on four conditions – no mayoral veto; no disadvantage to NECA councils as a result of mayoral voting; no disadvantage to Durham in relation to transport and highways funding; and fair funding (Durham County Council, 2016a: 17, 21; Durham County Council, 2016b).

These concerns expressed publicly at the cabinet reflected those voiced privately in the Labour group. There were undoubtedly concerns about a mayor, a fear that was exploited by CARD. In a social media post a week after Durham County Council decided to poll its residents, CARD conflated the threats of a Conservative imposition and rule from Newcastle with the misleading claim that local councillors would be side-lined and the mayor would be responsible for local government services across the area, as well as reminders of cuts and the paucity of the financial offer.

In spite of concerns about the mayor, however, it was funding which was the deciding factor. The Durham leadership, like Gateshead's, had thought carefully about the fiscal implications of more devolved responsibilities and was worried it would leave the North East worse off, leading to another round of cuts for which they would be blamed. According to one leading member, they would have been prepared, reluctantly, to accept a mayor if funding guarantees had been sufficient (Durham councillor. April 2018)

Alongside these articulated fears, however, other deep-seated factors were at work in the minds of Durham councillors, semi-conscious examples of the evolutionary economic geography discussed in 2.1.3. One was a historic unwillingness to deal with a Conservative government. According to one councillor:

There are people on this council who were arrested during the miners' strike, let alone anything else. So there's a lot of that folk memory, and it makes it – it
always made it all the way through – difficult to do anything (Durham councillor. April 2018).

Such people are the 'carriers of history', embodying the value framework of their territory (Keating, 1998; Martin, 2000) (2.1.3). The memory of the miners' strike and its continuing influence on politics continues to surface elsewhere in the region. Sunderland Council called in 2019 for a public inquiry into police actions at the 'Battle of Orgreave' during the strike 35 years previously (Binding, 2019). A North of Tyne politician said unsympathetically of his south of Tyne counterparts: 'They're political dinosaurs who can't see past what happened in the 1980s' (May 2018).

A second factor for Durham was the Gateshead effect. Gateshead's rejection of the deal changed the dynamic and unnerved Durham as the arguments spread across the border. The Durham leader witnessed strong opposition when he debated the deal at Blaydon (Gateshead) Constituency Labour Party. Word was passed on through links in the party and trade unions, informal networks and close family ties among elected representatives at the highest levels.

The third informal factor to influence Durham was an indifference in some parts of the county born of a belief that the deal was Newcastle's business and rural Durham could expect nothing from it. Durham shares few major public services with the conurbation to its north and many residents see no reason to go there; some in the south of the county look to Tees Valley rather than Tyne and Wear. A leading councillor described what he found as he travelled around to discuss devolution:

I wouldn't say it was a hostility, it was more a shrugging of the shoulders and 'What's that got to do with us?' and the further I got south and west, away from the metropolitan area and into rural areas the stronger that became. So it isn't political; actually it is stronger in some of the more Conservative areas. Some of the farming areas particularly say 'What have I got to do with Newcastle? Town for us is Barnard Castle; we don't even like Durham City very much' (Durham councillor. April 2018).

Others in County Durham, not politicians but with a direct interest, shared some of these views. A business support executive thought Durham was fearful of ending up

second cousin to Tyne and Wear: 'I think there was a fair bit of concern that that is what we would have ended up as if there had been a mayor appointed. I would certainly have been worried about that'. He, like councillors, was torn between concern about Newcastle predominance and the potential benefits of devolution. Ultimately, though, he came down for devolution, which he regarded as 'a lesser evil': 'I look at what's happening in Tees Valley and I think actually that's a good model' (October 2018).

#### 4.5.3 Sunderland and South Tyneside – following the leader

Sunderland Council met on 21 March to consider its advice to the cabinet. Views were generally favourable. The (ceremonial) mayor, summed up: 'Council appeared minded to support the devolution proposals and the creation of an elected mayor'. The deputy leader warned of the financial consequences of rejection and said 'the mayoral model proposed would ensure significant control for local leaders' (Sunderland Council, 2016b). Sunderland's cabinet met two days later, supported the deal, noted the outstanding issues and agreed to discuss the subject again in May (Sunderland Council, 2016a). Cabinet members remained supportive when they met on 16 May, but still cautiously retained the possibility of rejection if not satisfied with the final details (Sunderland Council, 2016c). As late as 31 August the cabinet still felt unable to commit itself unreservedly, so it authorised its leader to agree to the next steps if satisfied with further clarifications (Sunderland Council, 2016d). A councillor said later:

[Council leader] was very open with the group, and rather than having to delay anything that was happening he got the group's agreement that he was in a good place, he was clever enough, he knew what was going on, he lived and breathed it, and any decisions that were needed quite quickly that he could actually do that, make that decision (Sunderland councillor. March 2018).

In the event, the Sunderland leader, who also chaired NECA at the time, voted against the deal. Sunderland's records of its 31 August cabinet suggest why he did so: 'Since cabinet's deliberations in May, there have been the significant events of the European Union referendum result, and the change of Prime Minister and cabinet membership' (2016d). Whether anything else motivated the Sunderland leader to vote against the deal, with EU and national events just a rationalisation, is unknown; he died in 2017. A

campaigner against the deal told this study: 'His late conversion was strong and critical' (November 2017).

The Sunderland councillor quoted above believed it was funding, not the mayoral issue, that was the deal breaker. Given the changes of government that might occur, he did not believe 30-year funding promises could be guaranteed. Besides, the money did not compensate for the £400m of cuts that the council had already suffered (Sunderland councillor. March 2018).

In neighbouring South Tyneside they thought, and still think, NECA could have got a better deal in 2016, and still could if only the North of Tyne three had not broken away. A leading councillor said:

There was a perception from some people that once you don't agree this deal...that's it, the door is shut. I don't believe that. The North East is too important a part of the national economy not to be at the table discussing the agenda with government. So I think they would have come back to us in future to say: 'What is it about the previous deal you didn't like, what is it we can do to get you over that finish line?' So I was hopeful and I still think they would have done that (South Tyneside councillor. March 2018).

Given a good enough deal, a mayor might have been acceptable to South Tyneside, but it shared Sunderland's worries about the reliability of future funding. A business leader, however, argued pragmatically that local government had to work with central, even if it disagreed with it, to get the best it could for its area (Former NELEP board member. January 2018). A politician agreed: the south of Tyne councils made a major strategic mistake by not doing so (Conservative councillor. May 2018). A Labour supporter of the deal said, in response to fears that the money might be cut off, that keeping it flowing would depend on how successful one was in using it, and argued in frustration for a quite different approach:

One has to take opportunities to make things happen. Sometimes there's a bit risk. But it's about your passion and your enthusiasm and your strategy going forward and the vision you have; that's the way things happen. It's not about saying 'Oh well, this isn't any good, we're not going to do that'. We would never

do anything. No one would ever do anything if they weren't prepared and didn't have the enthusiasm to make things happen. So you find a way. And people have said to us 'Oh you're going in with the Tories'. What other way could I go to make things change in the area? That's the only thing that's on the table (Labour council member. March 2018).

South Tyneside's Labour group received regular updates from the leader. According to one account: 'Their view was this was a fast-moving agenda, it was changing daily, and that the leader needed to be given scope to use his judgement as to whether or not at the end to sign up on behalf of South Tyneside' (Interview, March 2018). The cabinet discussed the deal three times between March and August and left it in the hands of the leader. The last occasion saw a hint that members hoped the deal would proceed: subject to clarifications, and the approval of all the NECA authorities, they endorsed the requisite conclusion that the new arrangements 'would be likely to improve the exercise of statutory functions in the NECA area'. But in the end the South Tyneside leader voted with the majority against the deal (South Tyneside Council, 2016a; South Tyneside Council, 2016b; South Tyneside Council, 2016c). The Sunderland and South Tyneside votes are consistent with the view of a NELEP board member close to the talks who told this study:

If I was the Sunderland leader, why would I run the political risk, just for the greater North East? Sunderland has always been a problem like that. South Tyneside - the leadership in South Tyneside - just follow where Sunderland go to. It's connected (NELEP board member, private sector. January 2018).

This was rejected in South Tyneside, and the suggestion that South Tyneside, though considerably smaller, always follows Sunderland's lead is too crude a characterisation. The two areas are adjacent geographically and close socio-culturally. At the popular level, Sunderland Football Cub has three supporters branches in South Tyneside (Sunderland Assocation Football Club, undated). They share a post-industrial heritage with Newcastle but without the same level of commercial, cultural and architectural assets, and a common perception that economic growth in the regional capital does not ripple outwards to their areas. The two do a lot of cross-boundary working, said a

leading councillor, but on the basis of equality (South Tyneside councillor. March 2018). In as far as it is possible to judge, that seems a fair assessment.

#### 4.6 Postscript

There was surprise in Whitehall when news came through from Sunderland by text on 6 September 2016 that NECA had rejected the deal. Civil servants had been watching seven individual agents struggling with the historic, social, cultural and political structures of the North East and been expecting they would emerge from the battle to sign a deal. But the forces of the *status quo* were too strong, and they did not understand why. One civil servant said:

We knew what Gateshead's position was. We knew that Sunderland and South Tyneside had leeway, that their leaders had personal decision-making to go either way. We thought Durham were safe. In the end it was Durham we misread, because what we didn't have visibility of was discussion within Durham Labour Party; we only had official-level discussions so we missed that Labour Party policy. It is always difficult for an official to understand what goes on within Durham Labour Party, and with [Durham council leader] unable to back the deal it fell. We still could have got a majority but I don't think either [Sunderland leader] or [South Tyneside leader] were willing to take the risk...[Gateshead leader] was clear about what he was going to do although he also, if it looked like the vote was going to go, our understanding was that he would probably throw his hand in with the majority (Civil servant. January 2018).

This failure by the civil service to understand what was going on within NECA was matched by lack of insight on the part of the NECA leaders into the thinking of government, epitomised by a revealing incident in the middle of the 2016 summer holidays, recounted by a leading councillor (NECA leader. March 2018). On 6 August a report appeared in *The Times* that the Prime Minister was planning to ditch the *Northern Powerhouse* brand, switching the focus from Manchester and other northern cities in favour of a wider industrial strategy including areas further south. It threw the NECA leaders into confusion, and they and their chief executives held a two-hour conference call, one from the balcony of his holiday hotel, to discuss the implications.

Lacking Whitehall contacts – unlike their Tees Valley colleagues, as we will see in Chapter 5 - they were in the dark.

### 4.7 Conclusion

This chapter has narrated the roles of four of the North East's civic leadership groups identified by Hambleton in its devolution decision. This was the point at which the concepts discussed in the preceding chapters – formal and formal institutions, relations with government, political practices such as the private meetings of party groups, and the political tradition of collective in preference to mayoral leadership moved from the conceptual to the actual, empirical and real levels of critical realism and made their mark on the political governance and hence economic development of the NECA area.

The chapter has provided a narrative answer to research question 1 as it applies to the North East area: why did NECA reject a devolution offer from the government. That answer may be summed up as dissatisfaction with the £30m annual funding offer and resistance to an elected mayor. These grounds for rejection were facilitated by a sociocultural-political context in which rivalry was pervasive, trust and common purpose were absent and the power to decide was in the hands of a few hundred Labour councillors meeting in private. They were pressured to accept the deal by their professional advisers and business leaders, but pressure against the deal from within the Labour and trade union movement proved stronger.

NECA's institutional evolution lends itself to analysis in terms of the processes of layering, conversion and recombination discussed in connection with the work of Martin (2010) and Mahoney and Thelen (2012) (2.1.3). The North East's local authorities were initially able in 2014 to form a CA because it involved a recombination of existing authorities to take responsibilities which those authorities granted it and left each of their leaders with strong influence, if not an absolute veto. As part of the CA's procedures and in a layering process, they added co-operation with another structure, NELEP. Having established NECA, however, a majority were then resistant to seeing it displaced by or converted into an MCA for reasons explained by Mahoney and Thelen (2012: 19):

Instead of displacement or conversion, drift or layering [the addition of new rules, procedures or structures to existing institutions] are more promising as strategies of change in political environments with strong veto players. This is true because drift and layering do not require making any direct changes to the old institutions and do not rely on altering the rules themselves or actively shifting their enactment.

A veto player has been defined by Tsebelis (1999: 593) as an individual or collective actor whose agreement is necessary for a change of the *status quo*. While powerful veto players can protect old institutions, according to Mahoney and Thelen (2012: 20), they cannot necessarily prevent the addition of new elements. Thus, while four NECA councils were able to stop it becoming an MCA, they could not stop three other councils breaking away and establishing a new MCA alongside which the four would have to work. Only the three North of Tyne councils were able to recombine, in conjunction with new resources and properties, to form a new structure, a devolved CA with an elected mayor. The NECA majority had shown themselves to be 'opportunists' who, rather than trying to change a *status quo*, exploit it to achieve their ends and can be a source of institutional inertia:

Their preference of making use of existing possibilities over the riskier strategy of mobilizing for change makes opportunists – through their inaction – 'natural' (de facto) allies of an institution's supporters. Thus...opportunists help explain why changing an institutional status quo is often far more difficult than defending it (2012: 26-27).

In Chapter 5 we will see the very different institutional processes and outcome that resulted from the devolution process in Tees Valley, where trust and common purpose were present between key actors.

## Chapter 5. Turning Tees Valley blue

This case study, like that of the North East (Chapter 4) develops key concepts of section 1.3 and builds on the economic theories and political issues raised in Chapter 2 and contextualised regionally in Chapter 3 to provide a narrative answer to research question 1 as it relates to Tees Valley, asking why it accepted a devolution deal. The narrative of this chapter finds the informal institutions of Tees Valley, its socio-cultural background and political attitudes, to be subtly but significantly different to those of the North East, and they led to very different results from those to the north.

Trust, recognised conceptually as a vital governance ingredient in the literature review (2.2.1), is confirmed as having played exactly that role in the concrete reality of Tees Valley. The key concept of dominance (1.3.2) had an important reactive role in giving rise to resistance to the perceived leading regional position of Newcastle. It was supplemented by a positive concept of local place and identity, also introduced in section 1.3.2., contextualised in section 3.1.1 and now consummated in Tees Valley's devolution deal.

As well as being an institutional case study of Tees Valley this is also a study of the concept of leadership (1.3.3) and mayoral governance (2.2.4) in the person of the metro mayor elected in May 2017.

In Tees Valley, this case study finds, though the devolution decision lay with politicians, unlike in the North East they were not engaged in a struggle with business or in disagreement with officials. It was after the devolution deal had been done and a Conservative mayor elected that controversy arose.

It is notable that Tees Valley's civic leadership was and remains essentially a duopoly of politicians and business leaders rather than the partnership of five civic leadership groups envisaged by Hambleton. While public officials played their full role as professional advisers, there is no evidence that they steered policy, probably because they were in any case in agreement with the direction their political bosses were taking and so did not need to. Nor have either the documents examined for this thesis or the interviews conducted yielded any indication that either the trade unions or the

voluntary sector played significant roles in the devolution debate. Indeed, compared with the NECA area, there wasn't much debate at all at that stage. There was no equivalent to the CARD campaign against their deal.

The chapter describes how Tees Valley became separated from the wider North East in 2010, an essential precursor to the formation of a CA, and then signed a devolution deal and elected a Conservative mayor to lead a Labour dominated area, and the transformational effect that result had for Tees Valley politics. It describes the way the mayor has used popular politics and proactive public relations to impose himself on the area and implement his most high-profile manifesto pledge against the opposition of his cabinet. It describes how he has raised his own profile and that of Tees Valley at the national level. Finally, it discusses the possibility that Tees Valley may be reunited with the wider North East region.

#### 5.1 Tees Valley Unlimited: a basis for collaboration

Tees Valley Unlimited (TVU) was a public-private development organisation with roots in collaboration going back to 1996. It brought together council leaders and their officials, civil servants, higher and further education and private sector representatives including the chair. Five TVU groups dealt with economic development; place; transport and infrastructure; employment and skills; and innovation (TVU, 2014d: 1; TVU, 2016a). From 2010, as a formal and stand-alone LEP, TVU had a short life which, if not uneventful, was undramatic.

Its archived papers, covering 2014 to 2016, demonstrate a pragmatic strategy aimed at getting the most it could from the government for Tees Valley by doing whatever was required. TVU's minutes show a recognition from the start of the need for change and to tell the government what it wanted to hear. TVU's chair said that, following the securing of a city deal, TVU must look at how it could further improve its performance; times were changing and TVU would have to change with them, including more involvement of the private sector in developing the strategy and regeneration agenda (TVU, 2014a: 5). Ahead of a meeting with Communities Secretary Greg Clark, TVU noted that the government wanted evidence of solutions:

We are not asking for extra money but a policy change or a way of spending existing money in a better way... [We] can demonstrate to Greg Clark we are taking responsibility for managing rather than asking for money (TVU, 2014b: 3. 5).

TVU was effectively anticipating advice published the following year to those seeking devolution to 'give the confidence to devolve' by demonstrating ability to deliver within the framework of the government's approach (Grant Thornton and Localis, 2015). TVU certainly wanted all the help it could get but its approach, unlike the North East's, was based on a philosophy of 'ask not what the government can do for you, ask what you can do for the government'.

For Tees Valley, within the limits of what was available, it worked in the sense that it was able to carry on, and face the forthcoming changes brought by devolution, without institutional disruption – though its politics was transformed. That is not to say there were not setbacks, and unfortunately for TVU its accommodating approach did not bear immediate fruit. In an 'unsatisfactory' meeting, Clark 'attempted to dampen down expectations on what to expect regarding money'. TVU did not, however, dwell on its disappointment but continued to look for ways forward. Members noted the previous [Labour] government had focused on supply-side solutions to economic problems while the present [coalition] administration focused on the demand side. TVU should go forward with 'a couple' of transformational projects rather than a raft of supply-side projects. There was also a recognition that with an election due in 2015, the government was keen on projects that could visibly be delivered quickly, a message that was repeated following a meeting with the government, when there was mention of the importance of 'shovel-ready' projects which could demonstrate an early impact (TVU, 2014c: 2-3; TVU, 2014d: 4).

TVU's work throughout its lifetime was routine until, three months before it entered into a new relationship with TVCA, the closure was announced of the SSI steelworks in Redcar, a shock to the Tees Valley economy that cost 2,800 direct and supply chain jobs. TVU's involvement however was secondary, working with affected companies in the supply chain, with Redcar & Cleveland Council taking the lead (TVU, 2015b: 3-4; TVU, 2016a: 3; Redcar & Cleveland Council, 2017b: 4).

Devolution and possible changes in its own status first appeared on TVU's radar, as it did in the North East, in October 2014, following the Scottish referendum. With devolution, members were told by their officials, the private sector would sit at the heart of the strategic development of functions such as economic development, transport and skills. Thought needed to be given to the powers, functions and resources that the area may wish to see devolved, and in January 2015 TVU agreed to review its make-up, membership, support groups and relationship with the prospective new combined authority. Otherwise, Tees Valley's devolution deal and its implications for TVU and its relationship with TVCA caused few ripples as the process went through its formal stages over the following year.

In October 2015 the proposed devolution deal, which included the transfer of 'significant powers' over employment and skills, transport, planning and investment, the power to establish mayoral development corporations – which saw South Tees get the first such development corporation outside London - and a £15m annual grant for 30 years, was described by one of the council leaders at a TVU meeting as 'good for Tees Valley and would make it less reliant on others' (TVU, 2014f: 4-5; TVU, 2015a: 4; TVU, 2015b: 4). Unlike their NECA colleagues, the Tees Valley leaders did not change their minds.

#### 5.2 Tees Valley Combined Authority – normal politics

TVCA started life in 2016 with enthusiasm, as had NECA two years earlier. The opportunity for the CA was enormous and the plans that were in place or taking shape were an extremely exciting prospect, said the chair. This enthusiasm was given added impetus almost immediately by a report in June 2016 from Lord Heseltine following his appointment by the government to advise on the Tees Valley economy in the light of the SSI closure. Heseltine's report (6.4.2) was said to set out a bright future for the area, and an analysis demonstrated that it was largely complementary to the existing SEP, as had been hoped (TVCA, 2016b: 2; TVCA, 2016c: 13-14).

Enthusiasm for devolution did not, however, dispel all doubts, particularly in the wake of the Brexit referendum and subsequent changes in government. TVCA members, like their NECA neighbours, felt in August 2016 that given all that had happened reassurance was needed about devolution and the commitments that had been made. They were worried about the effects of Brexit on funding, regulation, exporting, FDI, the attraction and retention of talent and international knowledge transfer. But, unlike in the North East, doubts did not halt the process, and all the while plans were being laid for the establishment of a Tees Valley land commission to maximise the use of brownfield sites held by government departments and agencies to support economic development and housing supply (TVCA, 2016c: 3-13).

By early 2017, even before the metro mayor was elected, TVCA was starting to see the fruits of its deal. Proposals had been developed for a Tees Valley Housing Agreement under which the government would make available over £100 million for an accelerated home building programme enabling more than 2,500 good quality affordable homes to be built by 2021. However, said officials, the proposals went much further by providing a platform for working with the Homes and Communities Agency (HCA) and key local partners to regenerate communities, create jobs, support the SME sector (small and medium-sized enterprises), diversify market capacity, promote innovation in home building and attract new sources of investment (Waters, 2017h). The provision of housing funds was later to prove a bone of particularly bitter contention for one NECA leader because of the preference given to MCAs while the south of Tyne authorities had received nothing (NECA council leader. February 2018).

On 4 May 2017 a Conservative councillor from Stockton was elected as Tees Valley's first mayor. The narrowness of the Tory victory is evident from the statistics: the Conservative won by 48,548 votes to 46,400, counting second preferences, on a turnout of 21.3% on a day when the Conservatives gained 563 seats and Labour lost 382 across Britain (BBC, 2017; Stockton Council, 2017). It was the start of a new era, the novelty for much of North East England of a Conservative in power, and a new style of popular politics. That a change of control should have been a shock says much about the politics of North East England. A civil servant, watching Tees Valley devolution unfold at close quarters, regarded the election of a Tory mayor as having brought 'normal politics' to the area:

Political conversations in Tees Valley have tended to be within the Labour Party or between Labour and Independent. I think what it has done is say 'well actually there is a different political dynamic here that is more normal around

the country'. There are tensions obviously between a mayor and the leaders in terms of political outlook and way of doing things but actually it is normal politics. It is nothing untoward. It is probably something a little different from what you would expect in Tees Valley. It will probably chafen things up a bit. (Civil servant. January 2018).

5.3 Political pragmatism: no mayor, no deal; no deal, no money. A Tees Valley MP, asked why TVCA had accepted a devolution deal while NECA had not, blurted out instinctively: 'It was the money. It was the money'. Then he added more thoughtfully:

It's a relatively small amount of money, but facing an era when the local authorities have lost half their income and simply don't have the resource to invest in economic development promotional work, it was an opportunity for some money and of course it would be able to drive additional match funding over time. So that's why they did it (Labour MP. February 2018).

There is widespread agreement with this interpretation in Tees Valley, and the blow to the area's economy caused by the SSI steelworks closure only served to confirm local politicians in their pragmatic view that they needed every penny they could get, even if it meant a accepting a mayor. One council leader explained: 'We absolutely did not want a mayor. But the choice was "no mayor, no deal; no deal, no money"' (TVCA council leader. March 2018)

The SSI closure was a blow which affected prospects for a devolution deal in Tees Valley and the area's ability to regenerate the economy in contrary ways. On one hand, said a local official, local politicians didn't want the deal to be seen as a response to the steel works closure. They wanted the government to be on the hook for the consequences of the steel works closure over and above the deal. In that sense it could have been a disruptive factor (Public official, Tees Valley. January 2018). On the other hand – and in the end this weighed more heavily – the closure made councillors realise how badly they needed all the economic tools they could get.

A private sector representative close to the issue argued that the mayoral development corporation resulting from devolution was an important factor in dealing with the SSI closure because the compulsory purchase and planning powers that went with it enabled projects to be fast-tracked. Compulsory purchase was a Sword of Damocles which one hoped never to have to use, he said (LEP board member, private sector. March 2018) – though it was used in 2019 in relation to the SSI site.

While funding was the up-front reason for Tees Valley's pragmatic decision to accept a deal, it was made possible by the very underlying factors whose absence in the North East made a deal impossible there – the trust seen as so necessary for governance by Rhodes (1996) and Jessop (1998) (2.2.1) and the collaboration between at least two of the civic leadership groups of Hambleton's framework – politics and business (2.2.4). A private sector representative on Tees Valley LEP said business would not have been co-operating with the five local authorities for the previous 20 years if they had spent that time fighting each other (LEP board member, private sector. March 2018). An official said:

There's certainly good strong personal relationships between the leaders of the five councils; no one should underestimate that; I think they trust each other (TVCA official. January 2018).

If there was trust and collaboration between the Labour leaders and with the private sector, in an area with a significant number of Conservative politicians, as well as a Tory government in place, at least a modicum of trust across party lines was important as well, and its presence in Tees Valley meant the area had friends in high places when it needed them. Tees Valley actually has a significant number of Conservative councillors (3.1.1), and a Tory MP – at the time James Wharton, *Northern Powerhouse* Minister. It also experienced the involvement of Lord Heseltine, and Greg Clark, Communities Secretary and a Middlesbrough native. Though none of these was decisive in negotiating the devolution deal, they did help smooth the way. A Labour council leader said the involvement of Wharton and Clark was 'quite instrumental in enabling us to deliver the devolution deal here, because as a small area we probably would not have had one had it not been for those two people' (March 2018). A Labour MP said of Wharton: 'Despite all our political differences there are things that we have

worked on very closely together. The fact that he was a DCLG minister did actually help' (February 2018). According to a civil servant, Tees Valley 'had the right national politicians who were backing their case, who were willing to put in a bit of hard miles – Greg Clark and James Wharton – both prepared to put their political capital on the line to get something for Tees Valley' (January 2018). A business representative on the LEP said Wharton helped smooth the way when difficulties arose (LEP board member, private sector. March 2018)

The Labour leaders were suspicious when Heseltine was appointed to advise on regeneration in 2015, but his personality and their pragmatism created a working relationship. A civil servant who witnessed the process said: 'They made a friend in Lord Heseltine, who deployed his considerable personal charm' (Civil servant. January 2018).

As a result of all these factors – Tees Valley's history of collaboration between councils and with the private sector, political support across party lines and the spur to action provided by the SSI closure – the five council leaders confirmed their deal with the government just as their NECA neighbours were rejecting theirs.

The Tees Valley council leaders had, in theory, as much veto power as their NECA colleagues and could have prevented devolution. The fact that they did not do so is due to the fact that they were pragmatists and were prepared to make the most of a governance system not of their devising or choice. They were, to use the phraseology of Mahoney and Thelen (2012: 24), 'mutualistic symbionts'. This type of change agent 'thrive on and derive benefit from rules they did not write or design, using these rules in novel ways to advance their interests'. They are not linked to any specific mode of change (van der Heijden and Kuhlmann, 2016: 551).

#### 5.4 Devolution: business builds on the past

If Conservatives played a role in Tees Valley's devolution deal, it was a natural extension of the part they had fulfilled five years earlier in establishing the area as one of the coalition government's FEAs, separate from the NECA area and, importantly for them, from Newcastle. A Tory strategist and businessman recalls that even before the 2010 general election, in anticipation of a Conservative government, the local business community started pressing for the area to be in the forefront of any changes:

The Tees Valley business people said 'We always lose out. The North East is divided. Newcastle gets all the goodies' and so on. And [Eric] Pickles [Shadow Communities Secretary] said 'Look, we are going to have these LEPs'...[T]his group of businessmen... got themselves active, because in Teesside, it's such a small area, we all know each other, whether it's Labour leaders of the local authorities, Tory business people like me, we know each other, we are friendly, socially friendly, and we get on well (Conservative businessman. June 2018).

Tees Valley's history, with its well-established TVU, gave it an advantage. TVU was a ready-made vehicle to serve as a LEP and therefore, according to the account of the businessman quoted above:

Bang, we're in! And that's where the division [from the North East] happened. Speed, agility and a certain determination from a cohort of people in business and politics in Teesside – Tees Valley, I should say... And that's the forerunner of the subsequent devolution. (Conservative businessman. June 2018).

Tees Valley's momentum was maintained after the 2015 general election. Within days the Prime Minister flew in and named Wharton as the *Northern Powerhouse* Minister, opening the door for the area to choose devolution.

This version is a story of collaboration transcending party, crossing the boundary between politics and business, of urgency and enthusiasm. Nothing like it was evident in the NECA area. The same Conservative strategist said: 'I feel very passionate about it. I really, really want it to work'. There is approval too from the politically uninvolved. A local business support executive active under both the RDA and TVCA, said Tees Valley was the 'forgotten element' under the former whereas now, he suspected, Newcastle and the North East LEP were looking on with envy (Business development executive, Tees Valley. October 2018)

Events are seen very differently by some others. A regional business leader who was personally involved described his memory of events when Tees Valley was first separated from the North East:

[Cabinet minister] told me personally to my face that Eric Pickles was just philosophically determined to break up regions and believed – I'll never forget the word he used – regions are evil, which is an incredibly strong word, and of course had as a henchman Greg Clark, who was a Middlesbrough lad, who very much, they worked tirelessly to get Teesside to break away because they wanted to break up the region. So although they said this is about local determination, and we had come together totally of our own free will over the previous six months and agreed that we were better off staying, they worked tirelessly behind the scenes to break us apart, and it was quite a difficult time actually. (Regional business leader. August 2018)

That is a similar view to that, already noted, coming from a Labour activist opposed to NECA's devolution deal (3.1.1). But whatever the party political motives of the government may have been, Tees Valley's localism rather than regionalism is in line with the findings of Hambleton (2015a) and Kelly (2016a) (2.2.3) as well as the findings of local identity in the North East outlined in section 3.1.1 that people are attached to very local places.

If Labour councillors in Tees Valley took advantage of government policy for their own ends, to get what money they could for development, they equally had no illusions about the Conservative Party's political motives in offering Tees Valley a devolution deal. The Tories, one leader said, see Tees Valley as a starting point for their ambition to turn northern England blue (TVCA council leader. March 2018).

These contrasting interests – Labour's desire to win some extra funding and the Conservatives' electoral ambitions – both pulled Tees Valley in the same direction, towards a deal, for it meant both sides had something to gain. Pay-back for the Conservatives began on 4 May 2017, when Ben Houchen was elected as Conservative metro mayor.

## 5.5 Serious economics meets popular politics

It's a local delicacy, probably about 3,000 calories of pork escallop covered in cheese, deep-fried...best served up after people have had several gallons of beer at 2 o'clock in the morning (Labour MP. February 2018).

That mocking description of the Teesside parmo by a Labour MP symbolises the tensions between Tees Valley's Conservative mayor and the area's Labour establishment which provided the five council leaders who served in his first cabinet. For the parmo featured in the Tory candidate's mayoral campaign. He campaigned for the delicacy to be given Protected Designation of Origin (PDO) status, and it gained him headlines: 'We should be proud of the things we produce, including the world famous parmo', he told a local paper (Smiles, 2017). For an ally of the mayor, mockery of the parmo is replaced by admiration of the way that the mayor continued to exploit it in office:

The parmo is populist and fun, and he made the Chief Secretary to the Treasury [Liz Truss] eat a parmo in public (laughs) on video...The day he gets Theresa May to eat a parmo I think he'll have completely succeeded<sup>29</sup>. (Conservative businessman. June 2018).

He qualified his description of Houchen's campaign as populist by adding:

It certainly caught the mood of politics. It wasn't extreme; it wasn't populist on things like immigration and race and things like that. It was populist in economic terms though...Ben [Houchen] definitely ran a populist campaign, but populist should not be used as some kind of pejorative statement. Just because it's populist doesn't mean it's wrong (Conservative businessman. June 2018).

The parmo was not the only Conservative election issue which Labour politicians portrayed as far removed from the strategic economic plan (SEP) which it is the main

<sup>&</sup>lt;sup>29</sup> The word 'populist' during this interview was not and is not intended to imply association with radical right attitudes that have come to be connected with it, as is clarified in the quotation immediately below. 'Popular' is the word used elsewhere in the thesis in this context.

responsibility of the CA which the mayor heads to deliver. The mayor's manifesto priorities included buying back and revitalising the struggling Durham Tees Valley Airport (DTVA), in which the five local councils and Durham County Council had a minority share; and setting up an independent commission to review the structure of Cleveland Police, which the candidate regarded as a failing force (Lloyd, 2017). The police, like the parmo, continued to occupy the mayor's attention as he repeatedly criticised the [Labour] police and crime commissioner, holding him responsible when, in September 2019, the force was graded inadequate by HM Inspectorate in all three areas assessed – effectiveness, efficiency and legitimacy (Metcalfe, 2018a; HMICFRS, 2019; Brown, 2019a; Houchen, 2019b).

The parmo, the airport and the police were all, in the eyes of the mayor's political opponents, causes adopted purely to win a position to which they were irrelevant. But it was these touchstone issues, coupled with the Conservative Government's relative popularity at the time - after the Brexit referendum and the triggering of Article 50 but before the 2017 general election – that the mayor's ally quoted above believes resulted in the Tory victory. Of these, it was the plan for the airport that was most significant, and was to result in a striking political triumph for the mayor two years later.

The airport controversy draws attention to wider issues around elected mayors: their budgetary control, their power relations with their cabinets, and the extent to which they can dominate their area and transform its politics through force of personality and PR skills. It is therefore an episode that merits detailed attention and invites analysis in terms of the arguments for and against elected mayors listed by Hambleton (2015a) and noted in the literature review on leadership (2.2.4).

While the parmo issue does lend itself to ridicule, and the police do not fall within the mayor's remit, the airport is a legitimate concern of TVCA and its mayor. An official report in 2018 noted that the airport had a role in the SEP and forthcoming transport plan, and a continuation of its decline would be a significant risk to local economic growth. A significant ownership stake by TVCA could in principle provide substantial benefits (Lewis, 2018a: 2, 4). Disagreement over the mayor's plans can be seen simply as healthy politics, debated between parties, in public (as far as commercial

confidentiality allowed). It turned into a test case of the relative powers and, as decisively, political and PR skills, of the mayor and cabinet and as such is of more than local interest.

Post-election tensions over the airport first came to a head in early 2018 when the Labour leaders wanted to see £500,000 of airport-related spending included in the MCA budget go on support for developing new routes and retail investment in line with an agreement they had proposed, but not finalised, before the mayoral election. When cabinet members realised the money was to be spent on furthering the mayor's pledge to buy the airport, they blocked it and then voted through an amended budget under which the money would be used as they had originally intended (Lewis, 2018a: 3; TVCA, 2018b).

It was an important test of mayoral and cabinet powers because, as officials explained, as an amendment to the budget the spending could be agreed by a majority of the five council leaders excluding the mayor. However, this would achieve no more than set money aside in the budget; specific project approval would be required subsequently, and this would be subject to the TVCA's normal decision-making procedure requiring approval by both a majority of the leaders and the mayor, who thus had a veto (Wayman, 2018; TVCA, 2018b; Lewis, 2018d: 1-3). A council leader expressed confidence that the money would be spent as Labour intended, and regarded the budget as a battle won (TVCA council leader. March 2018).

Another leader anticipated a more nuanced outcome, however, with enough money both to support new routes and progress the mayor's purchase plan (TVCA council leader. April 2018) and this was reflected in a TVCA report that work was continuing on both purchasing a stake and supporting new routes (Lewis, 2018b: 2). In May 2018 TVCA agreed to spend £1m over three years on an 'air connectivity facility' to support routes to Europe (TVCA, 2018d).

If the cabinet thought after their budget success that they had curbed the mayor's ambitions, they under-estimated him. Late in 2018 talks to buy the airport were revealed to be heading to fruition, and the mayor used every PR lever to rally the public in support of the purchase and put pressure on his cabinet. With seven weeks still to go before the cabinet was to debate the purchase, the mayor tweeted a smiling

photo of himself at the airport over-stamped: 'Deal done; public ownership as promised'. In another characteristic move, he launched a poll to ask the public whether the airport's name should revert to its original 'Teesside International': 'Whatever you decide, I will deliver', he said. 'The people have spoken' the mayor commented when 93% of 14,000 participants voted for 'Teesside International'. Meanwhile, a local newspaper poll showed 'overwhelming' support for the mayor's purchase plan, with 76% backing among 1,966 participants (Wayman, 2018; Brown, 2018c; Houchen, 2018c; Lewis, 2018c; Brown, 2018d; TVCA, 2018d).

The PR blitz continued in 2019. In January, in quick succession, the *Northern Powerhouse* Minister visited Tees Valley and urged the public to get behind the mayor's plan; the airport's private owners said it faced an extremely uncertain future if the purchase did not go ahead; consultants reported that the area would take an immediate £57m hit if the airport closed, with 1,100 jobs at risk, but 1,800 new jobs in prospect if the purchase went ahead; and a local company employing 180 warned it might have to re-locate if the airport closed.

Forty businesspeople added to the pressure by writing to the Labour leaders urging them to support the purchase, which 'defines us as a first-class city region not the poor relation of Newcastle and Manchester'. Three days before the cabinet was to vote the mayor announced that the airport was in line for a heliport to serve a big new North Sea wind farm - if his deal was approved. A Labour leader complained that a gun was being held to their heads. But according to the mayor: 'History will not be kind to those who let emotions and personal politics get in the way of what the people voted for and rightly demand' (Metcalfe, 2019a; Price, 2019a; Tighe, 2019a; Metcalfe, 2019b).

The mayor was astute enough to know, however, that a major investment could not win approval solely on the back of a popular campaign. A commercial lawyer and businessman by background, he backed it up with a detailed business case, business plan and valuation report and introduced it personally, with visual aids, in a 35-minute presentation<sup>30</sup>. It convinced the overview and scrutiny committee (OSC), and it

<sup>&</sup>lt;sup>30</sup> Personal observation.

seemed the cabinet was splitting, with the Hartlepool leader announcing he was backing the plan (TVCA and Mayor, 2019; Metcalfe, 2019d).

Two crucial procedural decisions were issued before the cabinet finally voted on the acquisition: the cabinet won a ruling that its unanimous backing was needed - reinforcing its budgetary control – but were presented with the deal as part of a wider ten-year £588m investment package, all subject to a single take-it-or-leave-it vote.

With three days to the decisive cabinet meeting, Conservative, LibDem and Independent councillors representing a majority on Redcar & Cleveland Council, where Labour was running a minority administration and where the Labour leader was perhaps the most prominent of the TVCA cabinet to question the plan, wrote an open letter urging her to back the investment package. Simultaneously, with a threat to further mobilise public opinion, a campaign group, Save Teesside Airport, said it would stand in council elections across Tees Valley if the purchase was blocked.

The scene was set for a showdown, and a conference room in a local hotel was booked for the cabinet meeting in anticipation of a rare display of public interest in local government. Then, one by one, the remaining cabinet members announced they would back the plan; one of the last, two days before the vote, was the Redcar & Cleveland leader, who said she would do so 'despite real concerns'. The potential showdown turned into a triumph for the mayor as hundreds of members of the public applauded his plans in front of a bank of TV cameras<sup>31</sup>. He had combined political, professional and PR skills to find a way of spending £40m on a project which required the unanimous support of his cabinet and which they had all earlier opposed. He was publicly gracious in victory, defending his cabinet against criticism and saying the deal had been improved by their robust scrutiny. He had made the most of the opportunity to display leadership before the largest possible audience.

Tees Valley's local newspaper endorsed the deal, said the Labour leaders had performed their duty in properly examining the plan 'in the face of a clever PR campaign by the mayor's office' and congratulated all concerned. Within days the mayor was meeting potential commercial customers to promote the airport (Brown,

<sup>&</sup>lt;sup>31</sup> Personal observation.

2018b; Metcalfe, 2018b; McNeal *et al.*, 2019; TeessideLive, 2019; Cain, 2019a; Wayman, 2019a; Metcalfe, 2019c; Metcalfe, 2019e; Metcalfe, 2019f; Metcalfe, 2019g).

The mayor exploited the publicity value of his airport triumph to the full for months afterwards, achieving headlines in mainstream media and posting on social media, personally or through TVCA, at every opportunity, such as the announcement of new holiday flights. A video was posted on 'the story behind our new airport brand' explaining how the design was developed (BBC, 2019b; Houchen, 2019g; TVCA, 2019h; TVCA, 2019i).

It was not long before the full political significance of the mayor's victory was felt. Council elections in May 2019 saw Labour suffer heavy losses (3.1.1) and retain only one cabinet seat. Though these results were part of the Brexit-induced turmoil affecting councils throughout England, a local newspaper described the mayor plausibly as the biggest local winner of the day, even though he didn't stand: 'His unlikely success in buying back Teesside airport enthused the regional party during the campaign, and now his cabinet, which is made up of the five council leaders, has a very different political complexion' (Lloyd, 2019). Mayor Houchen may not have turned Tees Valley blue in one go, as Independents were also winners at Labour's expense, but he made a big start by diluting the red, and in the general election seven months later, again dominated by Brexit, Tees Valley contributed three new seats to the Conservative victory, giving them four to Labour's three in the area (3.1.1).

In relation to the advantages of a mayor listed by Hambleton (2015a) (2.2.4) Houchen has certainly achieved visibility, not least through his airport purchase and as is further demonstrated below (5.6); and can claim legitimacy, having been elected, and accountability on the same grounds, though with *caveats* discussed in section 7.1.1. His airport purchase displayed strategic focus, though not on an objective that all agreed with. His authority to decide has been greatly strengthened by his airport victory, for which he had to fight hard. The stability of his own leadership will be tested at the mayoral election in May 2021, while the Labour Party's collective leadership of Tees Valley has been overturned by its election losses of 2019. New people have been brought into cabinet roles, if not into politics, for the first time. Views on the mayor's propensity for coalition building and partnership working will be contested. Some will

see his garnering of popular support in that light; others, as we will see in sections 7.1.1 and 7.3, see him as open only to those who can serve his own interests.

As to the disadvantages of mayors listed by Hambleton, most are inapplicable or not evident in this case – the wrong geography (except to advocates of regionalism), scope for corruption, weak power of recall, cost and over-centralisation of the state. The mayor is, perhaps, open to criticism for celebrity posturing, though opinions will differ on this subjective question.

A senior Tees Valley official believes the mayor does have a legitimate role in providing leadership on a range of issues outside his direct responsibility, as Andy Burnham has done in Greater Manchester on the issue of homelessness and Houchen has done in relation to policing:

I think a legitimate role for a mayor is to have a position on something that they don't have direct responsibility for but because they want to lead a particular policy position within the area...So mayors are not going to be constrained by the core functions of a combined authority; they are going to range more widely than that. And they should do, because that's a legitimate part of their role. They have a sort of thought leadership and convening power as mayors that is an important part of the role (TVCA official. January 2018).

The tensions caused by these issues came to be one of the defining characteristics of the political life of Tees Valley after the mayoral election, just as collaboration and pragmatism had been the watchwords in the run-up to devolution. Those tensions found expression in the irritation of some at least of the mayor's opponents at the high profile which he attained in the media. But this profile for the individual was accompanied by a raised profile for the area as well, as Tees Valley rose from being a neglected part of England's most lagging region to a flagship for government policy and Conservative electoral hopes in the north.

## 5.6 Raised profiles – geographical and mayoral

Having a mayor has raised the external profile of Tees Valley as well as of the mayor personally; the two effects go together. For Tees Valley, they are most obviously visible in a constant string of ministerial visits, including by the Prime Minister, who visited in August 2017 to launch the South Tees Development Corporation (STDC), where TVCA hopes to create 20,000 higher skilled jobs in manufacturing innovation and advanced technologies (TVCA, 2018e: 3). Visits can have substance as well as show. A business representative on the LEP said he had spent a lot of time talking to a visiting minister who was 'personally supporting and will support a number of projects that I can't talk about in the Tees Valley' (March 2018). A special economic area was declared on the STDC site, allowing the local retention of business rates (TVCA, 2018g). By early 2019 the government had announced £137m for the site and the transfer of 1,420 acres to the Development Corporation (Bell, 2019). 'It has taken 18 months of blood, sweat and tears but I am delighted to announce that a deal has been done' the mayor tweeted (Houchen, 2019a).

A Labour MP said Tees Valley had hosted more ministerial visits since the mayor was elected than in the previous seven years. Their aim was to exploit the situation politically, but that was a price he was willing to pay if it brought investment (Labour MP, Tees Valley. February 2018). A local official said Tees Valley was not an area of the country that people knew much about, but it was starting to get a reputation with national government and the private sector which was going to pay dividends (TVCA official. January 2018). A Labour council leader, while recognising the political gain accruing to the Conservatives, could see the opportunity a Tory mayor offered to exploit his profile on behalf of the area:

This is the only place in the north where they [Conservatives] have made any significant inroads and on that basis they need to make that mayor a success, and that means we are seeing reams and reams of government ministers coming here ...It doesn't matter whether they [ministerial visits] are window dressing and PR or not. What it means is there is a focus on this area that there wouldn't be otherwise (TVCA council leader. March 2018).

Access to ministers and a seat at some tables is a result of having a mayor that quickly became evident. TVCA's cabinet minutes for June 2017, a month after the mayoral election, record tantalisingly: 'The mayor left the meeting for a short period to receive a 'phone call from the Secretary of State' (TVCA, 2017d). High-level contacts are frequent. An early opportunity came in September 2017 when the Chancellor visited

the Northern Powerhouse to discuss job creation, transport and housing. He met the north's three elected mayors, including Tees Valley's, but the North East was unrepresented, angering politicians who had campaigned against NECA's devolution deal (Walker, 2017). Weeks later the mayor was given the opportunity to introduce the Chancellor's keynote speech at the Conservative Party conference and seized the chance to promote Tees Valley's development opportunities (Price, 2017; TVCA, 2017b: item 4, p. 2). Days later it was again the northern mayors who met the Brexit Secretary to lobby over replacement grants for EU funding (Bounds, 2017). A year later the mayors were still lobbying over the issue, with a joint statement from Labour's mayors in Manchester, Liverpool and Sheffield and the Tees Valley Conservative. The government confirmed it was engaging with 'mayors and mayoral combined authorities' (Hughes, 2018). Soon afterwards the Tees Valley mayor joined his peers from London, Manchester and the other MCAs at a meeting to call on the government for more devolved powers (McNeal, 2017b), giving what Teesside's local paper called 'bragging rights' over Newcastle (McNeal, 2017c). The mayor joined his Labour counterparts in Manchester and Liverpool in urging his own government against a 'Whitehall power grab' for EU funding (McNeal, 2017d). Within weeks concerns were placated and November 2017 saw TVCA awarded special powers under its devolution deal to take extra control over its share of EU funds and the proposed UK Shared Prosperity Fund (TVCA, 2017c; TVCA, 2017e: 5). A North East Conservative believes the ability of mayors to have their voice heard is at the expense of MPs:

Increasingly government ministers are starting to talk to the mayors first, before the local MPs, because they are covering a wider area and more significant population... The elected mayor is now the go-to person for any sort of consultation (Conservative politician. May 2018).

The media, too, he said, regarded the mayor as the first person to talk to, even on issues not necessarily under his/her control.

If a high profile has benefited Tees Valley, the Conservatives were hoping it would do the same for them and help with their aim of turning the north blue. A Tory strategist said: 'A Tory mayor in Teesside! This is a man with a serious incentive to do well for Teesside people' (Conservative business leader. June 2018). A civil servant agreed,

saying that having a Conservative mayor in a relatively weak Conservative government was helpful both to Tees Valley and the government. It gave the mayor a bit of personal power that he probably would not have as a Labour elected mayor (Civil servant. January 2018)

But one irritated local MP felt the mayor was being given credit for initiatives long predating his election (Labour MP. February 2018). Another said it was not just local authorities, but other organisations, that were affected by the mayor 'riding in at the top to cream off the publicity'. But he warned pragmatically against allowing annoyance to prevent the area taking advantage of opportunities presented by devolution (Labour MP. February 2018)

The reason Tees Valley's Labour leaders had not wanted a mayor, said one, was that it would disrupt the collective approach traditional in English local government described by Wollmann (2008) (2.2.4). The relevance of collectivism to Tees Valley is not so much to that within individual local authorities, for Middlesbrough has a borough mayor and Hartlepool had one between 2002 and 2012, but that between the five councils.

When the Prime Minister brought the cabinet to Gateshead in 2018 (4.1), ministers gave the Tees Valley mayor a high-profile meeting and he was praised by the Prime Minister in an article for regional media (Bell, 2018; Price, 2018b; Walker, 2018d).

The picture of a Tees Valley mayor who combines advocacy for his area with promotion of the interests of his party is confirmed, if not impartially, by an ally, who described the mayor as 'a very high-calibre individual, very passionate, very active and very dynamic, and he has made that job his own'. He was seen by the government as 'a totem of the northern outreach of Conservative devolved power' (Conservative business leader. June 2018)

A regional business leader said having a devolution deal made a huge difference:

You only have to look at Tees Valley. It might not be that you get all the powers and responsibilities that you want, but it's that figurehead. I think Ben Houchen is doing a far greater job in the past year than I think most people would have given him credit for. He is one of a select few of mayors across the country, and there's a lot of focus and...you can really influence people to make decisions (Regional business leader. August 2018)'

Another regional business leader, taking a still more positive view of the mayor, and playing down the party politics, emphasized the importance of personality over geography in ensuring devolution was a success. Asked about Houchen, the somewhat star-struck response was: 'He is great – a fine upstanding young man and he has got his heart in his community (Regional business leader. September 2018).

The view is similar from just across the Tees Valley border in south Durham, with a tinge of envy. A business support leader wondered who had the ear of ministers on behalf of Durham. While the county council was doing a good job, he said, it would have been better to have a mayor, pushing issues of local concern like ensuring rural areas of the county had access to the proposed UK Shared Prosperity Fund (EU funding replacement):

He's managing to shout about the issues that matter for the people of Teesside. Some of them might be white elephants, airports included, but he's doing it, and it's got a line in and he's starting to make a noise. Who's doing that for our county here?' (Business development executive, Durham. October 2018).

Not everyone agrees with these assessments of the Tees Valley mayor. He has his critics, as we will see in Chapter 7.

## 5.7 The return of regionalism: a latent threat

Today Tees Valley feels secure in its separation from the remainder of the North East, with its own formal institutions. The possibility of a re-unified region is absent under a Conservative government, newly re-elected in 2019, but that is not to say it may not one day re-emerge as part of a move back to regionalism. Hildreth and Bailey (2013: 244) believe that some LEPs are too small and 'at some point an intermediate scale of geography will have to be back on the agenda', and 'there remains the question of what happens to the LEPs that are not connected to a core city'.

What might happen under a future Labour government remains far from reassuring for Tees Valley leaders wishing to remain independent of the wider region. While Labour's

2019 general election manifesto did not mention abolishing existing combined authorities, mayoral or not, or LEPs, its commitment to establishing government offices in nine regions, regional development banks and a local transformation fund in each region would bring a significant regional element back into the governance mix (Labour Party, 2019b: 14).

The very idea causes anguished mixed feelings in Tees Valley, where the latent possibility of regional re-unification has long been present in the mind. In July 2014 TVU discussed two reports published by Labour on its thinking on the future of LEPs. One contentious issue was said to be the scale/geography of LEPs, with potentially fewer. TVU agreed it had to 'align' its case for retaining a Tees Valley LEP. Its chair said Tees Valley was a functioning entity and in a recent telephone conversation with Lord Adonis (chair of the North East Independent Economic Review) had made this position clear. Local Labour MPs were said to be supportive of the LEP. Nevertheless, one Labour council leader and TVU member 'felt the Labour Party would reincarnate the regional work' (TVU, 2014e: 2).

A Labour economist and campaigner, interviewed for this study, explained the case for regional rather than sub-regional governance. Only regions would be large enough to handle the level of devolution, particularly fiscal devolution, that he envisaged: 'I don't know how long it is going to take but that has to be the approach if we do genuinely want to avoid a situation of central control in England' (Labour campaigner. November 2017).

A regional business leader took a similar view, more ominously for Tees Valley because not obviously politically motivated. Tees Valley was too small to punch its weight on the national and international scenes. 'But they feel like a threatened minority, and threatened minorities always fight harder (Business leader. August 2018). This view was reflected by another business leader: Tees Valley had felt bullied by the RDA (Former NELEP board member, private sector. January 2018).

A Labour frontbench MP interviewed for this study was ambivalent about Tees Valley under a Labour government. He recognised the strength of feeling in Tees Valley against Newcastle but spoke in general terms of dealing with that issue 'within

something that enables us to work together rather than quarrelling over how many subsets there should be' (Labour MP. January 2018).

There are trade unionists, too, who would welcome the return of regionalism and even another attempt to establish a regional assembly. One union leader said, however, that to gain acceptance by business and voters any assembly would have to be elected by proportional representation to stop it becoming another fiefdom; even though his union was affiliated to the Labour Party, he thought people, and certainly business, would not buy into another layer of Labour bureaucrats (Trade union leader. October 2018). Another union leader said:

In a kind of romantic world that would be so much easier for us, to have the region working together as a collective, as we did with the RDA. But in terms of devolution, we have been there in 2004 when we had a proposal for a regional assembly. I don't think there's any appetite to revisit that (Trade union leader. October 2018).

Some regard those proposing reuniting the North East with scorn; a Tees Valley business support executive suspected those advocating reunification were based in Newcastle, where they had been unable to get their act together [as NECA] and form their own mayoral CA (Business development executive. October 2018).

There is another latent threat to the established order, though it is barely discernible at present. Crossing the River Tees outside the TVCA headquarters one passes road signs announcing one's arrival – depending on direction of travel – in the 'historic' county of Durham or the North Riding of Yorkshire. Two tests of public opinion in Tees Valley have taken place in recent years which, though not directly related to the devolution process, touch on issues of place and identity and could hold the seeds of a new division. Referendums were held in Yarm and Thornaby in 2014 and 2015 respectively on the question of seceding from the borough of Stockton, which was formerly in County Durham, with a view to re-uniting in an undefined way with Yorkshire, of which both historically formed a part, lying as they do on the south bank of the Tees. The people of Yarm voted by 90% and of Thornaby by 72% in favour of leaving Stockton, though the ballots were not binding and have not been implemented. The two areas are very different and the results do not appear to have

had economic motives: Thornaby has three wards among the 20% most deprived in England while Yarm is among the 20% least deprived (Stockton Council, 2014: 120; Keane, 2017).

These popular votes may have led to nothing, but ties between Tees Valley and Yorkshire remain strong. One of the biggest economic developments in the area, reportedly involving investment of £3.2bn, links the two through the mining of natural fertilizer near Whitby, North Yorkshire, and its processing at Wilton on Teesside. It is creating more than 1,000 jobs and is expected to add 17% to the economic output of North Yorkshire and 18% to that of Tees Valley (Brown, 2018a). Teessiders look south to the North Yorks Moors National Park and Whitby for recreation<sup>32</sup>. A regional business leader said:

Tees Valley now have a view that they wouldn't want to join that rabble to the north. And also they are led by a mayor who is fundamentally a Yorkshireman, and his views are looking south rather than north (Business leader. August 2018).

A politician with the North East Party (NEP) believes the South Tees area might one day choose to return to its Yorkshire roots, splitting Tees Valley and making it unviable (NEP member. 2017). Alternatively, the Tees Valley mayor, who has publicly described himself as 'a proud Yorkshireman' has speculated about welcoming neighbouring Yorkshire towns into Tees Valley (Cain, 2018a). Nevertheless, any threat comes at present from the north, and most of those interviewed for this study believe that Tees Valley is secure. Still, a Tees Valley MP reflected in a nuanced way that:

[T]hat doesn't mean we shouldn't lift our heads a little bit when the opportunity presents, and hopefully once the north of the region can work out what it wants to do and becomes established, then there's going to be an opportunity (Labour MP. February 2018).

<sup>&</sup>lt;sup>32</sup> Personal knowledge

### 5.8 Conclusion

The narrative of this chapter has facilitated an understanding of how the subtly but significantly different socio-cultural-political environment in Tees Valley enabled it to accept a devolution while NEA rejected its similar offer. Though Tees Valley, like its northern neighbour, was completely controlled by the Labour Party at the time, its leaders benefited from an atmosphere of trust and collaboration, while their pragmatic approach enabled them to accept a mayor in order to access the funding and powers of devolution.

Separation from the North East has enabled Tees Valley to establish its own placebased identity, given it recognition on the national stage and revealed how distinct it has now become, socio-culturally and politically, from the rest of the region. Tees Valley's politics has been transformed both in style, with the election of a high-profile mayor, and substance as Labour has been swept from power in four of the five councils and lost three parliamentary seats (3.1.1). It feels secure in its new institutional arrangements, and it is hard to see a return to reunification with the rest of the North East even though latent pressure for regionalism remains in some quarters.

Tees Valley's institutional evolution, seen in terms of the analyses developed by Martin (2010) and Mahoney and Thelen (2012) (2.1.3) is very different from that of NECA (4.7). Tees Valley went through an initial process of layering in which TVU become an official LEP, quite a small institutional change but laying the ground for a much greater cumulative transformation. The local authorities, which had previously co-operated with each other and with the private sector through TVU, then recombined to form the new TVCA which, in turn, recombined in a close new relationship with the LEP. TVCA then underwent a process of layering by accepting a devolution deal, thus adding new powers and resources and an elected mayor to its existing structure.

In the following chapter we will see how the different decisions made by NECA and Tees Valley affected their approaches to economic development and the consequences for economic outcomes.

# Chapter 6. Contrasting approaches to governing economic geography

This chapter describes and explains the different approaches of the two functional economic areas (FEAs) of this study to pursuing economic development: technocratic in the North East - an approach questioned in the literature as undemocratic and elitist (2.2.1) - and highly political in Tees Valley following the election of its mayor. It thus provides an answer to research question 2. It finds that the explanation for the different approaches lies not in the economic problems or plans of the two areas, which are quite similar, but in the different institutional arrangements following the decisions on devolution described in the previous two chapters, and in the politics and personality of the Tees Valley mayor elected in 2017.

The chapter explores the consequences of the economic geography of North East England that emerged from the introduction of localism in 2010 (1.2) and the devolution processes described in the previous two chapters. It describes how one FEA can become two if politically expedient. These FEAs may still, however, not coincide with other possible definitions of economic geography used widely by international bodies. What is more, none of these alternatives matches the geography through which politicians administer their municipalities.

The chapter goes on to discuss the economic background against which the two LEPs prepared their strategic economic plans (SEPs) in 2014, including the key concepts of path dependence and path creation (1.3.4), and the controversy over Newcastle's dominant role as a growth hub and potential centre of agglomeration (1.3.2, 2.1.1). It then turns to the SEPs themselves and discusses issues arising, including the targets they set themselves. Finally, it discusses the economic outcomes at the half-way points of the ten-year SEPs.

## 6.1 The mismatch of political and economic geography

Though LEP geographies were selected ostensibly because they were FEAs, we have seen in the previous chapter that the establishment of Tees Valley LEP was heavily influenced by socio-cultural and political factors. Local politicians and business leaders were keen to break away from what they perceived as Newcastle's dominance, and

Conservative ministers were keen to break up the region for ideological reasons and in the hope of party advantage. Once Tees Valley had broken away, there inevitably followed implications for the remainder of the North East.

Geographically the North East is a large area, even without Tees Valley – 106 miles by road from Berwick-on-Tweed to Barnard Castle – and diverse, with urbanised, postindustrial Tyne and Wear located between the largely rural counties of Northumberland and Durham. Yet dividing it between two LEPs would have been problematic. Had County Durham, for example, sought to form its own LEP – a possibility in view of the detachment from NECA expressed by some in the county (4.5.2) - it would have been the second smallest, behind Cumbria, with a population of 523,000 (Durham County Council, undated; Nomis, undated).

NELEP considers that its geography covering Northumberland, Durham and the five councils in Tyne and Wear is an FEA, basing its view on the facts that it is the sixth largest LEP in area, 96% of jobs in the area are taken by people living there, 79% of those living in the area work there, its industrial sectors rely on supply chains and assets located in the area, the housing market is strongly related to the labour market, and it has integrated transport governance. As NELEP does not overlap with any neighbouring LEP it did not see any need to amend its geography to comply with a government requirement that it reflect a 'real' FEA (NELEP, 2018c: item 9, pp. 4, 5).

Yet, when the political governance of the NELEP area broke down and NECA split, as recounted in 0, it was found expedient and possible to divide the area into two along the line of the Tyne and claim both were FEAs, creating the North of Tyne CA (NTCA) from the three councils supporting a devolution deal while the remaining four continued to administer the rump NECA. NTCA succeeded not just in gaining the required consent of a majority of the NECA seven to the requisite boundary change, but in meeting the legal requirement to show both that the creation of NTCA would be likely to improve the exercise of statutory functions in its area and that the change would do the same for what remained of NECA.

The rationale to support this position was provided by officials in an argument of two parts – economic and political (Melia, 2017: 5, 23, 50). Economically, officials argued that 'whilst working at a wider geography has the advantage of capturing more

community and supply chain interactions, this comes at the expense of economic linkages between some parts of the area being weak, making it difficult to have shared priorities'. North of Tyne it was argued, was an economic area in its own right, a 'functional economic market area' within the seven-council NECA FEA, while the rump NECA also, on its own, 'would remain a coherent and functional economic geography, and could therefore operate effectively'.

The political advantage was that splitting would liberate both areas from the disagreements that had held them back and allow them more scope to pursue their respective policy objectives – economic development and transport in NECA and human capital, housing and rural growth in NTCA. Without explicitly mentioning the factional disputes described in Chapter 4, officials stated that having two CAs would ensure more closely aligned policy objectives in each authority than existed within NECA at that time, enabling each to pursue a more diverse policy agenda and make and implement decisions more efficiently and effectively (Melia, 2017: 50).

This rationalisation came at a time when the government was itself uncertain about the basis for FEA boundaries and in its review of LEPs in 2018 said the time was right to revisit them:

There is no universally accepted approach to measuring or defining functional economic areas and boundaries vary depending on the method used. However, we acknowledge that economic geographies often cross administrative boundaries and we want to see continued collaboration between local enterprise partnerships and local authorities where this is the case (MHCLG, 2018b: 22-23).

The CBI has called for a clear definition of FEAs taking account of political, geographical and decision-making boundaries (CBI, 2019: 24). When Heseltine reported on devolution in 2019 he looked at the problem from another angle, recommending that the Boundary Commission should review local authority boundaries, starting with CAs (Heseltine, 2019: 60). A North East economist took the pragmatic view that the most important requirement was stability, whatever the geography (NELEP economist. January 2018).

FEAs, however flexible they may be to meet the requirements of politics, are not the only possible or even actual economic geographies. The EU and OECD both divide North East England into seven units<sup>33</sup>, selected on economic grounds, taking account of contiguous built-up areas and commuting. They recognise a single economic territory of Tyneside, taking in the four neighbouring metropolitan municipalities of Gateshead, Newcastle, North Tyneside and South Tyneside and with a population of 854,000. County Durham, Northumberland and Sunderland, under this classification, remain as three separate units. Tees Valley is divided into three – Darlington; South Teesside (Middlesbrough and Redcar & Cleveland); and Hartlepool and Stockton (European Parliament and European Council, 2003: 53; Eurostat, 2019).

The OECD and some literature also recognize a 'Newcastle' economy which had a population of 1,166,000 in 2016 making it the fifth biggest city in the UK, and arguably a regional mini-agglomeration (McCann, 2016: 124-125; OECD, 2018). The Centre for Cities recognises a Newcastle primary urban area including Gateshead and North and South Tyneside; Sunderland as self-contained; and Middlesbrough as including Redcar & Cleveland and Stockton (Centre for Cities, 2019b). Looked at yet another way, North East England has five functional urban areas – Newcastle, Sunderland, Darlington, Hartlepool and Middlesbrough (OECD, 2016).

Whichever of these economic geographies is considered, it does not match the local government boundaries of North East England, which is divided into 12 municipalities. As Ward and Hardy (2013: 4) put it: 'The trouble...with the messy world of work, business and enterprise is that it obstinately refuses to acknowledge municipal boundaries'.

Councillors are elected primarily to serve their municipality and, it is argued in this thesis, are in some cases so constrained by their sense of local identity and perceived accountability to municipal voters that they are unable to take the wider view needed

<sup>&</sup>lt;sup>33</sup> EU: NUTS3 (Nomenclature of Territorial Units for Statistics); OECD: TL3 (small regions).
for dealing with economic issues that cross boundaries. One consequence of this, as we have seen in the North East (Chapter 4), was that insuperable political difficulties were caused. LEP geography, when CAs were added, embedded municipal-political thinking into what was intended to be a means of solving economic problems, and led to the NECA split.

The political difficulties that made a NECA devolution deal impossible have much in common with the problem revealed a few years earlier by Rees and Lord (2013) in their work on the Labour Government's *Northern Way* initiative, with its eight city-regions, including Tyne and Wear: a clash between abstract ideas of what city-region geography should look like, based on economic rationale, and the realities of fractious local politics. There is little doubt, they say:

that explaining the actual city regions that have emerged in the north of England and accounting for their operation can only be accomplished by reference to qualitative variables including culture, power and politics...Experience is so variegated and the geographies that have emerged are so fundamentally determined by context-specific characteristics...that there is very little in the way of a coherent logic that can be sensibly applied to understand all eight (Rees and Lord, 2013: 692).

It is a widespread problem which the business leader quoted in section 1.3.1 called balkanization, a feature found by Scott (2019) to be global:

[The challenge of governance] is exacerbated by the persistent tendency to balkanization of municipal government in probably the vast majority of city regions... The political geography of city regions... is composed out of multiple municipal governments that have strong incentives to focus on their own localized interests at the expense of the wider regional community, and this leads in turn to dysfunctional forms of municipal competition (Scott, 2019: 16).

The fact that North East England encompasses 12 local authorities is reflected in the constitution of its LEPs and CAs, where all are represented. It is possible, and doubtless common, for municipal politicians to look at economic statistics in terms of how they are reflected in their local council area. This is where their constituents reside and

vote, but not necessarily where they lead their economic lives. Recognition of this fact, if given greater weight by local politicians, as in Tees Valley (6.2.2), might help overcome barriers to cross-boundary trust and cooperation. As the World Bank points out:

Which spatial scale to use, or how best to define a spatial area, depends on the issue and the information available. But the choice can dramatically affect the conclusion drawn from studying social and economic conditions across different parts of a country (World Bank, 2009: 78).

Economic statistics are very different depending which geography is used. Unemployment levels, a popular measure of economic performance, are a good example. Taking the figures at the time of the region's devolution decisions, none of the geographies confirms a simple picture in which Newcastle had gained the lion's share of the benefits of development. **Figure 6-1** shows that unemployment in Newcastle as a local authority was above the regional average.



Figure 6-1: Unemployment (%) July 2015-June 2016 by local authority. Source: Nomis

The NUTS3 classification gives a picture in which unemployment on Tyneside, including Newcastle, at the time of the devolution decision was also above the regional average, though below that of South Teesside, while Durham and Sunderland were average (ONS, 2017a) (Figure 6-2). Measuring by travel-to-work area (TTWA) presents a similar picture, with Tyneside again with above average unemployment, along with Hartlepool and Sunderland (Figure 6-3).



Figure 6-2: Unemployment (%) July 2015-June 2016 by NUTS3 area. Source: Nomis



Figure 6-3: Unemployment (%) July 2015-June 2016 by travel-to-work area. Source: Nomis

Taking a different economic measure, Newcastle was the local authority with the highest gross value added (GVA)<sup>34</sup> in North East England, but again the picture is not that simple, for the city did not see the fastest GVA growth between 1998, when Labour established RDAs, and 2016, suggesting it did not disproportionately benefit. That distinction goes to North Tyneside (ONS, 2017b) **(Table 6-1)**.

<sup>&</sup>lt;sup>34</sup>GVA is a measure of the increase in the value of the economy due to the production of goods and services.

	1998 (£)	2016 (£)	% change <sup>36</sup>
Hartlepool	10152	16246	60.0
Stockton-on-Tees	13132	20638	57.1
Middlesbrough	10581	18575	75.5
Redcar and Cleveland	9865	15311	55.2
Darlington	13471	24381	80.9
County Durham	9423	16295	72.9
Northumberland	10512	16140	53.5
Gateshead	12960	21690	67.3
Newcastle upon Tyne	15008	26317	75.3
North Tyneside	10931	20786	90.1
South Tyneside	7936	14236	79.3
Sunderland	11556	20632	78.5

Table 6-1: Gross value added (balanced)<sup>35</sup> per head by local authority. Source: ONS

Nevertheless, there is some, mixed evidence of small-scale agglomeration of the Tyneside economy this century, with Newcastle leading the way. North East England lost population during and after the de-industrialisation of the 1980s and bottomed out around the turn of the century before starting to gain residents again, and by 2018 the region's population was 0.8% more than in 1981. Meanwhile, the number of people in employment has been climbing in the 21<sup>st</sup> century and is up regionally by 7.6% since 2004 (Nomis, undated).

**Table 6-2** shows Newcastle with one of the largest increases in population and significantly the biggest in the number in work. The other Tyneside municipalities have also seen employment growth, but Gateshead and South Tyneside have not recovered their population losses of the 1980s and 1990s. The picture elsewhere in the region is mixed in both population and employment.

These figures overall, including unemployment and GVA, suggest that looking at wider geographies would aid economic planning; economic planners know this, and it was the ostensible point of FEAs, CAs and LEPs. NELEP's constitution commits it to working across local administrative boundaries, and this is therefore what its board members, including council leaders, are formally obliged to do (NELEP, undated-a: 1).

<sup>&</sup>lt;sup>35</sup> GVA(B) is measured at current basic prices, which include the effect of inflation.

<sup>&</sup>lt;sup>36</sup> Author's calculation

Area	Resident population	Change 1981- 2018(%)	In employment	Change 2004-2018 (%)
Newcastle	300,200	5.6	140,500	19.7
Gateshead	202,500	-5.1	97,300	12.7
North Tyneside	206,000	3.7	95,000	5.4
South Tyneside	150,300	-7.1	64,400	7.0
Sunderland	277,400	-6.9	125,600	2.6
Northumberland	320,300	7.1	140,400	-0.6
County Durham	527,000	2.9	252,100	16.9
Middlesbrough	140,500	-6.7	57,000	0.2
Stockton	197,200	13.3	86,200	2.4
Hartlepool	93,200	1.8	37,600	5.9
Darlington	106,600	8.1	49,700	6.2
Redcar &	136,700	-9.4	57,100	-7.0
Cleveland				
Tees Valley	674,300	0.8	287,600	1.1
NELEP	1,983,600	0.8	915,400	9.8
North East Region	2,657,900	0.8	1,202,900	7.6

Table 6-2: Population and employment change by local authorit	Table 6-2: Pc	pulation and	employment	t change by	local authority
---	---------------	--------------	------------	-------------	-----------------

A NELEP official told this study different geographies made sense for different purposes: the NELEP area was suitable for labour market issues like access to training, and transport to get people into Newcastle for service sector jobs and from Newcastle to Sunderland for manufacturing jobs, while the North East region was appropriate for sector development in industries like chemicals and pharmaceuticals and the wider north of England for sectoral co-operation in, for example, motor manufacturing (Interview, January 2018). A researcher engaged in raising labour market participation agreed that working on the basis of a city-region TTWA made sense, but added: 'Inevitably as in all policy making there is a whole lot of compromises that take place' (Researcher. January 2018).

While the detail of these geographies may be confusing, the overall message is clear: economies do not fit neatly into local authority boundaries. Some particularly disadvantaged municipalities do suffer widespread problems, but there are challenges elsewhere too, including in Newcastle. What all the region's local authorities have in common, with one partial exception, is that they form parts of wider economic areas, suggesting cross-boundary co-operation would help. More than a third of Gateshead residents and almost a fifth of South Tynesiders cross the Tyne to work, for example (Swinney, 2017). The partial exception is Sunderland, which forms its own NUTS3 area and TTWA; for Sunderland, trying to win as much investment for itself as possible has a rationale. But even here there are *caveats*: for higher qualified workers, Sunderland is part of a larger TTWA encompassing most of the region (ONS, 2016). Its international advanced manufacturing park (IAMP), shared with South Tyneside, sits alongside the A19 corridor, and the 5,200 jobs it is expected to provide (Sunderland Council, 2018b) will, like the adjacent Nissan car plant, be within commuting distance of much of the region's population. Overall, for all councils, competing for investment with neighbouring authorities that share a common economic territory makes no economic sense. Nor *a fortiori* does splitting the Tyneside NUTS3 area and Newcastle TTWA along the Tyne, as has happened with the establishment of NTCA.

This is well understood in the region, even by politicians committed to their local municipalities, and it is hard to find anyone who thinks the NTCA geography is ideal. A Labour MP who opposed devolution at both NECA and NTCA levels told this study: 'In structural terms, it does not make any sense at all' (Labour MP. January 2018). A devolution supporter on the other side of the political fence said dividing the north of the Tyne from the south was 'absolute nonsense. I can't believe anybody believes it makes any sense at all apart from it's the only thing they can get through' (Former NELEP board member, private sector. January 2018).

# 6.2 Diverging approaches to economic development

This section describes how their different political arrangements after rejecting and accepting devolution deals respectively led the North East and Tees Valley to adopt different approaches to economic development: technocratic in the first case and democratic in the second.

### 6.2.1 North East – politics and business: uneasy partners

Economic development under NELEP fits what Tomaney (2014a: 134), calls 'a technical exercise which is disturbed by an excess of politics' and what Clarke and Cochrane (2013) worry can be anti-political in the sense of favouring apparently effective solutions and freezing out alternative approaches (2.2.1). In the North East, the work is shared between NELEP and the CAs, but business and politicians do not always work well together even at the technical level, as is made clear in the minutes of one of the forums where they meet regularly. The minutes of NECA's economic development and regeneration advisory board (EDRAB) for March 2016 record the frustration of the private sector over the apparent disconnect with local government; it had to be explained to the private sector by the chair, a councillor, that it was not realistic to expect local authorities to relocate their economic function into one regional pot (NECA, 2016b: 3). This minute indicates not only tensions between local government and business, but that the individual councils guard their resources from each other.

NELEP 's responsibilities are carried out mainly by three boards which oversee the three SEP themes for which NELEP has lead responsibility – the Innovation Board, the Business Growth and Access to Finance Board and the Skills Board. It administers the North East's enterprise zones and Invest North East England. There is an investment panel to advise on loan funding and a technical steering group to assess grant applications for new projects and monitor performance (NELEP, undated-b). Among its activities, it is engaged with the business community through several initiatives, such as its Growth Hub, which helps micro, small and medium-sized businesses find support and finance. It links schools with businesses to help prepare young people for work. NELEP receives updates from the CAs on the SEP themes for which the latter take lead responsibility: transport and digital connectivity; employability and inclusion; and economic development and regeneration. When appropriate, NELEP contributes to debates about economic policy, as when it responded with NECA to the government's 2017 green paper *Building our Industrial Strategy*, and co-ordinated the regional response to Brexit.

This is technical work. The political disturbance is provided by the council leaders, both in their relations with each other and collectively with NELEP. This is not new. In the early days of NELEP, according to one council leader, there were a lot of concerns about the accountability of the LEP, conflicts of interest and a lack of awareness by some business representatives of the standards of behaviour expected (Council leader, NECA. March 2018). Another shared this view:

One of the frustrations from the new partners who were appointed in 2010-2011 was 'why does it take so bloody long in the public sector; we want to appoint staff, appoint them tomorrow' and this type of thing, and we were having to say 'look you can't do that with public money; there has to be transparency, there has to be openness, there has to be adverts, there has to be proper interview techniques and so forth, because this is public money', and some of them that were on that board initially had a real genuine difficulty understanding how the public sector worked (Council leader, NECA. March 2018).

A former NELEP board member confirmed that NELEP and its council leaders got off to a bad start:

The local authorities around here first of all didn't want the LEP. They applied for it; they were forced to apply for it. It was not what they wanted. They consistently refused to provide anything more than the minimum funding...so we had no basic resources to do what was required. Over time I think the local authorities warmed to it more, but they found it very frustrating in the early years when business representatives would consistently try and do things that the local authorities didn't really like... The chairman would take soundings from everybody and then say, 'Well on the basis that there is a majority here we will do this'. And that was consistently unpopular with the local authorities (Former NELEP board member, private sector. January 2018).

The low opinion in which the council leaders were held by the NELEP business representatives is made crystal clear by a story told by one of the latter concerning what happened when a council leader was given a copy of the SEP. The contempt is unmistakeable:

He said he had an objection. What was that? He had been through and counted up the number of times [his council] was referred to and it was less than the number of times that the other local authorities were referred to. So we thought 'Great, OK, we'll change that. That's fine. Anything else? Strategic commitments to add? No?' (Former NELEP board member, private sector. January 2018). NELEP's board papers are available online, but NELEP's meetings are not open to the public, and this can lead to issues being kept from the public which, in the view of at least one former board member (the same one as quoted immediately above), ought not to be. On one occasion, he said, NELEP was asked by the government to redraw the boundaries of areas eligible for certain economic assistance, including – to save money - one option that would cover only 80% of existing areas, eliminating, perhaps, areas never likely to be used but including others that offered more potential. Council leaders, he said, refused, either because they were unwilling to implement Tory cuts or they feared their own area might lose out. When NELEP decided to go ahead anyway, believing it made sense otherwise the government would perform the exercise itself, the council leaders would not provide the staff, who were on their payroll. So NELEP hired consultants:

There's an example which would not have been in the public domain, which it should have been, where business people showed leadership. It was pretty aggressive stuff and never mind if it was a waste of public money (Former NELEP board member, private sector. January 2018).

That story did not come from a neutral observer but an insider who arguably had an interest in embarrassing Labour councillors. Nevertheless, it paints a picture of poor relations between the political and business members of NELEP, certainly in its early days, with accountability over public money one of the central issues.

The formation of NTCA quickly engendered the prospect of new intra-regional conflicts to come. Rump NECA now has its own representative on Transport for the North (TfN), the statutory body established in 2018 with responsibilities across the north of England, whereas the old NECA had been represented by the Newcastle leader, who now sits on the TfN board representing NTCA (TfN, undated). A County Durham official highlighted the local rivalries of North East politics when he said of the Newcastle leader: 'I know he is a County Durham boy, but is he really interested in rail stops in Chester-le-Street? Hopefully one of the benefits [of the rump NECA] is we have a new representative, so hopefully it won't be all about Newcastle'. A new representative, however, he acknowledged, would not eliminate competition for resources, for NTCA, which has a devolution grant, and NECA, which does not, sit alongside each other on a

joint transport committee: 'Whether we can agree on priorities when bidding for that pot will be interesting' (Harrison, 2018).

The prospect of increasing divergence between NTCA and NECA emerged when Heseltine published a report in 2019 calling for more powers for metro mayors. The North of Tyne mayor, like his Tees Valley comrades before him (5.3), became another North East Labour politician to unexpectedly find himself in agreement with Heseltine: 'Life takes you on some interesting journeys' the Momentum supporter told a local paper. But, as the same paper remarked, Heseltine's report may not be welcomed by rump NECA 'as the clear implication is that powers and funding should go to areas with mayors' (Walker, 2019g).

One result of NECA's collectivist yet fractious approach to development is that it lacks an identifiable figurehead either to be accountable to local people or as a point of contact for outsiders, whether ministers consulting on policy or overseas investors wanting a contact person. NECA has a chair who has normally served for two years and whose main job is elsewhere, leading his/her local council. NELEP's chair, meanwhile, is an unelected business leader. It is a situation that led an economic planner to adapt the question asked by former US Secretary of State Henry Kissinger about Europe: 'Who do I ring if I want to ring the North East?' It was important to be able to answer that question, he said, because if the nation's biggest conurbations ended up with mayors as their public face, as he thought likely, the North East could not afford to miss out (North East economist. January 2018)

### 6.2.2 Tees Valley – size isn't everything

Tees Valley has an answer to the Kissinger question, and the result has been a dramatic change in its approach to development since its mayoral election in 2017. Its SEPs, like the North East's, started as technical exercises, but the election of an assertive mayor in 2017 caused political disturbance of a different type to that experienced by its northern neighbour as the mayor imposed his highly political presence on the CA and the area.

There were early difficulties, before the mayoral election, when TVCA and TVU integrated, and they bore some similarities to those in the North East as public and

private sector cultures clashed. They were not as intense, but some staff who transferred from TVU to the CA had to adjust to moving from the private sector into local government (Council leader, TVCA. April 2018).

Such problems aside, the trust and collaboration built up over 20 years succeeded, at least to start with, in embracing the mayor, despite the shock of a Tory victory. A council leader said: 'Politics aside, I get on really well with Ben' (Labour council leader. Tees Valley. April 2018). Tees Valley's internal collaboration and new independence were validated from outside by a former NELEP board member who said they seemed to have got their act together after previously appearing to come second to the north of the region (Former NELEP board member, private sector. January 2018).

With devolution, a wider range of responsibilities, from transport to culture, combined with multi-year funding, has made it easier to take a strategic rather than municipal view while still keeping everyone happy they are getting their share. An official explained that individual authorities knew that while they might not get something they could call their fair share out of each area of activity and not necessarily win every year, they could have confidence that their priorities were being supported over a long period (Public official, Tees Valley. January 2018).

Overcoming potential local rivalries has been assisted by the compact geography of the area following its separation from the North East. A disadvantage for some because it means less status and resources than competing areas, its small size has the advantage that economic opportunities in any one place are potentially accessible to residents everywhere – though commuter transport needs improvements - helping to overcome the municipal outlook endemic in the North East. TVCA leaders, said one, did not think about five individual local authorities but about delivering the SEP with the transport infrastructure to make sure people could access the jobs wherever they were in Tees Valley (Council leader, TVCA. April 2018). An official agreed that Tees Valley was seen as a cohesive unit with a common economic geography (Public official, Tees Valley. January 2018).

Some have reservations. According to one business leader, Tees Valley has not always been immune to the North East temptation to put fair shares above strategic vision. He thought trust was superficial, motivated by fear of the consequences of disunity

(Regional business leader. August 2018). An official thought the collaboration could, on occasions in the past, have resulted in difficult decisions being ducked, while the partnership was about securing government resources (Tees Valley official. January 2018). The majority view in Tees Valley, however, is that collaboration is genuine.

Contacts with the Prime Minister, exemplified by a visit to launch the South Tees Development Corporation (5.6), and other ministers are more than flattering. They are an essential part of Tees Valley's and the mayor's high-profile approach to development. An executive with a background in industry and a role in Tees Valley governance said the external profile of the region has changed and access to government, including secretaries of state, had improved: 'You should never ever underestimate the power of being able to pick up the 'phone and do that' (Tees Valley LEP board member, private sector. March 2018). An ally of the mayor confirmed this picture: 'It's the ability to be taken seriously. Ministers realise that these individuals have big electorates. They do take the calls' (Conservative business leader, Tees Valley. June 2018).

The mayor's high-profile purchase of Teesside Airport (5.5), driven by him personally on the basis of his election manifesto, was just the most prominent example of the way economic development after devolution has become a political project, as opposed to the technocracy of the North East. From large-scale projects like the development corporation to routine visits to small companies, tweeted frequently by the mayor, Tees Valley's economic development has become a personal and politicised matter.

The mayor raised Tees Valley's profile another notch, to the international level, when he led a trade mission to the Far East to meet potential investors in the SSI site (McNeal, 2018). The high profile continued at COP24, the world climate talks in Poland, when the government announced £170m for the world's first carbon 'net zero' cluster for clean energy, with Tees Valley named as an area that could benefit (TVCA, 2018f). The plans to adapt the traditional industries of the past for the clean technologies of the future – a route out of path dependence – brought Tees Valley and its mayor favourable headlines, and not just locally (Tighe and Sheppard, 2018).

In a striking and characteristic initiative, the mayor invited President Trump to Tees Valley during his UK state visit of June 2019, speaking at the same time of attracting an unnamed major US investor to the area: 'This is not about Trump or his policies', he said. 'This is about Teesside jobs. It's a long shot, but if you don't ask you don't get'. The invitation was declined, but within weeks the mayor was welcoming the US ambassador (Robson, 2019; Brown, 2019b; Metcalfe, 2019j). Some are enthusiastic about the new approach:

Actually I think they are doing an incredible job. In Tees Valley we seem to be in the middle of a sort of halcyon period in terms of our profile being raised, and that's only down to I think what they have done at TVCA, but also having an elected mayor as well; it's made a hell of a difference (Business support executive, Tees Valley. October 2018).

The mayor's response to a challenge was tested with the threatened closure of two small steel plants in the area – though still employing 700 – when British Steel entered administration in May 2019. He reacted with characteristic vigour and outspokenness, chairing a cross-party emergency meeting including MPs and attending a public meeting at a local working men's club (Huntley, 2019; Manning, 2019; Houchen, 2019d). He also attacked British Steel's private equity owners with a detailed critique of their record on social media:

It's time local people know about the sharks behind British Steel's collapse. Greybull Capital is a private equity fund based in Knightsbridge. They are notorious for buying up firms, extracting money and then presiding over company failures. This is predatory capitalism at its absolute worst (Houchen, 2019c).

The mayor's outspoken attack on the British Steel 'sharks' displays another aspect of his political style. He is combative when challenged, and presents his pugnacity as championing the people of Tees Valley. His opponents in commercial dealings on behalf of his office were more than once accused to holding him or the people to ransom, and one named individual was called a profiteering opportunist. He did not shrink from branding the Chief Constable of Cleveland a liar or from accusing the

private owners of Durham Tees Valley Airport of attempted bullying, to which - of course - he would not yield (Metcalfe, 2018a; Cain, 2018b; Price, 2019b; Metcalfe, 2019h).

## 6.3 Economic strategy: the background

The SEPs published by the North East and Tees Valley LEPs in 2014 did not emerge from a vacuum. Regional economic planning, as Tomaney (2006) has recounted, predates the Second World War and has been continuous, though not undisrupted, since the advent of RDAs in 1998. Some issues raised in earlier reports remain relevant, and 2006 is an appropriate starting date, as it marks both the final regional economic strategy (RES) of One North East and an international perspective on the regional economy by the OECD. While advocating support for investment in the staples of economic development, such as innovation, business support, education and skills, inclusion and infrastructure, these reports also raise less straightforward questions. Issues include the realism of expecting the region to catch up with the rest of the UK; the extent to which it should aim to make a radical break with its industrial heritage and embark on a new path to a knowledge-based economy, the problem of balancing the supply of and demand for skills, the wisdom of selecting identified sectors for priority support and the focus on urban cores as drivers of growth.

The RDA in 2006 had a delicate line to tread between realism in acknowledging the region's poor record and ambition in promoting a new pathway to competitiveness and prosperity. Existing forecasts for growth were negative compared with other UK regions, it conceded, but this was because they were based on historic trends; the aim of the RES was to capture a sense of future ambition and buck the trend through a combination of raising productivity and the creation of 61,000 to 73,000 new jobs and 18,500-22,000 new businesses (ONE, 2006: 22-30).

The RDA aimed to achieve a restructuring of North East industry and its movement up the value chain by creating world class knowledge sectors where the region was already globally strong – chemicals and pharmaceuticals, health and social care, and energy. Other vital sectors were identified as automotive, defence and marine, food and drink, knowledge-intensive business services, tourism and hospitality, and commercial creative. Coupled with this focus on existing strengths and manufacturing,

the RDA also sought to make the region part of the knowledge-based economic world of the future. It was seeking, in short, to continue along some existing paths while simultaneously creating a new one (2006: 19-20).

In the OECD's view, this positioning of the region within the knowledge economy was suggestive of a radical rather than incremental path, necessary to raise GVA and reduce regional disparities but presenting a high risk of failure. The OECD was worried the region was trying to move too fast from an FDI strategy towards a knowledge economy. While an aspirational tone was to be applauded, it said, if it was not grounded in reality the likelihood of disillusionment was high and the region could lurch from strategy to strategy without advancing (OECD, 2006: 124).

In the OECD's alternative strategy, a mixed economy incorporating both a low-wage low-skill economy and a high-wage high-skill economy was considered achievable through the protection of the existing manufacturing base combined with growth in a range of sectors and the fostering of innovation and SME entrepreneurship over the entire economy. The OECD thus displayed a cautious approach towards the questions of path dependence and creation, and reluctance to attempt a radical change of direction (2006: 126).

When the *Adonis Report* (North East Independent Economic Review) appeared in 2013 it adopted the objective of 'more and better jobs', aiming at an extra 60,000, all in the private sector. It sought a balance between protecting existing manufacturing jobs and increasing the number of higher value service sector jobs linked to business services and the new economy. While building on the region's industrial strengths, it advocated moving away from sector-based initiatives to support that took account of new developments in the economy. At the same time, growth in lower skilled and lower paid jobs was said to be inevitable. The report's recommendations were based around the key elements of exporting, innovation, FDI, investment finance and investment in young people, as well as improvements in transport and the living environment (2013: 13). There were specific mentions for digital, media, telecoms and software; for tourism, heritage and culture; and for biomedicals, sub-sea and offshore engineering, automotive and science - a mix of continuing along existing paths and creating new ones (Adonis, 2013: 6, 13, 16, 21, 23).

The Adonis Report, which was commissioned by NELEP, did not, unlike the OECD, stress the significance of Newcastle as a growth hub, though it did identify it as an employment centre for corporate, financial and professional services and, with Gateshead, as an accelerated development zone for high value-added services (2013: 9, 29).

There was thus general agreement about the main priorities. The OECD, *Adonis* and One North East, as well as the later *Heseltine Report* on Tees Valley (6.4.2), were united in focusing on the need to build the region's business base, concentrating on the private sector, including FDI and exporting; on innovation; on the importance of education and skills and bringing the economically inactive into the labour market; and on infrastructure improvements. They highlighted, with varying degrees of emphasis, quality of life issues including housing, the natural environment, cultural and heritage assets. The later reports added digital connectivity to the list (OECD, 2006: 16-20, 175; ONE, 2006: 100-113, 126; Adonis, 2013: 6; Heseltine, 2016: 5-8).

The involvement of Adonis and Heseltine added prestige and authority to the North East and Tees Valley SEPs respectively. The North East SEP made the point explicitly that it had strong business and political endorsement and built on the workshop programme undertaken by the Adonis team, which had been supported by international experts bringing significant expertise and knowledge and a wider perspective in their role as critical friend (NELEP, 2014a: 2, 5). Heseltine launched Tees Valley's refreshed SEP in 2016 at Darlington (Walker, 2016).

When the LEPs produced their SEPs their choice of target measures was commensurate with the advice received – GVA, private sector enterprises and jobs, the employment rate, economic activity, and education and skills. Some interim and subsidiary targets, such as increased business density, innovation, and improvements in education, skills and employability, offered the advantage of increasing the areas' adaptability and resilience.

One notable outlier in the advice was the OECD's emphasis on Newcastle as the regional growth hub (OECD, 2006: 46, 172), which was not reflected elsewhere. The 'city as growth hub' debate (2.1.1) remains a live and divisive one in North East England, as we have seen in the case studies. It is a reflection at a smaller geographical

scale of the global and UK national debates about the benefits of agglomeration in relatively few places, expecting the benefits to trickle down. The OECD's lone position on Newcastle's central role may be explained by the fact that it was under no pressure to be tactful towards the region's rival interest groups.

A regional business leader said the rational approach was to do whatever was for the greater good, but that North East England needed help in some specific geographic areas – another regional example of the policy debates seen in the literature, this time over the controversy between place-based and spatially blind development (2.1.1). There was a similar debate to be had over scale-up and start-up, said this business leader – investing to help existing business to grow or helping to start new activity in areas of desperate need:

It's a tough call...In the North East there are areas of utter deprivation which will remain areas of utter deprivation to the detriment of the North East community, business and social, if they don't get some sort of help (North East business leader. September 2018).

OECD fears that the region was being too radical in switching from manufacturing, including FDI, to a knowledge-based economy, have turned out unfounded. NELEP and Tees Valley both see manufacturing and FDI as remaining important. NELEP plans 30% of new jobs to be outside the higher skilled, higher paid category and Tees Valley, similarly, sees only 49% of new jobs being at the higher level, with 17.9% at intermediate and 33.2% at lower levels. The region continues to pursue FDI, though the success rate is erratic (DIT, 2016; TVCA, 2016a: 25; DIT, 2017; DIT, 2018; DIT, 2019; NELEP, 2019a: 3; TVCA, undated-e; TVCA, undated-f).

In view of the weight of the accumulated inherited advice, it is not surprising that both the North East and Tees Valley LEPs went along with it in identifying the priority themes, key sectors and strategic objectives for their SEPs (NELEP, 2014a: 7; TVU, 2014g: 3, 12). Both the North East and Tees Valley display path dependence to the extent that they seek to maintain and adapt traditional industries like chemicals, marine and manufacturing, but also a willingness to create new paths into sectors such as the digital, the creative and professional services. The one piece of advice that is too

politically sensitive to be openly adopted anywhere in the region is to treat Newcastle, however defined, as a growth hub.

# 6.4 The Strategic Economic Plans

Having surveyed the economic plans providing the background to the SEPs produced by the North East and Tees Valley LEPs in 2014 and their subsequent revisions, this section turns to the SEPs themselves. It reveals a combination of approaches involving both the adaptation of existing paths, such as through advanced manufacturing and the development of low-carbon energy, and the creation of new paths in sectors such as digital, culture and business services.

# 6.4.1 North East – closing the gap

NELEP's SEP, *More and Better Jobs*, first published in 2014, built on the *Adonis Report*. Within a ten-year timescale, it aimed to add 60,000 private sector jobs to the 40,000 expected to be created in any case, with 60% being 'better', defined as high skilled and higher paid in the private sector (NELEP, 2014a: 6). The major growth sectors would be business and professional services, new economy (cultural, creative, technology, media and telecoms), low carbon (renewable technologies), tourism, and logistics (building on internet shopping and international trade) (2014a: 6-7, 29). Low carbon technology and sustainability were a new feature in the plan. Manufacturing remained important, including in export markets, and was said to be enjoying a renaissance. Growth was to be achieved by pursuing the six strategic themes of innovation; business support and access to finance; skills; employability and inclusion; economic assets and infrastructure; and transport and digital connectivity. Maintaining good levels of FDI was seen as key to future growth (NELEP, 2014a: 5-7, 28, 29, 43).

Skills was seen as essentially a supply-side problem, with an increase of 120,000 by 2020 in the number of jobs requiring educational level 4<sup>37</sup> and above, though on the demand side the plan did refer to creating a landscape where companies realised the benefits of training and saw returns on their investment; and it recognised low levels of graduate utilisation. The social inclusion agenda was aimed at removing or overcoming practical barriers to work such as skills, deprivation, ill health and poverty.

<sup>&</sup>lt;sup>37</sup> What qualification levels mean. See: <u>https://www.gov.uk/what-different-qualification-levels-mean/list-of-qualification-levels</u>

There was no special place for the Newcastle urban core as a growth centre (NELEP, 2014a: 9, 11, 24, 43, 51,52, 58, 60, 90).

NELEP set itself four targets for 2024 in addition to 100,000 new jobs. It would, it pledged, halve the gap between the North East and the national average (excluding London) on three measures: productivity (measured by GVA), private sector employment density and economic activity rate; and fully close the gap on employment rate. So encouraged was NELEP by progress on the jobs measures in 2017, with already 53,000 new jobs of which 63% were 'better', that it committed itself to a new target of 70% in this category (NELEP, 2014a: 5; NELEP, 2017h).

However, in the 2016 version of its plan, NELEP did acknowledge that closing the gap by 2024 between the North East's employment rate and the national rate as it had stood in 2014 was not a particularly valuable measure of performance. It changed the target to one of closing the gap with the rolling national figure, achieving which it said would be significantly more stretching (NELEP, 2017c: 6). This is an implicit acknowledgement that progress on jobs is to some extent a reflection of the national trend and not of NELEP's own activities. A North East England business representative said (at a time unemployment was falling):

The North East [region] has reduced unemployment significantly year on year on year. Now is that a direct result primarily of what the LEP has done? To some extent it will be; there will be programmes in there that have helped, but is there anything that you can tangibly directly point to? I don't think there is... How much is the business community just getting on and taking on more staff. Now you could arguably say that the LEPs and the combined authorities create those conditions for business to be able to flourish and to be able to expand, but again there's a shift in the labour market as well as that (Business leader. August 2018).

Much had changed politically when NELEP published a new SEP in 2019. Britain had voted to leave the EU and appeared on the verge of doing so; NECA had voted to reject a devolution deal and split; the new North of Tyne CA had been established, signed a devolution deal and was about to elect its first mayor; and the *Industrial Strategy* Green and White Papers had been published. The new SEP committed NELEP to

working with the government to develop a *Local Industrial Strategy* and noted that a number of sector deals under the *Strategy* were focused on key North East sectors: automotive, life sciences and offshore renewables. Global grand challenges listed in the national *Strategy* and offering business opportunities also included some sectors where the North East was active, including clean growth and the ageing society (NELEP, 2019a: 6, 17).

The 2019 plan identified four areas with strong opportunities for growth: digital, advanced manufacturing, health and life sciences, and energy. NELEP would work with four service sectors that supported the wider economy and offered significant opportunities for more and better jobs: education; financial, professional and business services; transport and logistics; and construction. There would be five programmes of delivery (equivalent to the six strategic themes of the 2014 plan): business growth; innovation; skills, employment, inclusion and progression; transport connectivity; and investment and infrastructure (2019a: 2). Priorities are sometimes expressed in varying, though not necessarily inconsistent, terms: NELEP's assurance framework published in December 2018 listed 'key priority areas' as innovation and business support; working with schools, colleges and universities to improve education outcomes; skills and economic inclusion; economic assets and infrastructure; and transport and digital connectivity (NELEP, 2018f: 8).

The extent to which the North East economy has been successful in path creation, not just under the SEP but in is recent decades, is made clear in NELEP's 2019 annual economic report. Manufacturing, while remaining important, is increasingly in highvalue, advanced sectors, and the regional economy has diversified into areas including energy, health and life sciences, digital, finance and professional services, and culture (NELEP, 2019e: 2).

This is encouraging in itself, but not without its problems. As the North East economy moves up the value chain the need for more highly qualified workers increases. In a region where we have seen (3.1.2) that educational attainment in secondary schools is comparatively low, this creates the danger that some young people will lack the qualifications needed for the jobs of the future, perpetuating perhaps the problems of the region's deprived neighbourhoods. This was one reason the OECD advocated a

mixed economy of both high and low skills (OECD, 2006: 126, 158). The skills equilibrium is a long-term issue that worried the RDA throughout its existence and featured in its regional economic strategies. 'Some believe that the region's low aspirations are rooted in the culture', it said in 1999 (ONE, 1999: 48; ONE, 2006).

#### 6.4.2 Tees Valley – the UK's 'greatest single development opportunity'

The Tees Valley SEP established six priorities when published in 2014 by Tees Valley Unlimited (TVU) – innovation and business growth; low carbon, high value; skills; increased capacity on the East Coast Main Line (rail); infrastructure; and making Tees Valley a preferred location to live, work and visit. It recognised that the area was overreliant on public sector jobs (TVU, 2014g: 3-4).

Like NELEP, TVU saw skills shortages as mainly a supply-side problem, with much of the predicted jobs growth being in higher skilled occupations. It identified specific local problems as including market failure in business accommodation, with achievable rents unable to cover development costs. Some key housing development was also unviable without public sector intervention due to infrastructure costs. Because of the nature of Tees Valley's heavy industry, power and water supplies and waste disposal infrastructure were issues, as was flood defence because of the natural geography. There was a suggestion of a cultural problem in the form of a lack of contact between schools and business. TVU reported inconsistency in careers information, advice and guidance for young people; schools did not have co-ordinated access to businesses (TVU, 2014g: 8, 41-43, 55-58, 66, 109).

The Tees Valley SEP 2014 set itself four measurable targets: £1bn extra GVA (10% increase); 25,000 new jobs, to match the national employment rate; 3,200 additional enterprises (25% increase); and 4,400 more residents achieving education levels NVQ3 and NVQ4 (TVU, 2014g: 6).

Just as NELEP had received the advice of an independent economic plan in the shape of the *Adonis Report*, so Tees Valley was given outside advice, in its case by Heseltine following his appointment to report on the SSI steel site closure (Redcar & Cleveland Council, 2017b: 4) (5.2). Heseltine's report, published in June 2016, came two years after the area's first SEP and was to lead to a revised SEP six months later.

When the second SEP appeared following the *Heseltine Report*, the familiar task of reducing the jobs deficit, with only 292,500 jobs in an area with a working age population of 417,000, was at the top of TVCA's new agenda (2016: 12), but Heseltine also highlighted education and skills issues and made a series of detailed recommendations in the fields of industrial and urban regeneration, growth opportunities, energy, housing, transport, leisure, environment and tourism. He identified key problems as low productivity linked to low rates of enterprise, high public sector employment, skills deficits and low levels of commercialisation. Eighty per cent of employers in Tees Valley had a skills shortage or struggled to recruit in the ICT sector. Heseltine was especially outspoken in his criticism of the performance of secondary education, with 25% of schools requiring improvement. Heseltine saw the formation of TVCA and its devolution deal as a significant opportunity to refocus the local economy. He thought TVCA would accelerate the growth in jobs and the election of a mayor would enhance this trend (Heseltine, 2016: 4-9, 19-20, 22, 42, 44-9).

The new SEP, which Heseltine launched in December 2016, combined ambition and realism. It reduced the number of targets to two. Its 25,000 jobs target was realistically in line with recent employment growth of over 2,500 a year, but if the area were to match the anticipated national rate over the next ten years the extra jobs would total only 11,000; so it needed to out-perform, which was ambitious; this challenge was noticed by TVCA's overview and scrutiny committee (TVCA, 2016e: item 7, p. 3). The other target, GVA, was increased from £1bn extra in 2014 to £2.8bn. extra, a seemingly very ambitious rise, though the figure was updated to 2016 prices and the timescale extended to 2026 (TVCA, 2016a: 4, 18).

In addition to these two targets, the new plan listed six measurable 'impacts' for 2026, including 2,000 new enterprises (reduced from the 2014 target of 3,200), jobs density growth (+55%), GVA per hour worked (+30%), population (+6%) and residents with NVQ4 (+20%) (2016a: 39).

TVCA based its ambitions on 'growth enablers' resulting from devolution, the *Heseltine Report* and the *Northern Powerhouse*. These enablers were itemised as new powers over adult education, housing, regeneration and transport, a proposed mayoral development corporation (MDC), increased influence over national policy, the Single

Funding Pot, integrated business advice and support, increased sectoral and supply chain linkages and a skilled workforce (2016a: 6-8).

Tees Valley also updated its key themes in the light of the SSI closure, receipt of the *Heseltine Report* and signing of its devolution deal, though in substance they were little changed. They were now business growth; research, development, innovation and energy; education, employment and skills; place (to attract businesses and people); culture; and transport and infrastructure. Key sectors were identified as chemicals, health innovation, energy, advanced manufacturing, logistics, digital and creative, culture and leisure, and business and professional services. Logistics was a new key sector in both North East and Tees Valley plans (TVCA, 2016a: 4-9).

The MDC created under the devolution deal and the first outside London, enabled Tees Valley to begin the regeneration of 4,300 acres of industrial land including the former SSI steel site (TVCA, 2017a: item 7). Its formal launch in summer 2017 was marked by a visit by the Prime Minister (Duncan, 2017), followed in October 2017 by the launch of a masterplan for the site, with plans to create 20,000 jobs over 25 years in sectors such as steelmaking, metals recycling, bulk materials processing/manufacture, offshore energy manufacturing, energy storage, major power generation, submarine cable manufacture, rail related industries and waste management (McNeal, 2017a). The plan was described by the *Northern Powerhouse* Minister as 'a crucial step to stimulate economic growth and regeneration across Tees Valley' and by the mayor as 'the single greatest development opportunity in the UK right now' (Hill, 2017).

## 6.5 The economic outcomes – still struggling, still lagging

The quarterly labour market statistics for February-April 2019, as the study period for this thesis was ending, were disappointing. Though not broken down between the North East and Tees Valley, they cannot have promoted optimism in either. Regional unemployment was up 0.6% on the previous quarter and – a more reliable indicator – by 1.1% on the year at a time when the UK figures were down (Nomis, undated). It left the region in its accustomed position with the UK's highest unemployment rate, and going in the wrong direction.

#### 6.5.1 North East

At the half-way point in NELEP's ten-year SEP progress was mixed and the outlook remained challenging. In December 2018 progress towards the headline target of 'more and better jobs' was on track, with 71,600 new jobs towards the 100,000 target and 49,800 of these (70%) in the 'better' jobs category. But employment had been rising nationally too, so the gap was only 34% of the way towards the target of being totally closed. The economic activity rate showed disappointing progress of only 15% towards halving the gap (NELEP, 2019e: 6-7).

Progress towards halving the gap in private sector employment density was particularly disappointing, though somewhat uncertain because of a dataset change, and the most recent statistics were for 2017. The 2014 SEP drew attention to the area's shortage of private sector jobs needed for a balanced and sustainable economy (NELEP, 2014a) and by 2015 the gap had closed by 16%. Then the dataset changed, broadening the range of businesses included, and by 2017 the gap had actually widened by 5%, reflecting a 1.5% increase in the North East compared with a 3.2% increase in England excluding London. The productivity measure was encouraging however, though it too had undergone a change in the dataset and was now measured in terms of GVA per hour rather than per full-time equivalent worker. By 2017 the gap was down by 29% towards its 50% target (NELEP, 2019a: 59; NELEP, 2019e: 7; NELEP, undated-e).

For North East performance to match England excluding London it still, in 2019, according to NELEP, needed to add 93,000 jobs, 25,500 businesses and 40,000 individuals qualified to degree level or above. NELEP's brief overall conclusion on its performance so far was that there was more to do to close the gaps: 'We have made progress in each of these areas but the rate of improvement will need to increase if we are to achieve our 2024 targets' (NELEP, 2019a: 3, 15, 59).

Overall, across a wide range of indicators, almost two-thirds are showing improvements since 2014 and roughly one-third have seen the gap closing with England excluding London. Nevertheless, NELEP reported to NECA's overview and scrutiny committee in March 2019 that the North East continued to underperform on many indicators that still required action, including growing the jobs base and

increasing the quality of available jobs; ensuring individuals have the skills to take up available jobs; continuing to grow the business base and ensure the right support is available for businesses to grow; and investing and innovating to drive improvements (NECA, 2019d: 44-45; NELEP, 2019e: 1).

#### 6.5.2 Tees Valley

When TVCA reported on its own progress in 2019 its assessment was generally downbeat, though mitigated by recalling the damage done by the SSI closure of 2015. GVA per head increased from 2014 to 2015, fell back in 2016 and increased again in 2017, reflecting, said TVCA, a significant loss of high paid jobs from the SSI closure and subsequent supply chain effects. GVA per head fell from 79.9% of UK average in 2009 to 71.5% in 2017 (TVCA, 2019a: 3).

Towards its target of 25,000 new jobs, Tees Valley was broadly in the same place after five years of the SEP as it had been before the Plan started, though with some variations on the way. In the year before the SEP was introduced (April 2013-March 2014) the number of people in employment was 287,600. Figures thereafter fluctuated between a low of 284,500 in January-December 2014 and a high of 293,500 in January-December 2016. In the 12 months to March 2019, the half-way point of the SEP and the conclusion of the study period for this thesis, the number in employment was 287,300 – down by 300 on five years previously (Nomis, undated).

It is notable that employment was at its highest in January-December 2016, the year immediately following the SSI closure, and this was in fact the employment high point for Tees Valley since 2004, the period for which statistics are available. National employment was on a rising trend at the same time, indicating that the performance of the national economy may have had a greater effect on local job numbers, though not necessarily quality, than even a significant economic shock at local level (Nomis, undated). The figures, though, say nothing about what the Tee Valley employment number would have been had the steelworks not closed.

Towards its targets of 3,200 new businesses set in 2014 (or 2,000 set in 2016), Tees Valley gained enterprises annually from 2014 to 2017, up from a total of 14,580 to 17,500 in 2017. But it then suffered a loss to 17,230 before recovering to 17,765 in 2019. Overall, therefore, there were 3,185 more businesses in 2019 than five years earlier, already ahead of even the more ambitious, earlier target. But the bulk of the increase had come in the early years, with only 665 more enterprises since 2016. The CA was told in early January 2019 that Tees Valley's business density was the lowest of all LEP areas (TVCA, 2019a: 4; Nomis).

Population rose 1.1% from 667,000 to 674,000 between 2014 and 2018 towards a hoped-for impact of 6%. Against the new target for educational attainment set in 2016 of a 20% increase in residents with NVQ4, the number actually fell from 126,600 in 2016 to 125,700 in 2018, though it had risen earlier from 116,600 in 2014 (Nomis).

As Tees Valley's mayor focused public attention on his successful airport venture, the SEP was thus struggling. Nevertheless, in spite of the generally disappointing statistics, Tees Valley's leaders remained optimistic about devolution and the election of a mayor. A regional business leader said there was more purpose and direction that previously (Business leader, August 2018). An official, asked if having a mayor would make a difference to economic outturns, replied:

Absolutely, and I think we can point to that now already, bringing more resources in, getting bigger scale projects, more interest from international investors, a more strategic approach to our infrastructure requirements, more ready access to government money, a greater reputation with the business community, indigenous and potential investors – all of these things are very clear benefits already from having gone down the road we have gone down. Is it enough? No probably not; absolutely not. It's a bit of a journey, but we are more likely to succeed in the long term if we head off in that direction, I think, than resisting on the basis that it's not adequate at the moment (Public official, Tees Valley. January 2018).

### 6.6 Conclusion

This chapter has provided a clear answer to research question 2, asking what effect the different institutional arrangements in the North East and Tees Valley have made to their approaches to development. It shows that the two areas' economic problems and plans are quite similar and do not account for the technocratic approach in the former

and the high-profile, political style in the latter. It is the political and institutional situation, coupled with the highly proactive personality and style of the Tees Valley mayor, which make the difference. While the everyday duties of development are performed by professional officials in both areas, in Tees Valley the mayoral combined authority as a formal institution has placed control firmly in the hands of elected politicians, while the mayor has imposed his personality on Tees Valley in a high-profile manner which is unknown in the North East.

This chapter and the two case studies preceding it (Chapter 4 and Chapter 5) therefore together provide an additional case study of the key concept of leadership introduced in section 1.3.3, contrasting the collective and individual approaches discussed in the literature (2.2.4).

In relation to other questions of special interest to this study, this chapter reveals that path dependence remains evident but its grip is weakening as the capital and human assets of past industries are either adapted to new related industries like energy and offshore or replaced by new industries, technologies and skills in sectors like digital and creative which the CAs and LEPs are keen to support in an attempt to create new paths. The developments in energy and offshore industries are examples of the incremental adaptations within a path theorised by Sunley *et al.* (2017) (2.1.3).

The chapter has also discussed the key concept of agglomeration (1.3.2) and the linked controversy over place-based *versus* spatially blind development raised in the literature in connection with new economic geography (2.1.1) and has found some evidence of limited agglomeration around Newcastle, though the city still shares the unemployment of other parts of the region (6.1).

Whether the different approaches will result in different economic outcomes is another matter. After five years implementing their ten-year SEPs, two-and-a-half years after their devolution decisions and, in Tees Valley, two years after the election of a mayor, the economies of the two LEP areas continued to struggle. That in itself is not a sign of failure. It takes a long time to turn round an economy. North East England was lagging in 2010, after 12 years of the RDA. It had lagged for decades before that, in spite of a succession of plans (Board of Trade, 1963; NEPC, 1966; NRST, 1977). Tees Valley's major projects in particular are for the long term: the South Tees Regeneration

Master Plan is a 25-year vision (STDC, 2019). Nevertheless, both LEPs set themselves aggregate targets for their ten year plans and can expect questions to be asked about progress at the half-way stage.

The LEPs and their SEPs have other significance for the local economies too, by way of steering the strategy for their areas and supporting the ways in which they are adapting old paths and creating new ones. While private sector businesses may be the main drivers of this evolution, the SEP plays an important role by identifying sectors for support and promoting big projects like the International Advanced Manufacturing Park (IAMP) and the South Tees Master Plan (STDC, 2019).

Disappointing short-term outcomes do not, therefore, invalidate the devolution decisions that were taken in either area. Tees Valley's decision to accept devolution was pragmatic, and some philosophical pragmatists see the efficacy of ideas consisting in their mere adoption by the community rather than any success the community may or may not gain from them (Rescher, 2005). Either accepting or rejecting devolution could therefore be the right thing to do, depending on the circumstances, irrespective of the economic outcomes. Tees Valley has made a choice that has given it a raised profile, a feeling of liberation from its northern neighbour and an enhanced sense of identity. Whether North of Tyne and the rump NECA, if their economies continue to lag, will find similar pragmatic consolation in their respective decisions seems doubtful. Neither of them is where it is by choice, but to make the best it can of an unsatisfactory situation.

The next chapter discusses accountability, public consultation and inclusion and addresses research question 3. It supplements the study of the Tees Valley mayor's leadership already conducted by considering it in the context of Hambleton's distinction between facilitative leadership and the 'city boss' (1.3.3).

# Chapter 7. Holding power to account

This chapter addresses research question 3: to what extent can the system of LEPs and CAs be described as accountable and inclusive and finds significant problems in both areas. It does so by examining in the specific circumstances of the North East and Tees Valley and their formal governance institutions the concepts of accountability and inclusion discussed in section 2.2.4. Accountability is discussed in terms of electoral accountability to voters and internal accountability by leading councillors to their backbenchers through overview and scrutiny (O&S) as discussed by Stoker *et al.* (2004), Wollmann (2008) and Leach (2010). The chapter situates the leadership provided by the Tees Valley mayor in the context of the distinction drawn by Hambleton (2015a) between facilitative leadership and the 'city boss' (1.3.3). It examines the public consultation carried out in relation to devolution and the extent to which its results were taken into account when decisions were made, complementing concerns expressed by the PSA (2016) about the reluctance of elites to broaden participation in the devolution process.

Inclusion is considered as an economic concept concerned with ensuring that everyone can benefit from participation in the labour market, and as a political concept concerned with enabling all to be involved in decision-making procedures. These two aspects are seen to be distinct, but with a causal connection, as argued by Young (2000).

### 7.1 Accountability

The following sections deal in turn with accountability as it applies to Tees Valley, NELEP, and NECA and North of Tyne. In Tees Valley and, since 2019 in North of Tyne, accountability is focused on a directly-elected mayor, whereas rump NECA's rejection of a devolution deal means its CA's electoral accountability is indirect, *via* the leaders of the four councils who sit on its leadership board. Section 7.1.1 includes a discussion of the Tees Valley mayor's leadership style in the context of the distinction drawn by Hambleton (2015a) between facilitative leadership and the 'city boss' (1.3.3). The North of Tyne mayor's leadership is not considered as his election occurred after the period on which this thesis is focused. There is a discussion of internal accountability through the O&S system, which is found to be problematic in both Tees Valley and the

North East. The performance of NECA council leaders in contributing to NELEP's democratic accountability is also problematic, to the point of having come to the government's attention.

#### 7.1.1 Tees Valley

On 7 May 2021 (postponed from May 2020 due to the coronavirus emergency) the mayor of Tees Valley is due to stand for re-election after three years in post. It will be a crucial test of the claim – intrinsic to the government's devolution agenda - that metro mayors are a way of bolstering the accountability of local governance in England (Osborne, 2014). Mayor Houchen is one of only two people in North East England to have been elected specifically to lead a CA with devolved powers and increased funding, and the first to face re-election.

Tees Valley's mayor has a public profile probably as high as possible in local media and among voters. He won the position in 2017 in favourable electoral circumstances by a tiny margin. If he can hold it in 2021, particularly if there is a national swing against the Conservatives, some support will be provided for the claim for mayoral accountability. If local voting follows the national trend, and particularly if the turnout is again low, the claim will be thrown into doubt. It is an election meriting close attention, and by 2019 Mayor Houchen was already preparing, with the support of Boris Johnson at a fundraising dinner. The mayor, a Brexit supporter, went on to back Johnson for the Conservative Party leadership (Houchen, 2019e; Walker, 2019f; Houchen, 2019j).

Mayor Houchen's popular style, with a high profile and personal identification with big projects like the airport purchase and the regeneration of the SSI site, means he relies on a continuing series of visible successes, and is in permanent campaigning mode, with frequent use of social media. In 2019 he adopted another high-profile cause, advocating free ports for post-Brexit Britain. In a characteristically ambitious move, he produced a 'white paper' for the two Conservative Party leadership candidates setting out plans for six ports (including the Tees) creating 70,000 jobs and adding £4bn to the economy. He tweeted it on a day he campaigned with Boris Johnson on the Conservative Party leadership trail (Giles *et al.*, 2019; Houchen, 2019f). When Johnson was appointed Prime Minister and mentioned free ports in his first public statement, Houchen was able to boast 'Boris Johnson...is backing my policy to roll out free ports'

(Houghton, 2019; Houchen, 2019i). Days later he was celebrating again when the International Trade Secretary, Liz Truss, undeterred by her earlier encounter with a parmo (5.5), visited Tees Valley again to confirm the free port policy (Walker, 2019a).

The Tees Valley mayor is evidence that Hambleton's distinction between a facilitative leader and a city boss is not a binary choice (Hambleton, 2015a: 11). Houchen combines characteristics of both. His facilitative, political and professional skills were never better displayed than in the controversy over the airport purchase as he won over his cabinet yet, on the day of his triumph in front of the TV cameras and hundreds of enthusiastic voters, defended the cabinet members, his political rivals, from criticism (5.5).

Nevertheless, the 'city boss' aspects of the mayor's political persona are there for all to see in his dominance of Tees Valley's public and political discourse, its media reporting, his discussions with ministers, his occupation of the chair of the MDC, his airport victory, his self-identification and combative style as defender of the rights of local people against private equity 'sharks', 'profiteering opportunists' and others (6.2.2), and in the answer he provides to the Kissinger question (6.2.1, 6.2.2). All this does mean that he has some critics:

It's becoming more about the Ben Houchen Show and what is he trying to build for, for himself, long term rather than genuinely looking to help the wider economy. I hear a number of people saying he won't meet with people unless he feels that it's in his [personal] interests. (Business leader. August 2018)

Members of the TVCA cabinet are also democratically elected, but indirectly. They are there because of their positions as leaders of their councils. They are not personally identified with TVCA as the mayor is.

CAs, like councils, have internal mechanisms to hold their executive arms to account. TVCA, its cabinet and mayor are overseen by two committees. The overview and scrutiny committee (OSC) reviews the most important strategic decisions and the direction of the CA to ensure any decisions are in line with agreed policies. It meets once every six weeks (TVCA, undated-h). It has 15 members, of whom (until the 2019 council elections) 11 were Labour, three Conservative and one LibDem. The role of the

audit and governance committee (AGC) is to assure sound governance, effective internal control and financial management, and it meets at least three times a year (TVCA, undated-i). It had four Labour members and a Conservative. Until 2019, meetings of both committees usually took place as scheduled, though between the mayoral election in May 2017 and March 2019, of 16 meetings of the OSC one was cancelled and two were inquorate, and of 11 meetings of the AGC one was cancelled due to bad weather and three were inquorate. Some councils' representatives were less diligent in attendance than others', and the CA found it necessary to feed this back to leaders and chief executives in the hope of improvement (TVCA, 2017g; TVCA, 2017h; TVCA, 2018j; Tees Valley OSC, 2019; TVCA, 2019f; TVCA, undated-g).

The OSC paid close attention to the mayor's most controversial plan - to buy Durham Tees Valley Airport. In January 2018, when it heard £46,000 had been spent on financial and legal advice with more set aside, the OSC wanted to be fully informed of financial and legal issues. There was a 'full discussion' on whether £500,000 set aside was reasonable. In March 2018 the OSC held a special, closed meeting devoted to the airport and held a 'wide-ranging discussion', followed by requests for confidential information, and for regular updates. Another special, closed meeting was held on 20 December 2018 at which questions were put to the mayor and proposed airport operators. The mayor made a presentation and answered the committee's questions again, in public, in January 2019, before cabinet finally approved the purchase – again in public (5.5) (TVCA, 2018k; TVCA, 2018l; TVCA, 2018m; TVCA, 2019b).

But that was not the end of the story. The OSC was soon asking critical questions of its own role in the airport affair. It should have used its call-in powers, said one Labour member; including the airport deal within the much wider investment plan at a late stage (5.5) had been 'gerrymandering' and a ploy to make call-in more difficult: 'I don't think we did enough'. It must also have been difficult if not impossible for backbench councillors, without independent expert advice, to evaluate the more than 200 pages of business case, business plan and valuation report with which they were presented (TVCA and Mayor, 2019). The OSC was not denied the information it needed; it was overwhelmed by it. No further action was taken, but the OSC chair, referring to mayor and cabinet, noted: 'It's not our job to work for them, it's our job to scrutinise and make sure they're doing their jobs' (Metcalfe, 2019i).

The OSC continued to be uneasy about the airport. At an inquorate meeting in April 2019 the chair said the committee must play a significant role in future overview and scrutiny of the airport. It was stressed that serious debate was needed about the role of O&S and that an arrangement needed to be in place to allow effective and timely scrutiny (TVCA, 2019g).

This episode marked a troubling and unexplained development in the work of scrutiny at TVCA. The OSC's meetings of March and April 2019 and an extraordinary meeting in July were all inquorate. The OSC chair acknowledged the recent poor attendance when presenting the committee's annual report to the cabinet in June, and noted one result was that a report on diversity (7.3) had not yet been approved. The mayor, in a remark without further explanation, said 'it was his personal belief that some committee members had been told not to attend, particularly in relation to the diversity report'. Poor attendance was reported by Hartlepool members in particular (TVCA, 2019j). The diversity report was on the OSC's agenda at a quorate meeting on 12 July but members voted not to progress it at that stage; they were concerned how TVCA could make changes if its five constituent councils did not do so (TVCA, 2019k). A year later, in June 2020, the report was still in limbo (TVCA, 2020d).

When the researcher asked to see notes that were taken of the inquorate OSC meetings (in place of minutes) he had to submit a request under the Freedom of Information Act, even though in at least one case a media report of the meeting appeared the following day (Metcalfe, 2019i). One item to emerge was that the inquorate meeting of 1 March 2019 agreed to hold a special meeting four days later (if possible) to discuss the diversity report. There is no record of such a meeting having been held (TVCA, 2019o).

This was not the only development affecting the OSC's work at the time. TVCA's cabinet was told by the chief executive in June 2019 that amendments were being made to the TVCA constitution, including to the way the OSC worked. Among these, the requirement to send all decisions to the OSC within two days was being replaced by a requirement to send only key decisions within that period (Gilhespie, 2019). This prompted the OSC to use its call-in power, and on 12 July – its first quorate meeting

since discussing the airport in January – some members expressed concern that they had not been consulted, and about other constitutional changes as well (TVCA, 2019k).

Among them, the OSC was told that the rules had been amended to make it clear that it was the chair (i.e. the mayor) who agrees items for the cabinet agenda, prompting the response that O&S should retain its independence to put reports to the Cabinet: 'There may be instances in the future where the mayor may not want a specific report to be considered and could refuse to submit; this goes against the purpose of overview and scrutiny' (2019k).

It is apparent from the OSC's minutes that these worries caused some division among members. The call-in is recorded as having been supported by five committee members, and the committee's decision following its discussion of the issues - to defer the matter so all members could digest the information - is recorded as having been agreed with a majority vote of six (out of 13 recorded as in attendance). The authority's monitoring officer advised that deferral was possible, but only once, and members were told another meeting must be held within 14 days. Yet a week later, when an extraordinary meeting was held to decide what to do about these issues considered important enough to have prompted a call-in, it was inquorate (TVCA, 2019k). The call-in subsequently expired and the constitutional changes were adopted by default (TVCA, 2019I: 2).

When this researcher asked a question about a discrepancy in the new constitution between the mayor's right to determine the cabinet agenda and the OSC's right to present reports, it was confirmed that OSCs' rights are established in law (TVCA, 2019p).

Members of the public concerned about accountability and reliant on the CA's published documents were left to wonder what was going on. The OSC appeared to have been transformed from vigilant watchdog to sleepy lapdog in six months. The events of 2019 over the airport and its scrutiny, the spate of inquorate meetings, the constitutional changes and their unconsummated call-in, and the delay to the diversity report suggest Tees Valley's cherished trust and collaboration might be breaking down. The OSC was inquorate again in January 2020, and in the committee's annual report for 2019-20 its chair reported that she had obtained consent from the Communities

Secretary to allow substitute members in the hope of overcoming the problem (TVCA, 2020d; TVCA, 2020e).

Tees Valley's LEP, as with other LEPs, is much less accountable to the public than the CAs. It is not elected at all. LEPs represent the local business community, not the general population. As an integrated part of the TVCA, the LEP takes part in its public meetings, normally held every two months. But it also meets monthly in private to enable 'commercially confidential items to be discussed and for open and frank exchanges of information and views to be expressed that might not otherwise be expressed in an open forum'. The minutes of these meetings are published (TVCA, 2019c: 22). A LEP board member interviewed for this study made no claim to be publicly accountable; quite the contrary. He and the other members, he said, were collectively accountable as representatives of the business community; it was the job of the politicians to represent voters. LEP members were an expert sounding board for the CA but did not need to be accountable to the public because they did not vote (Business member of Tees Valley LEP. June 2018).

#### 7.1.2 North East Local Enterprise Partnership

NELEP, like its Tees Valley counterpart, argues that as an unelected, business-led, single-purpose organisation it is not in a position to answer to wider society, which should be represented through elected councillors. The only time it seriously considered widening its remit to encompass social issues, in 2017, it eventually backed off (4.2). A statement on NELEP's website makes clear that it regards its democratic accountability as being exercised through NECA, including its OSC (NELEP, 2018a: item 6, pp. 2-3).

Accountability went up NELEP's agenda following a report for the government into LEP governance and transparency, the Ney Report, in 2017. Although NELEP had been given a clean bill of health, it responded to Ney with some procedural changes to keep up with best practice, including a new policy on whistle blowers and a revised form for declarations of interest. But the board still meets in private, believing that the amount of commercially confidential business it deals with makes public proceedings impractical (NELEP, 2017g: 4,19).

Nevertheless, one of NELEPs roles, according to its constitution, is to provide local accountability and its attempts to square this circle include the online publication of its official papers (subject to confidentiality). Among these are regular, detailed and informative programme delivery updates against the SEP. NELEP operates according to the Nolan Principles of Public Life and the Langlands Principles of Good Governance, under which it is committed to giving partners, funders and local people information that meets their needs, being open about what it does and publishing information about its activities wherever possible. It holds stakeholder events to launch new initiatives and an annual state-of-the-region event, issues an e-bulletin, sends an enewsletter to 2,000 subscribers and has more than 9,000 followers on Twitter, the second highest of any LEP. Openness and accountability are written into the members' code of conduct. As with all LEPs, it is required by the government to publish an assurance framework to demonstrate standards of governance, transparency and accountability consistent with its role. It is subject to the Freedom of Information Act via NECA (NELEP, 2018f: 19, 21-22; MHCLG, 2019d: 18; NELEP, 2019d: item 4; NELEP, undated-a).

Whether these arrangements make the system accountable is a matter of opinion, and interviews revealed a wide range, with no pattern. Views included: there is accountability for specific funding but not strategic decision-making (Researcher, voluntary sector. January 2018); and, the government says it is devolving power but will not devolve the governance and accountability structures (Backbench Labour councillor, Gateshead. February 2018). A business leader, referring to NELEP, said: 'There is a sense that whatever the LEP will do the LEP will do' (Regional business leader. September 2018). A business support leader in County Durham said he had good informal relationships with NELEP staff but no official contact. He expressed high regard for NELEP's professionalism but did not think accountability was high on their agenda (Business support executive, County Durham. October 2018).

A survey of stakeholders, delivery partners and board members commissioned by NELEP from an independent research agency resulted in positive feedback. Fifty people took part in an online survey (out of 200 asked) and 18 gave in-depth interviews (out of 50 asked). The results showed that NELEP's role was well understood and most respondents thought it made a significant contribution to
economic growth. The leadership and wider team received recognition and there was an awareness of the challenges they faced. A question was raised (but no answer provided) about whether the LEP should be more vocal on political issues (NELEP, 2019d: item 8; DRG, undated).

One major cloud hangs over the NELEP board's performance in overseeing its own responsibilities, and it relates to a matter under the complete control of its own members – attendance in sufficient numbers to ensure a quorum, which consists of ten members including at least five from the private sector and four council leaders (NELEP, undated-a). It was the council leaders who were mainly responsible for inquorate meetings.

Between 2014, when NECA was established in an atmosphere of optimism, and May 2016, when it was going through the drawn-out process of rejecting its devolution deal and splitting in two, all but one NELEP meetings were quorate. But at that point inquorate meetings became a problem, with an inquorate meeting in July 2016. It was at its most serious over a period of 20 months, from 25 May 2017 to 31 January 2019. Between those two quorate meetings the NELEP board held nine consecutive inquorate meetings. In seven of those it was the absence of council leaders that caused the meetings to be inquorate. In the period covered by this study, from September 2016 to March 2019, of 16 board meetings ten were inquorate, and it was the council leaders who were inquorate on eight of these occasions (**Table 7-1**). This is a serious failing of democratic oversight if the government's view is accepted:

As autonomous local partnerships, local enterprise partnerships are primarily accountable to the communities within their area. In practice, the full and active role of senior local authority representatives on these boards provides a strong and direct link back to local people and are one part of the local enterprise partnership's democratic accountability (MHCLG, 2018b: 19)

All seven councils share responsibility for inquorate meetings. Of a total of 28 board meetings, quorate and inquorate, over the life of NELEP between NECA's establishment in 2014 and March 2019, Durham missed 11 meetings, Gateshead 7, Newcastle 17, North Tyneside 16, Northumberland 11, South Tyneside 10 and Sunderland 12.

Table 7-1: Attendance at NELEP board meetings, September 2016-March 2019.Source: NELEP website

Source: NELEP				
Date	Quorate or	Private sector attendance	Local Authority attendance	Local Authority absentees
	inquorate			
29/09/2016	Quorate	8	4	Gateshead, Newcastle, North Tyneside
24/11/2016	Quorate	7	4	Gateshead, Newcastle North Tyneside
26/01/2017	Inquorate	10	3	Gateshead, Newcastle Northumberland, Sunderland
23/03/2017	Quorate	9	5	Gateshead, North Tyneside
25/05/2017	Quorate	8	6	North Tyneside
27/07/2017	Inquorate	6	3	Durham, Newcastle South Tyneside, Sunderland
28/09/2017	Inquorate	7	3	Durham, Northumberland North Tyneside, Sunderland
30/11/2017	Inquorate	4	4	Newcastle, North Tyneside
25/01/2018	Inquorate	7	2	Durham, Newcastle, North Tyneside, Northumberland, South Tyneside
22/03/2018	Inquorate	7	2	Durham, Newcastle, North Tyneside, Northumberland, South Tyneside
07/06/2018	Inquorate	5	4	Durham, Newcastle, Northumberland
26/07/2018	Inquorate	6	2	Durham, Newcastle, North Tyneside, Northumberland, South Tyneside
27/09/2018	Inquorate	7	2	Durham, Gateshead, North Tyneside, Northumberland, Sunderland
29/11/2018	Inquorate	8	3	Northumberland, Newcastle, South Tyneside, Sunderland
31/01/2019	Quorate	7	5	Newcastle, Sunderland
21/03/2019	Quorate	8	5	Newcastle, North Tyneside

Inquorate meetings were the main governance issue raised by the government when NELEP underwent its annual appraisal on 14 January 2019 at a meeting attended by a

senior civil servant and NELEP leaders. They assured the MHCLG [Ministry of Housing, Communities and Local Government] that the matter was being raised with board members and constituent local authorities and would also be raised at the next board away day (NELEP, 2019b).

Subsequent board minutes record that 'the chair noted that it had been made clear that there was a very good written representation process in place which ensured that decisions were made in a timely fashion' (NELEP, 2019c: 5). Written representations may result in timely decisions, but whether they facilitate 'a full and active role' for local authority representatives and provide 'a strong and direct link back to local people' is at the very least open to question.

Nor has the problem been solved, in spite of government intervention. Board meetings were inquorate again in May and September 2019, as this thesis was being written, though in September it was the private sector that did not muster the required numbers. NELEP was still looking for a solution to the problem of inquorate meetings when it held its AGM in September 2019 (Hodgson, 2019). In November only one councillor was present.

The failure of the NECA leaders to attend regularly at NELEP board meetings may be interpreted in terms of their lack of veto power (4.7) compared with their positions of strength in their own local authorities and, to a lesser degree, on NECA. Actors with strong veto capabilities on one institution may lack them on another. The NECA leaders, in a minority on NELEP, thus face difficulties in trying to displace or convert it, and drift (along with layering) are theoretically more promising strategies, not requiring any direct changes, as Mahoney and Thelen (2012: 19) point out.

### 7.1.3 North East and North of Tyne Combined Authorities

As the North of Tyne councils prepared to follow Tees Valley down the devolution route, one of their leading figures expressed faith in the system of local democracy it would entail, combining a directly elected mayor and indirectly elected council leaders:

The mayor would be elected. We are elected - the leaders...People can get rid of you very, very quickly. They don't have to stay with you for a lifetime if you are

not successful. I think that's the accountability, isn't it? (Council leader. March 2018).

But another NECA leader, alluding to the Tees Valley mayoral election, was more sceptical:

Twenty-one per cent of the electorate electing you; does that make you accountable to anybody in particular? I don't think so. (Council leader. February 2018)

While this leader made a valid point in referring to low turnout, he also had a vested interest in undermining the credibility of the mayoral model, for he represents one of the south of Tyne councils which rejected devolution in part because of opposition to a mayor.

As in Tees Valley, one of the bodies in a position to question NECA's leadership board, and with a duty to do so, is the overview and scrutiny committee (OSC). Effective scrutiny arrangements are said to be an essential component of local democracy, enhancing accountability and transparency of decision making and enabling councillors to represent the views of constituents. They are in place 'to enable local councillors, on behalf of their communities, to scrutinise and challenge all matters within the remit of the combined authority' (NECA, undated-a). NECA's OSC consists of two members of each constituent authority, reflecting political balance – 12 Labour, a Conservative and a LibDem in 2016 as NECA was making its devolution decision (NECA, 2016t: 28)

NECA's OSC also has co-opted non-voting members as chair and vice-chair, who are described as independent and impartial. OSC members were told by officials in 2017 that these roles would be advertised, the appointment process open and transparent. There were set criteria to determine whether or not an applicant could be considered non-political. Non-political did not necessarily mean non-party, provided the applicant was not a councillor of the majority party. The chair appointed in 2017 (after the devolution debate featured in this thesis) is a Labour Party member but not a councillor (NECA, 2017e: 28; NECA, 2017f: 6; NECA, 2017g; NECA, 2017p: item 6; Gooby, undated).

The relationship between the OSC and the body it scrutinises can sometimes appear cosy, as noted by Wollmann (2008: 285). On one occasion (before the appointment of the independent chair) NECA's OSC noted that 'there should be more publicity generally of the positive things that were happening within NECA' (NECA, 2017f: 3).

Devolution came to the attention of NECA's OSC in March 2015, when the presence of the BBC at its meeting was an indication of rising public interest. This was at a time when NECA was pursuing devolution with enthusiasm, and the committee shared its approach; members thought this was 'a really exciting time'. But one OSC concern, surprisingly in the light of subsequent events, was that an official report made no mention of a mayor: 'In Manchester the agenda being worked towards was predicated on an elected mayor. The North East could potentially have seven leaders and there was a danger that it lacked a figurehead'. By July however the OSC was on message alongside the leadership, to the point of offering support rather than critique over the issue. It noted among members' comments that 'the support of this committee in expressing reservations at the possibility of an elected mayor may be welcomed by the leadership board' (NECA, 2015g: item 4, pp. 5, 8; NECA, 2015h: 5).

The committee returned to devolution in December 2015 and met again at 2.00pm on 22 March 2016, the day Gateshead's cabinet had come out against the deal at its 10.00am meeting. The OSC expressed worries mirroring those of the NECA leadership board: the role of the mayor, the importance of democratic decision-making, and general uncertainties including over funding. On the other hand were the benefits of joint working and the successful example of Manchester. Comment was also made on 'unfair criticism of the proposals by the media'. It was noted that the government was keen to transfer a lot of responsibility to NECA, but this was discussed in the context of risk, presumably because it was felt resources would be inadequate. The committee decided nothing, but to keep devolution as a standing agenda item (NECA, 2016u: item 4, pp. 10-11).

By the time of its September 2016 meeting, when the leadership board had already decided not to proceed, the OSC was fully on side with the deal's opponents. It 'considered the correct approach had been adopted and informally endorsed the decision of the leadership board as the government had failed to provide the assurances sought for a properly funded devolution deal'. In March 2017 the OSC

recognised the prospect of NECA splitting into two was a worry; if that happened, it was suggested, there would be a problem of the authorities competing against each other rather than co-operating (NECA, 2016r: item 3, pp. 5-6; NECA, 2016s: item 3, p. 2; NECA, 2017h: item 3, p. 3).

This record raises questions. Four of the nine councillors present in September 2016 when the OSC informally endorsed the decision not to proceed represented councils which had supported the deal, including a Northumberland Conservative and a Newcastle Liberal Democrat; if any of them did not agree that the correct approach had been adopted, the minutes do not say so (NECA, 2016r: 6; NECA, 2016s: item 3).

And why was the committee's endorsement informal? The answer is that its meeting was not quorate and its decision had to wait until the next meeting to be formalised (NECA, 2016s: item 3). The previous meeting, in July, was also inquorate (NECA, 2016i). In fact, as the devolution controversy raged, the OSC did not hold a single quorate meeting between 22 March and 1 November 2016, and was without a chair from NECA's annual meeting on 21 June until 1 November. At a critical moment in NECA's history the body supposed to scrutinise its actions was not fully functioning.

The leadership board was aware of this situation, and at its 2016 annual meeting 'considered the challenges in holding a quorate meeting' and decided to allow substitute members. That did not solve the problem though; the OSC was inquorate again at two meetings out of five in 2017, and four out of five in 2018, meaning there was only one quorate meeting between September 2017 and March 2019 including none for a year between 15 March 2018 and 14 March 2019 as NECA decided to consent to its own division into two and in its new four-council form adopted a new constitution and approved its first budget – though informal discussions were always held. There was then no meeting at all for six months (NECA, 2016j: 1tem 3; NECA, 2018b: item 3b; NECA, 2018d; NECA, 2018h; NECA, 2018i; NECA, 2018j; NECA, 2018k; NECA, 2018m; NECA, 2019c; NECA, 2019d). As we have seen (7.1.2), at the time the NECA leaders tried to deal with this problem at the OSC they were themselves about to embark on a period when they failed to muster a quorum at most meetings of the NELEP board.

The other body with a role in the oversight of NECA is its audit and standards committee (ASC) (previously governance committee). It is said to be 'a key component' of NECA's corporate governance arrangements and 'an important source of assurance about the organisation's arrangements for managing risk, maintaining an effective control environment and reporting on financial and other performance, and for the promotion and maintenance of high standards of conduct by its elected and co-opted members'. It has an independent chair and vice chair and an independent person as observer as well as, under the rump NECA, eight councillors of whom six are Labour and two LibDems (NECA, undated-b). This committee too has an incomplete record of quorate meetings: of ten scheduled between September 2016 and March 2019, one was cancelled, three inquorate and one part inquorate (NECA, 2017i; NECA, 2018e; NECA, 2018f; NECA, 2018g).

Lack of public knowledge of, or interest in, governance was indicated when NTCA sought to recruit two independent persons for the purpose of its standards regime, with website and press adverts, and there was only one applicant, who was successful. There were however two applicants to be independent chair of the ASC (NTCA, 2019c: 8; NTCA, 2019e: 5).

For many of the intended beneficiaries of economic development, discussion about CA and LEP governance may appear remote. One politician, referring to a deprived town, said LEP-funded activity was taking place there which people liked, but without being aware of the LEP. Did that make the LEP accountable to ordinary people in a meaningful sense? 'Absolutely not' (North East Party member. December 2017)

# 7.2 Public consultation

Devolution is circumscribed by statutory public consultation at every stage. In the North East and Tees Valley it involved a very small proportion of the population (apart from in Durham (7.2.1). It could be, and was, effectively ignored. One backbench councillor interviewed for this study spoke of talking to local people informally about the issues, and there is no reason to doubt councillors do consult in this way, but such conversations are unsystematic and, in this particular case, resulted in established views being confirmed rather than challenged:

I asked people in my community, in my ward, what do you think to this [devolution], and people were very suspicious. They like what works now, on the whole (Labour councillor, North East. March 2018).

The following sections deal in turn with public consultation as conducted in NECA, North of Tyne and Tees Valley.

## 7.2.1 North East Combined Authority

NECA's creation in 2014 followed a consultation by the Communities Secretary, who wanted to be sure that the opposition of voters to governance changes when they rejected a regional assembly in 2004 did not extend to the NECA proposal. There were 73 responses, of which 21 were from local organisations and their representative bodies (including ten from local government), 11 from local businesses and their representative bodies, nine from elected representatives, two from the voluntary sector and one each from a community newspaper and a trade union; 28 were from individual residents. Of total responses, 64% supported establishing a CA, 30% opposed and four commented without coming down on one side. The seven councils all supported the idea, which is hardly surprising as they had proposed it, though perhaps not with enthusiasm, as suggested by a former NELEP board member (6.2.1) (Gateshead Council, 2013: item 3, 2; DCLG, 2014: 4-8)<sup>38</sup>.

The breakdown of responses does not make clear whether the opposition came from institutions, elected representatives or residents. However, further light on public attitudes resulted from consultations carried out by each of the councils individually, covering 650 interested parties including around 450 residents. An online survey received 444 responses including 324 from residents, with 75% supportive of the proposal; there were 16 focus groups, meetings with interested parties, and local press articles. Consultation by the local authorities, according to an official report, showed 'overwhelming support from stakeholders'. MPs and members of the House of Lords

<sup>&</sup>lt;sup>38</sup> The Tees Valley councils, consulted about the NECA proposal, did not want the name 'North East' to be used for fear of causing confusion. They were right, as readers of this thesis will recognise.

were 'very supportive' (Gateshead Council, 2013: item 3, appendix 1; appendix 3, 49-50; DCLG, 2014: 4-8).

NECA's decision in 2015 to seek a devolution deal sparked another round of consultation, with 'inclusive' events in all seven council areas which showed 'clear support' from the public and business. MPs and peers were supportive of the broad principle and keen to champion the NECA proposals. The consultation feedback was used in framing the proposals which NECA put to the government. The government's requirement that devolution be accompanied by a mayor was recognised, so in going ahead councillors must have thought it was not a deal breaker, presumably because either the government would drop its insistence or they would drop their opposition, or perhaps a compromise would result in a mayor with few powers (NECA, 2015a: item 3, 2-3; NECA, 2015d: 8-10; NECA, 2016d: 41, 54-55).

The signing of NECA's devolution deal was the trigger for another round of consultation, including six public events involving 374 people and sessions with the voluntary sector, business and trade unions. Altogether there were 750 responses and they were 'positive overall'. There was said to be strong support among a wide range of stakeholders from communities, businesses and partners for the principle of devolution and broad positive support for the proposals in NECA's agreement with the government (NECA, 2016d: 4-5, 28, 42-47).

The only test of public opinion to involve a large number of members of the public was a poll of all residents by Durham County Council. There were 81,964 responses representing 21.7% of the electorate, coincidentally almost exactly the same as the turnout in the Tees Valley mayoral election (3.1.1). This was described by the council as 'evidence of a significant amount of interest in what is essentially a policy issue'. A majority of 59.5% thought that devolving some power and resources to the North East would be a step in the right direction; 14.9% thought it would not; the remainder thought it would make little difference or didn't know. On mayoral powers, 47.8% thought that the mayor should have limited power and influence, 40.3% that the mayor should have quite a lot of power and influence, and 11.9% were unsure. On the possibility of more devolution in future, the most popular response was 'let's wait and see' (42.9%); 28.3% would want to see more devolution and 22.4% would not. As to

whether the agreement would bring more prosperity and jobs, 40.5% thought it would, 9.1% thought it would not, and 36.2% thought it would make little or no impact; 14.2% did not know (Henig, 2016: paragraphs 25-27).

These results are open to interpretation. Followed as they were by the Durham County Council leader's vote at NECA on 6 September 2016 not to proceed to the next stage of the deal, effectively killing it, they prompted one fellow leader to comment to this study:

Durham had a referendum and the majority of people wanted to go ahead with it. Yet he voted against it. So that £250,000 or whatever it was that he spent on a referendum was meaningless. Because the Labour group or whoever they were put the pressure on him (NECA board member. March 2018).

A Durham backbench councillor, though, interpreted the referendum result as support for devolution in principle but not for the proposed governance structure, including a mayor (Labour councillor. April 2018).

In other consultation by the seven councils, 85 businesses attended an event in Durham, and the overall tone was in support of the deal. Gateshead Council undertook a website survey which elicited 81 responses, with 62% agreeing with the focus of the proposed agreement. Gateshead also hosted an event, the official report of which gives no clear indication of support or opposition. Newcastle Council's consultation elicited feedback which 'demonstrated very significant support for the principle of devolution and for using the current agreement as a platform for further opportunities'. Events in North Tyneside indicated 'broad support'. Northumberland County Council sent an online questionnaire to its 1,654-strong People's Panel, with 350 responses; the results showed majorities of between 70% and 78% thought it was important for the North East to take control of a range of issues. The council held meetings with a variety of organisations and 'on balance the view given was that devolution should broadly be supported, albeit *caveated* by the fact that the government's approach was rather unstructured and in danger of leaving places in England behind'. South Tyneside Council had a devolution consultation page on its website and held a number of events, but no feedback was supplied at the time NECA received reports from the other councils. Sunderland Council received 582 responses

to a website survey and newssheet delivered to every home, with 61% agreeing that some strategic decisions should be made at a more regional level. The disadvantages noted were the elected mayor, the feeling that the government was ignoring the 2004 referendum, the additional tier of bureaucracy and inadequate funding (NECA, 2016d: 47-50).

NECA's decision not to proceed with its deal was therefore taken in spite of the 'positive overall' support referred to above. The public consultation results were reported to the NECA leadership board on 24 March 2016, two days after Gateshead Council had come out against the deal. A council leader explained his view of public consultation, expressed in relation to the North of Tyne deal but applicable generally:

In North of Tyne they've carried out a public consultation and 1,086 people have responded on line. By my calculation I think it's 0.04% of the population of the affected area, and it's a permanent governance change. I think that's inherently undemocratic (NECA council leader. February 2018).

## 7.2.2 North of Tyne Combined Authority

Public consultation on the North of Tyne deal ran from 14 December 2017 to 5 February 2018, with public meetings in all seven NECA areas, and official literature struck an upbeat note. With references to the £600m investment fund being offered over 30 years, the consultation document described the moment as 'a once in a lifetime opportunity to help to grow the economy, retain the region's talent and support residents to access new and better jobs'. As well as the powers and resources in the deal, the public was told there was a commitment to explore new opportunities in future (NTCA, 2017a).

Perhaps the most robust debate to occur in public took place during a consultation event in Gateshead, where the council had been the first to reject the NECA deal. Newcastle's leader, in presenting the NTCA case, faced a series of critical questions, including from two Gateshead Labour councillors, around the divisiveness of the devolution process, the issue of mayoral powers and democracy, and the alleged

paucity of the financial offer, as well as the suggestion that the consultation was meaningless as the decision had already been made<sup>39</sup>.

According to a report on the consultation, of about 1,400 people and organisations that took part, a majority supported the overall plans. Actually, the report's accompanying chart shows that of 1,082 people who answered the question on governance – 'Do you agree or disagree with our proposals to change the way councils in the North East work together in order for devolution in the North of Tyne to be implemented?' - 50% precisely (541) agreed while 412 disagreed and 129 neither agreed nor disagreed (NTCA, 2018a: 5, 10).

Feedback from stakeholders was supportive in almost all cases, though with a variety of reservations. Among them, the business community saw North of Tyne devolution as a first step towards its preferred option covering the whole of NECA. The TUC wanted to be involved in consultation on the deal as it developed. Unison supported devolution in principle but not the North of Tyne deal because of the proposed level of funding, and Unite was also concerned about funding and the impact on the rump NECA. Newcastle University backed the deal, including an elected mayor, enthusiastically and wanted to work alongside the new combined authority and NELEP (2018a: 22-27).

While stakeholders were generally supportive, and 50% of respondents backed the new way of working together, the results from those who provided more detailed feedback revealed more people worried about the deal than were content, though the numbers were smaller. Of those agreeing, 25 thought the deal made sense, 20 that it would give greater access to government funding, 11 that the area should not be left behind and 11 that an elected mayor would raise the region's profile and be key to unlocking government funding. Of those disagreeing, 58 wanted to retain NECA as it was, 48 thought the funding was not enough and the proposals would not have the desired impact and 40 were against an elected mayor. Thirty-two respondents

<sup>&</sup>lt;sup>39</sup> Personal observation.

'thought the proposals were designed to meet the personal interests and ambitions of local decision makers or political parties' (2018a: 37-38).

Three of the four south of Tyne authorities responded. Durham was most concerned to ensure continuation of effective joint working on transport. South Tyneside's overarching concern was that the North of Tyne deal 'must be on the explicit basis that it does not lead to any detriment to the economic, social and environmental wellbeing of the residents and businesses of South Tyneside', with special mention of education and training. Sunderland was also insistent on 'no detriment' and wanted the same assurance as South Tyneside over education and training. It was worried too that accelerated house-building north of the Tyne might result in a slow-down to the south. Overall, it welcomed the North of Tyne deal and was disappointed something similar could not have been achieved for NECA. Gateshead did not respond. (2018a: 49-59).

Individuals accounted for almost 1,100 of the responses, and 902 of those were residents of the NECA area, ranging from 316 from Northumberland to 18 from Sunderland. Of those prepared to say, 698 were aged over 45; there were 629 males and 377 females; 940 gave their ethnicity as white British, with 13 white Irish as the next biggest group (2018a: 42-43).

The consultation leads to several plausible conclusions. The North East establishment in the form of the organisations taking part in various aspects of governance, including business, supported the proposed deal. Most local people either did not know about the consultation or did not care enough to participate. Of those who did, most were middle-aged to elderly white men, of whom half were in support and the other half either disagreed or neither agreed or disagreed. Those who disagreed were more active in giving feedback, suggesting they felt more strongly.

## 7.2.3 Tees Valley Combined Authority

The first test of public opinion in Tees Valley came when consultation was carried out between 10 December 2014 and 31 January 2015 on the proposal to create the original (non-mayoral) TVCA. Of 1,911 responses, 1,638 were from residents. Of the responses, 74.4% agreed with the idea of partnership working between councils and business and 64.8% supported the idea of a CA. The mayoral issue was not part of the

consultation, though it could and arguably should have been, for it was known by then in the light of Osborne's *Northern Powerhouse* speech that a mayor was likely to be an element in any future devolution package (DCLG, 2015a: 12).

The Communities Secretary carried out statutory consultation between 28 October and 9 December 2015, after the Tees Valley councils and LEP had signed their provisional deal providing for a CA with an elected mayor. Of 28 responses, four were irrelevant to the issues; 11 were from residents and one from a councillor, and the remainder from local authorities and organisations. While there was institutional support, most individuals were opposed. Three of the 11 were from Hartlepool and said they did not associate themselves with Tees Valley. Faced with this negligible response, the Communities Secretary in approving the proposal took into account the consultation carried out by the local authorities a year earlier, even though the devolution deal, with its mayoral element, had not been included (DCLG, 2016: 5-8).

A third consultation between 11 July and 22 August 2016 on new governance arrangements to accompany devolution raises other serious concerns. The consultation elicited 1,160 responses, of which 1,067 were from residents. Once again, people were not asked a question about the principle of having a mayor. That did not stop them providing an answer. As TVCA's report on the consultation notes, 'many respondents used this consultation to express opposition to the principle of an elected mayor, notwithstanding that this was not something on which we were seeking views'. Principal objections to having a mayor concerned centralization of power, cost and disregard for the previously expressed views of people in referendums and consultations. Of 547 who answered a question about specific powers and responsibilities, 67 took the opportunity to give their view on having an elected mayor, with over 90% opposed. Of 535 who responded to the question about how the mayor should work, 133 focused on whether there should be a mayor at all and the 'vast majority' were negative (TVCA, 2016d: paragraphs 3, 4, 30, 33).

Hartlepool was disproportionately represented in this consultation. It is the smallest of the five boroughs with a population of 92,500, or 13.8% of the Tees Valley total (TVCA, undated-b), yet produced 501 (46.9%) of the 1,067 resident responses. Hartlepool had a mayor between 2002 and 2012, when it voted in a referendum for abolition, and

opposition to a Tees Valley mayor during this consultation was particularly strong in Hartlepool; views about the geographical unsuitability of the Tees Valley area were also predominant in Hartlepool (TVCA, 2016d: 21, 31, 33).

The TVCA report following the consultation argued that the nature of the exercise and the diversity of the views expressed meant it was not possible to draw a simple statistical conclusion. Where clear views expressed did not directly refer to the consultation - for example the views opposing a mayor - they had been noted. However, of those people who gave views on the consultation questions, a majority overall supported the proposals; many mentioned the need for checks and balances on the mayor's powers and this emphasised the importance of the measures in the governance scheme. In an apparent reference to Hartlepool, the summary noted that individual communities emerged with clear views, but these individual interests did not negate the overall conclusions drawn from Tees Valley as a whole. The same was true of certain business sectors: it was not possible to draw conclusions suggesting the responses were statistically representative of the population. The official response claimed that the consultation had taken place at the formative stage of the devolution proposal, prior to any final decision and at a time when the decision could still be influenced by the outcome. However, given that the powers being consulted on were devolved from central government rather than being centralised from local authorities, it said, the proposals did not harm and should enhance local community powers and identity (TVCA, 2016d: paragraphs 10, 15, 17, 35-37).

A possibility of influencing the outcome through the consultation may have existed theoretically, but in fact the MCA went ahead, and on the geographical basis always envisaged. Tees Valley thus got its mayor after three consultations, the first of which did not raise the issue of an elected mayor at all; the second elicited only 28 responses; and the third attempted to side-step the principle of a mayoralty but was ambushed by a few score voters from the smallest of the five boroughs. It was a far-from-ideal way to end an unsatisfactory process. A Labour campaigner against the current devolution deals, observing events in Tees Valley from the North East, commented:

As far as the population as a whole is concerned they haven't got a clue what is happening. They have had a metro mayor foisted on them without having a referendum (Labour anti-devolution deal campaigner. November 2017).

The consultations described in this section illustrate the problems experienced more widely in such exercises, notably the challenge identified by the NAO in getting responses to governance issues (Morse, 2017) as well as the need expressed by the PSA (2016) to reinvigorate public participation (2.2.4). The devolution consultations also had a drawback of their own: because the government was making mayors an essential condition, the public were not asked about the contentious principle of having a mayor but only about the powers s/he should have. Previous questions on this point had produced largely but not wholly negative responses. Middlesbrough and North Tyneside have borough mayors and Hartlepool had a mayor between 2002 and 2012, when it voted to abolish the post (Mulholland, 2012). Referendums were held in 2012 in ten English cities, including Newcastle in the North East, on whether to introduce mayors and all but Bristol rejected the idea (BBC, 2012).

It is noteworthy that those opposing devolution shared two of their councillors' reasons for doing so, and one of those was opposition to a mayor. The other reason for rejecting devolution common to councillors and the public was the inadequacy of funding. Consultation participants also expressed opposition on two grounds not prominent among councillors - bureaucracy and a belief that the project was designed to serve personal ambitions.

In spite of these opponents, however, the balance among the public was in support of devolution in NECA, where it was rejected, while in Tees Valley there was initial support for a non-mayoral CA followed by inconclusive results when the mayoral element was added. While the numbers taking part (apart from Durham's poll of all residents) often numbered in the hundreds and sometimes over 1,000, they were still a very small proportion of the total electorate. Equally concerning is that stakeholders (apart from the unions) were also overruled in the NECA area. Stakeholders work with and are familiar with local government and are the partners it needs if the joint civic leadership envisaged by Hambleton (2015a) (2.2.4) and the common purpose found by the OECD (2006) (3.1.1) to be lacking are to be realised.

# 7.3 Inclusion

Inclusion in one sense is concerned with whose voice is included in the decisionmaking process. Inclusion is 'a core of the democratic ideal' (Young, 2000: 13). But this is not the sense in which most people engaged in economic development understand it. They mean inclusive growth – giving everyone the opportunity to participate in the labour market. This distinction between political and economic inclusion (or exclusion) is one drawn by Young (2000), though she also recognises that economic exclusion helps to account for political exclusion (2.2.4).

LEPs and CAs take economic inclusion seriously. Inclusion stands alongside employability as a joint objective aimed at ensuring all have both the skills and the opportunities to access the labour market. Inclusion will be achieved, according to NELEP

by addressing high levels of youth unemployment, reducing inequalities, improving economic wellbeing and by ensuring that growth is inclusive and assists those most distant or disadvantaged in the labour market and living in areas of persistent and entrenched deprivation (NELEP, 2014a: 58).

Inclusion was one of six strategic themes in NELEP's 2014 SEP, allocated £196m to be spent by 2020/21. By 2019 NELEP had spent £18.4m with a further £28.5m from other public and private sources on skills and inclusion, including careers guidance and support for 170 schools and colleges (NELEP, 2014a: 7, 87; NECA, 2019d: 50). Programmes include initiatives to provide more joined-up support for individuals with moderate mental health issues return to work; work with employers to improve takeup of the Better Health at Work Award Scheme; building capacity in the voluntary and social enterprise sectors to support people who are excluded from work; and adopting a community-led local development approach in communities suffering significant deprivation (NELEP, 2019b: item 4).

Inclusion in the labour market has improved markedly since 2014 for young people. Unemployment in the NELEP area for those aged 16-24 was 20% and has fallen to 11.5% and, moreover, has come into line with the England (excluding London) average. But it is still higher than any other age group. Another priority now is the 50-64 age

group, where unemployment remains above average, though it too has fallen since 2014, down from 5.6% to 4.1% compared with the 2.8% national average. The problem is exacerbated by the area's older age profile (NELEP, 2019e: 15).

However, some of NELEP's programmes to promote employability and inclusion are facing obstacles. A scheme to support primary care professionals help people access help to get back to work has, according to NELEP, been struggling to get referrals from Jobcentres and faces closure following a decision by the Department for Work and Pensions (DWP) not to extend the contract. Monitoring and evaluation of the Work and Health Programme faces 'significant challenges'. Proposed projects intended to focus on up-skilling and reskilling to aid job progression for those in work face possible delays. A successful project to support third sector organisations work with unemployed learners has a waiting list but needs more funding for which no decision date has been given at national level. Lack of support or delay pose risks to a project to support people with protected characteristics under the Equality Act who are furthest from the labour market. And a substantial amount of EU funding to support disadvantaged groups in the workplace was not yet allocated, with time running out<sup>40</sup> (NELEP, 2019b: item 4).

Tees Valley's SEP speaks similarly of 'increasing the lifetime opportunities for young people and older workers which will help ensure a more inclusive economy'. TVCA is working with its constituent councils to run a DWP Routes to Work pilot project for people over 30 facing barriers to employment, and is developing programmes to tackle long-term unemployment and create a careers and enterprise initiative (TVCA, 2016a; TVCA, undated-j; TVCA, undated-k; TVCA, undated-l).

Tackling exclusion from the labour market is difficult, as NELEP has found. By September 2018, four-and-a-half years into its SEP, it had reduced the gap with England excluding London in the economic activity rate by only 15% compared with a target of 50% by 2024 (6.5.1).

Most interviewees for this study saw inclusion in terms of the labour market. A policy institute researcher said:

<sup>&</sup>lt;sup>40</sup> January 2019.

We think of it as growth that benefits everybody, so that is about people's ability to be able to participate in the labour market, to be able to work in a job where they are actually able to see their standard of living improve; it is about having the skills and capabilities to enable you to do that. (Researcher. January 2018)

But those on the receiving end of services see inclusion differently. Those who feel they have no voice inevitably want one. Grassroots development workers in a deprived town spoke of being ignored by those at the strategic level. Even if they were consulted, it had no effect. One said:

Inclusive to them means consultation, the fact that they can say 'there's some information, there's consultation, we are being inclusive because we are asking people's opinion'. Inclusivity to me means about having direct contact and a voice that's heard (Business development worker, Northumberland. January 2018).

Perceptions differ. Those involved in business support at a rather higher level were more positive. One, operating in County Durham, felt he had access to NECA through the county council: 'I know if I'm unhappy about something that's taken place within the combined or any other authority I know who I can go to; I know I can email [council leader's first name]' (Business support executive, County Durham. October 2018). A counterpart in Tees Valley said: 'As a business I feel like we are listened to more. I also feel that they [TVCA] canvass for opinion quite a lot as well, which is something that didn't happen before.' The same interviewee saw inclusion as meaning having one's voice heard, but pointed to a problem with the process as a small, self-selecting group could come to dominate, with the same people appearing repeatedly in social and mainstream media, usually with their own agenda (Business development executive, Tees Valley. October 2018)

About three-quarters of people in all parts of North East England feel unable to have much say over local decisions, according to a 2018 survey for the BBC – though it was not concerned with CAs and LEPs specifically, and respondents may have had their

local council in mind when saying how much influence they felt they could have (Table

**7-2**).

Source: YouGov for BBC				
Area	Can influence local decisions fair amount or a lot/not very much or not at all (%)			
England average	21/73			
North East regional average <sup>41</sup>	20/74			
North East (NECA) average	20/73			
Tees Valley average	19/74			
County Durham	20/74			
Darlington	21/73			
Gateshead	20/74			
Hartlepool	18/75			
Middlesbrough	18/74			
Newcastle	23/71			
North Tyneside	21/73			
Northumberland	21/74			
<b>Redcar &amp; Cleveland</b>	19/75			
South Tyneside	19/74			
Stockton	20/74			
Sunderland	19/74			

 Table 7-2: Influence over local decisions.

 Source: YouGov for BBC

LEPs and CAs are arguably constitutionally unsuited for inclusive decision-making. They have formal roles for three of the five leadership groups identified in the analytic framework for this study: elected politicians constitute the CAs, where they are advised by professional public servants, and business representatives lead the LEPs. But there is no guaranteed place for the trade union movement, as there was on the board of One North East. Higher and further education have seats at the LEP table (and one such representative on NELEP is a Labour Party and trade union member (NELEP, 2018g)), and the NHS has a representative on Tees Valley LEP, but otherwise community organisations are unrepresented at the top level, though there are places for them on various committees and subordinate bodies of both LEPs and CAs. Faith

<sup>&</sup>lt;sup>41</sup> Regional, NECA and Tees Valley averages are rounded

leaders and other voluntary sector representative are outside the system, at least at the highest levels.

NELEP in its early days was up-front about the fact that it is 'a public, private and education sector partnership'; for NELEP it is this 'triple helix' leadership of public and private sectors and higher education that is important (NELEP, 2017d: 10; NELEP, undated-c). Below board level, it has a business growth board with 14 members including two from education and one from social enterprise; an employment and skills board with 15 members including the TUC and two from the charity sector; and an innovation board with 10 members from the private sector, education and local government.<sup>42</sup> NECA's economic development and regeneration advisory board (EDRAB) has places for the chamber of commerce (NEECC), TUC, CBI and Federation of Small Businesses<sup>43</sup>.

The inclusiveness of LEP governance is, however, evolving. The government is encouraging LEPs to recruit a voluntary sector champion to their boards. NELEP was told at its annual appraisal with the government that it was 'required' to have voluntary and community sector representation on the board as a matter of good practice, and to nominate a diversity champion. It says it has 'a strong partnership with... the third sector, who are represented on its advisory boards as appropriate', and in early 2020 a voluntary sector representative was appointed to the main board (NELEP, 2018e: item 9, p. 2; NELEP, 2020a). However, some groups still feel ignored. One interviewee told this study:

Despite the fact that we are the only black-led women's organisation that's around, that's survived for 25 years, and our focus has been very much on economic independence until recently...there's never been even the slightest bit of attempt, whether it's by the authorities or the others, to really identify organisations that they needed to include to make sure that that voice was part of decision making (Voluntary sector leader, Tyneside. December 2018).

<sup>&</sup>lt;sup>42</sup> Source: NELEP website, March 2019.

<sup>&</sup>lt;sup>43</sup> Source: NECA website, March 2019.

LEPs must have 33% of women on the board by 2020 and 50% by 2023 (NELEP, 2018e: item 9; NELEP, 2019c: 5). The NELEP board has five women out of 18, but the 50% target will be a problem unless more councils elect women leaders.<sup>44</sup>Councillors take seven seats on the board and six are men, and if that does not change by 2023 eight of the other 11 will have to be women, though the numbers will be affected if total membership of the board changes. The government has said it will support boards of up to 20 members and wants two-thirds to be from the private sector (MHCLG, 2018b: 5).

Tees Valley had a 21-member LEP board in March 2019 – 14 private sector including higher and further education representatives, and seven public sector comprising the mayor, five council leaders and an NHS representative<sup>45</sup>. It planned to reduce the number to 20 that May when the terms of office of several members expired. Six board members were women, including three appointed in 2018, and the aim was to increase this to 50% by 2020. This, said TVCA, 'reflects the combined authority's commitment to diversity which is not just about the gender balance but ensuring that the combined authority is reflective of the local community' (TVCA, 2019c: 12; TVCA, 2019n)

TVCA commissioned an O&S review of equality and diversity within the CA, its constituent local authorities and the LEP in 2018. It found a 'universal desire' to more closely represent local communities and a recognition that this was not being achieved. Among findings, although women made up an above average percentage of councillors, they appeared not to be progressing into leadership roles (TVCA, 2019d). The diversity report was affected by delays caused by problems in the O&S system, as we have seen (7.1.1).

The trade unions are well represented and influential in the North East's local government (4.4), but have no designated place in the CA/LEP system. Opportunities to make their voices heard are patchy and depend on local arrangements and even personalities. A union movement representative compared access to power in Tees Valley unfavourably with that in NECA and North of Tyne. In NECA, trade unions and

<sup>&</sup>lt;sup>44</sup> Source: NELEP website, March 2019.

<sup>&</sup>lt;sup>45</sup> Source: TVCA website, March 2019.

others are represented on sub-groups and can use them as channels to access decision-making boards. According to the union leader, this works quite well and enables them to have input on important issues like Brexit and employment and skills. In North of Tyne, even before its statutory establishment, talks were taking place on issues of concern to the unions. But the situation is different in Tees Valley, according to this union leader:

We don't have a seat on anything on Tees Valley. I have tried to engage with the mayor by investing a significant amount of money into some research around carbon capture and storage and trying to complement their strategic economic plan for that area. I have worked with industry; I have worked with politicians, with business, the unions, leaders from the local authorities to develop that piece of work<sup>46</sup>...But to get round the table with the mayor...has never happened, despite asking twice for a meeting. It does very much depend on the personality of the mayor, and their particular goals and ambitions... It's very much about who you know, and leverage to get in, rather than: 'Actually this is the make-up of our region, these are the voices, these are actually the diverse voices we want in our governance structures'. Relations at officer level, they're all OK, but is it the be-all and end-all if you haven't got the ear of the mayor? No (Trade union leader. October 2018).

TVCA has a selection of around 70 representatives of various stakeholder groups, including business, education and government agencies, on partnership boards dealing with issues like education, employment and skills; culture and tourism; innovation; and transport (TVCA, 2018c). A senior Tees Valley official commented cautiously that while this reflected an embedded Team Tees Valley approach among many local organisations, absent in the North East, it did not necessarily mean inclusive decisionmaking (TVCA official. January 2018). This approach may, however, it is hoped, at least exert influence. Tees Valley has set up a Business Engagement Forum - with 91 members in March 2019 - to help it keep in touch with the wider community. TVCA and the LEP

<sup>&</sup>lt;sup>46</sup> For transparency: a reference to research carried out at the Centre for Urban and Regional Development Studies, Newcastle University.

recognise that the private sector members [of the LEP] cannot represent the views of 17,230 businesses in Tees Valley. Therefore a variety of engagement mechanisms are utilized to ensure that the broader business community has the ability to influence strategy and policy development, our investment priorities and to be actively engaged in the delivery of some of our activities, particularly around supporting careers development with schools (TVCA, 2019c: 14).

As we have seen in section 2.2.4, a critique of the inclusiveness of regional approaches to economic development is that the focus on growth and the creation of 'better' jobs comes at the expense of improving existing jobs (Sissons *et al.*, 2018). A policy institute interviewee for this study worried that mass-employing, low-paying sectors like retail and hospitality were poorly represented on LEP boards, which were skewed toward sectors like legal and manufacturing (Researcher. January 2018). The care sector is another that came to prominence during the coronavirus emergency.

Neither retail nor hospitality is represented on North East England's two LEP boards, and MPs have found that these industries have also been overlooked by the government for sector deals under the *Industrial Strategy* (BEIS Select Committee, 2019). The NELEP board includes representatives of the engineering, utilities, legal, automotive, ports, architecture and business development and transformation sectors. Tees Valley LEP has representatives of chemicals, brewing, telecoms, ports, manufacturing, PR and digital entertainment<sup>47</sup>. The researcher quoted above believes that increasing emphasis on inclusive growth is bringing more pressure for voluntary sector participation in inclusive decision-making through involvement in LEPs:

I suppose you could argue that that is what a big part of what the local authorities' role is, because they are also there as well as representatives of the people. But...now that inclusive growth as an idea has started to gain a bit of traction...it becomes more difficult for them to maintain there is no role for third sector organisations. As you start to think about things like how do we engage marginalised people in the labour market then actually that is the kind

<sup>&</sup>lt;sup>47</sup> Source: LEP websites, September 2018.

of area where organisations potentially can bring a lot of expertise to the table (Researcher. January 2018).

Beel *et al.* (2018) argue on the basis of research in Manchester and Sheffield that there is a clear divide between those who have a voice in the devolution, city-region agenda and those who have been marginalized. Civil society groups have been placed outside the decision-making process while being expected to deal with the fallout from uneven development, inequality and austerity.

Beel *et al.* (2017: 572) see the devolution of health and social care in Greater Manchester as an opportunity for greater voluntary sector involvement, and something similar might have happened in the North East. In an early initiative under its proposed devolution deal, which could have seen it getting involved in multi-agency work in dealing with social issues wider than the purely economic, NECA co-operated with local NHS organisations to set up the North East Commission for Health and Social Care Integration. The commission investigated the health and wellbeing gap between the North East and the rest of the UK and health inequalities within the area. Though eventually devolution did not proceed, the commission completed its work in 2016 and, according to its chair, its recommendations could be implemented through existing structures (Selbie *et al.*, 2016: 1). But an interviewee who took a close interest told this study it was a wasted opportunity:

Because of the failure to sign off as a properly devolved authority that's all gone; it's just not there. Nothing is happening...There has been no support for this from any of the authorities in the combined authority. (NHS governance volunteer. April 2018)

Lessons from Greater Manchester include the need to include the voluntary sector as actors in the devolution process (Beel *et al.*, 2017: 566) – a view supported by the experience of the leader, quoted above, of a voluntary organization which helps ethnic minority women into employment in the North East. Their situation, she said, had not improved in spite of all efforts:

For 30 years unemployment for African, Pakistani and Bangladeshi women had stagnated and that reflects our experience, and despite a lot of public policy and development (Voluntary sector leader, Tyneside. December 2018).

In November 2018 around 150 of North East England's voluntary organisations met at a Collaboration Conference to discuss how they could better work together. 'We are people who feel that collaboration is a great way to drive social change and who are motivated to do something to help this to happen', said an organiser<sup>48</sup>. A conference report found from seven case studies that shared goals, trust and established relationships were central to motivating people to work together. 'Sadly', it said 'collaboration is not yet the norm across the North East' (Cole, 2018; Webb *et al.*, 2018)

# 7.4 Conclusion

This chapter has provided an answer to research question 3, asking to what extent the system of LEPs and CAs can be described as accountable and inclusive, by showing that North East England's two CAs are both democratically accountable but not equally so. Democracy, as Young (2000: 5) says, is not an all-or-nothing affair but a matter of degree; societies can vary in both the extent and the intensity of their commitment to democratic practice.

So it is with CAs and LEPs. Tees Valley is headed by a directly-elected mayor and NECA by indirectly-elected council leaders. LEPs are much less democratic, unelected but subject to the input of democratically-elected politicians. This is an effective process in Tees Valley, where the mayor and cabinet are firmly in control, but much less so in the North East, where councillors are in a minority and less than assiduous in attending meetings, demotivated probably by their lack of control, as suggested in section 7.1.2, and distracted by their municipal duties.

Inclusion is also problematic in North East England. The normative legitimacy of a democratic decision, says Young (2000), depends on the degree to which those affected by it have been included in the decision-making process and had the opportunity to influence the outcomes. Representation is most inclusive when it

<sup>&</sup>lt;sup>48</sup> Personal observation.

embraces relatively marginalised or disadvantaged groups (2000: 6, 8). Yet it is just such groups, such as the black women's group referred to above (7.3), which do feel excluded.

Political inclusion is evolving, with the voluntary sector gaining some involvement. But there is still no formal place for the trade unions in the governance system. They are strong in the North East's councils, as we have seen (4.4) but that is because of Labour's electoral strength; a change in the political balance could see them excluded, as a union leader feels has happened in Tees Valley. Business is included through its leadership of the LEPs, where it has significant influence over technocratic decisions in the North East, but NECA's rejection of its devolution deal shows the limits of business's power – its proper limits, some will say. In Tees Valley business is subservient to the mayoral CA, embraced by it but not empowered. So in both areas locally elected politicians remain in final charge, subject to the important *caveats* about continuing government control discussed in section 2.2.2.

One big step towards addressing issues of both accountability and inclusion could be taken by broadening the democratic debate. While councillors, being elected, can justly claim the right to decide, others in society can fairly claim the right to participate meaningfully, on an informed basis, and to influence. This is the model of collective civic leadership envisaged by Hambleton (2015a) and referred to throughout this thesis, and it is also advocated by Young (2000: 8):

I look to the vast range of activity often brought under the label 'civic society' for important forms of participation, of expression from a situated perspective, and forms of holding power accountable that a strong communicative democracy needs.

One step is being taken in this direction with the inclusion of a representative of the voluntary sector on LEP boards. But it is a small one. Recommendations for further reform will be made in the final chapter of this thesis.

We have thus seen that there are problems in both accountability and inclusion. Judging by yardsticks used by Young (2000), democracy is limited in both extent, particularly in the North East where no one is directly elected to sit on NECA, and in

the intensity of commitment to democratic practice, as evidenced by ineffective scrutiny and, in NELEP, persistent inability to muster quorate numbers.

Inclusion has been seen until recently almost exclusively as an aspect of the labour market, and LEPs are working hard to overcome the obstacles to achieving it. Inclusion in the decision-making process is evolving, but some groups continue to feel marginalized.

Chapter 8 will draw together the difficult issues that have been discussed throughout this thesis in both the economic and political spheres and present, by no means a solution to the problems, but a suggested approach to fulfilling the necessary precondition of working together to tackle them, and some detailed recommendations for incremental reform.

# Chapter 8. Conclusions and recommendations

This chapter draws together answers to the three research questions set out in section 1.1. It then details the contribution the thesis makes to its field, dealing in turn with the empirical, conceptual and methodological contributions.

Empirically, it concludes that the lesson of the research is that for devolution to work local politics, shaped by the socio-cultural environment, must be amenable. Conceptually and, it argues that the best way to achieve this viable devolution governance is a bottom-up approach. Methodologically, the chapter commends detailed research of the type evident throughout the thesis, which digs deep into case study reality to supplement, though not replace, theorising about how to achieve the most appropriate formal institutions for local economic governance.

The thesis provides a valuable addition to the case study literature of the era of localism in England, thus providing empirical evidence to support or critique the theory. It is one of only a few studies in its field and of its era to have interviewed significant numbers of key players from the political, official, business, trade union and voluntary sectors. Using a critical realist approach, it has dug deep into the underlying reality of the devolution process in North East England. In doing so, it contributes a detailed study of how the informal institutional environment of a place in the form of socio-cultural values and political traditions and practices can shape its formal institutional arrangements and its response to the political and economic challenges of development.

A further contribution throws light on the workings of the formal institutions of economic governance, including the power relations between five groups of civic leaders – politicians, officials, business, trade unions and the voluntary sector - which are analysed using the framework developed by Hambleton (2015a) (2.2.4). His model of civic leadership as a collaboration between these five groups is found to be still a distant ideal in North East England.

The thesis gives a rare insight into the 'back stage' machinations of local governance (Ayres *et al.*, 2017b) where these power relations are played out, and the internal

workings of the party group system discussed by Copus (2004), affording it the attention which he says it deserves and complains it has not been given. It examines the shortcomings of accountability in the system, both externally by voters due to low turnouts and the tendency for local voting to follow national trends, as noted by Copus (2004) and Newton (1976), and internally through the overview and scrutiny (O&S) system, confirming the doubts of Stoker *et al.* (2004), Wollmann (2008) and Leach (2010) about its effectiveness. Additionally, it reveals problems in achieving diversity in governance.

A further contribution is that the thesis provides a case study of leadership, contrasting the mayoral model that has developed in Tees Valley since 2017 with the traditional, collectivist approach still operating in the North East.

Economically, the thesis finds that after five years of their SEPs, the North East and Tees Valley provide only limited and inconclusive evidence at aggregate level of an economic dividend, in line with findings by Pike *et al.* (2012). It therefore does not provide clarification of the contentious question of place-based *versus* spatially blind development discussed in the literature (2.1.1).

What the thesis can offer in the area of economic governance, however, is a suggested way forward which could encourage collaboration between municipalities and civic leadership groups and potentially facilitate the combination of economic efficiency and social justice hoped for by Martin (2008) and others (2.1.1). This suggested approach, based on building economic governance from the bottom, is outlined in section 8.2.3, with detailed recommendations in section 8.3. These recommendations are designed to make local governance more fit for the purpose of economic development by improving the legitimacy, transparency, accountability and inclusion of the formal institutions of devolution.

## 8.1 Answers to the research questions

The following section summarises the answers to the three research questions and how they were arrived at.

#### 8.1.1 Research question 1

Research question 1 asks why the combined authorities in the North East (NECA) and Tees Valley, so similar in many ways, took opposite decisions over devolution. Finding an answer involved consideration of the region's socio-cultural environment including the key concepts of place and identity, agglomeration and dominance. In Tees Valley and some parts of the North East these issues were reflected in resentment and resistance to Newcastle's perceived receipt of favoured treatment because of its position as regional capital. This factor operated with opposite effects in the two areas. In the North East it tended to pull the seven NECA councils apart, divided between those willing to accept devolution and those fearing a continuation of Newcastle's perceived preferential treatment. In Tees Valley it encouraged the five councils to remain united, create a devolved authority and consolidate their collective separate identity as distinct from the territory to their north.

Other factors, the research found, all springing from the socio-cultural history of the region and its political traditions, contributed to the different devolution decisions. In NECA there was disdain for the proffered devolution grant, especially in the midst of a period of austerity, and resistance to the concept of an elected mayor in place of the traditional collective leadership, as noted in the literature (2.2.4), as well as lack of trust in a Conservative government. Leaders in Tees Valley, in contrast, were pragmatically prepared to accept a mayor and do a deal in order to access whatever money they could.

Two very different informal institutional environments thus pulled the North East and Tees Valley in opposite directions. North East unity was fatally undermined by historic local rivalries, while the area's lack of effective political opposition meant there was no countervailing pressure except from council officials and business leaders. Private decision-making in party groups and a virtual absence of public debate allowed complacency and factionalism to flourish. Above all, and permeating these factors, was a lack of the mutual trust identified by Rhodes (1996) (2.2.1) as so essential to governance.

In Tees Valley, in contrast, the institutional environment was very different. Trust between council leaders and with the private sector was built on the strong

foundations of collaboration dating back 20 years, while familiarity with Conservative politicians enabled business to be done with a Tory government.

The North East and Tees Valley have much in common, yet reacted very differently to devolution offers because their informal institutions and hence their politics are different in important respects. The North East's devolution story has been of a struggle between political forces based on history, geography and the local identities they create on one hand, and economic interest on the other. The politics was against a deal and the counterbalancing economic rationale was not strong enough to prevail. The result has been a technocracy that no one likes but everyone can live with. In Tees Valley, where politics and economics were pulling in the same direction, devolution was accepted comparatively easily.

### 8.1.2 Research question 2

Where the answer given to research question 1 focused on the influence of the informal institutions which shape the formal, it is the nature of the formal institutions and the personality of the Tees Valley mayor which explain the different approaches to economic development. It is not, it should be noted, the economies of the two areas or their strategic economic plans (SEPs), which are quite similar, which caused the divergence of approach.

Following its acceptance of devolution, Tees Valley's formal institutions include a directly elected mayoralty to which significant economic powers are attached. The mayor, a proactive personality with an ambitious political as well as an economic agenda, dominates Tees Valley's governance and media scene, maintaining a regular presence in social and mainstream media on a virtually daily basis and on a range of issues, and he has succeeded in achieving the political transformation of the area. He is making the most of his democratic mandate, however narrowly achieved in 2017, to pursue high-profile projects like the purchase of Teesside Airport and the development of the SSI steelworks site. He wins headlines in part by presenting himself as the champion of local people and stoking public grievance against others perceived to be frustrating their interests – Thai banks, the private equity owners of British Steel and the owner of the 'Teesside Airport' name (6.2.2). Meanwhile, the routine tasks of implementing the SEP, while not neglected, are left largely to officials with the advice

of the LEP – but all subject to the democratic oversight of the mayor and his cabinet, where he is the dominant figure.

In the North East, in contrast, there is no single dominant figure (apart perhaps, since 2019, the North of Tyne mayor, who is outside the temporal scope of this study). No politician has taken ownership of economic development, and the front man for the SEP has been the NELEP chair, conscious of his position as an unelected business leader and reluctant to stray beyond economics. NECA has retained the collective leadership discussed in section 2.2.4, but it is riven by the divisions that led to its rejection of devolution in 2016 and subsequent formal split into two parts. The historic, place-based rivalries discussed in section 3.1.1. remain strong. The result is a technocratic approach to economic development led by the business sector on NELEP and with weak and demotivated input by political leaders whose prime focus is on their municipalities.

The North East and Tees Valley share many economic problems but have thus adopted different political approaches to tackling them. In the North East, economic development under localism has been an example of what Tomaney (2014a: 134) has called a technical exercise disturbed by an excess of politics and which Clarke and Cochrane (2013) worry can be anti-political in the sense of favouring apparently effective solutions and freezing out alternative approaches, favouring the wealthy and educated and excluding the inarticulate and poor (2.2.1).

Economic governance in the North East fits this picture. It has been led by the business community and disturbed by politics. Politicians have accepted business leadership because they had to. They have worked with the SEP and played their part in implementation by taking lead roles in transport, economic regeneration and inclusion and employability. But no politician has created a high public profile as the SEP's champion or the voice of the North East. The semi-detached relationship of the council leaders is exemplified by their irregular attendance at NELEP board meetings (7.1.2).

The similarities of the two SEPs, considered alongside the contrasting approaches to their implementation, indicate that while the leaderships of both the North East and Tees Valley are open to path creation in the economic field and the development of new industries, whether in energy, technology or services (Chapter 6), the politics of

the North East (but not Tees Valley) retains a path dependence rooted in the industrial history of the 20<sup>th</sup> century.

This answer, finally, with its focus on the contrasting mayoral and technocratic approaches to development of the North East and Tees Valley, contributes to the comparative case studies in place leadership which have emerged from the research – business-led and technocratic in the former case, political, democratic and popular in the latter - as well as of the places themselves, their socio-cultural environment, their politics and their economies.

### 8.1.3 Research question 3

The subject of this question and its answer are the key concept of the accountability of the formal institutions of economic governance in North East England - the CAs and LEPs of the North East and Tees Valley as well as the mayoral leadership of the latter (1.3.3). The inclusiveness of the institutions is also examined, and the related issues raised in the literature of legitimacy, scrutiny and the role of party groups in local government (2.2.3, 2.2.4) are discussed.

Democratic accountability of the formal institutions exists in both the North East and Tees Valley, but not to the same extent and in both cases is imperfect. The Tees Valley mayor has a direct democratic mandate and will account personally to his voters again in 2021 (delayed from 2020 due to the coronavirus emergency). But he was elected on a turnout of only 21.3%. His cabinet and all the members of NECA's leadership board are elected only indirectly. The legitimacy of elected members is undermined by low turnouts generally throughout local government, while in much of the North East the vigour of the democratic process is sapped by entrenched voting habits.

This public lack of engagement reflects the low level of engagement of the NECA leaders, though it is not directly caused by it, for leaders like all councillors are elected for their work on their local authority, not on their CA or LEP. Demotivated and distracted by the factors discussed in this thesis, including their responsibilities in their own councils, discontented over funding and frustrated by their inability to control NELEP, the NECA leaders simply display no enthusiasm for CA/LEP governance. They suffer from some of the downsides of governance networks identified by Torfing

(2007) (2.2.1), including low expectation of joint benefits and frustration at the absence of clear results, as well as (at the CA/LEP level) lack of leadership. How different it is in Tees Valley, where the mayor, driven by motives including party political and probably personal ambition, has thrown himself into his role, keeping the Ben Houchen Show permanently on the road.

The lack of any direct election in the NECA area and the low turnout in Tees Valley in 2017 are symptomatic of the fact that CAs and LEPs have not caught the public imagination. As the Political Studies Association concluded, CAs need to move quickly to drive public engagement and wider stakeholder collaboration (PSA, 2016: 3). Uninvolved voters do not hold office holders to account. But there is a vicious circle which will not be easy to break. Low levels of engagement and election turnouts will not encourage governments to devolve more, while low levels of devolved power and funding will not encourage public engagement.

With the above *caveats*, in Tees Valley the directly elected mayor and indirectly elected cabinet are in democratic control of their brief – economic development. Democratic control is less in the North East, where the indirectly-elected NECA leaders make up only a minority of the NELEP board. What is more, they fulfil their democratic duties on the board by attending so irregularly that inquorate meetings have become a significant problem (7.1.2).

Scrutiny of the leaders by their own backbench councillors is flawed in both the North East and Tees Valley by a cosy relationship in the former and emerging but opaque signs of discontent in Tees Valley (7.1.1), where inquorate meetings of the overview and scrutiny committee have become a problem.

Inclusion has two aspects, economic and political. As to the former, work is being done to bring disadvantaged groups into the labour market which needs to be continued and stepped up. It is not just a matter of resources, but bureaucratic obstacles, which make the task a struggle, as described in section 7.3. Ideas for developing this work are proposed in section 8.3, as are recommendations for increasing political inclusion.

Inclusion in the decision-making process is already on the way to being broadened to some extent with the inclusion of the voluntary sector, but there is more to be done.

The absence of trade unions and, until early 2020, the voluntary sector on the NELEP board has deprived these civic leadership groups of a voice except through the indirectly-elected and sometimes absent council leaders. Some voluntary organisations, like the black women's group referred to (7.3) feel marginalised. Not all, clearly, can have seats at the table and there will be a heavy responsibility on the single voluntary sector representative to channel their concerns and provide feedback. Diversity remains a problem in both areas, with NELEP struggling to meet gender targets and Tees Valley's diversity report delayed (7.3).

# 8.2 The contribution of the thesis

The following sections describe the contribution made by this thesis empirically, methodologically and conceptually. Empirically, we consider what has been learned about economic governance in North East England. Methodologically, we consider how the way this study has been conducted might inform similar research into the same or similar topics in other areas. Conceptually, we think about a range of issues that arise throughout the thesis, from the key concepts and the literature review to the regional context, the case studies, the approaches to development, accountability and inclusion.

## 8.2.1 Empirical contribution

The overarching empirical lesson of the case studies is that selecting areas for development for purely economic reasons (functional economic areas) without taking account of the politics will not necessarily work. In a democratic country the collaboration of locally elected politicians is required, and it cannot be taken for granted just because some extra powers and funding are on offer. In Tees Valley collaboration was forthcoming but in the North East not. The local politics, in turn, depends on the socio-cultural environment, rooted in history, place and identity. If two such superficially similar areas as the North East and Tees Valley cannot be relied upon to take similar approaches, where can such an expectation by confidently held? This lesson is therefore of wider interest - potentially in every part of England and perhaps even more widely.

The institutional environment of the North East meant that the local councils' (NECA's) debate with themselves over whether to accept the devolution deal on offer was set
against a backdrop of lack of trust. Trust and solidarity, noted in the literature as essential to governance (Rhodes, 1996; Rhodes, 2007) (2.2.1), are historically absent from the North East's local government community, expressed most markedly in resentment at the perceived preferential treatment of Newcastle as regional capital. This fact laid the stony ground on which efforts to encourage devolution withered, throttled by the weeds of rivalry, resentment, jealousy, unwillingness to take a risk and the absence of effective challenge that comes with one-party rule and ineffective scrutiny.

Lack of trust of the NECA members among themselves was compounded by their shared distrust of a Conservative government as a result of the still-remembered and deeply resented injustices of de-industrialisation of the coalfields and shipyards, coupled with current austerity. These feelings remain strong in influential political circles within the region and were forcefully expressed by interviewees. They deprive the area of ministerial allies unless there is a Labour government. No similar feelings over the de-industrialisation of the late 20<sup>th</sup> century were expressed by interviewees in Tees Valley, where these particular manifestations of the influence of informal institutions and evolutionary economic theory, of actors as 'carriers of history' (2.1.3), are weaker. In fact, Tees Valley interviewees barely looked backwards at all beyond the SSI closure in 2015 and the era of RDA domination, as they saw it, from Newcastle.

The fact the devolution debate in NECA was held virtually entirely within the Labour and trade union movement meant the opposing political case went almost unheard, while the public's voice was muted and could be safely ignored. The only robust argument in favour of the deal came from council officials, whose advice ultimately could be, and was, overruled, and from the business community, which was not influential enough in the committee rooms. Both these groups were entirely unrepresented in the forums where the final decisions were made – the Labour groups.

What influenced the Labour groups has been subject to inevitable speculation, but the interviews carried out for this study suggest very strongly that two factors were crucial: money and the mayoralty. And these are indeed the two issues which emerged publicly as matters of concern between October 2015 and September 2016, and by the

time the deal came to be rejected were openly cited as being the deal breakers (NECA, 2016p).

The money problem for NECA was the persistence of councillors in viewing the £30m annual devolution grant in the context of the much greater austerity cuts of recent years, as though the former were intended to compensate for the latter, and they were encouraged by CARD to treat it with disdain (4.4). Endemic distrust of government on money matters was then vindicated in the eyes of NECA at a crucial moment when it failed to get the guarantees it wanted over EU funding (4.1). Coming as it did in the summer of 2016, this rebuff was fresh in councillors' minds when they voted on the deal. Brexit may thus be seen as having played an indirect role in the death of the deal.

The funding issue leaves an intriguing question unanswered: if the money had been deemed sufficient, would the mayoralty alone have been enough to kill the deal? The attitudes of the four councils which voted against are crucial in answering this question, and collectively they suggest probably not. A Durham councillor said: 'For us the key was funding, not the mayor' (Interview: April 2018); a South Tyneside colleague: 'If the deal is good enough and is worthy of the North East region then it is worthy to consider new governance arrangements as part of that' (Interview: March 2018). In Sunderland the view was that they would put up with a mayor provided there were sufficient checks on his/her powers (Sunderland Council, 2016b).

So the key was in the hands of Gateshead, where there were three fears. One was that a mayor would over-ride collective democracy and another that the job might be hijacked for personal ambition – an example of the distrust that permeates regional politics. The third fear was that the mayoral election might be won by a Conservative. In spite of these fears, there are those who believe that had the other six councils stood firm for devolution, Gateshead would have conceded in the end (4.6). This mix of factors was astutely exploited and persuasively presented by CARD, the campaign for real devolution. Opponents of the deal could read CARD's literature and feel their case was intellectually justified.

Of the five civic leadership groups in Hambleton's model, public servants were restricted to their proper role of advising but not deciding, business leaders were

guests in the corridors but without the power, and the voluntary sector was virtually invisible. In an area dominated by one party, the result was an in-house debate within the Labour and trade union movement, and the decision was left in the hands of a few hundred, often elderly, Labour councillors with long memories, meeting in private. These are not the institutional arrangements to facilitate devolution.

Tees Valley's devolution deal did not stir the controversy of that in NECA, and it was at later stages of the process – the mayoral election and subsequent governance – that tensions arose. The factors operating at the critical realist level of the real which facilitated Tees Valley's deal took the form of an underpinning layer of trust and collaboration built over 20 years and a shared, pragmatic desire to access whatever funding was available, reinforced at a critical stage by the SSI closure. Trust was shared between the five councils and with the private sector, and extended when necessary to include the Conservative government represented by ministers with local connections and symbolised in the active presence of Lord Heseltine. A further unifying factor was a desire to consolidate Tees Valley's independence from Newcastle.

A further consideration, perhaps one essential to any deal, is that potentially there was something in it for everybody; both sides felt they had something to gain. Labour council leaders saw an opportunity for additional funding, and to get it were prepared to tolerate a mayor, while the Conservatives saw an opportunity – if they could after all win the mayoralty – to establish a foothold in Tees Valley as a step towards turning more of northern England blue. When they succeeded in winning the mayoralty local politics was swiftly transformed.

While Tees Valley and North of Tyne historically are majority Labour-voting areas, the social and political structures in place did not hold them in such an iron grip that politicians were unable to relate pragmatically to a Conservative government. South of the Tyne, though, while pragmatism and adaptability are present in the economic arena, enabling new paths to be created, as discussed in Chapter 6, adaptability has its limits. Politicians faced with decisions with consequences that were principally political and only secondarily or incidentally economic, were locked in by their values and traditions, structurally trapped in institutional rigidity.

That is not to say that structure was the only factor channelling these politicians towards rejection of their devolution offer. Some had rational concerns about being handed further responsibilities without the necessary resources – concerns noted in section 1.2 (Prud'homme, 1995; Rodríguez-Pose and Gill, 2003) as well as locally in sections 4.5.1 and 4.5.2. But structure channelled councillors south of the Tyne towards rejection more forcibly than in North of Tyne and Tees Valley. Following the shocks of de-industrialisation, the North East economy has been gradually transforming for four decades, but a large part of its politics remains firmly rooted in the 20<sup>th</sup> century.

Agency was arguably strongest of all in North of Tyne, where councillors had to defy pressure to conform to anti-Tory, anti-deal norms, marshalled by CARD, as well as the arguments – sometimes very forcefully expressed – of their NECA colleagues south of the river. Tees Valley's leaders, in contrast, were supported in making the change they did by a history of collaboration and a balancing pressure in support of devolution from their long-standing private sector partners, as well as their own pragmatic inclination to take whatever money was available. The structural constraints binding them were not as tight.

Attitudes among the North East's elected politicians are aligned with and reinforced by the views of their voters and have deep historical roots. For some, voting against a Conservative proposal is so deeply ingrained as to be habit. A leading member of one NECA council, asked why some councillors had voted against devolution, replied:

I haven't got a clue really. I think some people are like sheep. I shouldn't say that. They're feeling as though 'it's the case I have to do this'. I find that quite worrying (Labour council member, NECA. March 2018).

While the specific factors discussed above determined why the North East and Tees Valley took the decisions they did, one further factor may have been lying even more deeply embedded in the minds of those in the four south of Tyne councils that rejected devolution. After generations as a lagging region and decades of failure to fully recover from 1980s de-industrialisation, after repeated economic plans (Board of Trade, 1963; NEPC, 1966; NRST, 1977; ONE, 1999; ONE, 2006) – and with a local industrial strategy on the way (NELEP, 2019b: 5b) – all so far followed by repeatedly

disappointed hopes - some may have lost faith in any prospect of catching up. What difference would another reorganization in the town halls make? Anyway, most people were getting by, weren't they, and had other, community values as well? It was a doubt that had been expressed in the past by Robinson (2002: 326) and was repeated to this study by a young Newcastle graduate:

Sadly, do I think the North East will be a great prosperous region for my grandchildren? I don't think so. It's a kind of eternal war that we know we are not going to win. I don't think we can ever be the best area of the UK unless there was some kind of seismic change, and I don't think most people even think about this. It just seems that it's an ever-long situation where the North East...will probably always be the least prosperous area. But it does not mean that it is doomed or it doesn't mean that we should give up on it. It kind of means the antithesis of that in the sense that we have got to make the best of things however which we can (Private sector professional. Newcastle. November 2017).

#### 8.2.2 Methodological contribution

This thesis has followed a step-by-step process to reach answers to its research questions. The first step, after some reading for the purpose of familiarisation with the subject, was to identify key concepts (1.3) and then discuss them, together with issues they raised, in the light of relevant literature (Chapter 2). Following this conceptual framing, the next step was to bring these concepts and issues closer to home (so to speak) by considering them in the context of North East England but still without relation to specific cases or institutions (Chapter 3). The final step was to examine the specific institutions and actors of the case studies, their natures, procedures and actions, in the light of the conceptual framing and contextualisation that had gone before, and draw conclusions leading to answers to the research questions.

These steps were carried out by means of a methodology comparing two cases in North East England using mixed methods involving official documents, interviews with key actors, news reports, social media and official statistics. The critical realist philosophical approach to this research was judged particularly appropriate to the task. Critical realism enables events to be assessed at the levels of the actual, empirical

and real, which is an approach well suited to the investigation of the world of local governance, as is the narrative presentation. The supplementary *phronetic* approach focuses practical and evaluative rationality on the judgement of concrete situations such as those in the case studies (1.4.1).

The analytic framework for civic leadership developed by Hambleton (2015a), adapted for this thesis, identifies five groups of leading actors in a four-sided framework which is ideal for analysing the events being studied. The four-sided framework was used to place North East England in its relevant context (Chapter 3) while the five leadership groups formed the basis of the NECA case study (Chapter 4), where four of them were major players, and the study in Tees Valley (Chapter 5), where two of them had significant roles.

This case study methodology forms part of what Martin and Sunley (2002: 149) have called 'economic geography proper', though it might equally be referred to as 'political geography proper', so closely connected are the two subjects. This methodology has enabled the research to fulfil its critical realist mission of exploring the institutions and events of North East England at the levels of the actual, empirical, and real (1.4.1) and to progress its search for answers step by step from the concepts and theories of the literature to their regional context and finally to their realisation in the causal mechanisms that resulted in the different devolution decisions of local politicians in the North East and Tees Valley.

If this is a postmodernist age, as referred to in section 2.1.2, metanarratives should be out of favour. The philosopher Jean-Francois Lyotard defined postmodernism as incredulity towards metanarratives (Lyotard, 1984: xiv), and a metanarrative is exactly what a top-down, standardized, all-England regional or local economic governance system is or would be. It is not argued here that theorising about ideal devolution governance is without value, but the claim is made that the detailed, ground-level research used in this study is a useful reminder that an understanding of local conditions is also necessary if governance that works in practice is to be established.

#### 8.2.3 Conceptual contributions

This thesis has shown, in answer to its own title question, that institutions can adapt sometimes, but not always, to new models of economic governance. Whether they can

do so depends heavily on their institutional arrangements which, in turn, are formed by their institutional environment. The fundamental answer to the question, then, lies deep in the political and socio-cultural environment of the areas in question, arising from their geography and history.

That informal environment may not necessarily conform to its superficial appearance. As the empirical findings of this study show, while North East England may appear homogenous, the two areas into which it was initially split by localism are divided by differences more profound than the semblances of unity that kept them together until 2010 in what can now be seen, with hindsight, to have been uneasy collaboration. While Tees Valley's five councils were able to maintain their unity in the face of a new political situation, adapt and do a deal, in the NECA area seven apparently similar councils were unable to hold together when faced with a devolution offer which turned out for them to be a challenge rather than an opportunity. Not even the five Labour-controlled councils of metropolitan Tyne and Wear could maintain solidarity.

Institutions, formal and informal, we thus find, can be enablers or obstacles to the collaborative governance of development, adaptable or inflexible depending on their environment. These outcomes show in the North East, but not Tees Valley, a political path dependence which belies the openness to economic and industrial path creation which is shared throughout the region. It is a path dependence with the objective of safeguarding not particular industries or sectors but the political *status quo* on behalf of a governing establishment.

Whether such an establishment could and would adapt to a new governance scheme if it believed the result would be its own loss of power is impossible to say on the basis of the case studies. We have seen that a majority of councils in the NECA area were unwilling to take even what must be judged a small risk of losing power, bearing in mind the strength of their political stranglehold in the area. The fact that their Tees Valley counterparts were willing to accept devolution and a mayor is inconclusive on this question, for the risk must have seemed relatively slight; they can hardly have imagined the transformation of the area's politics that was to occur within four years. Based on these two cases, therefore, the political will to devolve must be considered an attribute peculiar to each area, not something to be simply assumed by

governments on the basis of economic areas defined centrally and the offer of some extra funding.

Hambleton (2015a) (2.2.4) was right to say that successful civic leadership requires the co-operation of leadership groups across society, not just politicians. But this thesis shows that it is not necessarily the case that all five groups identified in Hambleton's model must be involved. Tees Valley's institutions – mayoralty, combined authority and local enterprise partnership – are working with contributions from politicians, officials and business, but unions and the voluntary sector are largely or wholly absent from governance. The model is an ideal, but viable governance that falls short of it is still possible.

Without at least some degree of cross-group co-operation, however, adaptation may well fail, as it did in the North East. There, relations among politicians from neighbouring municipalities, and between politicians and business, never close, broke down over devolution, resulting in the splitting of the area along the Tyne. Officials were never whole-heartedly behind the rejection of the devolution offer, the trade unions exercised their influence elsewhere, in the Labour groups of the local authorities, and the contribution of the voluntary sector is only now being belatedly recognised with a seat on the NELEP board. Collaboration in pursuit of a common purpose was almost totally lacking, and adaptation proved impossible.

In describing these developments, the thesis contributes a contextual, place-based understanding of devolution, how local actors responded to government policies and what conditions must be met to enable institutions to adapt to new models of economic governance. The value of the case studies of this thesis is that they show that different places, even if superficially similar, respond to devolution in different ways, and governments that fail to understand the localities to which they are seeking to devolve are likely to encounter obstacles. That is why this thesis argues for a *phronetic*, or practically rational, context-dependent approach to devolution.

Leadership is one institution, and an important one, whose adaptability, this thesis shows, is dependent upon circumstances. The form it takes – mayoral or collective – depended in North East England in 2015-16 on the willingness of existing governing establishments to accept devolution. Resistance to an elected mayor was one of the

two most important reasons that a majority of the NECA councils rejected a devolution deal, while acceptance of a mayoralty, albeit reluctantly, was an essential step which the Tees Valley leaders agreed to take. The study therefore finds that the traditional English preference for collective leadership over mayoral leadership in local government remains strong though it can sometimes be overcome if the circumstances are right. As with institutions generally, adaptation can sometimes but not always be achieved.

Once established, the style that a mayoralty takes depends in part on the mayor's formal powers, particularly his/her control over the combined authority's budget. But the Tees Valley case study demonstrates that the personality of the mayor is a still more important factor. A less determined and resourceful character than Mayor Houchen could not have forced through the acquisition of Teesside International Airport against the opposition of his entire cabinet (5.5). He also played a significant, though immeasurable, personal role in turning Tees Valley blue in both local and parliamentary elections in 2019.

The case study of mayoral leadership which this thesis provides demonstrates that in the England of the Brexit era a proactive, combative mayor can impose himself on an area by pursuing popular policies and a high media profile, even with a narrow electoral mandate on a low turnout and with, initially, a cabinet consisting entirely of his political opponents. But, as demonstrated in section 6.5.2, such political success, even coupled with the significant institutional adaptations brought by devolution, do not necessarily translate into significant economic improvement, at least in the short term. Whether mayoral charisma, relentlessly positive messaging and a heightened sense of local identity will prove enough to secure more than transient support if economic progress remains difficult to discern will be seen in the mayoral election of 2021.

What all these findings demonstrate is that that governance for economic development in England cannot necessarily be successfully established on the basis of economics alone. Local governance requires the collaboration of local politics as well, and that means taking account of the socio-cultural environment in which politics is embedded. The top-down definition of geographies for governance based on

functional economic areas will work institutionally if local culture and politics are amenable, as in Tees Valley, but not if they are resistant, as in the North East.

The lesson to be taken from the findings is therefore that local economic governance should be built from the bottom up, not imposed from the top according to a one-size-fits-all blueprint. It requires input from more than elected politicians. Local leaders from all sections of society should come together and agree which area they wish to encompass, with its local authorities, and what powers they want devolved. This will enable those on the ground who understand the local socio-cultural and political context to develop a model that works. Governments may cajole. As Rhodes (1996) (2.2.2) points out, they control funding and wield influence; they can indirectly and imperfectly steer. But they should not impose.

Crucially, local leaders must agree collectively that they are willing to accept the government's conditions for devolution, which means, in effect, an elected mayor.

Unless this happens, regions and/or localities will not be fit for the purpose of devolution. They will reveal themselves as lacking the informal institutional environment which would allow them to make the necessary adaptations, and as a result they will not be able to establish the requisite formal institutional arrangements. NECA is a case in point.

This lesson is applicable at any and every level of sub-national governance. Loose but wide co-operation may emerge from the Convention of the North, or from NP11, the group of *Northern Powerhouse* LEPs, or the UK2070 Commission, or some process resulting from them. The IPPR think tank has suggested, and the UK2070 Commission considered, the idea of four economic provinces for England, including one covering the three northern regions, for certain functions, each governed by an indirectly-elected regional council (IPPR, 2018: 244-246; UK2070, 2019b: 63). It may make economic sense, but northern England is no more immune from geographic rivalries than North East England. A province with a centre of gravity in the Manchester-Leeds corridor would not necessarily be any more acceptable in the North East than a region headquartered in Newcastle was in Tees Valley. It has already provoked competing claims to resources, including from the Tees Valley mayor (Bounds, 2019a; Parsons, 2019c; Houchen, 2019k).

If, as is the widespread demand throughout northern England, regional and/or local government is to be given control over significant new resources, it must show itself to have the solid local support from the public and other leadership groups needed to ensure it has the necessary capacity. That means governance for development must take account of regional/local socio-political circumstances as well as economic functionality. A bottom-up approach is more likely to achieve this, for it is those on the ground who understand the socio-political situation, as argued by Giovannini (2015) and McCann (2019).

Building from the bottom certainly has disadvantages too, at least unless and until the process picks up pace and geographies expand. Small units do not command large resources, and for the *grands projets* envisaged by ambitious regionalists, such as infrastructure, would have to rely on higher levels of governance, such as the panregional Transport for the North (TfN), or *ad hoc* arrangements between neighbours. A sceptical former One North East board member, a businessman, asked:

What are we going to go back to: parish councils? At the end of the day, particularly in a global world, we are competing globally. You need a scale of economy which allows you to punch your weight in a global market, and the more we splint the more difficult it becomes (Former private sector member of RDA board. December 2017).

While that may be correct, splinting is not what is suggested here but an attempt to prevent it. What is argued is that it is the division and acrimony caused by attempting to force political communities together against their will – or at least, the will of their elected local leaders - to fit a top-down model that causes splinting, as happened to NECA. It either does not work or, if imposed, is illegitimate and unstable. Instead, let local authorities combine as they wish, with the support of other local leaderships, give them whatever devolved powers and resources they can handle, and let them grow in size, powers and resources when and if they and their voters are ready. It's not ideal but it is practical; it's *phronetic*. Besides, small scale economies can develop successfully too, as argued in section 2.2.3.

A bottom-up solution could find a place within the context of the *Industrial Strategy* White Paper, which envisages working in some cases with smaller geographical units

than the city-regions dominating the debate so far: Grimsby and Harlow are given as examples (BEIS, 2017: 225-226). The UK government reinforced this message in 2018 when the minister responsible, announcing what he dubbed *Northern Powerhouse 2.0,* said the project had been too focused on Manchester and other big cities and would be extended to smaller cities and towns, and spread to rural areas which, moreover, would not have to have a mayor (Bounds, 2018). A £394m growth deal for the Borderlands Partnership, covering mainly rural areas in England and Scotland, was signed in 2019 (Borderlands Partnership, 2019).

Whatever model is chosen, councillors who want the added powers and funding that go with devolution will have to show understanding of why mayors matter so much to ministers, and respond accordingly. This study has explored the reasons the NECA devolution deal broke down from the perspective, mainly, of the local authorities and other North East actors. But the government has its reasons for caution too. As McCann (2016: 501) puts it: 'Resistance to devolution from the centre arises from the fact that central government lacks trust in sub-national government competence or accountability'. Councillors need to show government why they can be trusted with large sums of public money. A report published in 2015, just as the North East was embarking on its ill-fated devolution journey, found that demonstrable trust and unity of purpose were key in giving the government confidence to devolve:

It is up to local government and related partners to continue to show that they are prepared to take tough decisions regarding scarce resources. And that when they do take those decisions, they are informed and driven by economic necessity and the best long-term outcomes for an area, not temporary local political accommodation...The government, understandably enough, does not want to see deals which are reluctantly patched together, but robust proposals which will not crack under the first sign of local tension (Grant Thornton and Localis, 2015: 1, 2, 32).

### 8.3 Recommendations

This thesis now recommends institutional reforms to make regional/local economic governance more accountable, inclusive, legitimate and transparent, as well as to make it more responsive to local needs and capable of promoting collaboration in

support of a common purpose. These are prerequisites for regions/localities with devolutionary ambitions. The recommendations are designed to achieve these objectives without foisting new structures or new governance geographies on places that have not asked for them. They would facilitate limited but useful change through a process of layering, as described by Martin (2010) (2.1.3), without radically altering structures or redrawing geographical boundaries.

Five of the recommendations are intended for application to all LEPs, one to those like the North East LEP which operate more or less independently of local political control (independent LEPs), one to CAs and two to local authorities.

- Membership of all LEPs should be reformed to give seats to the voluntary sector, as is already starting to happen, and the trade unions. This would bring together all five civic leadership groups and provide a forum for the development of a sense of common purpose;
- Consideration should be given to increasing the diversity of the private sector presence on all LEP boards to include more representatives of large, low-pay sectors such as retail, hospitality and social care;
- 3. On independent LEPs with CA representation, the business sector should retain the chair and remain the largest single group. However, other LEP board members combined should outnumber the business sector by one. This would enable council leaders, the democratically elected element, to prevail in disputes if they can persuade all others to support them. It would incentivize councillors to participate fully and collaboratively;
- 4. All LEPs should meet in public for that part of their business not subject to statutory confidentiality. This would encourage media attendance and thus public engagement. LEP involvement in economic strategy, policy debates and dealing with economic shocks would receive a higher profile. Contacts between the LEPs and local media would be strengthened, facilitating reporting at the critical realist level of the real rather than just the actual, 'back stage' as well as 'front stage' (Ayres *et al.*, 2017b). This is not always a comfortable process, but an essential part of local democracy;
- 5. In all cases, LEP responsibilities for inclusion and employability, with their budgets, should be transferred to the local authorities, which are already

involved in much of the work. With their networks of councillors, they are in closer touch with areas of need and the voluntary organisations that are also often involved. This would also give the local authorities a direct stake in the system;

- 6. In all cases, local authority economic development departments should move in the opposite direction, centralising the economic function in the LEPs and avoiding the type of fragmentation of which the private sector complains (6.2.1). Recommendations 5&6 together would help to realise the hope of combining efficiency with equality, or justice: not exactly the equality between regions sought by Martin (2008) but greater equality for deprived communities within regions. It would be an approach that would see the LEPs focusing on economic efficiency while the local authorities work on social justice;
- Local authorities should have the option, if they wish, to revert to the local government model preceding the reform of 2000, with government by executive committee rather than restricting backbench councillors to scrutiny. This would restore, where desired, the collectivist approach identified by Wollmann (2008) (2.2.4) as traditional to the English system;
- 8. The overview and scrutiny committees of CAs should include two non-voting, external members from the voluntary and trade union sectors, mirroring these sectors' representation on LEP boards. This would strengthen these committees' independence, further embed these two civic leadership groups in economic governance, and act as a spur to backbench councillors;
- 9. Party groups on local authorities should publish their agendas, reports and advice they receive, and minutes, including any voting figures. Groups cannot in practice, and arguably should not, be prevented from meeting privately, but if they are in effect determining or potentially determining council policy, the process should be as transparent as possible.

These recommendations would make the institutions concerned fit, or at least fitter, to implement the general lessons of this thesis and fulfil their ambition for their areas. Those lessons are, to sum up in a single sentence, that whatever governance model emerges for English devolution, if significant funding is ever to be placed under regional/local democratic control, it must be sought in each region or locality by all

civic leadership groups, acting together in pursuit of a common purpose, with public support, building from the bottom, showing willingness to overcome socio-cultural and political obstacles, and instilling in central government the confidence that they can be trusted to deliver.

### 8.4 Further research

There are 38 LEPs in England and devolution deals have been signed covering 11 areas (Sandford, 2019c). Every case study would add valuable insights into how local economic governance can best work, less with a view to developing a blueprint applicable everywhere, which would be contrary to the *phronetic* approach of this thesis, but rather to see if any general lessons can be learned and to identify pitfalls to be avoided in individual cases.

Most SEPs will complete their 10-year lifespans in 2024, and there will be not just an opportunity but a pressing need for research into what has worked well and what has not. Meanwhile, local industrial strategies are coming on stream and will equally require constant monitoring.

There are philosophical issues too that merit discussion, especially around the related issues of place-based and spatially blind development and the trade-off, if there is one, between growth and social justice (2.1.1). Perhaps the measure of justice should be based on some form of community wellbeing, not simply economic metrics. Why devolve at all – or at least, why go beyond the present limited model - when there is little obvious demand from the public, as opposed to local political elites, and limited discernible evidence of an economic dividend? (2.1.1) For while it may make sense to place decisions over the relatively modest amounts available to LEPs in the hands of local leaders who know their areas, shouldn't control over sums like £250bn (1.2, 2.1.1) be in the hands of the government, acting in the national interest?

These questions demand answers to which there are no 'right' or 'wrong' answers. They demand practicality, rationality and ethical judgement: they are *phronetic*'.

# Appendix A: List of events attended in the course

## of research

Date	Venue	Organiser	Event
6/9/2016	Sunderland	North East Combined Authority	Leadership Board
10/3/2017	Newcastle	North East Local Enterprise Partnership	North East Strategic Economic Plan: Refresh
31/3/2017	Newcastle	Institute for Social Renewal, Newcastle University	Thriving in Turbulent Times
16/6/2017	Sunderland	Campaign for Real Devolution	North East Devolution Conference
21/6/2017	Newcastle	North East Local Enterprise Partnership and Ward Hadaway	Brexit Conference
11/7/2017	North Shields	Carnegie UK Trust and Institute for Social Renewal	Turning Around Towns in the North East
2/11/2017	Newcastle	Faculty of Humanities and Social Sciences, Newcastle University	Social Renewal and Social Justice in the North East region
7/11/2017	Newcastle	Tyne and Wear Citizens	Launch Assembly
22/1/2018	South Shields	North of Tyne councils	Combined Authority public consultation
31/1/2018	Gateshead	North of Tyne councils	Combined Authority public consultation
1/2/2018	Sunderland	North of Tyne councils	Combined Authority public consultation
6/3/2018	Gateshead	North East Local Enterprise Partnership	State of the Region
17/4/2018	Newcastle	The Journal	Transport breakfast
6/7/2018	Newcastle	Great Exhibition of the North	Northern Powerhouse Great Futures
6/9/2018	Gateshead	Newcastle City Council	Convention of the North
7/9/2018	Gateshead	Baltic Centre for Contemporary Art	ldea of North
2/10/2018	Newcastle	Café Philosophique	Powerhouse or Underdog? Is Northern Devolution running out of Steam?
14/11/2018	Newcastle	North East Collaboration Working Group	North East Collaboration Conference
16/11/2018	Gateshead	Sovereign Strategy	North East Economic Forum
27-29/11/2018	Lyon/St Etienne	Universities Jean Monnet/Sciences Po	Urban Narratives
17/1/2019	Stockton	Tees Valley Combined Authority	Overview and Scrutiny Committee
24/1/2019	Middlesbrough	Tees Valley Combined Authority	Cabinet
30/1/2019	Sunderland	Sunderland City Council	Council

14/2/2019	Newcastle	North East Local Enterprise Partnership	Strategic Economic Plan: Update
30/4/2019	Newcastle	Tyne and Wear Citizens	Mayoral Assembly
10/5/2019	Newcastle	North East Local Enterprise Partnership	Our Economy 2019
18/6/2019	Newcastle	IPPR North	Economic Justice for the North East
13/9/2019	Rotherham	Leeds and Newcastle Councils	Convention of the North and NP11
26/9/2019	Newcastle	North East Local Enterprise Partnership	Annual General Meeting
17/10/2019	Stockton	Tees Valley Combined Authority	Overview and Scrutiny Committee

# Appendix B: Coding

NVivo code headings and the number of sources and references for each:

Code	Sources	References
Accountability	16	32
Attitudes – miscellaneous	15	25
Attitudes – political	23	63
Attitudes to government	15	37
Attitudes to money	24	62
Brexit	9	18
CARD campaign	14	27
Culture	7	18
Democracy	25	83
Devolution – general	20	60
Durham	13	75
Economic theory	18	42
Education and training	16	27
Entrepreneurship	19	51
Fair shares	15	35
Gateshead	13	35
Geography	22	48
Getting by	2	2
Hartlepool	1	5
Health	1	4
Inclusiveness	11	36
Labour Party	9	15
Leadership	11	21
LEPs	31	76
Local identity	15	17
Mayors	35	133
MPs	8	10
NECA	29	106

Newcastle	27	58
North of Tyne	26	53
North Tyneside	5	13
Northern Powerhouse	6	7
Northumberland	6	13
Parochialism	4	8
Personal relationships – North East	20	45
Personal relationships – Tees Valley	12	34
Political opinions	3	4
Populism	6	18
Pragmatism	13	30
Prospects	11	23
Public opinion	10	24
RDAs	22	45
Regional assembly	7	13
Regional identity	11	26
Reunification	23	35
Risk	4	5
Rivalry	21	47
Role of officials	3	3
Rural	3	8
South of Tyne	6	9
South Tyneside	7	8
Sunderland	14	34
Tees Valley devolution	25	111
Trade unions	11	26
Transport	18	42
Trust	14	22
Values	4	5

### References

Adonis, A. (2013) North East Independent Economic Review. Newcastle: North East Local Enterprise Parrtnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-</u> <u>content/uploads/2014/11/North-East-Independent-Economic-Review-April-2013.pdf</u>

(Accessed: 14 February 2020).

Allen, G. (2002) *Regional Development Agencies* (02/50). London: House of Commons Library. [Online]. Available at:

http://researchbriefings.parliament.uk/ResearchBriefing/Summary/RP02-50 (Accessed: 29 November 2016).

Amin, A. and Thrift, N.J. (1994) *Globalization, Institutions, and Regional Development in Europe*. Oxford ; New York: Oxford University Press.

Angula-Guerrero, M.J., Perez-Moreno, S. and Abad-Guerrero, I.M. (2017) 'Disparaties in Entrepreneurial Activity and Attitude across EU Countries', *Journal of European Planning Studies*, 25(4), pp. 680-702 [Online] DOI: 10.1080/09654313.2017.1287163 Archer, M., Collier, A. and Porpora, D.V. (2004) *Transcendence*. London: Routledge. Aristotle (c350BC (1953)) *Ethics*. Translated by Thomson, J.A.K. Harmondsworth, Middlesex: Penguin.

Ayres, S., Flinders, M. and Sandford, M. (2017a) 'Territory, Power and Statecraft: Understanding English devolution', *Regional Studies*, 52(6), pp. 853-864 [Online] DOI: 10.1080/00343404.2017.1360486

Ayres, S., Sandford, M. and Coombes, T. (2017b) 'Policy making 'front' and 'back' stage: Assessing the implications for effectiveness and democracy', *The British Journal of Politics and International Relations*, 19(4), pp. 861-876 [Online] DOI:

10.1177/1369148117721842

Bailey, D. (2018) 'The Recomposition of the Tax System: Exacerbating uneven development through the Northern Powerhouse agenda', in Berry, C. and Giovannini, A. (eds.) *Developing England's North: the political economy of the Northern Powerhouse*. Palgrave Macmillan. Available at:

https://link.springer.com/content/pdf/10.1007%2F978-3-319-62560-7.pdf (Accessed: 2018).

Bailey, D., Bellandi, M., Caloffi, A. and De Propis, L. (2010) 'Place-renewing Leadership: Trajectories of change for manufacturing regions in Europe', *Policy Studies*, 31(4), pp. 457-474 [Online] DOI: 10.1080/01442871003723408.

Barker, M. (2016) North East Combined Authority: Proposed devolution agreement and elected regional mayor. Gateshead: Gateshead Council. [Online]. Available at:

http://democracy.gateshead.gov.uk/documents/s2197/Item%201 (Accessed: 22 February 2017).

BBC (2004) North East votes 'no' to assembly [TV news webpage]. Available at: <u>http://news.bbc.co.uk/1/hi/uk\_politics/3984387.stm</u> (Accessed: 22 December 2016). BBC (2012) Nine cities reject elected mayors. Available at:

http://www.bbc.co.uk/news/uk-politics-17949950 (Accessed: 19 June 2017).

BBC (2017) Local Elections 2017: The results mapped. Available at:

http://www.bbc.co.uk/news/election-2017-39795422 (Accessed: 23 May 2017). BBC (2018) Local election results 2018: The results in maps and charts. Available at:

http://www.bbc.co.uk/news/uk-politics-43992681 (Accessed: 11 May 2018).

BBC (2019a) *Election Results: Labour wins North of Tyne mayor but loses votes eleswhere*. Available at: <u>https://www.bbc.co.uk/news/uk-england-tyne-48119196</u> (Accessed: 4 May 2019).

BBC (2019b) *Durham Tees Valley Airport becomes Teesside International*. Available at: <u>https://www.bbc.co.uk/news/uk-england-tees-49108623</u> (Accessed: 29 July 2019). BBC (2019d) *Results: Conservatives make gains in Labour heartlands*. Available at: <u>https://www.bbc.co.uk/news/election/2019/results</u> (Accessed: 13 December 2019). Beecham, J. (2018) *Newcastle upon Tyne, North Tyneside and Northumberland Combined Authority (Establishment and Functions) ) Order 2018*. London: House of Lords Hansard. [Online]. Available at: <u>https://hansard.parliament.uk/Lords/2018-10-</u> 30/debates/2531A764-E9F9-4D93-B63D-

4EB11B438049/NewcastleUponTyneNorthTynesideAndNorthumberlandCombinedAut hority(EstablishmentAndFunctions)Order2018?highlight=gateshead#contribution-122F1DFD-4634-4B86-9285-2ABB2E80B265 (Accessed: 1 November 2018).

Beel, D., Jones, M., Jones, I.R. and Escadale, W. (2017) 'Connected Growth: Developing a framework to drive inclusive growth across city regions', *Local Economy* 32(6), pp. 565-575 [Online] DOI: 10.1177/0269094217727236

Beel, D., Jones, M. and Rees Jones, I. (2018) 'Regionalisation and Civil Society', in Berry, C. and Giovannini, A. (eds.) *Developing England's North; the political economy of the Northern Powerhouse*. Palgrave Macmillan. Available at:

https://link.springer.com/content/pdf/10.1007%2F978-3-319-62560-7.pdf (Accessed: 2018).

BEIS (2017) Industrial Strategy: Building a Britain Fit for the Future. London: Department for Business, Energy and Industrial Strategy. [Online]. Available at: <u>https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/662</u> <u>508/industrial-strategy-white-paper.pdf</u> (Accessed: 26 June 2019).

BEIS Select Committee (2019) *Industrial Strategy: Sector Deals*. Available at: <u>https://publications.parliament.uk/pa/cm201719/cmselect/cmbeis/663/66303.htm# i</u> dTextAnchor000 (Accessed: 19 March 2019).

Bell, R. (2018) 'Teesside set for Jobs Boost as Firms plan £250m Waste-to-energy Plant', *Bdaily North East*edn), 23 July 2018. [Online] Available at:

https://bdaily.co.uk/articles/2018/07/23/teesside-set-for-jobs-boost-as-firms-plot-250m-waste-to-energy-

<u>plant?utm\_source=bulletin&utm\_medium=email&utm\_campaign=2018-07-24-north-east&utm\_content=readmore</u> (Accessed: 25 July 2018).

Bell, R. (2019) 'Govt Deal to help create 500 Jobs and new Industrial Quarter on former Redcar Steel Works', *Bdaily North East*edn), 7 January 2019. [Online] Available at: <u>https://bdaily.co.uk/articles/2019/01/07/govt-deal-to-help-create-500-jobs-and-new-</u> industrial-quarter-at-former-redcar-

steelworks?utm source=bulletin&utm medium=email&utm campaign=2019-01-08north-east&utm content=readmore (Accessed: 8 January 2019).

Berry, C. and Giovannini, A. (2018) 'Introduction: Powerhouse Politics and Economic Development in the North ', in Berry, C. and Giovannini, A. (eds.) *Developing England's North: The Political Economy of the Northern Powerhouse*. Palgrave Macmillan. Available at: <u>https://link.springer.com/content/pdf/10.1007%2F978-3-319-62560-</u>

7.pdf (Accessed: 4 February 2019).

Bevir, M. (2011) 'Government and Governmentality after Neoliberalism', 39(4), pp. 457-471 [Online] DOI: 10.1332/030557310X550141.

Binding, C. (2019) "Battle of Orgreave': Fresh call for probe in 1984 miners' clash', *ChronicleLive*edn), 24 January 2019. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/battle-orgreave-fresh-callsprobe-15727589 (Accessed: 24 January 2019). BIS (2010) Local Growth: Realising Every Place's Potential (Cmnd 7691). London: Department for Business, Innovation and Skills. [Online]. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/320 76/cm7961-local-growth-white-paper.pdf (Accessed: 12 July 2017).

BIS (2013b) European Regional Development Fund and European Social Fund: Allocations 2014-2020. London: Department for Business, Innovation and Skills. [Online]. Available at: <u>https://www.gov.uk/government/speeches/european-regional-development-fund-and-european-social-fund-allocations-2014-to-2020</u> (Accessed: 9 May 2017).

Blackman, T. and Ormston, C. (2005) 'Discourses of Accountability: Policy scrutiny of an English Regional Development Agency', *Regional Studies*, 39(3), pp. 375-386 [Online] DOI:

https://rsa.tandfonline.com/doi/pdf/10.1080/00343400500087448?needAccess=true. Board of Trade (1963) *The North East: A programme for regional development and growth*. London: Board of Trade.

Bolton, T. and Hildreth, P. (2013) *Mid-sized Cities: Their role in England's economy*. London: Centre for Cities. [Online]. Available at: <u>http://www.centreforcities.org/wp-content/uploads/2014/08/13-06-18-Mid-Sized-Cities.pdf</u> (Accessed: 19 February 2017).

Borderlands Partnership (2018) *Borderlands Inclusive Growth Deal: An ambitious apporach to cross border working* Available at: <u>http://www.borderlandsgrowth.com/</u> (Accessed: 20 February 2019).

Borderlands Partnership (2019) *Transformational £394.5m Borderlands Growth Deal signed off.* Carlisle: Borderlands Partnership,. [Online]. Available at:

http://www.borderlandsgrowth.com/Portals/0/Documents/News/Borderlands%20Pre ss%20Release%20-%201.7.19.pdf (Accessed: 2 July 2019).

Borraz, O. and John, P. (2004) 'The Transformation of Urban Political Leadership in Western Europe', *International Journal of Urban and Regional Research*, 28(1), pp. 107-120 [Online] DOI: 10.1111/j.0309-1317.2004.00505.x/epdf

Boschma, R.A. and Frenken, K. (2006) 'Why is Economic Geography not an Evolutionary Science? Towards an evolutionary economic geography', *Journal of Economic Geography*, 6(3), pp. 273-302 [Online] DOI: 10.1093/jeg/lbi022.

Bounds, A. (2017) 'UK's Northern Cities lobby Brexit Minister over EU Funding', *Financial Times*edn), 5 October 2017. [Online] Available at:

https://www.ft.com/content/77524126-a9dc-11e7-ab55-

27219df83c97?emailId=59d5efb18df16b00042150f8&segmentId=488e9a50-190e-700c-cc1c-6a339da99cab (Accessed: 6 October 2017).

Bounds, A. (2018) 'Northern Powerhouse 'too focused on Manchester'', *Financial Times Group*edn), 21 |March 2018. [Online] Available at:

https://www.ft.com/content/38d29d08-2c3e-11e8-a34a-

7e7563b0b0f4?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

2437841d178a#myft:notification:instant-email:content (Accessed: 22 March 2018).

Bounds, A. (2019a) 'Northern leaders angered by ministers' stance on rail promises', *Financial Times*edn), 3 October 2019. [Online] Available at:

https://www.ft.com/content/8c46f878-e46e-11e9-9743-

db5a370481bc?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

<u>2437841d178a#myft:notification:instant-email:content</u> (Accessed: 3 October 2019). Bounds, A. (2019b) 'Greater Manchester needs more Devolved Powers says Study',

*Financial Times*edn), 8 February 2019. [Online] Available at:

https://www.ft.com/content/afe87034-2a44-11e9-a5ab-

ff8ef2b976c7?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

<u>2437841d178a#myft:notification:instant-email:content</u> (Accessed: 8 February 2019). Brown, M. (2018a) 'Sirius Minerals: full 'transformational' impact of £3.2bn mining project on Teesside is revealed', *GazetteLive*edn), 29 May 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/sirius-minerals-full-</u> <u>transformational-impact-</u>

<u>14716520?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content=</u> <u>EM\_GazetteLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story2&utm\_campaign=daily\_newsletter</u> (Accessed: 29 May 2018).

Brown, M. (2018b) "£40m Plan' to buy Airport described as a Vanity Project ahead of Mayor's announcement', *TeessideLive*edn), 3 December 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/labour-leaders-claim-mayor-buy-15498932</u> (Accessed: 3 December 2018).

Brown, M. (2018c) 'Ben Houchen will stop houses at airport and says £35m deal 'won't cost taxpayers a penny'.', *TeessideLive*edn), 4 December 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/ben-houchen-stop-houses-airport-</u>

<u>15502363?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content=</u> <u>EM\_GazetteLive\_Nletter\_News\_Home\_largeteaser\_Text\_Story1&utm\_campaign=daily\_newsletter</u> (Accessed: 4 December 2018).

Brown, M. (2018d) 'Overwhelming Backing for Ben Houchen's Plan to buy back Airport', *TeessideLive*edn), 8 December 2018. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/overwhelming-backing-benhouchens-40m-15521250 (Accessed: 17 December 2018).

Brown, M. (2019a) 'Mayor accuses Cleveland Police of 'bullying and harassing' critics after being cleared of wrongdoing', *TeessideLive*edn), 6 February 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/mayor-accuses-cleveland-police-bullying-15790007</u> (Accessed: 6 February 2019).

Brown, M. (2019b) 'President Donald Trump invited to Teesside as Mayor talks up 'special relationship'', *Teesside Live*edn), 1 May 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/president-donald-trump-invited-</u>

teesside-16207533 (Accessed: 1 May 2019).

Brown, N. (2015) *Nick Responds to North East Devolution Deal*. Available at: <u>http://www.nickbrownmp.com/</u> (Accessed: 12 January 2018).

Burcher, S. and Mayer, H. (2017) 'Are there Differences in Social Capital related to Corporate Regional Engagement in Dynamic and less Dynamic Non-core Regions?', *European Planning Studies*, 10.1080/09654313.2017.1361604 [Online]. Available at:

http://www.tandfonline.com/doi/pdf/10.1080/09654313.2017.1361604?needAccess= true DOI: 10.1080/09654313.2017.1361604 (Accessed: 17 December 2017).

Cabinet Office, Deputy Prime Minister's Office and Clark, G. (2014) *Tees Valley Growth Deal*. London: Cabinet Office. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment\_data/file/398878/34\_Tees\_Valley\_Growth\_Deal.pdf (Accessed: 8 April 2019).

Cain, J. (2018a) 'Tees Mayor Ben Houchen would welcome North Yorkshire 'With Open Arms'', *GazetteLive*edn), 22 April 2018. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/tees-mayor-ben-houchen-would-14558083 (Accessed: 24 April 2018). Cain, J. (2018b) 'Durham Tees Valley Airport: Recap updates as council leaders veto Tees mayor's budget', *TeessideLive*edn), 16 February 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/durham-tees-valley-airport-live-14297629</u> (Accessed: 3 September 2019).

Cain, J. (2019a) 'Redcar and Cleveland leader urged to back airport deal ahead of statement', *TeessideLive*edn), 22 January 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/redcar-cleveland-labour-leaderurged-

<u>15712868?utm source=gazette live newsletter&utm medium=email&utm content=</u> <u>EM GazetteLive Nletter News Home smallteaser Text Story6&utm campaign=daily</u> newsletter (Accessed: 22 January 2019).

Cameron, D. (2014) *Scottish Independence Referendum: Statement by the Prime Minister*. London: HM Government. [Online]. Available at:

https://www.gov.uk/government/news/scottish-independence-referendumstatement-by-the-prime-minister (Accessed: 17 May 2017).

CBI (2019) *Powering up Places: Unlocking regional growth through devolution*. London: Confederation of British Industry. [Online]. Available at:

https://www.cbi.org.uk/media/3578/cbi-report-powering-up-places.pdf (Accessed: 8 October 2019).

Centre for Cities (2019a) *Cities Outlook 2019: A decade of austerity*. London: Centre for Cities. [Online]. Available at: <u>https://www.centreforcities.org/reader/cities-outlook-</u>2019/a-decade-of-austerity/ (Accessed: 28 January 2019).

Centre for Cities (2019b) *Primary Urban Areas: Spatial definitions*. London: Centre for Cities. [Online]. Available at: <u>https://www.centreforcities.org/wp-</u>

<u>content/uploads/2014/07/12-03-19-Primary-Urban-Areas-deffinitions.pdf</u> (Accessed: 12 April 2019).

Chronicle (2013) "Ellington's Closure was Over-reaction", *ChronicleLive*edn), 27 February 2013. [Online] Available at: <u>http://www.chroniclelive.co.uk/news/north-east-news/ellingtons-closure-was-over-reaction-1530880</u> (Accessed: 11 Noember 2017).

Clark, G.L. (1998) 'Stylized Facts and Close Dialogue: Methodology in economic geography', *Annals of the Association of American Geographers*, 88(1), pp. 73-87 [Online] DOI: 10.1111/1467-8306.00085.

Clarke, N. and Cochrane, A. (2013) 'Geographies and Politics of Localism: The localism of the United Kingdom's coalition government', *Political Geography*, 34, pp. 10-23 [Online] DOI: 10.1016/j.polgeo.2013.03.003.

Cochrane, A. (1998) 'Illusions of power: interviewing local elites', *Environment and Planning A*, 30, pp. 2121-2132.

Cole, L. (2018) *North East Collaboration Conference: Inspire, Explore and Act for Social Change*. Newcastle, 14 November 2018. Newcastle: North East Collaboration Working Group. Available at: <u>https://www.vonne.org.uk/events/north-east-collaboration-</u>

<u>conference-inspire-explore-and-act-social-change</u> (Accessed: 14 November 2018). Colomb, C. and Tomaney, J. (2016) 'Territorial Politics, Devolution and Spatial Planning in the UK: Results, prospects, lessons', *Planning Practice and Research*, 31(1), pp. 1-22 [Online] DOI: 10.1080/02697459.2015.1081337.

Commission for Economic Policy (2019) *Implementing a Place-based Approach to EU Industrial Policy Strategy*. Brussels: Regions, E.C.o.t. [Online]. Available at: <u>https://cor.europa.eu/en/engage/studies/Documents/CoR\_Industry.pdf</u> (Accessed: 31 July 2019). Copus, C. (2004) *Party Politics and Local Government*. Manchester: Manchester University Press.

Copus, C. and Erlingsson, G. (2012) 'Politics in Local Government: A review', *Representation*, 48(2), pp. 235-247 [Online] DOI: 10.1080/00344893.2012.683489. Copus, C., Roberts, M. and Wall, R. (2017) *Local Government in England: Centralisation, autonomy and control*. Palgrave Macmillan. Available at:

https://link.springer.com/content/pdf/10.1057%2F978-1-137-26418-3.pdf (Accessed: 8 March 2019).

Core Cities UK (2017) *A Unique and Influential Voice for Cities*. Available at: <u>https://www.corecities.com/</u> (Accessed: 19 February 2017).

Coyle, D., Flanders, S., Glaeser, E.L., Mazzucato, M., Overman, H. and Singh, D. (2019) *Greater Manchester Independent Prosperity Review*. Manchester: Greater Manchester Combined Authority. [Online]. Available at:

https://issuu.com/greatermcr/docs/gmipr\_reviewersreport\_web\_20190208?e=35861 904/67626379 (Accessed: 8 February 2019).

DCLG (2010) *Pickles announces Plans to abolish Regional Government*. London: Department for Communities and Local Government. [Online]. Available at:

https://www.gov.uk/government/news/pickles-outlines-plans-to-abolish-regionalgovernment (Accessed: 12 July 2017).

DCLG (2014) The Durham, Gateshead, Newcastle upon Tyne, North Tyneside, South Tyneside and Sunderland Combined Authority: Summary of responses to consultation on proposal to establish a combined authority. London: Department for Communities and Local Government [Online]. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/286 409/140228 Final\_summary\_of\_responses.pdf (Accessed: 20 February 2017).

DCLG (2015a) *Proposal to Establish a Combined Authority for Tees Valley: Consultation*. London: Department for Communities and Local Government. [Online]. Available at: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach</u> <u>ment\_data/file/472044/Tees\_Valley\_Consultation.pdf</u> (Accessed: 27 March 2019).

DCLG (2015b) English Indices of Deprivation 2015. London: Department for Communities and Local Government. [Online]. Available at:

https://www.gov.uk/government/statistics/english-indices-of-deprivation-2015 (Accessed: 15 July 2017).

DCLG (2016) Tees Valley Combined Authority: Summary of responses to the government consutation on proposal to establish a combined authority for Tees Valley London: Department for Communities and Local Government. [Online]. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/500 840/160209 - Summary of consultation responses.pdf (Accessed: 18 June 2017). Demographia (undated) England Local Authorities: Population, area and density. Available at: http://www.demographia.com/db-engla.htm (Accessed: 27 August 2017). Department for Education (2018) National Funding Formula Tables for Schools and High Needs: 2019 to 2020. London: Department for Education,. [Online]. Available at: https://www.gov.uk/government/publications/national-funding-formula-tables-forschools-and-high-needs-2019-to-2020 (Accessed: 5 August 2019).

DIT (2016) Foreign inward investment (FDI) by UK region (tax year 2011 to 2012 to tax year 2015 to 2016). London: Department for Internatiional Trade. [Online]. Available at: <u>https://www.gov.uk/government/publications/foreign-direct-investment-projects-by-ukti-regions-201011-to-201415/foreign-direct-investment-projects-by-uk-region-201011-to-201415</u> (Accessed: 11 September 2019).

DIT (2017) *Inward Investment Results 2016-17*. London: Department for Internatiional Trade. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment\_data/file/625682/DIT\_Investment\_Results\_2016-17-A5-digital.pdf (Accessed: 11 September 2019).

DIT (2018) *Inward Investment Results 2017-18*. London: Department for International Trade Trade, D.f.I. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment data/file/730041/Department for International Trade inward investment res ults 2017 to 2018.pdf.

DIT (2019) *Inward Investment Results 2018-19*. London: Department for International Trade. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment\_data/file/812252/dit-inward-investment-results-2018-to-2019.pdf (Accessed: 11 September 2019).

DRG (undated) *About DRG*. Available at: <u>https://www.drg.global/about/</u> (Accessed: 13 May 2019).

Duncan, K. (2017) 'Theresa May tells Teessiders to 'look to the future' as 25-year jobs plan is launched', *GazetteLive*edn), 23 August 2017. [Online] Available at:

http://www.gazettelive.co.uk/news/teesside-news/theresa-may-south-teesdevelopment-13520446?ptnr rid=431380&icid=EM GazetteLive Nletter News (Accessed: 24 August 2017).

Durham County Council (2016a) *Durham Cabinet Agenda: 11 May 2016*. Durham: Durham County Council [Online]. Available at:

https://democracy.durham.gov.uk/documents/g7932/Public%20reports%20pack%20 Wednesday%2011-May-2016%2010.00%20Cabinet.pdf?T=10 (Accessed: 16 February 2018).

Durham County Council (2016b) *Durham Cabinet Minutes: 11 May 2016*. Durham: Durham County Council. [Online]. Available at:

https://democracy.durham.gov.uk/documents/g7932/Printed%20minutes%20Wednes day%2011-May-2016%2010.00%20Cabinet.pdf?T=1 (Accessed: 16 February 2016). Durham County Council (undated) *Durham Insight*. Available at:

https://www.durhaminsight.info/population/ (Accessed: 28 October 2019).

Education Policy Institute (2019) *Analysis: 'Levelling up' - What it really means for school funding*. London: Education Policy Institute. [Online]. Available at:

https://epi.org.uk/publications-and-research/analysis-the-prime-ministers-promise-tolevel-up-school-funding/ (Accessed: 5 August 2019).

Elcock, H. (2016) 'Regional Identity in the North East: Death and rebirth', in Hayton, R., Giovannini, A. and Berry, C. (eds.) *The Politics of the North: Governance, territory and identity in Northern England*. White Rose University Consortium. Available at: <a href="http://speri.dept.shef.ac.uk/wp-content/uploads/2019/01/Politics-of-the-North-Hayton-Giovannini-Berry.pdf">http://speri.dept.shef.ac.uk/wp-content/uploads/2019/01/Politics-of-the-North-Hayton-Giovannini-Berry.pdf</a>.

Elder-Vass, D. (2010) *The Causal Power of Social Structures: Emergence, structure and agency*. Cambridge University Press. Available at:

https://www.cambridge.org/core/books/causal-power-of-social-

structures/DC90FA9DAB2FDCFFF9E3D8E98A7D9585 (Accessed: 3 April 2019).

Electoral Commission (2017a) *Voting in 2017: Understanding public attitudes towards elections and voting*. London: Electoral Commission. [Online]. Available at:

https://www.electoralcommission.org.uk/ data/assets/pdf\_file/0011/234893/Voting -in-2017-Final.pdf (Accessed: 11 January 2018).

Electoral Commission (2017b) *Combined Authority Mayor (CAM) elections in England May 2017*. London: Electoral Commission. [Online]. Available at:

https://www.electoralcommission.org.uk/ data/assets/pdf file/0011/237980/May-2017-CAM-electoral-data-report.pdf (Accessed: 11 January 2018).

Electoral Commission (2017c) *Local Elections in England, May 2017*. London: Electoral Commission. [Online]. Available at:

https://www.electoralcommission.org.uk/ data/assets/pdf\_file/0010/237979/May-2017-England-locals-electoral-data-report.pdf (Accessed: 11 January 2017).

Etikan, I. (2016) 'Comparison of Convenience Sampling and Purposive Sampling', *American Journal of Theoretical and Applied Statistics*, 5(1), pp. 1-4 [Online] DOI: 10.11648/j.ajtas.20160501.11.

European Commission (2018) *European Quality of Governance Index*. Brussels: European Commission. [Online]. Available at:

https://ec.europa.eu/regional\_policy/en/newsroom/news/2018/02/27-02-2018european-guality-of-government-index-2017 (Accessed: 16 May 2019).

European Parliament and European Council (2003) *Regulation (EC) No 1059/2003 of the European Parliament and of the Council*. Brussels: European Parliament, European Commission,. [Online]. Available at: <u>https://eur-lex.europa.eu/legal-</u>

content/EN/TXT/PDF/?uri=CELEX:02003R1059-20180118&from=EN (Accessed: 12 April
2019).

Eurostat (2019) *Population at 1 January [2019] by age group, sex and NUTS3 region*. Brussels: Eurostat. [Online]. Available at:

<u>http://appsso.eurostat.ec.europa.eu/nui/show.do?dataset=demo\_r\_pjangrp3&lang=e</u> <u>n</u> (Accessed: 12 April 2019).

EY and Centre for Towns (2018) *Bridging the Gap*. London: EY. [Online]. Available at: <u>https://www.ey.com/Publication/vwLUAssets/ey-ukas-towns-report/\$File/ey-ukas-towns-report.pdf</u> (Accessed: 29 November 2018).

Filipetti, A. and Sacchi, A. (2016) 'Decentralisation and Economic Growth reconsidered: The role of regional authority', *Environment and Planning C: Government and Policy*, 34(8), pp. 1793-1824 [Online] DOI: 10.1177/0263774X16642230.

Flyvbjerg, B. (2001) *Making Social Science Matter*. Translated by Sampson, S. Cambridge: Cambridge University Press.

Flyvbjerg, B. (2006) 'Five Misunderstandings about Case-study Research', *Qualitative Inquiry*, 12(2), pp. 219-245 [Online] DOI: 10.1177/1077800405284363.

Forbes, N. and Lewis, A. (2016) *Devolution to the North East*. Newcastle: Newcastle City Council. [Online]. Available at:

https://democracy.newcastle.gov.uk/documents/g6011/Public%20reports%20pack%2 023rd-Mar-2016%2017.30%20Cabinet.pdf?T=10 (Accessed: 31 March 2018).

Fraser, C.M. (ed.) (1966) *The Publications of the Surtees Society*. Durham and London: Andrews and Co. and Bernard Quaritch.

Fujita, M. and Thisse, J.-F. (2009) 'New Economic Geography: An appraisal on the occasion of Paul Krugman's 2008 Nobel Prize in economic sciences', *Regional Sciences and Urban Economics*, 39(2), pp. 109-119 [Online] DOI:

10.1016/j.regsciurbeco.2008.11.003.

Gamble, A. (1990) 'Theories of British Politics', *Political Studies*, 38(3), pp. 404-420 [Online] DOI: 10.1111/j.1467-9248.1990.tb01078.x/pdf. Garcia-Rodriguez, F.J., Gil-Soto, E., Ruiz-Rosa, I. and Guttierez-Tano, D. (2016) 'Entrepreneurial Process in Peripheral Regions: The role of motivation and culture ', *Journal of European Planning Studies*, 25(11), pp. 2037-2056 [Online] DOI: 10.1080/09654313.2016.1262827.

Gateshead Council (2013) *Gateshead Cabinet Agenda: 17 December 2013*. Gateshead: Gateshead Council. [Online]. Available at:

http://democracy.gateshead.gov.uk/Data/Cabinet/20131217/Agenda/Cabinet%20-%2020131217%20-%20Complete%20agenda.pdf (Accessed: 14 June 2017).

Gateshead Council (2016a) *Gateshead Cabinet Agenda and Draft Minutes: 6* September 2016. Gateshead: Gateshead Council,. [Online]. Available at:

http://democracy.gateshead.gov.uk/ieListDocuments.aspx?Cld=138&Mld=1275&Ver= <u>4</u> (Accessed: 28 March 2019).

Gateshead Council (2016b) *Gateshead Cabinet Agenda: 22 March 2016*. Gateshead: Gateshead Council [Online]. Available at:

http://democracy.gateshead.gov.uk/documents/g1146/Public%20reports%20pack%20 22nd-Mar-2016%2010.00%20Cabinet.pdf?T=10 (Accessed: 16 February 2018).

Gateshead Council (2016c) *Gateshead Cabinet Minutes: 22 March 2016*. Gateshead: Gateshead Council. [Online]. Available at:

http://democracy.gateshead.gov.uk/documents/g1146/Printed%20minutes%2022nd-Mar-2016%2010.00%20Cabinet.pdf?T=1 (Accessed: 31 March 2019).

Gateshead Council (2016d) *Gateshead Cabinet Agenda: 6 September 2016*. Gateshead: Gateshead Council. [Online]. Available at:

http://democracy.gateshead.gov.uk/documents/g1275/Public%20reports%20pack%20 06th-Sep-2016%2009.00%20Cabinet.pdf?T=10 (Accessed: 12 July 2017).

Gertler, M.S. (2003) 'A Cultural Economic Geography of Production', in Anderson, K., Domosh, M., Pile, S. and Thrift, N. (eds.) *Handbook of Cultural Geography*. London: Sage Publications, pp. 131-146.

Giles, C., Wright, R., Pickard, J. and Tighe, C. (2019) 'Experts sceptical on Johnson's plans for regional free ports', *Financial Times*edn), 5 July 2019. [Online] Available at: <u>https://www.ft.com/content/f9c41a54-9f1b-11e9-9c06-</u>

a4640c9feebb?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

2437841d178a#myft:notification:instant-email:content (Accessed: 6 July 2019).

Gilette, B. (2004) 'Small Town Success Stories can serve as Models for Others',

Mississippi Business Journaledn), 25 October 2004. [Online] Available at:

http://msbusiness.com/2004/10/small-town-success-stories-can-serve-as-models-forothers/ (Accessed: 20 July 2017).

Gilhespie, J. (2019) *Report of the Chief Executive*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2019/06/8-Constitution-amendments-report.pdf</u> (Accessed: 30 July 2019).

Giovannini, A. (2015) 'Devolution in the North of England: Time to bring the people into the debate', *Democratic Audit UK*. Available at:

http://www.democraticaudit.com/2015/03/30/devolution-in-the-north-of-englandtime-to-bring-the-people-into-the-debate/ (Accessed: 8 November 2017).

Giovannini, A. (2017) *Democracy Matters: Citizens' Assembly Pilots on English Devolution: Assembly North*. Presentation. North East Devolution Conference, Sunderland: 16 June 2017.

Gooby, D.T. (undated) 'David Taylor-Gooby', *Linked In*. Available at: <u>https://www.linkedin.com/in/david-taylor-gooby-16800229/?ppe=1</u> (Accessed: 8 October 2017).

Grabher, G. and Stark, D. (1997) 'Organising Diversity: Evolutionary theory, network analysis and postsocialism', *Regional Studies*, 31(5), pp. 533-544 [Online] DOI: <a href="http://www.tandfonline.com/doi/pdf/10.1080/00343409750132315?needAccess=true">http://www.tandfonline.com/doi/pdf/10.1080/00343409750132315?needAccess=true</a>

Graham, H. (2016b) 'Refusing to accept Devolution 'Is like Pretending the Earth is Flat' says Nick Forbes', *ChronicleLive*edn), 8 March 2016. [Online] Available at: http://www.chroniclelive.co.uk/news/north-east-news/refusing-accept-devolution-

like-pretending-11005372 (Accessed: 12 July 2017).

Graham, H. (2017) 'Devolution one step closer: Newcaste and North Tyneside cabinets give North of Tyne plan go ahead', *ChroncleLive*edn), 11 December 2017. [Online] Available at: <u>http://www.chroniclelive.co.uk/news/north-east-news/devolution-one-step-closer-newcastle-14025203</u> (Accessed: 15 December 2017).

Graham, H. and Eden, T. (2018) 'Theresa May in Newcastle Live: Tory cabinet visits Tyneside for Great Exhibition away day', *ChronicleLive*edn), 23 July 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/theresa-newcastle-live-tory-cabinet-</u>

<u>14938813?utm source=chronicle live newsletter&utm medium=email&utm content</u> <u>=EM ChronicleLive Nletter News Home largeteaser Text Story1&utm campaign=d</u> <u>aily newsletter</u> (Accessed: 23 July 2018).

Grant Thornton and Localis (2015) *Making Devoution Work: A practical guide for local leaders*. London: Localis. [Online]. Available at: <u>http://www.localis.org.uk/wp-content/uploads/2015/10/gt1171-making-devolution-work-final.pdf</u> (Accessed: 16 January 2018).

Gray, M. and Barford, A. (2018) 'The depths of the cuts: the uneven geography of local government austerity', *Cambridge Journal of Regions, Economy and Society*, 11(3), pp. 541-563 [Online] DOI: 10.1093/cjres/rsy019 (Accessed: 29 October 2018).

Gray, N., Pugalis, L. and Dickinson, D. (2018) 'The Northern Powerhouse meets the Cities and Local Growth agenda: Local economic policy making and agglomeration in practice', in Berry, C. and Giovannini, A. (eds.) *Developing England's North: the Political Economy of the Northern Powerhouse*. Palgrave Macmillan. Available at:

https://link.springer.com/content/pdf/10.1007%2F978-3-319-62560-7.pdf (Accessed: 2018).

Grix, J. (2002) 'The Generic Terminology of Social Research', *Politics*, 22(3), pp. 175-186 [Online] DOI: 10.1111/1467-9256.00173 (Accessed: 12 December 2016).

Gustafsson, M. (2019) *Mapping Millennials' Living Standards*. London: Resolution Foundation. [Online]. Available at:

https://www.resolutionfoundation.org/app/uploads/2019/08/Mapping-millennialsliving-standards.pdf (Accessed: 29 August 2019).

Halliday, J. (2017) 'Jeremy Corbyn helps draw Record Crowds to Durham Miners' Gala', *The Guardian*edn), 8 July 2017. [Online] Available at:

https://www.theguardian.com/politics/2017/jul/08/jeremy-corbyn-helps-draw-recordcrowds-to-durham-miners-gala (Accessed: 11 November 2017).

Hambleton, R. (2015a) *Leading the Inclusive City: Place-based innovation for a bounded planet*. Bristol: Policy Press.

Hambleton, R. (2015b) 'Place-based collaboration: leadership for a changing world', *Administration*, 63(3), pp. 5-25 [Online] DOI: 10.1515/admin-2015-0018.

Harrison, J. (2018) "Hopefully it won't be all about Newcastle': South of Tyne transport improvement hopes', *ChronicleLive*edn), 11 December 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/hopefully-wont-newcastle-south-tyne-15531357</u> (Accessed: 11 December 2018).

Hassink, R. and Gong, H. (2016) 'New Economic Geography', in Orum, A.M., Garcia, M., Judd, D., Roberts, B. and Pow, C.P. (eds.) *Wiley-Blackwell Encyclopedia of Urban and Regional Studies*. Wiley. Available at:

https://www.researchgate.net/publication/308782681 New Economic Geography (Accessed: October 2016).

Henig, S. (2016) *North East Combined Authority: Devolution update and poll results.* Durham: Durham County Council [Online]. Available at:

https://democracy.durham.gov.uk/documents/s59801/FINAL%20-

<u>%20NECA%20Devolution.pdf</u> (Accessed: 17 June 2017).

Heseltine, M. (2016) *Tees Valley: Opportunity Unlimited*. London: Department for Communities and Local Government HMSO. [Online]. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/527

<u>649/Tees Valley Opportunity Unlimited.pdf</u> (Accessed: 13 January 2017). Heseltine, M. (2019) *Empowering English Cities*. Birmingham: West Midlands Combined Authority. [Online]. Available at:

https://englishcitiesmichaelheseltine.premediastudio.com/MichaelHeseltine/ (Accessed: 5 July 2019).

Hildreth, P. and Bailey, D. (2013) 'The Economics behind the Move to 'Localism' in England', *Cambridge Journal of Regions, Economy and Society*, 6(2), pp. 233-249 [Online] DOI: 10.1093/cjres/rst004.

Hill, L. (2016a) 'North East Mayoral Hopeful Jeremy Middleton quits the Conservative Party', *ChronicleLive*edn). [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/north-east-mayoral-hopefuljeremy-11423859 (Accessed: 13 January 2017).

Hill, L. (2016b) 'Durham MPs call on councillors to reject 'unsound' North East devolution deal', *ChronicleLive*edn), 24 April 2016. [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/durham-mps-call-councillorsreject-11233164 (Accessed: 17 May 2017).

Hill, L., Graham, H. and Holland, D. (2018) 'North of Tyne Devolution Deal not 'under powered' and will bring in 10,000 Jobs', *ChronicleLive*edn), 20 March 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/north-tynedevolution-deal-not-</u>

14433257?utm source=chronicle live newsletter&utm medium=email&utm content =EM ChronicleLive Nletter News Home smallteaser Text Story8&utm campaign=d aily newsletter (Accessed: 20 March 2018).

Hill, N. (2017) '25-year plan to create 20,000 new jobs at 'UK's greatest single development opportunity' in Redcar ', *Bdaily North East*edn), 18 October 2017.

[Online] Available at: <u>https://bdaily.co.uk/articles/2017/10/18/25-year-plan-to-create-</u> 20-000-new-jobs-at-uk-s-single-greatest-development-opportunity-in-

<u>redcar?utm\_source=bulletin&utm\_medium=email&utm\_campaign=2017-10-19-north-east&utm\_content=lead-story-title</u> (Accessed: 19 October 2017).

HMICFRS (2019) *Cleveland 2018/19*. London: HM Inspectorate of Constabulary and Fire and Rescue Services. [Online]. Available at:

https://www.justiceinspectorates.gov.uk/hmicfrs/peel-assessments/peel-2018/cleveland/ (Accessed: 30 September 2019). Hodgson, A. (2016) North East LEP reaffirms Business Leadership Commitment to Economic Growth. Available at: <u>http://www.nelep.co.uk/north-east-lep-reaffirms-</u> <u>business-leadership-commitment-to-economic-growth/</u> (Accessed: 13 July 2017). Hodgson, A. (2017) Andrew Hodgson gives his view on devolution. Available at: <u>http://www.nelep.co.uk/andrew-hodgson-gives-view-devolution/</u> (Accessed: 19 February 2017).

Hodgson, A. (2019) North East Local Enterprise Partnership Annual General Meeting. Newcastle, 26 September 2019. North East Local Enterprise Partnership(Accessed: 26 September 2019).

Hodgson, G.M. (2006) 'What are Institutions?', *Journal of Economic Issues*, XL(1), pp. 1-25 [Online]. Available at:

http://citeseerx.ist.psu.edu/viewdoc/download;jsessionid=CFE756C956983F107C7C44 A43867E711?doi=10.1.1.121.8035&rep=rep1&type=pdf,

http://citeseerx.ist.psu.edu/viewdoc/download;jsessionid=CFE756C956983F107C7C44 A43867E711?doi=10.1.1.121.8035&rep=rep1&type=pdf.

Holland, D. (2018a) 'North of Tyne Combined Authority: What to expect as leaders meet for first time', *ChronicleLive*edn), 8 November 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/north-tyne-combined-authority-what-15384515</u> (Accessed: 8 November 2018).

Holland, D. (2018b) 'North of Tyne Mayor Candidate revealed by LibDems', *ChronicleLive*edn), 20 December 2018. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/north-tyne-mayor-candidaterevealed-15577987 (Accessed: 29 January 2019).

Horlings, L.G., Roep, D. and Wellbrock, W. (2018) 'The Role of Leadership in Placebased Development and Building Institutional Arrangements', *Local Economy*, 33(3), pp. 245-268 [Online] DOI: 10.1177/0269094218763050 (Accessed: 28 May 2018).

Houchen, B. (2018c) 'Deal Done: Public ownership as promised', *[Twitter]*. Available at: <u>https://pbs.twimg.com/media/DtkcLTiW0AA-dT3.jpg:large</u> (Accessed: 19 June 2019). Houchen, B. (2019a) 'Land Deal agreed for former Tata Steel site', *[Twitter]*. Available at: at:

https://twitter.com/i/topics/tweet/1082226851035123712?cn=ZmxleGlibGVfcmVjc18 y&refsrc=email (Accessed: 8 January 2019).

Houchen, B. (2019b) 'Empty words, as usual', [Twitter]. Available at: https://www.justiceinspectorates.gov.uk/hmicfrs/peel-assessments/peel-

2018/cleveland/ (Accessed: 30 September 2019).

Houchen, B. (2019c) 'British Steel - the Facts', *[Facebook]*. Available at:

https://www.facebook.com/1934566156780439/posts/2351110005126050?sfns=mo (Accessed: 25 May 2019).

Houchen, B. (2019d) 'British Steel: Emergency summit', *[Facebook]*. Available at: <u>https://www.facebook.com/1934566156780439/posts/2351704605066590?sfns=mo</u> (Accessed: 25 May 2019).

Houchen, B. (2019e) 'Boris Johnson will fight for those who feel left behind', *The Times*edn), 19 June 2019. [Online] Available at:

https://www.thetimes.co.uk/article/boris-has-a-record-of-fighting-for-those-who-feelleft-behind-5mf5cbxgt (Accessed: 19 June 2019).

Houchen, B. (2019f) 'Ben Houchen: Which candidate will show leadership on free ports?', [Twitter]. Available at:

https://www.conservativehome.com/localgovernment/2019/07/ben-houchen-whichcandidate-will-show-leadership-on-free-ports.html (Accessed: 5 July 2019). Houchen, B. (2019g) 'It's been seven years since local people could go from Teesside to poolside', [Twitter]. Available at:

https://twitter.com/BenHouchen/status/1151435824942243841 (Accessed: 29 July 2019).

Houchen, B. (2019i) 'As Prime Minister @Boris Johnson will deliver Brexit, he is backing my policy to roll out free ports', [Twitter]. Available at:

https://twitter.com/BenHouchen/status/1153979914971951110?cn=ZmxleGlibGVfcm Vjcw%3D%3D&refsrc=email (Accessed: 24 July 2019).

Houchen, B. (2019j) 'Leave Means Leave', [Twitter]. Available at:

https://twitter.com/LeaveMnsLeave/status/1083648563853344769 (Accessed: 30 July 2019).

Houchen, B. (2019k) 'Manchester does not represent the North and this shows why', *[Twitter]*. Available at: <u>https://twitter.com/BenHouchen</u> (Accessed: 10 August 2019). Houghton, A. (2019) 'Boris Johnson's message to business: satellites, free ports and GM crops', *BusinessLiveedn*), 24 July 2019. [Online] Available at:

https://www.business-live.co.uk/economic-development/boris-johnsons-messagebusiness-satellites-16640138 (Accessed: 29 July 2019).

Hudson, R. (2000b) *Production, Places and Environment: Changing Perspectives in Economic Geography*. Prentice Hall. Available at:

http://NCL.eblib.com/patron/FullRecord.aspx?p=1798365 (Accessed: 20 June 2019).

Huggins, R. and Thompson, P. (2017) *Human Behaviour and Economic Growth: A psychocultural perspective and local and regional development* Cardiff: Cardiff University, Nottingham Business School. [Online]. Available at:

https://orca.cf.ac.uk/107148/1/Huggins%20and%20Thompson%20-

%20Human%20Behaviour%20and%20Economic%20Competitivness%20Report.pdf (Accessed: 17 September 2018).

Huggins, R., Thompson, P. and Obschonka, M. (2018) 'Human Behaviour and Economic Growth: A psychocultural perspective on local and regional development',

*Environment and Planning A*, 50(6), pp. 1269-1289 [Online] DOI:

10.1177/0308518X18778035.

Hughes, L. (2018) 'Metro Mayors lobby for Control over Replacement for EU Cash', *Financial Times Group*edn), 8 October 2018. [Online] Available at:

https://www.ft.com/content/67969552-c88f-11e8-ba8f-

ee390057b8c9?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

2437841d178a#myft:notification:instant-email:content (Accessed: 8 October 2018).

Huntley, D. (2019) "Don't panic' over British Steel as Tees leaders hold public meeting', *TeessideLive*edn), 9 June 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/dont-panic-over-british-steel-16401695?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content= EM\_GazetteLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story6&utm\_campaign=daily

newsletter (Accessed: 10 June 2019).

Hutchinson, A., Robinson, D., Carr, D., Hunt, E., Crenna-Jennings, W. and Akhal, A. (2018) *Education in England: Annual Report 2018*. London: Education Policy Institute. [Online]. Available at: <u>https://epi.org.uk/publications-and-research/annual-report-2018/</u> (Accessed: 4 August 2019).

lammarino, S., Rodríguez-Pose, A. and Storper, M. (2019) 'Regional Inequality in Europe: Evidence, theory and policy implications', *Journal of Economic Geography*, 19(2), pp. 273-298 [Online] DOI: 10.1093/jeg/lby021

Industrial Strategy Commission (2017) *Final Report of the Industrial Strategy Commission*. Sheffield: Sheffield Political Economy Rsearch Institute. [Online]. Available at: <u>http://industrialstrategycommission.org.uk/wp-content/uploads/2017/10/The-Final-Report-of-the-Industrial-Strategy-Commission.pdf</u> (Accessed: 9 January 2020). IPPR (2018) *Prosperity and Justice: A plan for the new economy*. Cambridge: Polity Press.

Jessop, B. (1998) 'The Rise of Governance and the Risks of Failure: the Case of Economic Development', *International Social Science Journal*, 50(155), pp. 29-45 [Online] DOI: 10.1111/1468-2451.00107

Jessop, B. (2005) *Critical Realism and the Strategic-relational Approach*. Researchgate. [Online]. Available at:

https://www.researchgate.net/publication/263691122 Critical Realism and the Stra tegic-Relational Approach (Accessed: 13 July 2017).

Jessop, B. (2016) 'State Theory', in Ansell, C. and Torfing, J. (eds.) *Handbook on Theories of Governance*. Edward Elgar.

Jones, G.W. (1969) *Borough Politics: A Study of Wolverhampton Borough Council 1888-1964*. Basingstoke Macmillan.

Jones, M. (2001) 'The Rise of the Regional State in Economic Governance: 'Partnerships for prosperity' or 'new scales of state power'', *Environment and Planning A*, 33(7), pp. 1185-1211 [Online] DOI: 10.1068/a32185

Jones, R., Goodwin, M., Jones, M. and Simpson, G. (2004) 'Devolution, state personnel and the production of new territories of governance in the United Kingdom',

Environment and Planning A, 36, pp. 89-109 [Online] DOI:

http://epn.sagepub.com/content/36/1/89.full.pdf

Keane, D. (2017) 'Yorkshire and County Durham signs re-ignite row over power and identity in Stockton Borough', *GazetteLive*edn), 16 January 2017. [Online] Available at: <a href="http://www.gazettelive.co.uk/news/teesside-news/yorkshire-county-durham-signs-reignite-12463982">http://www.gazettelive.co.uk/news/teesside-news/yorkshire-county-durham-signs-reignite-12463982</a> (Accessed: 14 December 2017).

Keating, M. (1998) *The New Regionalism in Western Europe : Territorial restructuring and political change*. Cheltenham, UK ; Northampton, MA: Elgar.

Kelly, J.-F. (2016a) *Turnaround Towns: International Evidence*. Dunfermline: Carnegie UK Trust.

Kelly, M. (2016b) 'Gateshead's decision to reject devolution decision stuns critics, who describe move as 'mad', *ChronicleLive*edn), 22 March 2016. [Online] Available at: <u>http://www.chroniclelive.co.uk/news/north-east-news/gatesheads-decision-reject-devolution-deal-11080752</u> (Accessed: 13 July 2017).

Kelly, M. (2018) 'Minister assures North East over post-Brexit cash to help make up lost EU millions', *ChronicleLive*edn), 3 January 2018. [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/minister-assures-north-eastover-14106299 (Accessed: 3 January 2018).

Kelly, M. (2019) 'General election results prove Labour should never take the North East for granted again', *ChronicleLive*edn), 13 December 2019. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/general-election-results-</u> <u>prove-labour-17414722</u> (Accessed: 13 December 2019).

Kinossian, N. (2018) 'Planning strategies and practices in non-core regions: a critical response', *European Planning Studies*, 26(2), pp. 265-275 [Online] DOI: 10.1080/09654313.2017.1361606.

Krugman, P. (2011) 'The New Economic Geography: Now middle aged', *Regional Studies*, 45(1), pp. 1-7 [Online] DOI: 10.1080/00343404.2011.537127.

Labour Party (2019a) 'Unfair Funding Formula', *North Jesmond Voice*edn), April 2019. (Accessed: 9 April 2019).

Labour Party (2019b) *It's Time for Real Change: The Labour Party Manifesto 2019*. London: Labour Party. [Online]. Available at: <u>https://labour.org.uk/wp-</u>

<u>content/uploads/2019/11/Real-Change-Labour-Manifesto-2019.pdf</u> (Accessed: 5 December 2019).

Leach, K. (2013) 'Community Economic Development: Localisation , the key to a resilient and inclusive local economy', *Local Economy*, 28(7-8), pp. 927-931 [Online] DOI: 10.1177/0269094213500912.

Leach, S. (2010) 'The Labour Government's Local Government Agenda 1997-2009: The impact on member-officer relationships', *Local Government Studies*, 36(3), pp. 323-339 [Online] DOI:

http://www.tandfonline.com/doi/pdf/10.1080/03003931003738207?needAccess=true

Leckie, Prior, L. and Goldstein, H. (2019) *Report for the Northern Powerhouse Partnership on Adjusted Progress 8*. Manchester: Northern Powerhouse Partnership

University of Bristol Northern Powerhouse |Partnership. [Online]. Available at: <u>http://npp.maginfrastructure.com/media/1221/report-for-the-northern-powerhouse-partnership-on-adjusted-progress-8.pdf</u> (Accessed: 29 October 2019).

Leeke, M., Sear, C. and Gay, O. (2003) *Introduction to Devolution in the UK* (03/84). London: House of Commons Library. [Online]. Available at:

http://researchbriefings.parliament.uk/ResearchBriefing/Summary/RP03-84#fullreport (Accessed: 22 December 2016).

Leunig, T. and Swaffield, J. (2007) *Cities Unlimited: Managing urban regeneration work*. London: Policy Exchange. [Online]. Available at:

https://www.policyexchange.org.uk/wp-content/uploads/2016/09/cities-unlimitedaug-08.pdf (Accessed: 15 February 2017).

Lewis, A. (2018a) *Durham Tees Valley Airport*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2018/02/DTVA-Background-Paper-1.pdf</u> (Accessed: 12 June 2018). Lewis, A. (2018b) *Managing Director's Update: 28 March 2018*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

content/uploads/2018/03/6MD-Update.pdf (Accessed: 12 June 2018).

Lewis, A. (2018c) *Establishment of an Air Connectivity Facility*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>content/uploads/2018/03/11-Sector-Action-Plan-Advanced-Manufacturing-Report.pdf

(Accessed: 12 June 2018).

Lewis, A. (2018d) *Budget 2018/19 and Investment Plan*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2018/02/Budget-201819-Report.pdf</u> (Accessed: 26 February 2018). LGA (undated-d) *Combined Authorities*. London: Local Government Association,.

[Online]. Available at: <u>https://www.local.gov.uk/topics/devolution/devolution-online-hub/devolution-explained/combined-authorities</u> (Accessed: 28 May 2020).

Lidstrom, A. (1999) 'The Comparative Study of Local Government Systems: A research agenda', *Journal of Comparative Policy Analysis: Research and Practice*, 1(1), pp. 97-115 [Online] DOI: 10.1080/13876989808412617.

Lijphart, A. (1971) 'Comparative Politics and the Comparative Method', *The American Political Science Review*, 65(3), pp. 682-693 [Online]. Available at:

http://www.jstor.org/stable/pdf/1955513.pdf,

http://www.jstor.org/stable/pdf/1955513.pdf (Accessed: 28 June 2017).

Lloyd, C. (2017) 'Profile: Tees Valley Mayor candidate Ben Houchen', *Northern Echo*edn), 6 April 2017. [Online] Available at:

http://www.thenorthernecho.co.uk/features/15207767.Profile Tees Valley Mayor candidate Ben Houchen/ (Accessed: 23 May 2017).

Lloyd, C. (2019) 'Labour's North East catastrophe: pummelled in protest', *Northern Echo*edn), 4 May 2019. [Online] Available at:

https://www.thenorthernecho.co.uk/news/17618258.labours-north-east-catastrophepummelled-in-protest/ (Accessed: 4 May 2019).

Lloyd, M., (2015) 'Fiscal Federalism in the UK'. December 2015. London: The Federal Trust for Education and Research.

Los, B., McCann, P., Springford, J. and Thissen, M. (2017) 'The Mismatch between Local Voting and Local Economic Consequences of Brexit', *Journal of Regional Studies*, 51(5), pp. 786-799 [Online] DOI: 10.1080/00343404.2017.1287350.

Lyotard, J.-F. (1984) *The Postmodern Condition: A report on knowledge*. Translated by Bennington, G. and Massumi, B. Manchester: Manchester University Press.

MacKinnon, D. (forthcoming) 'Northern Powerhouse'.

MacKinnon, D. and Cumbers, A. (2011) *Introduction to Economic Geography*. 2nd edn. Harlow, Essex: Pearson Education.

Magrini, E. (2018) 'Mayoral Cities are already a Step Ahead in preparing for the Rise of Robots', *Centre for Cities*. Available at: <u>http://www.centreforcities.org/blog/mayoral-cities-already-step-ahead-preparing-rise-</u>

robots/?utm\_source=Centre+for+Cities+Newsletter&utm\_campaign=4a6615404e-EMAIL\_CAMPAIGN\_2018\_02\_14&utm\_medium=email&utm\_term=0\_2a9c9d5ef9-4a6615404e-156019569&mc\_cid=4a6615404e&mc\_eid=333f9bcecf (Accessed: 17 February 2018).

Mahoney, J. and Thelen, K. (eds.) (2012) *Explaining Institutional Change: Agency, ambiguity and power*. Cambridge: Cambridge University Press.

Manning, J. (2018) 'Could Gateshead be set to become the Silicon Valley of the North?', *ChronicleLive*edn), 11 November 2018. [Online] Available at:

https://www.chroniclelive.co.uk/business/business-news/could-gateshead-setbecome-silicon-15396141 (Accessed: 12 November 2018).

Manning, J. (2019) 'Business leaders, unions and politicians hit out at Government over British Steel collapse', *ChronicleLive*edn), 22 May 2019. [Online] Available at:

https://www.chroniclelive.co.uk/business/business-news/business-leaders-unions-politicians-hit-

<u>16313255?utm source=chronicle live newsletter&utm medium=email&utm content</u> <u>=EM TheJournalNEC Nletter News Business mediumteaser Text Story1&utm cam</u> <u>paign=business newsletter</u> (Accessed: 25 May 2019).

Martin, R. (1999) 'The new 'geographical turn' in economics: Some critical reflections', *Cambridge Journal of Economics*, 23(1), pp. 65-91 [Online]. Available at:

https://www.jstor.org/stable/23600667, https://www.jstor.org/stable/23600667 (Accessed: 15 February 2019).

Martin, R. (2000) 'Institutionalist Approaches to Economic Geography', in Sheppard, E. and Barnes, T. (eds.) *Companion to Economic Geography*. Oxford: Blackwell, pp. 75-89. Martin, R. (2008) 'National growth versus spatial equality? A cautionary note on the new 'trade-off' thinking in regional policy discourse', *Regional Science Policy and Practice*, 1(1), pp. 3-13 [Online] DOI: 10.1111/j.1757-7802.2008.00003.x.

Martin, R. (2010) 'Roepke Lecture in Economic Geography: Rethinking regional path dependence: beyond lock-in to evolution', Economic Geography, 86(1), pp. 1-27 [Online] DOI: 10.1111/j.1944-8287.2009.01056.x.

Martin, R. (2011) 'The 'New Economic Geography': Credible models of the econmic landscape?', in Leyshon, A., Lee, R., McDowell, L. and Sunley, P. (eds.) Sage Handbook of Economic Geography. Sage. Available at:

https://ebookcentral.proguest.com/lib/ncl/reader.action?docID=689508&ppg=70 (Accessed: 28 February 2011).

Martin, R. (2015) 'Rebalancing the Spatial Economy: The challenge for regional theory', *Territory Politics Governance*, 3(3), pp. 235-272 [Online] DOI:

10.1080/21622671.2015.1064825.

Martin, R. and Gardiner, B. (2018) 'Reviving the Northern Powerhouse and Spatially Rebalancing the British Economy: The scale of the challenge', in Berry, C. and Giovannini, A. (eds.) Developing England's North: the Political Economy of the Northern Powerhouse. Palgrave Macmillan. Available at:

https://link.springer.com/content/pdf/10.1007%2F978-3-319-62560-7.pdf (Accessed: 2018).

Martin, R., Pike, A., Tyler, P. and Gardiner, B. (2016) 'Spatially Rebalancing the UK Economy: towards a New Policy Model', Regional Studies, 50(2), pp. 342-357 [Online] DOI: 10.1080/00343404.2015.1118450.

Martin, R. and Sunley, P. (2002) 'Re-thinking the 'Economic' in Economic Geography: Broadening our vision or losing our focus', Antipode, 33(2), pp. 148-161 [Online]. Available at: <a href="https://onlinelibrary.wiley.com/doi/epdf/10.1111/1467-8330.00173">https://onlinelibrary.wiley.com/doi/epdf/10.1111/1467-8330.00173</a> DOI: 10.1111/1467-8330.001.

Martin, R. and Sunley, P. (2006) 'Path Dependence and Regional Economic Evolution', Journal of Economic Geography, 6, pp. 395-437 [Online] DOI: 10.1093/jeg/lbl012. Mason, S. and Davey, G. (2016) North East Combined Authority : Proposed devolution agreement and elected regional mayor. Morpeth: Northumberland County Council. [Online]. Available at:

http://committeedocs.northumberland.gov.uk/MeetingDocs/20675 M3930.pdf (Accessed: 16 February 2017).

McCann, P. (2016) The UK Regional-national Economic Problem. Abingdon: Routledge. McCann, P., (2019) 'UK Research and Innovation: A Place-Based Shift?'. 2019. Sheffield: University of Sheffield Management School. (Accessed: 16 September 2019). McNeal, I. (2017a) 'South Tees Development Corporation Masterplan launched in Redcar', GazetteLiveedn), 18 October 2017. [Online] Available at:

http://www.gazettelive.co.uk/news/teesside-news/live-updates-south-teesdevelopment-13777949 (Accessed: 18 October 2017).

McNeal, I. (2017b) "Give us greater powers' demands Tees Mayor as regional chiefs pile pressure on government', GazetteLiveedn), 1 November 2017. [Online] Available at: http://www.gazettelive.co.uk/news/teesside-news/give-greater-powers-demandstees-13839142?ptnr rid=431380&icid=EM GazetteLive Nletter News (Accessed: 1 November 2017).

McNeal, I. (2017c) 'Regional Bragging Rights', GazetteLiveedn), 2 November 2017. [Online] Available at: http://www.gazettelive.co.uk/news/teesside-news/live-updatesben-houchen-discusses-13845237http://www.gazettelive.co.uk/news/teessidenews/live-updates-ben-houchen-discusses-13845237 (Accessed: 2 November 2017). McNeal, I. (2017d) "Give us Control over Brexit millions' says Mayor amid Fears of

London 'Power Grab' for Tees Cash', GazetteLiveedn), 10 November 2017. [Online]
Available at: <u>http://www.gazettelive.co.uk/news/teesside-news/give-control-over-brexit-millions-13884696</u> (Accessed: 10 November 2017).

McNeal, I. (2018) 'How a Letter from the Chancellor played a crucial role in the talks over the future of the SSI site ', *GazetteLive* edn), 31 May 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/how-letter-chancellor-playedcrucial-14725227</u> (Accessed: 31 May 2018).

McNeal, I., Cain, J. and Metcalfe, A. (2019) 'Live reaction as five Labour councils leaders back Teesside Airport buyout', *TeessideLive*edn), 22 January 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/live-updates-council-leaders-</u> <u>reveal-15713334</u> (Accessed: 22 January 2019).

Melia, P. (2016) North East Combined Authority - Proposed Devolution Agreement and Elected Regional Mayor. North Tyneside: North Tyneside Council. [Online]. Available at:

http://www.northtyneside.gov.uk/pls/portal/NTC\_PSCM.PSCM\_Web.download?p\_ID= 564786 (Accessed: 31 March 2018).

Melia, P. (2017) *North Tyneside Council: Report to Cabinet: Devolution to North of Tyne: Governance Review: 11 December 2017.* North Tyneside: North Tyneside Council [Online]. Available at:

http://my.northtyneside.gov.uk/sites/default/files/meeting/related-

documents/item%206%20g%20%2011%2012%202017.pdf (Accessed: 18 February 2018).

Metcalfe, A. (2018a) "If the mayor believes liars should resign then he should look at himself", *TeessideLive*edn), 20 September 2018. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/cleveland-police-chief-constablecomes-

<u>15176482?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content=</u> <u>EM\_GazetteLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story4&utm\_campaign=daily\_newsletter</u> (Accessed: 20 September 2018).

Metcalfe, A. (2018b) 'Operators? Name Change? Back-up plan? Key questions on airport deal answered', *TeessideLive*edn), 5 December 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/operators-name-change-back-up-15506792?utm source=gazette live newsletter&utm medium=email&utm content= EM GazetteLive Nletter News Home largeteaser Text Story1&utm campaign=daily newsletter (Accessed: 5 December 2018).</u>

Metcalfe, A. (2019a) 'Airport boss says it is unlikely to last beyond 2021 as crucial vote looms ', *TeessideLiveedn*), 10 January 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/airport-boss-says-unlikely-last-

<u>15653713?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content=</u> <u>EM\_GazetteLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story4&utm\_campaign=daily\_newsletter</u> (Accessed: 11 January 2019).

Metcalfe, A. (2019b) 'Minister urges public to back airport plans during 'important month in Teesside's history'', *TeessideLive*edn), 7 January 2019. [Online] Available at: <a href="https://www.gazettelive.co.uk/news/teesside-news/minister-urges-public-back-airport-15642509">https://www.gazettelive.co.uk/news/teesside-news/minister-urges-public-back-airport-15642509</a> (Accessed: 11 January 2019).

Metcalfe, A. (2019c) 'Bold Teesside Airport plan to attract airlines and over a million passengers revealed', *TeessideLive*edn), 16 January 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/bold-teesside-airport-plan-attract-</u>

15685166?utm source=gazette live newsletter&utm medium=email&utm content=

<u>EM GazetteLive Nletter News Home largeteaser Text Story1&utm campaign=daily</u> <u>newsletter</u> (Accessed: 16 January2019).

Metcalfe, A. (2019d) 'Airport plan is about weighing benefits and risks: People want something more on their doorstep', *TeessideLive*edn), 17 January 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/airport-plan-</u>weighing-benefits-risks-15695391 (Accessed: 18 January 2019).

Metcalfe, A. (2019e) 'Tees Valley Mayor Ben Houchen faces public grilling over plans to buy back airport', *TeessideLive*edn), 17 January 2019. [Online] Available at:

https://www.gazettelive.co.uk/live-updates-mayor-ben-houchen-15692541 (Accessed: 18 January 2019).

Metcalfe, A. (2019f) 'Teesside Airport activisits pledge: 'We'll fight it through the ballot boxes' ', *TeessideLive*edn), 22 January 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/teesside-airport-activists-pledgewell-15713939 (Accessed: 22 January 2019).

Metcalfe, A. (2019g) "We're putting Teesside first!": mayor's airport deal is approved to cheers', *TeessideLive*edn), 24 January 2019. [Online] Available at:

https://www.gazettelive.co.uk/were-putting-teesside-first-mayor-15727738 (Accessed: 24 January 2019).

Metcalfe, A. (2019h) 'Plan for Teesside Airport name change sparks bitter row between mayor and advertising chief ', *TeessideLive*edn), 18 April 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/plan-teesside-airport-namechange-16151805</u> (Accessed: 19 April 2019).

Metcalfe, A. (2019i) "Stop moaning' mayor urges as claims 'mistakes' made in Durham Tees Valley Airport deal ', *TeessideLive*edn), 26 April 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/stop-moaning-mayor-urges-</u> <u>claims-16182995</u> (Accessed: 26 April 2019).

Metcalfe, A. (2019j) 'President Trump's US ambassador vitits Teesside to help bring 'comfort to major investor'', *TeessideLive*edn), 30 July 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/president-trumps-ambassador-visits-teesside-16671947</u> (Accessed: 31 July 2019).

MHCLG (2018a) *Business Rates Retention Reform*. London: Ministry of Housing, Communities and Local Government. [Online]. Available at:

https://www.gov.uk/government/consultations/business-rates-retention-reform (Accessed: 5 April 2019).

MHCLG (2018b) *Strengthened Local Enterprise Partnerships*. London: Ministry of Housing, Communities and Local Government. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment\_data/file/728058/Strengthened\_Local\_Enterprise\_Partnerships.pdf (Accessed: 4 March 2019).

MHCLG (2018c) *North of Tyne Combined Authority Devolution Deal*. London: Ministry of Housing, Communities and Local Government. [Online]. Available at:

https://www.gov.uk/government/publications/north-of-tyne-combined-authoritydevolution-deal (Accessed: 25 December 2018).

MHCLG (2019d) *National Local Growth Assurance Framework*. London: Ministry of Housing, Communities and Local Government. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment\_data/file/768356/National\_Local\_Growth\_Assurance\_Framework.pdf (Accessed: 11 March 2019). Milligan, D. and O'Donoghue, D. (2016) 'Gateshead Council leader Mick Henry steps down as council elect new leader Martin Gannon', *ChronicleLive*edn), 20 May 2017. [Online] Available at: <u>http://www.chroniclelive.co.uk/news/north-east-</u>

<u>news/gateshead-council-leader-mick-henry-11363436</u> (Accessed: 12 January 2017). Molotch, H. (1976) 'The City as a Growth Machine: Toward a political economy of place', *American Journal of Sociology* 82(2), pp. 309-332 [Online]. Available at: <u>https://www.jstor.org/stable/pdf/2777096.pdf?refreqid=excelsior%3A54176f99371ed</u>

<u>d458311f8efd053581f</u>,

https://www.jstor.org/stable/pdf/2777096.pdf?refreqid=excelsior%3A54176f99371ed d458311f8efd053581f (Accessed: September 1976).

Momentum (undated) *Momentum Gateshead and Blaydon*. Available at: <u>https://www.facebook.com/permalink.php?story\_fbid=951160944981346&id=944653</u> <u>188965455&\_tn\_=K-R</u> (Accessed: 10 April 2019).

Moore-Cherry, N. and Tomaney, J. (2019) 'Spatial Planning, Metropolitan Governance and Territorial Politics in Europe: Dublin as a case of metrophobia', *European Urban and Regional Studies* 26(4), pp. 365-381 [Online] DOI: 10.1177/0969776418783832. Morgan, K. (2006) 'Devolution and Development: Territorial justice and the northsouth divide', *Journal of Federalism*, 36(1), pp. 189-206 [Online] DOI: 10.1093/publius/pjj003.

Morse, A. (2017) *Progress in Setting Up Combined Authorities* (HC240). London: National Audit Office. [Online]. Available at: <u>https://www.nao.org.uk/wp-</u> <u>content/uploads/2017/07/Progress-in-setting-up-combined-authorities.pdf</u> (Accessed:

20 July 2017).

Mossberger, K. and Stoker, G. (2001) 'The Evolution of Urban Regime Theory', *Urban Affairs Review*, 36(6), pp. 810-835 [Online] DOI: 10.1177/10780870122185109.

Mulholland, H. (2012) 'Mayor H'Angus the Monkey finally loses his Hartlepool habitat', *The Guardian*edn), 16 November 2012. [Online] Available at:

https://www.theguardian.com/politics/2012/nov/16/mayor-elections-hangusmonkey-hartlepool-habitat (Accessed: 6 Dcember 2019).

Muncaster, M. (2018) 'North of Tyne devolution: North East councils vote in favour of landmark deal', *ChronicleLive*edn), 26 April 2018. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/north-tyne-devolution-northeast-

<u>14582517?utm\_source=chronicle\_live\_newsletter&utm\_medium=email&utm\_content</u> =EM\_ChronicleLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story8&utm\_campaign=d aily\_newsletter (Accessed: 27 April 2018).

National Archives (undated) *Local Government Act 1972*. London: National Archives. [Online]. Available at: <u>https://www.legislation.gov.uk/ukpga/1972/70</u> (Accessed: 14 March 2019).

Neal, S. and McLaughlin, E. (2009) 'Researching up? Interviews, emotionality and policy-making elites', *Journal of Social Policy*edn), pp. 689-707. [Online] Available at: <u>https://www.cambridge.org/core/services/aop-cambridge-</u>

<u>core/content/view/5144A6FF4E95E1B09281FAA137E61E32/S0047279409990018a.pdf</u> /<u>div-class-title-researching-up-interviews-emotionality-and-policy-making-elites-</u> <u>div.pdf</u> (Accessed: 27 January 2017).

NECA (2014a) North East Combined Authority Leadership Board Supplemental Agenda: 21 October 2014, Newcastle. North Tyneside: North East Combined Authority [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-</u> <u>Board-21-October-2014-Supplemental-Agenda-Pack.pdf</u> (Accessed: 20 February 2017). NECA (2014b) North East Combined Authority Leadership Board Agenda: 15 July 2014, Newcastle. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/03/Leadership-Board-15-July-2014-Agenda-Pack.pdf</u> (Accessed: 20 February 2017).

NECA (2014c) North East Combined Authority Leadership Board Supplemental Agenda: 15 July 2014, Newcastle. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/03/Leadership-Board-15-July-2014-Supplemental-Agenda-Pack-</u> <u>1.pdf</u> (Accessed: 20 February 2017).

NECA (2015a) North East Combined Authority Leadership Board Agenda: 14 July 2015, Morpeth. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-14-July-</u>2015-Agenda-Pack.pdf (Accessed: 21 February 2017).

NECA (2015b) North East Combined Authority Leadership Board Agenda: 20 January 2015, North Tyneside. North Tyneside: North East Combined Authority. [Online]. Available at: https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-

Board-20-January-2015-Agenda-Pack.pdf (Accessed: 21 February 2017).

NECA (2015c) North East Combined Authority Leadership Board Supplemental Agenda: 20 January 2015, North Tyneside. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

content/uploads/2018/03/Leadership-Board-20-January-2015-Supplemental-Agenda-Pack-1.pdf (Accessed: 21 February 2017).

NECA (2015d) North East Combined Authority Leadership Board Agenda: 15 September 2015, Newcastle. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-15-September-2015-Agenda-Pack.pdf</u> (Accessed: 21 February 2017).

NECA (2015e) North East Combined Authority Leadership Board Supplemental Agenda 3: 17 November 2015, Gateshead. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/03/Leadership-Board-17-November-2015-Supplemental-</u> <u>Agenda-Pack-3.pdf</u> (Accessed: 21 February 2017).

NECA (2015f) North East Combined Authority Overview and Scrutiny Committee Agenda: 10 February 2015, Morpeth. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2015/02/Overview-and-Scrutiny-Committee-10-February-2015-</u> <u>Agenda-Pack.pdf</u> (Accessed: 19 August 2017).

NECA (2015g) North East Combined Authority Overview and Scrutiny Committee Agenda: 22 June 2015, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2015/06/Overview-and-Scrutiny-Committee-22-June-2015-Agenda-</u> <u>Pack.pdf</u> (Accessed: 19 August 2017).

NECA (2015h) North East Combined Authority Overview and Scrutiny Committee Agenda: 8 September 2015, Gateshead. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2015/09/Overview-and-Scrutiny-Committee-8-September-2015-</u> <u>Agenda-Pack.pdf</u> (Accessed: 19 August 2017).

NECA (2016a) North East Combined Authority Leadership Board Agenda: 19 January 2016, North Tyneside. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-19-January-2016-Agenda-Pack.pdf</u> (Accessed: 22 February 2017). NECA (2016b) North East Combined Authority Economic Development and Regeneration Advisory Board Minutes: 25 October 2016, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wpcontent/uploads/2019/02/EDRAB-25-October-2016-Approved-Minutes.pdf</u> (Accessed: 28 February 2019).

NECA (2016c) *Chief Finance Officer's Narrative Report for the Year ended 31 March 2016*. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/04/Narrative-Report-2015-16.pdf</u> (Accessed: 16 March 2019).

NECA (2016d) North East Combined Authority Leadership Board Supplemental Agenda 2: 24 March 2016, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-24-March-2016-Extraordinary-Meeting-Supplemental-Agenda-Pack-2.pdf</u> (Accessed: 22 February 2017).

NECA (2016e) North East Combined Authority Leadership Board: Devolution agreement supplementary report: 24 March 2016, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/03/Leadership-Board-Report-24-March-2016-Extraordinary-</u> Meeting-Supplementary-Report.pdf (Accessed: 23 February 2017).

NECA (2016f) North East Combined Authority Leadership Board Agenda: 19 April 2016, Sunderland. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-19-April-2016-Agenda-Pack.pdf</u> (Accessed: 23 February 2017).

NECA (2016g) North East Combined Authority Leadership Board Supplemental Agenda: 17 May 2016, Durham. North Tyneside: North East Combined Authority. [Online].

Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-</u> Board-17-May-2016-Supplemental-Agenda-Pack-1.pdf (Accessed: 23 February 2017).

NECA (2016h) North East Combined Authority Leadership Board Supplemental Agenda 2: 17 May 2016, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-

Board-17-May-2016-Supplemental-Agenda-Pack-2.pdf (Accessed: 23 February 2017). NECA (2016i) North East Combined Authority Overview and Scrutiny Committee Minutes: 7 July 2019, Durham. North Tyneside: North East Combined Authority.

[Online]. Available at: https://northeastca.gov.uk/wp-

<u>content/uploads/2019/02/Overview-and-Scrutiny-Committee-7-July-2016-Approved-</u> <u>Minutes.pdf</u> (Accessed: 6 December 2019).

NECA (2016j) North East Combined Autthority Leadership Board Agenda: 19 July 2016, Morpeth. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-19-July-</u> 2016-Agenda-Pack.pdf (Accessed: 30 December 2018).

NECA (2016k) North East Combined Authority Leadership Board (Extraordinary Meeting) Supplemental Agenda: 4 July 2016, Chester-le-Street (23 February 2017). North Tyneside: North East Combined Authority. [Online]. Available at:

https://northeastca.gov.uk/wp-content/uploads/2019/02/Leadership-Board-4-July-2016-Supplemental-Agenda-Pack.pdf (Accessed: 23 February 2017).

NECA (2016l) North East Combined Authority Leadership Board Supplemental Agenda 1: 19 July 2016, Morpeth. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-</u> <u>Board-19-July-2016-Supplemental-Agenda-Pack-1.pdf</u> (Accessed: 23 February 2017). NECA (2016m) North East Combined Authority Leadership Board Supplemental Agenda 2: 19 July 2016, Morpeth. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-19-July-2016-Supplemental-Agenda-Pack-2.pdf</u> (Accessed: 23 February 2017). NECA (2016n) North East Combined Authority Leadership Board Agenda: 20 September 2016, Sunderland. North Tyneside: North East Combined Authority. [Online]. Available

at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-20-</u> September-2016-Agenda-Pack.pdf (Accessed: 23 February 2017).

NECA 2017 (2016o) 'North East Combined Authority Leadership Board (Extraordinary Meeting) Agenda: 6 September 2016, Sunderland'. North Tyneside: North East Combined Authority. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/03/Leadership-Board-6-September-2016-Agenda-Pack.pdf</u> (Accessed: 23 February 2017).

NECA (2016p) North East Combined Authority Leadership Board Agenda: 15 November 2016, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: <a href="https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-15-November-2016-Agenda-Pack.pdf">https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-15-November-2016-Agenda-Pack.pdf</a> (Accessed: 24 February 2017).

NECA (2016g) North East Combined Authority Leadership Board Supplemental Agenda

*2, 20 September 2016, Sunderland:* . North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/03/Leadership-Board-20-September-2016-Supplemental-</u> <u>Agenda-Pack-2.pdf</u> (Accessed: 28 May 2017).

NECA (2016r) North East Combined Authority Overview and Scrutiny Committee Agenda: 1 November 2016, Jarrow. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2016/11/Overview-and-Scrutiny-Committee-1-November-2016-</u> <u>Agenda-Pack.pdf</u> (Accessed: 20 August 2017).

NECA (2016s) North East Combined Authority Overview and Scrutiny Committee Agenda: 14 December 2016, Gateshead. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2016/12/Overview-and-Scrutiny-Committee-14-December-2016-</u> <u>Agenda-Pack.pdf</u> (Accessed: 20 August 2016).

NECA (2016t) North East Combined Authority Leadership Board Annual Meeting: 21 June 2016, Chester-le-Street. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-</u> Board-21-June-2016-Agenda-Pack.pdf (Accessed: 30 December 2018).

NECA (2016u) North East Combined Authority Overview and Scrutiny Committee Agenda: 7 July 2016, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: https://northeastca.gov.uk/wp-

<u>content/uploads/2016/07/Overview-and-Scrutiny-Committee-7-July-2016-Agenda-</u> <u>Pack.pdf</u> (Accessed: 20 August 2017).

NECA (2017a) *Chief Finance Officer's Narrative Report for the Year ended 31 March 2017.* North Tyneside: North East Combined Authority. [Online]. Available at: <a href="https://northeastca.gov.uk/wp-content/uploads/2018/04/2016-17-Narrative-Report.pdf">https://northeastca.gov.uk/wp-content/uploads/2018/04/2016-17-Narrative-Report.pdf</a> (Accessed: 16 March 2019).

NECA (2017b) North East Combined Authority Leadership Board Supplemental Agenda 2: 17 January 2017, North Tyneside. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u> content/uploads/2018/03/Leadership-Board-17-January-2017-Supplemental-Agenda-Pack-2.pdf (Accessed: 24 February 2017).

NECA (2017c) North East Combined Authority Leadership Board Agenda: 21 March 2017, Morpeth. North Tyneside: North East Combined Authority. [Online]. Available at: <a href="https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-21-March-2017-Agenda-Pack.pdf">https://northeastca.gov.uk/wp-content/uploads/2018/03/Leadership-Board-21-March-2017-Agenda-Pack.pdf</a> (Accessed: 24 May 2017).

NECA (2017d) The Durham, Gateshead, Newcaste upon Tyne, North Tyneside, Northumberland, South Tyneside and Sunderland Combined Authority Constitution. North Tyneside: North East Combined Authority. [Online]. Available at:

https://northeastca.gov.uk/wp-content/uploads/2018/08/NECA-Constitution.pdf.pdf (Accessed: 12 June 2017).

NECA (2017e) North East Combined Authority Leadership Board Annual Meeting Agenda: 20 June 2017, North Tyneside. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

content/uploads/2018/03/Leadership-Board-20-June-2017-Agenda-Pack.pdf (Accessed: 18 August 2017).

NECA (2017f) North East Combined Authority Overview and Scrutiny Committee Agenda: 10 July 2017, Gateshead. North Tyneside North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2017/07/Overview-and-Scrutiny-Committee-10-July-2017-Agenda-</u> <u>Pack.pdf</u> (Accessed: 8 October 2017).

NECA (2017g) North East Combined Authority Vacancy: Independent chairperson: Overview and Scrutiny Committee. Available at:

https://www.northeastca.gov.uk/news/neca-vacancy-independent-chairpersonoverview-and-scrutiny-committee (Accessed: 8 October 2017).

NECA,

https://www.northeastca.gov.uk/sites/default/files/minutes\_document/NECA%20Ove rview%20%26%20Scrutiny%20Committee%2028%20March%202017%20-

<u>%20agenda%20pack.pdf</u> (2017h) 'Overview and Scrutiny Committee Agenda: 28 March 2017, Sunderland' *Overview and Scrutiny Committee*. North Tyneside North East Combined Authority. Available at:

https://www.northeastca.gov.uk/sites/default/files/minutes\_document/NECA%20Ove rview%20%26%20Scrutiny%20Committee%2028%20March%202017%20-

<u>%20agenda%20pack.pdf</u> (Accessed: 20 August 2017).

NECA (2017i) North East Combined Authority Governance Committee: 4 April 2017, Sunderland. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2019/02/Governance-Committee-4-</u> <u>April-2017-Approved-Minutes.pdf</u> (Accessed: 6 March 2019).

NECA (2017p) North East Combined Authority Overview and Scrutiny Committee Agenda, 14 February 2017, Newcastle. North Tyneside: North East Combined Authority, North East Combined Authority. [Online]. Available at:

https://northeastca.gov.uk/wp-content/uploads/2017/02/Overview-and-Scrutiny-Committee-14-February-2017-Agenda-Pack.pdf (Accessed: 12 May 2020).

NECA (2018a) North East Combined Authority Leadership Board Annual Meeting Agenda: 19 June 2018, South Shields. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/06/Leadership-Board-19-June-2018-Agenda-Pack.pdf</u> (Accessed: 20 June 2018). NECA (2018b) North East Combined Authority Overview and Scrutiny Committee Agenda: 5 February 2018, Morpeth. North Tyneside: North East Combined Autthority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/02/Overview-and-Scrutiny-Committee-5-February-2018-</u> <u>Agenda-Pack.pdf</u> (Accessed: 30 December 2018).

NECA (2018c) *The Durham, Gateshead, South Tyneside and Sunderland Combined Authority Constitution: November 2018.* Sunderland: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/12/NECA-Constitution.pdf</u> (Accessed: 29 December 2018).

NECA (2018d) North East Combined Authority Overview and Scrutiny Committee Agenda:, 20 September 2018, Gateshead. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2018/09/Overview-and-Scrutiny-Committee-20-September-2018-</u> <u>Agenda-Pack.pdf</u> (Accessed: 30 December 2018).

NECA (2018e) North East Combined Authority Audit and Standards Committee Agenda: 4 December 2018, Sunderland. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/11/Audit-and-Standards-Committee-4-December-2018-Agenda-Pack.pdf</u> (Accessed: 6 March 2019).

NECA (2018f) North East Combined Authority Audit and Standards Committee Minutes: 3 April 2018, Sunderland. North Tynesdide: North East Combined Authority. [Online]. Available at: https://northeastca.gov.uk/wp-content/uploads/2018/07/Governance-

<u>Committee-3-April-2018-Approved-Minutes.pdf</u> (Accessed: 6 March 2019).

NECA (2018g) North East Combined Authority Audit and Standards Committee Approved Minutes: 30 July 2018, North Tyneside. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

<u>content/uploads/2019/01/Audit-and-Standards-Committee-30-July-2018-Approved-</u> <u>Minutes.pdf</u> (Accessed: 6 March 2019).

NECA (2018h) North East Combined Authority Leadership Board Approved Minutes: 20 March 2018, South Shields. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2019/01/Leadership-</u> <u>Board-20-March-2018-Approved-Minutes.pdf</u> (Accessed: 11 March 2019).

NECA (2018i) North East Combined Authority Leadership Board Approved Minutes: 26 April 2018, Gateshead. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2019/01/Leadership-</u> Board-26-April-2018-Approved-Minutes.pdf (Accessed: 11 March 2019).

NECA (2018j) North East Combined Authority Leadership Board Approved Minutes: 31 July 2018, South Shields. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2019/01/Leadership-</u> Board-31-July-2018-Approved-Minutes.pdf (Accessed: 11 March 2019).

NECA (2018k) North East Combined Authority Leadership Board Decisions: 13 November 2018, Gateshead. Newcastle: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2018/11/Leadership-Board-13-November-2018-Decisions.pdf</u> (Accessed: 11 March 2019).

NECA (2018I) Chief Finance Officer's Narrative Report for the Year ended 31 March 2018. North Tyneside: North East Combined Authority. [Online]. Available at: https://northeastca.gov.uk/wp-content/uploads/2018/04/2016-17-Narrative-Report.pdf (Accessed: 16 March 2019). NECA (2018m) North East Combined Authority Overview and Scrutiny Committee Agenda: 15 March 2018, Durham. North Tyneside: North East Combined Authority. [Online]. Available at: <u>https://northeastca.gov.uk/wp-</u>

content/uploads/2018/03/Overview-and-Scrutiny-Committee-15-March-2018-Agenda-Pack.pdf (Accessed: 30 December 2018).

NECA (2019a) North East Combined Authority Economic Development and Regeneration Advisory Board Membership. Available at:

<u>https://northeastca.gov.uk/decision-making/economic-development-and-</u>regeneration-advisory-board/membership/ (Accessed: 11 December 2018).

NECA (2019b) North East Combined Authority Leadership Board Supplemental Agenda: 5 February 2019, Gateshead. Newcastle: North East Combined Authority,. [Online]. Available at: <u>https://northeastca.gov.uk/wp-content/uploads/2019/01/Leadership-</u> Board-5-February-2019-Supplemental-Agenda-Pack.pdf (Accessed: 3 March 2019).

NECA (2019c) North East Combined Authority Leadership Board Decisions: 5 February 2019, Gateshead. Newcastle: North East Combined Authority. [Online]. Available at: https://northeastca.gov.uk/wp-content/uploads/2019/02/Leadership-Board-5-

February-2019-Decisions.pdf (Accessed: 11 March 2019).

NECA (2019d) North East Combined Authority Overview and Scrutiny Committee Agenda: 14 March 2019, Sunderland. Newcastle: North East Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2019/03/DRAFT-Minutes-010319.pdf</u> (Accessed: 11 March 2019). NECA (undated-a) *Overview and Scrutiny Committee*. Available at:

https://northeastca.gov.uk/decision-making/overview-and-scrutiny-

committee/#papers-section (Accessed: 25 March 2019).

NECA (undated-b) North East Combined Authority Audit and Standards Committee (previously Governance Committee). Available at: <u>https://northeastca.gov.uk/decision-making/audit-and-standards-committee-previously-governance-committee/#dates-section</u> (Accessed: 6 March 2019).

NECA (undated-c) *Economic Development and Regeneration Advisory Board Membership*. Available at: <u>https://northeastca.gov.uk/decision-making/economic-</u> <u>development-and-regeneration-advisory-board/membership/</u> (Accessed: 30 March 2019).

NELEP (2014a) *More and Better Jobs: A strategic economic plan for the North East.* Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-content/uploads/2014/11/North-East-Strategic-</u>

<u>Economic-Plan-More-and-Better-Jobs.pdf</u> (Accessed: 4 November 2016). NELEP (2015a) *North East Local Enterprise Partnership Board Minutes: South Tyneside, 14 May 2015*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: http://www.nelep.co.uk/wp-content/uploads/2015/09/NE-LEP-Docs.-15-05-14-

Board-Minutes.pdf (Accessed: 27 February 2017).

NELEP (2015b) North East Local Enterprise Partneship Board Minutes: Ashington, 23 July 2015, . Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <a href="http://www.nelep.co.uk/wp-content/uploads/2015/10/Board-Minutes-July-2015.-15-07-23-Redacted-Version.pdf">http://www.nelep.co.uk/wp-content/uploads/2015/10/Board-Minutes-July-2015.-15-07-23-Redacted-Version.pdf</a> (Accessed: 27 February 2017).

NELEP (2015c) North East Local Enterprise Partnership Board Minutes: Peterlee, 29 March 2015. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: http://www.nelep.co.uk/wp-content/uploads/2015/05/North-East-LEP-Board-Minutes-March-2015.pdf (Accessed: 27 February 2017). NELEP (2016a) North East Local Enterprise Partnership Board Agenda: Gateshead, 24 November 2016. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-content/uploads/2016/11/LEP-board-</u> minutes-24-November-2016.pdf (Accessed: 5 March 2017).

NELEP (2016b) North East Local Enterprise Partnership Board Minutes: Chester-le-Streeet, 28 January 2016. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-content/uploads/2014/10/16-01-28-</u> <u>Redacted-Version-2.pdf</u> (Accessed: 27 February 2017).

NELEP (2016c) North East Local Enterprise Partnership Board Minutes: Wallsend, 13 April 2016. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: http://www.nelep.co.uk/governance/ (Accessed: 27 February 2017).

NELEP (2016d) North East Local Enterprise Partnership Board Agenda: South Shields, 26 May 2016. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <a href="http://www.nelep.co.uk/wp-content/uploads/2015/03/North-East-LEP-Board-Agenda-and-Papers-26-05-2016.pdf">http://www.nelep.co.uk/wp-content/uploads/2015/03/North-East-LEP-Board-Agenda-and-Papers-26-05-2016.pdf</a> (Accessed: 27 February 2017).

NELEP (2016e) North East Local Enterprise Partnership Board Minutes: Sunderland, 28 July 2016, Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <a href="http://www.nelep.co.uk/wp-content/uploads/2017/02/July-board-minutes.pdf">http://www.nelep.co.uk/wp-content/uploads/2017/02/July-board-minutes.pdf</a> (Accessed: 27 February 2017).

NELEP (2017b) North East Local Enterprise Partnership Board Agenda: Gateshead, 26 January 2017. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-content/uploads/2017/01/North-East-LEP-Board-</u> Papers-January-2017.pdf (Accessed: 27 February 2017).

NELEP (2017c) *More and Better Jobs: the North East strategic economic plan.* Newcastle: North East Local Enterprise Parrtnership. [Online]. Available at:

http://www.nelep.co.uk/wp-content/uploads/2017/04/North-East-SEP-FINAL-March-2017.pdf (Accessed: 21 April 2017).

NELEP (2017d) Building our Industrial Strategy: A response from the North East Local Enterprise Partnership and the North East Combined Authority. Newcastle: North East Local Entrprise Partnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-</u> <u>content/uploads/2017/04/North-East-LEP-Industrial-Strategy-Response-Apr-2017.pdf</u>

(Accessed: 19 May 2017).

NELEP (2017e) North East Local Enterprise Partnership Board Agenda: Boldon, 27 July 2017. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <a href="http://www.nelep.co.uk/wp-content/uploads/2017/07/Board-agenda-and-papers-July-2017.pdf">http://www.nelep.co.uk/wp-content/uploads/2017/07/Board-agenda-and-papers-July-2017.pdf</a> (Accessed: 18 August 2017).

NELEP (2017f) *Annual Review 2016-2017*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>http://www.nelep.co.uk/wp-</u>

<u>content/uploads/2017/05/North-East-LEP-Agenda-and-Board-Papers-May-2017.pdf</u> (Accessed: 7 June 2017).

NELEP (2017g) North East Local Enterprise Partnership Assurance Framework (February 2017). Available at: <u>http://www.nelep.co.uk/wp-content/uploads/2016/11/North-East-LEP-Assurance-Framework-Feb-2017.pdf</u> (Accessed: 8 September 2017).

NELEP (2017h) *More and Better Jobs. The North East strategic economic plan: Executive summary, January 2017*. Newcastle: North East Local Economic Partnership (Accessed: 6 April 2018).

NELEP (2018a) North East Local Enterprise Partnership Board Agenda: Gateshead, 25 January 2018. Newcastle: North East Local Enterprise Board. [Online]. Available at: https://www.nelep.co.uk/wp-content/uploads/2018/01/january-agenda-andpapers.pdf (Accessed: 26 February 2018).

NELEP (2018b) *Our Economy 2018*. Newcastle: North East Local Enterprise Partnership. NELEP (2018c) *North East Local Enterprise Partnership Board Agenda: South Tyneside, 27 September 2018*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>https://www.nelep.co.uk/wp-content/uploads/2018/09/september-</u> <u>agenda-and-papers.pdf</u> (Accessed: 1 November 2018).

NELEP (2018d) *News: A statement from the North East Brexit Group*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at:

https://www.nelep.co.uk/news/a-statement-from-the-north-east-brexit-group/ (Accessed: 7 December 2018).

NELEP (2018e) North East Local Enterprise Partnership Board Agenda: Durham, 29 November 2018. Newcastle: North East Local Enterprise Parrtnership. [Online]. Available at: <u>https://www.nelep.co.uk/wp-content/uploads/2018/11/agenda-and-papers.pdf</u> (Accessed: 7 December 2018).

NELEP (2018f) North East Local Enterprise Partnership Assurance Framework (December 2018). Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>https://www.nelep.co.uk/wp-content/uploads/2018/12/north-east-lep-assurance-framework-dec-2018.pdf</u> (Accessed: 11 March 2018).

NELEP (2018g) *Board Member Registers of Interest: Last updated 29 March 2018*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>https://www.nelep.co.uk/wp-content/uploads/2018/04/register-of-interests-2018-19.pdf</u> (Accessed: 11 March 2019).

NELEP (2019a) *The North East Strategic Economic Plan: Creating more and better jobs*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>https://www.nelep.co.uk/the-plan</u> (Accessed: 20 February 2019).

NELEP (2019b) North East Local Enterprise Partnership Board Agenda: 31 January 2019, Ponteland. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>https://www.nelep.co.uk/wp-content/uploads/2019/01/january-agenda-</u>

and-papers.pdf (Accessed: 22 February 2019).

NELEP (2019c) North East Local Enterprise Partnership Minutes: Ponteland, 31 January 2019. Newcastle: North East Local Enterprise Partnership [Online]. Available at: <a href="https://www.nelep.co.uk/wp-content/uploads/2019/02/19.01.31-redacted-version.pdf">https://www.nelep.co.uk/wp-content/uploads/2019/02/19.01.31-redacted-version.pdf</a> (Accessed: 3 March 2019).

NELEP (2019d) North East Local Enterprise Partnership Board Agenda: Newcastle, 21 March 2019. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: https://www.nelep.co.uk/wp-content/uploads/2019/03/agenda-and-papers-march-2019.pdf (Accessed: 18 March 2019).

NELEP (2019e) *Our Economy 2019: Full report*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at: <u>https://www.nelep.co.uk/wp-</u>

<u>content/uploads/2019/05/our-economy-2019 full report north-east-local-</u> <u>enterprise-partnership.pdf</u> (Accessed: 10 May 2019).

NELEP (2020a) The North East Local Enterprise Partnership (North East LEP) has announced three new appointments to its board. Available at:

https://www.northeastlep.co.uk/news/north-east-lep-appoints-three-new-membersto-its-board (Accessed: 2 February 2020).

NELEP (undated-a) *North East Local Enterprise Partnership Constitution*. Newcastle: North East Local Enterprise Partnership. [Online]. Available at:

https://www.nelep.co.uk/wp-content/uploads/2018/04/north-east-lep-constitutionincludes-board-member-codes-of-conduct.pdf (Accessed: 3 March 2019).

NELEP (undated-b) North East LEP Governance. Available at:

http://www.nelep.co.uk/governance/ (Accessed: 17 May 2017).

NELEP (undated-c) *About Us: What is the North East Local Enterprise Partnership?* Available at: <u>https://www.nelep.co.uk/about/</u> (Accessed: 24 June 2018).

NELEP (undated-d) *Our Board: Meet our board members, view their register of interests and download meeting minutes*. Available at:

https://www.nelep.co.uk/about/board/ (Accessed: 3 March 2019).

NELEP (undated-e) *Our Progress: What do we want to achieve and how are we doing?* Available at: <u>https://www.nelep.co.uk/the-plan/our-targets</u> (Accessed: 22 February 2019).

NELEP (undated-f) *Local Growth Funding: Funding a range of strategic capital projects in the North East* Available at: <u>https://www.nelep.co.uk/funding/local-growth-funding</u> (Accessed: 8 April 2019).

NEPC (1966) *Challenge of the Changing North*. Newcastle: Northern Economic Planning Council.

NewcastleGateshead Initiative (undated) *NewcasteGateshead Initiative*. Available at: <u>http://www.ngi.org.uk/</u> (Accessed: 11 October 2019).

Newton, K. (1976) *Second City Politics: Democratic processes and decision-making in Birmingham*. Oxford: Clarendon Press.

Nomis (undated) *Official Labour Market Statistics*. Durham: Office for National Statistics. [Online]. Available at: <u>https://www.nomisweb.co.uk/default.asp</u> (Accessed: 25 March 2019).

North, D.C. (1991) 'Institutions', *Journal of Economic Perspectives*, 5(1), pp. 97-112 [Online] DOI: <u>10.1257/jep.5.1.97</u> (Accessed: 1991).

Northumberland County Council (2017) *Elections*. Available at:

<u>http://www.northumberland.gov.uk/Councillors/Vote.aspx</u> (Accessed: 23 May 2017). NRST (1977) *Strategic Plan for the Northern Region*. Newcastle: Northern Region Strategy Team (Accessed: 10 March 2017).

NTCA (2017a) *Consultation*. North Tyneside: North of Tyne Combined Authority. [Online]. Available at: <u>https://northoftynedevolution.com/consultation/</u> (Accessed: 18 December 2017).

NTCA (2017e) About North of Tyne. Available at:

https://northoftynedevolution.com/about/ (Accessed: 8 February 2018).

NTCA (2018a) *North of Tyne Consultation Report*. North Tyneside: North of Tyne Combined Authority I. [Online]. Available at: <u>https://northoftynedevolution.com/wp-content/uploads/2018/02/North-of-Tyne-devolution-consultation-report.pdf</u> (Accessed: 27 February 2017).

NTCA (2018b) *NTCA Constitution*. North Tyneside: North of Tyne Combined Authority. [Online]. Available at:

https://static1.squarespace.com/static/5bbf08bdc2ff616708156a58/t/5bf7e31a352f5 34626b7129b/1542972191368/NTCA+Constitution+v0+1.pdf (Accessed: 11 December 2018).

NTCA (2018c) Cabinet, North of Tyne Combined Authority. Available at:

https://www.northoftynedevolution.com/cabinet/ (Accessed: 11 December 2018).

NTCA (2019a) North of Tyne Combined Authority Cabinet Agenda: Newcastle, 5 February 2019. North Tyneside: North of Tyne Combined Authority. [Online]. Available at: https://static1.squarespace.com/static/5bbf08bdc2ff616708156a58/t/5c4f461e562fa7 fb256a0dd1/1548699171244/NTCA+Cabinet+5+February+2019+Agenda+Pack.pdf (Accessed: 3 March 2019).

NTCA (2019b) North of Tyne Combined Authority Cabinet Supplemental Agenda 1: Newcastle, 5 February 2019. North of Tyne Combined Authority.

NTCA (2019c) North of Tyne Combined Authority Cabinet Supplemental Agenda 2: Newcastle, 5 February 2019. North Tyneside: North of Tyne Combined Authority. [Online]. Available at:

https://static1.squarespace.com/static/5bbf08bdc2ff616708156a58/t/5c7d3119f4e1fc 6a4a8f2529/1551708442601/NTCA+Cabinet+12+March+2019%2C+Agenda+Pack.pdf (Accessed: 10 March 2019).

NTCA (2019e) North of Tyne Combined Authority Cabinet Agenda: North Tyneside, 12 March 2019. North Tyneside: North of Tyne Combined Authority. [Online]. Available at: https://www.nelep.co.uk/wp-content/uploads/2019/02/19.01.31-redacted-

version.pdf (Accessed: 10 March 2019).

NTCA (2019f) *Election of a Combined Authority Mayor for Newcastle upon Tyne, North Tyneside and Northumberland Combined Authority*. North Tyneside: North of Tyne Combined Authority Returning Officer. [Online]. Available at:

<u>https://www.northumberland.gov.uk/NorthumberlandCountyCouncil/media/Councillo</u> <u>rs-and-Democracy/Electoral%20Services/Election-of-a-Combined-Authority-Mayor.pdf</u> (Accessed: 9 April 2019).

O'Donnell, L. and Henig, S. (2016) *North East Combined Authority: Devolution agreement*. Durham: Durham County Council [Online]. Available at:

https://democracy.durham.gov.uk/documents/g7932/Public%20reports%20pack%20 Wednesday%2011-May-2016%2010.00%20Cabinet.pdf?T=10 (Accessed: 30 March 2018).

O'Donoghue, D. (2016) 'Former Mayoral hopeful Jeremy Middleton sets up North East Devolution Commission ', *ChronicleLive*edn). [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/former-mayoral-hopefuljeremy-middleton-11910846 (Accessed: 13 January 2017).

OECD (2006) OECD Territorial Reviews: Newcastle and the North East in the UK. Paris: OECD Publishing.

OECD (2015a) *Local Economic Leadership*. Paris: OECD Publishing. [Online]. Available at: <u>http://www.oecd.org/cfe/leed/OECD-LEED-Local-Economic-Leadership.pdf</u> (Accessed: 28 May 2018).

OECD (2015b) The Metropolitan Century: Understanding urbanisation and its consequences. OECD Publishing. Available at: <u>https://www.oecd-</u>

ilibrary.org/docserver/9789264228733-

en.pdf?expires=1571824824&id=id&accname=ocid194704&checksum=3851688DEC12 5602D0ADAA1F9F8158A2 (Accessed: 23 October 2019).

OECD (2016) Functional Urban Areas by Country. Available at:

http://www.oecd.org/cfe/regional-policy/functionalurbanareasbycountry.htm (Accessed: 3 December 2018).

OECD (2018) Regional Demography. Available at:

https://stats.oecd.org/Index.aspx?DataSetCode=REGION\_DEMOGR (Accessed: 15 December 2018).

Ofsted (2018) Changes to Ofsted's Statistical reporting of Inspection Outcomes for State-funded Schools: Data, charts and tables. London: Ofsted. [Online]. Available at: https://www.gov.uk/government/statistics/changes-to-ofsteds-statistical-reporting-ofinspection-outcomes-for-state-funded-schools-an-analysis-of-the-changes (Accessed: 7 April 2019).

ONE (1999) *Unlocking Our Potential: Regional economic strategy for the North East.* Newcastle: One North East.

ONE (2006) *Leading the Way: Regional economic strategy 2006-2016*. Newcastle: One North East. [Online]. Available at:

http://webarchive.nationalarchives.gov.uk/20090302144023/http:/www.onenortheas t.co.uk/lib/liReport/9653/Regional%20Economic%20Strategy%202006%20-2016.pdf (Accessed: 7 January 2017).

ONE (2011) *Annual Report and Accounts 2010-11*. London: One North East. [Online]. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attach ment\_data/file/247367/1166.pdf (Accessed: 29 May 2018).

ONE (2012) Annual Report and Accounts 2011-2012. Newcastle: One North East ONS (2016) Travel-to-work Areas by Qualification: High. Newport: Office for National Statistics. [Online]. Available at:

https://ons.maps.arcgis.com/home/webmap/viewer.html?useExisting=1&layers=a26af 939ffb8419aa6069f9083d9f5a0&layerId=0 (Accessed: 29 December 2018).

ONS (2017a) *L104 Regional Labour Market: Local indicators for NUTS3 areas*. Newport: Office for National Statistics. [Online]. Available at:

https://www.ons.gov.uk/employmentandlabourmarket/peopleinwork/employmentan demployeetypes/datasets/locallabourmarketindicatorsfornuts3areasli04/current (Accessed: 25 March 2019).

ONS (2017b) *Regional Gross Value Added (Balanced) by Local Authority in the UK.* Newport: Office for National Statistics. [Online]. Available at:

https://www.ons.gov.uk/economy/grossvalueaddedgva/datasets/regionalgrossvaluea ddedbalancedbylocalauthorityintheuk (Accessed: 12 April 2019).

ONS (2018a) Wealth in Great Britain Wave 5: 2014 to 2016. Available at:

https://www.ons.gov.uk/peoplepopulationandcommunity/personalandhouseholdfina nces/incomeandwealth/bulletins/wealthingreatbritainwave5/2014to2016 (Accessed: 21 August 2019).

ONS (2018b) *Estimates of the Population of the UK, England and Wales, Scotland and Northern Ireland*. Newport: Office for National Statistics. [Online]. Available at:

https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populationestimates/datasets/populationestimatesforukenglandandwalesscotlandandno rthernireland (Accessed: 7 April 2019).

ONS (2018c) *Regional Household Final Consumption Expenditure*. Newport: Office for National Statistics. [Online]. Available at:

https://www.ons.gov.uk/economy/regionalaccounts/grossdisposablehouseholdincom e/datasets/regionalhouseholdfinalconsumptionexpenditure (Accessed: 23 December 2018).

ONS (2018d) Business Demography, UK 2017: Change in the number of UK businesses broken down by sector of the economy. Newport: Office for National Statistics. [Online]. Available at:

https://www.ons.gov.uk/businessindustryandtrade/business/activitysizeandlocation/b ulletins/businessdemography/2017#which-regions-have-the-highest-business-birthsand-deaths (Accessed: 5 April 2019). Ord, M. (2016) 'North East Devolution Commission launches Youth Employment Policy', *Insider Media Limited*edn), 3 October 2016. [Online] Available at: https://www.insidermedia.com/northeast (Accessed: 17 June 2017).

Osborne, G. (2014) *Chancellor: We need a Northern Powerhouse*. London: HM Treasury. [Online]. Available at:

https://www.gov.uk/government/speeches/chancellor-we-need-a-northernpowerhouse (Accessed: 14 July 2017).

Overman, H. (2012) 'Investing in the UK's most successful Cities is the surest Recipe for National Growth', *LSE Research Online*. Available at: <u>http://eprints.lse.ac.uk/44073/</u> (Accessed: 2 January 2017).

Page, W. (ed.) (1891) *Three Early Assize Rolls of the County of Northumberland*. Durham: The Surtees Society.

Parliament (2018a) *Register of Members' Financial Interests as at 5 February 2018: Twist, Liz (Blaydon)*. London: Parliament. [Online]. Available at:

https://publications.parliament.uk/pa/cm/cmregmem/180205/twist\_liz.htm.

Parsons, R. (2019a) 'Exclusive: Greater Manchester mayor's bid for £6bn underground station could delay arrival of high speed rail in the North ', *Yorkshire Post*edn), 9 August 2019. [Online] Available at: <u>https://www.yorkshirepost.co.uk/news/uk/exclusive-greater-manchester-mayor-s-bid-for-6bn-underground-station-could-delay-arrival-of-high-speed-rail-in-the-north-1-9927112</u> (Accessed: 11 August 2019).

Parsons, R. (2019c) 'Ministers must concentrate on towns and not just Manchester if the North is to succeed, says metro mayor Ben Houchen', *Yorkshire Post*edn), 2 October 2019. [Online] Available at:

https://www.yorkshirepost.co.uk/news/politics/ministers-must-concentrate-ontowns-and-not-just-manchester-if-the-north-is-to-succeed-says-metro-mayor-benhouchen-1-10028062 (Accessed: 3 October 2019).

Pearce, G. (2008) 'Institutional Capacity in the English Regions', in Bradbury, J. (ed.) *Devolution, Regionalism and Regional Development*. London: Routledge, pp. 97-116. Pickard, J., Bounds, A. and Tighe, C. (2019) 'Transport to be devolved to mayors in northern England', *Financial Times*edn), 13 September 2019. [Online] Available at: <u>https://www.ft.com/content/a3cc221a-d605-11e9-a0bd-</u>

ab8ec6435630?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

2437841d178a#myft:notification:instant-email:content (Accessed: 14 September 2019).

Pike, A., Dawley, S. and Tomaney, J. (2010) 'Resilience, Adaptation and Adaptability', *Cambridge Journal of Regions, Economies and Societies*, 3, pp. 59-70 [Online] DOI: 10.1093/cjres/rsq001

Pike, A., Marlow, D., O'Brien, P. and Tomaney, J. (2015) 'Local Institutions and Local Economic Development: The local enterprise partnerships in England, 2010–', *Cambridge Journal of Regions, Economies and Societies*, 8(2), pp. 185-204 [Online] DOI: 10.1093/cjres/rsu030.

Pike, A., Rodriguez-Pose, A. and Tomaney, J. (2017) *Local and Regional Development*. Second edn. Oxford: Routledge.

Pike, A., Rodríguez-Pose, A., Tomaney, J., Torrisi, G. and Tselios, V. (2012) 'In search of the 'Economic Dividend' of Devolution: Spatial disparities, spatial economic policy, and decentralization in the UK', *Environment and Planning C: Government and Policy*, 30, pp. 10-28 [Online] DOI: 10.1068/c10214r.

Price, K. (2017) 'Mayor uses Moment in Spotlight to sell SSI site', *GazetteLive*edn), 2 October 2017. [Online] Available at: <u>http://www.gazettelive.co.uk/news/teesside-news/mayor-uses-moment-spotlight-sell-13707623</u> (Accessed: 3 October 2017).

Price, K. (2018a) 'It's Official: Teesside workers are more productive than Geordies', *Gazette Live* edn), 12 March 2018. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/its-official-teesside-workersmore-14401063 (Accessed: 14 March 2018).

Price, K. (2018b) 'Wave of Jobs as Energy Plant announced for South Tees site', *Teesside Live*edn), 24 July 2018. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/wave-teesside-jobs-energy-plant-14943711?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content= EM\_GazetteLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story2&utm\_campaign=daily \_\_newsletter (Accessed: 25 July 2018).

Price, K. (2019a) 'Teesside woud take 'immediate' £57m hit and lose 1,100 jobs if Airport closes, Mayor claims', *TeessideLive*edn), 10 January 2019. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/teesside-would-take-immediate-57m-15658009</u> (Accessed: 11 January 2019).

Price, K. (2019b) "Thai Banks are taking us for a Ride' says Mayor as deal for SSI Land collapses', *TeessideLive*edn), 15 March 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/thai-banks-taking-ride-says-15975640?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content= EM\_GazetteLive\_Nletter\_News\_Home\_largeteaser\_Text\_Story2&utm\_campaign=daily \_\_newsletter (Accessed: 15 March 2019).

Prud'homme, R. (1995) *On the Dangers of Decentralisation* (2). The World Bank Research Observer. [Online]. Available at:

http://documents.worldbank.org/curated/en/218141468739288067/pdf/multipage.pdf (Accessed: 3 April 2019).

PSA (2016) Political Studies Association Research Commission: Examining the role of 'informal governance on devolution to England's cities London: Political Studies Association. [Online]. Available at: <u>https://www.psa.ac.uk/sites/default/files/page-</u> files/PSA%20Informal%20Governance%202016.pdf (Accessed: 26 October 2017).

PwC (2016) Bringing the Citizen View into the Devolution Debate. London: PwC. [Online]. Available at: <u>https://www.pwc.co.uk/industries/government-public-sector/what-does-the-public-want-from-devolution.html</u> (Accessed: 8 November 2017).

Queen (2019b) The Queen's Speech December 2019: Her Majesty's most gracious speech to both Houses of Parliament London: UK Government,. [Online]. Available at: <a href="https://www.gov.uk/government/speeches/queens-speech-december-2019">https://www.gov.uk/government/speeches/queens-speech-december-2019</a> (Accessed: 28 May 2020).

Quinn, M. (2013) 'New Labour's Regional Experiment: Lessons from the East Midlands', *Local Economy*, 28(7-8), pp. 738-751 [Online] DOI: 10.1177/0269094213500124. Raco, M. (1999) 'Researching the New Urban Governance: An examination of closure, access and complexities of institutional research', *Area*, 31(3), pp. 271-279. Raikes, L. (2020) *The Devolution Parliament*. Manchester: Institute for Public Policy Research Research, I.f.P.P. [Online]. Available at: <u>https://www.ippr.org/files/2020-02/the-devolution-parliament-feb-20-summary.pdf</u> (Accessed: 19 May 2020). Rallings, C. and Thrasher, M. (2006) ''Just another expensive talking shop': Public attitudes and the 2004 regional assembly referendum in the north east of England', *Regional Studies*, 40(8), pp. 927-936 [Online] DOI: 10.1080/00343400600929069. Ramuni, L. (2019) 'Fast Growth Cities: Why one size rarely fits all in urban policy making', *Centre for Cities*. Available at: <u>https://www.centreforcities.org/blog/fast-growth-cities-why-one-size-rarely-fits-all-in-urban-policy-</u>

making/?utm\_source=Centre+for+Cities+Newsletter&utm\_campaign=63296a4a5f-EMAIL\_CAMPAIGN\_2018\_12\_19\_09\_47\_COPY\_01&utm\_medium=email&utm\_term=0 2a9c9d5ef9-63296a4a5f-156019569&mc\_cid=63296a4a5f&mc\_eid=333f9bcecf

(Accessed: 28 January 2019).

Redcar & Cleveland Council (2017b) *SSI Task Force: Legacy report*. Redcar: Redcar & Cleveland Council. [Online]. Available at: <u>https://www.redcar-</u>

<u>cleveland.gov.uk/SSITaskForce/Documents/SSI%20Task%20Force%20Legacy%20Repor</u> t%20Two%20Years.pdf (Accessed: 1 January 2019).

Rees, J. and Lord, A. (2013) 'Making Space: Putting politics back where it belongs in the construction of city regions in the north of England', *Local Economy*, 28(7-8), pp. 679-695 [Online]. Available at: <u>https://www.scopus.com/inward/record.uri?eid=2-s2.0-</u>84887549272&partnerID=40&md5=b9a264e7d5037c0a03315566576ae299,

https://www.scopus.com/inward/record.uri?eid=2-s2.0-

Rescher, N. (2005) 'Pragmatism', in Honderich, T. (ed.) *The Oxford Companion to Philosophy*. Oxford: Oxford University Press, p. 748.

Rhodes, R.A.W. (1996) 'The New Governance: Governing without government', *Political Studies*, 44(4), pp. 652-667 [Online] DOI: 10.1111/j.1467-9248.1996.tb01747.x.

Rhodes, R.A.W. (2007) 'Understanding Governance: Ten years on', *Organization Studies*, 28(8), pp. 1243-1264 [Online] DOI:

http://oss.sagepub.com/content/28/8/1243.full.pdf+html.

Robinson, F. (2002) 'The North East: A journey through time', *City*, 6(3), pp. 317-334 [Online] DOI:

http://www.tandfonline.com/doi/pdf/10.1080/1360481022000037751?needAccess=tr ue

Robinson, F., Shaw, K. and Regan, S. (2017) *Who Runs the North East Now? Governance and governing in an English region*. Durham: Northumbria University and St Chad's College Durham University. [Online]. Available at:

https://www.stchads.ac.uk/wp-content/uploads/2015/12/Who-Runs-the-North-East-Now-Main-Report-Oct-2017-FINAL-09-10-17-2.pdf (Accessed: 21 October 2017).

Robinson, J. (2016a) North East Combined Authority: Proposed devolution agreement and elected regional mayor. Gateshead: Gateshead Council [Online]. Available at: http://democracy.gateshead.gov.uk/documents/g1146/Public%20reports%20pack%20

22nd-Mar-2016%2010.00%20Cabinet.pdf?T=10 (Accessed: 30 March 2018).

Robinson, J. (2016b) North East Combined Authority: Devolution and governance update Gateshead: Gateshead Council [Online]. Available at:

http://democracy.gateshead.gov.uk/documents/g1101/Public%20reports%20pack%20 12th-Jul-2016%2010.00%20Cabinet.pdf?T=10 (Accessed: 1 April 2018).

Robson, D. (2018) 'Airline KLM replaces Geordie-Land picture for Flights to Teesside...with image of Durham', *TeessideLive*edn), 18 November 2018. [Online] Available at: <u>https://www.gazettelive.co.uk/news/teesside-news/airline-klm-replaces-geordie-land-</u>

<u>15432241?utm\_source=gazette\_live\_newsletter&utm\_medium=email&utm\_content=</u> <u>EM\_GazetteLive\_Nletter\_News\_Home\_smallteaser\_Text\_Story6&utm\_campaign=daily\_newsletter</u> (Accessed: 19 November 2018). Robson, D. (2019) 'The 'robust' reason Donald Trump turned down invitation to Teesside', *TeessideLive*edn), 5 June 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/robust-reason-donald-trumpturned-16381771 (Accessed: 5 June 2019).

Rodriguez-Pose, A. (2013) 'Do Institutions Matter for Regional Development?', *Regional Studies*, 47(7), pp. 1034-1047 [Online] DOI: 10.1080/00343404.2012.748978 Rodríguez-Pose, A. and Gill, N. (2003) 'The Global Trend towards Devolution and its Implications', *Environment and Planning C: Government and Policy*, 21(3), pp. 333-351 [Online] DOI: 10.1068/c0235 (Accessed: 26 June 2019).

Rodríguez-Pose, A. and Sandall, R.B. (2008) 'From Identity to the Economy: Analysing the evolution of the decentralisation discourse', *Environment and Planning C:* 

Government and Policy, 26(1), pp. 54-72 [Online] DOI: 10.1068/cav2. Rogers, A., Castree, N. and Kitchin, R. (2013) Oxford Dictionary of Human Geography.

Oxford University Press. Available at:

https://www.oxfordreference.com/view/10.1093/acref/9780199599868.001.0001/acr ef-9780199599868-e-1686 (Accessed: 10 May 2020).

Rogers, M. (2018) *Governing England: Devolution and Public Services*. Manchester: Association for Public Service Excellence. [Online]. Available at:

https://www.apse.org.uk/apse/index.cfm/news/articles/20181/governing-englanddevolution-and-public-services/ (Accessed: 21 April 2019).

Rowell, C., <u>http://www.recognitionpr.co.uk/clients/id/19959</u> (2016) 'North East Devolution Commission launched'. Darlington: Middleton Enterprises. Available at: <u>http://www.recognitionpr.co.uk/clients/id/19959</u> (Accessed: 17 June 2017).

Sandford, M. (2013) *The Abolition of Regional Government* (SN/PC/05842). London: House of Commons Library. [Online]. Available at:

https://commonslibrary.parliament.uk/research-briefings/sn05842/ (Accessed: 27 May 2020).

Sandford, M. (2016a) *Devolution to Local Government in England* (07029). London: House of Commons Library. [Online]. Available at:

http://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN07029 (Accessed: 11 December 2016).

Sandford, M. (2016b) *Combined Authorities* (06649). London: House of Commons Library. [Online]. Available at:

http://researchbriefings.files.parliament.uk/documents/SN06649/SN06649.pdf (Accessed: 11 December 2016).

Sandford, M. (2017a) *Directly-elected Mayors* (05000). London: House of Commons Library. [Online]. Available at:

http://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN05000#fullrepor t (Accessed: 19 June 2017).

Sandford, M. (2017b) 'Signing up to Devolution: The prevalance of contract over governance in English devolution', *Regional and Federal Studies*, 27(1), pp. 63-82 [Online] DOI: 10.1080/13597566.2016.1254625.

Sandford, M. (2019a) 'Money Talks: The Finances of English Combined Authorities', *Local Economy*, 34(2), pp. 106-122 [Online] DOI: 10.1177/0269094219839609 Sandford, M. (2019b) *Reviewing and Reforming Local Government Finance* (07538). London: House of Commons Library. [Online]. Available at:

https://www.gov.uk/government/consultations/business-rates-retention-reform (Accessed: 5 April 2019).

Sandford, M. (2019c) *Devolution to Local Government in England* (07029). London: House of Commons Library. [Online]. Available at:

https://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN07029#fullrepo rt (Accessed: 6 December 2019).

Sayer, A. (1992) *Method in Social Science: A realist approach*. Routledge. Available at: <u>http://www.tandfebooks.com/isbn/9780203163603</u> (Accessed: 21 June 2019).

Schoenberger, E. (1992) 'Self-criticism and Self-awareness: A reply to Linda McDowell', *Professional Geographer*, 44(2), pp. 215-218.

Scott, A.J. (2019) 'City Regions Reconsidered', *Economy and Space A*, 51(3), pp. 554-580 [Online] DOI: 10.1177/0308518X19831591.

Seddon, S. (2016) 'North East devolution doubts after Gateshead Council rejected the proposals', *ChronicleLive*edn), 22 March 2016. [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/north-east-devolution-doubtsafter-11079111 (Accessed: 8 February 2017).

Seddon, S. (2017) 'Durham County Council local elections results 2017 in full', *ChronicleLive*edn), 5 May 2017. [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/durham-county-council-localelection-12994807 (Accessed: 23 May 2017).

Seddon, S. (2018a) 'One of these four men will be Labour's North of Tyne mayor candidate', *ChronicleLive*edn), 28 November 2018. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/north-tyne-mayor-labourcandidate-15473670 (Accessed: 29 November 2018).

Seddon, S. (2018b) 'Labour in disarray over North of Tyne mayor race as party bosses demand fresh blood', *ChronicleLive*edn), 12 December 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/labour-north-tyne-mayor-election-15533427</u> (Accessed: 12 December 2018).

Seddon, S. (2018c) 'Parliamentary logjam could derail North of Tyne devolution, Newcastle leader warns', *ChronicleLive*edn), 22 July 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/parliamentary-logjam-could-</u> derail-north-14939135 (Accessed: 2018-07-22).

Seddon, S. (2019b) 'Labour candidate for North of Tyne mayor unveiled', *ChronicleLive*edn), 20 February 2019. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/labour-candidate-north-tynemayor-

15857819?utm source=chronicle live newsletter&utm medium=email&utm content =EM ChronicleLive Nletter News Home smallteaser Text Story2&utm campaign=d aily newsletter (Accessed: 20 February 2019).

Seddon, S. (2019c) 'Who is Jamie Driscoll? Labour's left-wing North of Tyne mayoral candidate', *ChronicleLive*edn), 21 February 2019. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/jamie-driscoll-north-tynemayor-15860074 (Accessed: 5 March 2019).

Seddon, S. (2019d) 'Labour's Jamie Driscoll becomes first North of Tyne Mayor and sets out socialist agedna ', *ChronicleLive*edn), 3 May 2019. [Online] Available at:

https://www.chroniclelive.co.uk/news/labours-jamie-driscoll-becomes-first-16224168 (Accessed: 4 May 2019).

Selbie, D., Bhargava, A., Black, C., Whiteman, R. and Wright, T. (2016) *Health and Wealth: Closing the gap in the north east*. North Tyneside: North East Combined Authority and National Health Service. [Online]. Available at:

http://www.northeastca.gov.uk/sites/default/files/file\_attachments/Health%20and%2

OWealth%20Closing%20the%20Gap%20in%20the%20North%20East%20-

<u>%20Executive%20Summary.pdf</u> (Accessed: 14 May 2017).

Shaw, K. and Robinson, F. (2018) 'Whatever happened to the North East? Reflections on the end of regionalism in England', *Local Economy*, 33(8) [Online] DOI: 10.1177/0269094218819789

Shipley, J. (2018) Newcastle upon Tyne, North Tyneside and Northumberland Combined Authority (Establishment and Functions) Order 2018 (Vol. 793). London: Hansard. [Online]. Available at: <u>https://hansard.parliament.uk/Lords/2018-10-</u>

30/debates/2531A764-E9F9-4D93-B63D-

<u>4EB11B438049/NewcastleUponTyneNorthTynesideAndNorthumberlandCombinedAut</u> <u>hority(EstablishmentAndFunctions)Order2018?highlight=gateshead#contribution-</u> 122F1DFD-4634-4B86-9285-2ABB2E80B265 (Accessed: 1 November 2018).

Shutt, J. and Liddle, J. (2019) 'Are combined authorities in England strategic and fit for purpose?', *Local Economy*, 34(2), pp. 196-207 [Online] DOI:

10.1177/0269094219839956

Sissons, P., Green, E.E. and Broughton, K. (2018) 'Inclusive Growth in English Cities', *Regional Studies*, 53(3), pp. 435-446 [Online] DOI: 10.1080/00343404.2018.1515480. Slack, E. and Cote, A. (2014) *Future of Cities: Comparative urban governance*. London: Government Office for Science. [Online]. Available at:

https://www.gov.uk/government/publications/future-cities-comparative-urbangovernance (Accessed: 26 December 2016).

Smiles, M. (2017) 'Could it soon be Illegal to make a Parmo anywhere other than Teesside?', *GazetteLive*edn), 29 March 2017. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/could-soon-illegal-make-parmo-12808825 (Accessed: 4 April 2018).

Smith, K.E. (2005) 'Problematising power relations in 'elite' interviews', *Geoforum*, 37(4), pp. 643-653.

Snyder, R. (2001) 'Scaling Down: The subnational comparative method', *Studies in Comparative International Development*, 36, pp. 93-110 [Online] DOI: 10.1007/BF02687586.

South Tyneside Council (2016a) *South Tyneside Cabinet Minutes: 24 March 2016*. South Shields: South Tyneside Council. [Online]. Available at:

https://www.southtyneside.gov.uk/article/60220/Committee-

meeting?formid=147785&pageSessionId=1efae3a1-53b2-45a4-bf24-

<u>9476d769c87b&fsn=29f08f0f-7b85-413f-8a7b-382ef455299b</u> (Accessed: 17 February 2018).

South Tyneside Council (2016b) *South Tyneside Cabinet Minutes: 17 May 2016*. South Shields: South Tyneside Council [Online]. Available at:

https://www.southtyneside.gov.uk/article/60220/Committee-

meeting?formid=147785&pageSessionId=57fc7506-c466-4421-9246-

<u>550731771353&fsn=688d6023-e132-447c-9e62-6232168dd7fa</u> (Accessed: 17 February 2018).

South Tyneside Council (2016c) *South Tyneside Cabinet Minutes: 26 August 2016* South Shields: South Tyneside Council [Online]. Available at:

https://www.southtyneside.gov.uk/article/60220/Committee-

meeting?formid=147785&pageSessionId=5e591246-2194-4cd8-b996-

<u>38cbca9a0f0c&fsn=2b619068-9305-45bf-beca-df08c2cd1eb2</u> (Accessed: 17 February 2018).

Stake, R.E. (2000) 'Case Studies', in Denzin, N. and Lincoln, Y. (eds.) *Handbook of Qualitative Research*. Second edn. Thousand Oaks: Sage, pp. 435-454.

STDC (2019) South Tees Regeneration Master Plan: Executive summary. Stockton: South Tees Development Corporation, Tees Valley Development Corporation and Tees Valley Mayor. [Online]. Available at: <u>https://www.southteesdc.com/wp-</u> content/uploads/2019/04/Summary-Masterplan-March-2019-LowResolution.pdf

(Accessed: 29 October 2019).

Stockton Council (2014) *Local Economic Assessment 2014*. Stockton: Stockton Council. [Online]. Available at: <u>https://www.stockton.gov.uk/media/3636/local-economic-assessment-2014.pdf</u> (Accessed: 14 December 2017).

Stockton Council (2017) *Tees Valley Combined Authority Mayoral Election Results* 2017. Available at: <u>https://www.stockton.gov.uk/news/2017/may/tees-valley-</u>combined-authority-mayoral-election-results-2017/ (Accessed: 12 June 2017).

Stoker, G. (1998) 'Governance as Theory: Five propositions', *International Social Science Journal*, 50(155), pp. 17-28 [Online] DOI: 10.1111/1468-2451.00106 Stoker, G., Gains, F., Greasley, S., John, P. and Rao, N. (2004) *Operating the New Council Constitutions in English Local Authorities: A process evaluation*. London: Office of the Deputy Prime Minister. [Online]. Available at:

<u>https://www.bipsolutions.com/docstore/pdf/7794.pdf</u> (Accessed: 16 February 2019). Stoker, G. and Mossberger, K. (1995) 'The post-Fordist Local State: The dynamics of its development', in Stewart, J. and Stoker, G. (eds.) *Local Government in the 1990s*. Basingstoke: Macmillan, pp. 210-227.

Storper, M., Kemeney, T., Makarem, N.P. and Osman, T. (2015) *The Rise and Fall or Urban Economies: Lessons from San Fransisco and Los Angeles*. Stanford: Stanford University Press.

Sunderland Assocation Football Club (undated) *Supporters Branches*. Available at: <u>https://www.safc.com/fans/supporters-branches/uk-branches/uk-branches-1</u> (Accessed: 27 October 2019).

Sunderland Council (2016a) *Sunderland Cabinet Minutes: 23 March 2016*. Sunderland: Sunderland City Council. [Online]. Available at:

https://www.sunderland.gov.uk/committees/cmis5/Meetings/tabid/73/ctl/ViewMeetingPublic/mid/410/Meeting/8598/Committee/1890/Default.aspx (Accessed: 16 February 2018).

Sunderland Council (2016b) *North East Combined Authority: Proposed devolution agreement and elected regional mayor*. Sunderland: Sunderland City Council. [Online]. Available at:

https://www.sunderland.gov.uk/committees/cmis5/Meetings/tabid/73/ctl/ViewMeetingPublic/mid/410/Meeting/8598/Committee/1890/Default.aspx (Accessed: 16 February 2018).

Sunderland Council (2016c) *Sunderland Cabinet Minutes: 16 May 2016*. Sunderland: Sunderland City Council. [Online]. Available at:

https://www.sunderland.gov.uk/committees/cmis5/Meetings/tabid/73/ctl/ViewMeetingPublic/mid/410/Meeting/8890/Committee/1890/Default.aspx (Accessed: 17 February 2018).

Sunderland Council (2016d) *Sunderland Cabinet: Record of executive decisions*. Sunderland: Sunderland City Council. [Online]. Available at:

https://www.sunderland.gov.uk/committees/cmis5/Meetings/tabid/73/ctl/ViewMeetingPublic/mid/410/Meeting/9111/Committee/1953/Default.aspx (Accessed: 17 February 2017).

Sunderland Council (2018a) *Royal Boost for Northern Spire Bridge*. Available at: <u>https://www.makeitsunderland.com/about</u> (Accessed: 19 March 2018). Sunderland Council (2018b) *IAMP*. Available at:

https://www.makeitsunderland.com/investment-hotspots/iamp (Accessed: 29 December 2018).

Sunley, P., Martin, R. and Tyler, P. (2017) 'Cities in Transition: Problems, Processes and Policies', *Cambridge Journal of Regions, Economy and Society*, 10(3), pp. 383-390 [Online] DOI: 10.1093/cjres/rsx018.

Swales, M. (2016) *Devolution: A report to consider devolution proposals for the North East Combined Authority* South Shields: South Tyneside Council [Online]. Available at: <u>https://www.southtyneside.gov.uk/article/60220/Committee-</u>

<u>meeting?formid=147785&pageSessionId=4f8f243f-b9a2-42e4-a7f4-</u> 06cf4c23550a&fsn=64a355af-0a57-4c1f-9d50-4c911d8ee276 (Accessed: 30 March

2018).

Swinney, P. (2017) 'North of Tyne Devolution: A victory for pragmatism over perfection', *Centre for Cities*. Available at: <u>https://www.centreforcities.org/blog/north-tyne-devolution-victory-pragmatism-perfection/</u> (Accessed: 28 January 2019).

Tees Valley OSC (2019) *Report of the Chair of the Overview and Scrutiny Committee*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/06/7-OS-annual-report-2018-19.pdf</u> (Accessed: 8 July 2019).

TeessideLive (2019) 'Teesside Airport deal is a risk worth taking but the real work starts now', *TeessideLive*edn), 23 January 2019. [Online] Available at:

https://www.gazettelive.co.uk/news/teesside-news/teesside-airport-deal-risk-worth-15716761 (Accessed: 23 January 2019).

TfN (2018) *Gaining statutory status*. Available at:

<u>https://transportforthenorth.com/about-transport-for-the-north/sub-national-transport-body/</u> (Accessed: 7 December 2018).

TfN (undated) *Our Board*. Available at: <u>https://transportforthenorth.com/about-</u>transport-for-the-north/our-board/ (Accessed: 7 November 2019).

Tighe, C. (2019a) 'Why a Conservative mayor is trying to take over a local airport', *Financial Times Group*edn), 16 January 2019. [Online] Available at:

https://www.ft.com/content/62cd391c-0459-11e9-9d01-

cd4d49afbbe3?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

2437841d178a#myft:notification:instant-email:content (Accessed: 16 January 2019). Tighe, C. (2019b) 'Corbyn supporter wins Labour nomination for north-east metro mayor', *Financial Times*edn), 20 February 2019. [Online] Available at:

https://www.ft.com/content/8ae2edc0-3523-11e9-bd3a-

8b2a211d90d5?desktop=true&segmentId=d8d3e364-5197-20eb-17cf-

2437841d178a#myft:notification:instant-email:content (Accessed: 20 February 2019).

Tighe, C. and Sheppard, D. (2018) 'Clean technology offers hope for Teesside transformation', *Financial Times*edn), 23 December 2018. [Online] Available at: <u>https://www.ft.com/content/c9b0abc4-f6f2-11e8-af46-2022a0b02a6c</u> (Accessed: 25 December 2018).

Tomaney, J. (2006) 'North East England: A brief economic history', *NERIP Research Conference*. Newcastle, 6 September 2006. Newcastle: North East Research and Information Partnership. Available at:

http://www.nerip.com/reports\_briefing.aspx?id=241 (Accessed: 16 July 2017).

Tomaney, J. (2014a) 'Region and Place 1: Institutions', *Progress in Human Geography*, 38(1), pp. 131-140 [Online] DOI: 10.1177/0309132513493385.

Tomaney, J. (2014b) Reappraising Labour's Loss in the North East Elected Regional Assembly Referendum: Was the 2004 'no' vote a critical juncture in the long-term trajectory of regionalism in the north east's public administration? London: London School of Economics. [Online]. Available at:

http://www.lse.ac.uk/government/degreeProgrammes/programmes/undergraduate/p df/GV390-Joseph-Anderson.pdf (Accessed: 8 December 2016).

Tomaney, J. (2016) 'Limits of Devolution: Localism, economics and post-democracy', *Political Quarterly*, 87(4), pp. 546-552 [Online] DOI: 10.1111/1467-923X.12280. Torfing, J. (2007) 'Introduction: Democratic network governance', in Marcussen, M. and Torfing, J. (eds.) *Democratic Network Governance in Europe*. Basingstoke, Hants.: Palgrave Macmillan, pp. 1-22.

Torr, G. (2018) "If this was a Leeds takeover I wouldn't touch it with a Barge Pole': Mayor Dan Jarvis on future Yorkshire devolution proposals ', *Sheffield Staredn*), 17 August 2018. [Online] Available at: <u>https://www.thestar.co.uk/news/if-this-was-aleeds-takeover-i-wouldn-t-touch-it-with-a-barge-pole-mayor-dan-jarvis-on-futureyorkshire-devolution-proposal-1-9307467 (Accessed: 25 February 2019).</u>

Treasury (2011) *Public Expenditure Statistical Analyses 2011*. London: HM Treasury. [Online]. Available at: <u>https://www.gov.uk/government/statistics/public-expenditure-statistical-analyses-2011</u> (Accessed: 18 October 2017).

Treasury (2013) *Investing in Britain's Future*. London: HM Treasury. [Online]. Available at: <u>https://www.gov.uk/government/publications/investing-in-britains-future</u> (Accessed: 8 May 2017).

Treasury (2015) North East Devolution Agreement London. [Online]. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment\_data/file/472 187/102915 DEVOLUTION TO THE NORTH EAST signed\_pdf.pdf (Accessed: 6 December 2016).

Tsebelis, G. (1999) 'Veto Players and Law Production in Parliamentary Democracies: An empirical analysis', *Americal Political Science Review*, 93(3), pp. 591-608 [Online]. Available at:

https://search.proquest.com/docview/214235461?OpenUrlRefId=info:xri/sid:primo&a ccountid=12753,

https://search.proquest.com/docview/214235461?OpenUrlRefId=info:xri/sid:primo&a ccountid=12753.

TVCA (2016a) *Tees Valley Strategic Economic Plan: An industrial strategy for Tees Valley 2016-2026*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <a href="https://teesvalley-ca.gov.uk/wp-content/uploads/2016/12/TVCA207-SEP-Document-full-WEB.pdf">https://teesvalley-ca.gov.uk/wp-content/uploads/2016/12/TVCA207-SEP-Document-full-WEB.pdf</a> (Accessed: 16 July 2017).

TVCA (2016b) *Tees Valley Combined Authority Board: Business meeting minutes: Middlesbrough, 7 June 2016*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Minutes-</u> <u>TVCA-Board-7th-June-2016-Business-Meeting.pdf</u> (Accessed: 2 March 2017).

TVCA (2016c) *Tees Valley Combined Authority Board Minutes: Stockton, 24 August 2016*. Stockton: Tees Valley Combined Authority. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Minutes-TVCA-Board-24th-August-2016.pdf (Accessed: 2 March 2017).

TVCA (2016d) Summary Report of the Outcomes of the Consultation on the Powers of the Tees Valley Mayor and Mayoral Combined Authority. Stockton: Tees Valley

Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/11/Summary-consultation-report-amended-after-CLG-comments-27-Sept-16.pdf</u> (Accessed: 18 June 2017).

TVCA (2016e) *Tees Valley Overview and Scrutiny Committee Agenda: Stockton, 2 October 2016.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/TVCA-Overview-and-</u> <u>Scrutiny-Committee-Agenda-and-Papers-12th-October-2016.pdf</u> (Accessed: 19 August 2017).

TVCA (2017a) *Tees Valley Combined Authority Board Agenda: Stockton,13 January 2017.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <a href="https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/TVCA-Board-Agenda-https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/TVCA-Board-Agenda-https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/TVCA-Board-Agenda-

Papers-130117.pdf (Accessed: 2 March 2017).

TVCA (2017b) *Tees Valley Overview and Scrutiny Committee Agenda: Redcar, 17 October 2017.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2017/05/Overview-Scrutiny-17th-October-2017.pdf</u> (Accessed: 13 October 2017).

TVCA (2017c) Intermediate Body Status for Tees Valley Combined Authority. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-</u> ca.gov.uk/wp-content/uploads/2017/05/10-IB-Status-Report.pdf (Accessed: 15

January 2018).

TVCA (2017d) *Tees Valley Combined Authority Cabinet Minutes: Stockton, 13 June 2017.* Stockton: Tees Valley Combined Authority. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2017/05/Item-3-Cabinet-Minutes-13th-June-2017.pdf (Accessed: 26 February 2018).

TVCA (2017e) *Tees Valley Combined Authority Cabinet Minutes: Stockton, 23 November 2017*. Stockton: Tees Valley Combined Authority. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2017/12/3-Cabinet-Minutes-231117.pdf (Accessed: 26 February 2018).

TVCA (2017f) *Statement of Accounts 2016-17*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2016/03/TVCA-Statement-of-Accounts-2016-17-Web.pdf</u> (Accessed: 16 March 2019).

TVCA (2017g) *Tees Valley Overview and Scrutiny Committee Minutes: Redcar, 17 October 2017.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2017/05/Minutes-17th-October.pdf</u> (Accessed: 6 March 2019).

TVCA (2017h) *Tees Valley Audit and Governance Committee Agenda: Stockton, 5* September 2017. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/meetings/archive-papers-and-minutes/audit-governance-committee-archive/2017-18/</u> (Accessed: 6 March 2019).

TVCA (2018a) *Tees Valley Combined Authority (and Group): Annual financial statement 2017/18*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <a href="https://teesvalley-ca.gov.uk/wp-content/uploads/2018/07/Statement-of-Accounts-">https://teesvalley-ca.gov.uk/wp-content/uploads/2018/07/Statement-of-Accounts-</a>

2017-18-GROUP-Signed.pdf (Accessed: 16 March 2019).

TVCA (2018b) *Tees Valley Combined Authority Cabinet Decision Record: 16 February 2018.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <a href="https://teesvalley-ca.gov.uk/wp-content/uploads/2018/02/Decision-Records-16-">https://teesvalley-ca.gov.uk/wp-content/uploads/2018/02/Decision-Records-16-</a>

February.pdf (Accessed: 26 February 2018).

TVCA (2018c) *Tees Valley Combined Authority Cabinet: Annual general meeting agenda: Stockton, 31 May 2018.* Stockton: Tees Valley Combined Authority, Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2017/05/Combined-Authority-Cabinet-Agenda-Papers-AGM.pdf</u> (Accessed: 24 June 2018).

TVCA (2018d) *Tees Valley Combined Authority Cabinet Decision Record: 15 May 2018*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2018/05/Decision-Records-14-May.pdf</u> (Accessed: 12 June 2018).

TVCA (2018e) *Annual Financial Statements 2017/18*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2018/07/7-App-1-Statement-of-Accounts-2017-18.pdf</u> (Accessed: 17 September 2018).

TVCA (2018f) *Tees Valley in 'Pole Position to be UK's Clean Growth Capital'*. Available at: <u>https://teesvalley-ca.gov.uk/tees-valley-in-pole-position-to-be-uks-clean-growth-capital/</u> (Accessed: 14 December 2018).

TVCA (2018g) *Tees Valley Mayor's Update*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2018/11/3-</u> Agenda-item-5-Mayors-Update-Report.pdf (Accessed: 9 January 2019).

TVCA (2018h) *Governance and Appointments*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2018/11/4-Agenda-Item-6-Governance-and-Appointments.pdf</u> (Accessed: 9 January 2019).

TVCA (2018i) *Budget 2019-20 and Medium Term Financial Plan*. Stockton: Tees Valley Combined Authority [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2018/11/14-Agenda-item-13-Combined-Authority-Budget-Report-</u> 2019-20.pdf (Accessed: 9 January 2019).

TVCA (2018j) *Tees Valley Overview and Scrutiny Committee Minutes: Stockton, 6 December 2018.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/01/APPROVED-Minutes-6-12-18.pdf</u> (Accessed: 6 March 2019).

TVCA (2018k) *Tees Valley Overview and Scrutiny Committee Minutes: 20 December 2018*. Stockton: Tees Valley Combined Authority, Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2019/01/APPROVED-Minutes-201218.pdf</u> (Accessed: 6 March 2019). TVCA (2018I) *Tees Valley Overview and Scrutiny Committee Minutes: Stockton,16 January 2018*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2018/04/OS-Minutes-16th-January-2018.pdf</u> (Accessed: 6 March 2019).

TVCA (2018m) *Tees Valley Overview and Scrutiny Committee Minutes: 15 March 2018*. Stockton: Tees Valley Combined Authority, Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2018/04/4-Minutes-15th-March-2018.pdf</u> (Accessed: 6 March 2019).

TVCA (2019a) *Tees Valley Investment Plan 2019-2029*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

content/uploads/2019/01/Appendix-1-Tees-Valley-Combined-Authority-Investment-Plan-2019-29-1.pdf (Accessed: 2019-01-17).

TVCA (2019b) *Tees Valley Overview and Scrutiny Committee Minutes: Stockton,17 January 2019.* Stockton: Tees Valley Combined Authority, Authority, T.V.C. [Online].

Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/01/APPROVED-</u> <u>Minutes-201218.pdf</u> (Accessed: 6 March 2019).

TVCA (2019c) *Tees Valley Combined Authority Assurance Framework*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/03/15-Appendix-1-Tees-Valley-Assurance-Framework.pdf</u> (Accessed: 10 March 2019).

TVCA (2019d) Scrutiny Study: Diversity in the Tees Valley Combined Authority Membership. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/03/12-Overview-and-Scrutiny-Study.pdf</u> (Accessed: 18 March 2019).

TVCA (2019e) *Tees Valley Combined Authority. Tees Valley Mayor: About.* Available at: <u>https://teesvalley-ca.gov.uk/about/leadership-senior-management-team/tees-valley-combined-authority-cabinet/</u> (Accessed: 13 June 2019).

TVCA (2019f) *Tees Valley Overview and Scrutiny Committee: Stockton, 25 April 2019* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/06/7-OS-annual-report-2018-19.pdf</u> (Accessed: 8 July 2019).

TVCA (2019g) *Tees Valley Overview and Scrutiny Committee Agenda: Stockton, 12 July 2019.* Stockton: Authority, T.V.C. [Online]. Available at: <u>https://teesvalley-</u>

ca.gov.uk/wp-content/uploads/2019/07/Agenda-and-papers-12th-July-2019.pdf (Accessed: 8 July 2019).

TVCA (2019h) 'The story behind our new airport brand', *Tees Valley CA*. Available at: <u>https://twitter.com/TeesValleyCA/status/1154336613994041344?cn=ZmxleGlibGVfcm</u> Vjc18y&refsrc=email (Accessed: 29 July 2019).

TVCA (2019i) 'Passengers set to fly to Spain once again from Durham Tees Valley Airport', *teesvalley-ca.gov.uk*. Available at: <u>https://teesvalley-ca.gov.uk/passengers-</u> <u>set-to-fly-to-spain-once-again-from-durham-tees-valley-airport/</u> (Accessed: 29 July 2019).

TVCA (2019j) *Tees Valley Combined Authority Draft Minutes: 28 June 2019*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-</u> <u>ca.gov.uk/wp-content/uploads/2019/07/3-Draft-Cabinet-Minutes-280619.pdf</u>

(Accessed: 30 July 2019).

TVCA (2019k) *Tees Valley Combined Authority Overview and Scrutiny Committee Draft Minutes: Stockton, 12 July 2019.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/07/Minutes-12-07-19.pdf</u> (Accessed: 7 August 2019).

TVCA (2019I) *Tees Valley Combined Authority Cabinet Draft Minutes: Stockton, 26 July 2019.* Stockton: Tees Valley Combined Authority. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2019/08/Draft-Minutes-frommeeting-on-26-July-2019.pdf (Accessed: 29 August 2019).

TVCA (2019n) *Local Enterprise Partnership*. Available at: <u>https://teesvalley-ca.gov.uk/lep/</u> (Accessed: 10 March 2019).

TVCA (2019o) *Tees Valley Combined Authority Overview and Scrutiny Committee Informal Notes (Fol): Stockton, 1 March 1 2019.* Stockton: Tees Valley Combined Authority (Accessed: 5 November 2019).

TVCA (2019p) *Tees Valley Combined Authority Overview and Scrutiny Committee: Informal Notes (Fol), Stockton 19 July 2019*. Stockton: Tees Valley Combined Authority (Accessed: 5 November 2019). TVCA (2020d) *Tees Valley Overview and Scrutiny Committee Annual Report 2019-20*. Stockton: Tees Valley Combined Authority. [Online]. Available at:

http://www.egenda.stockton.gov.uk/akstvca/users/public/admin/kab14.pl?operation= SUBMIT&meet=33&cmte=SCR&grpid=public&arc=71 (Accessed: 14 June 2020).

TVCA (2020e) *Tees Valley Overview and Scrutiny Committee Minutes, Stockton, 16 January 2020.* Stockton: Tees Valley Combined Authority. [Online]. Available at: <a href="http://www.egenda.stockton.gov.uk/akstvca/users/public/admin/kab71.pl?cmte=SCR">http://www.egenda.stockton.gov.uk/akstvca/users/public/admin/kab71.pl?cmte=SCR</a> (Accessed: 14 June 2020).

TVCA (undated-a) *Ben Houchen - Tees Valley Mayor*. Available at: <u>https://teesvalley-ca.gov.uk/tees-valley-combined-authority/tees-valley-mayor/</u> (Accessed: 23 May 2017).

TVCA (undated-b) *Demographics: ONS estimates: total population: mid 2015*. Available at: <u>https://teesvalley-ca.gov.uk/InstantAtlas/area\_profile/atlas.html</u> (Accessed: 18 June 2017).

TVCA (undated-c) *Tees Valley: Working together for more than 20 years*. Available at: <u>https://teesvalley-ca.gov.uk/about/tees-valley-working-together-for-20-years/</u> (Accessed: 11 December 2018).

TVCA (undated-d) *Local Enterprise Parrtnership*. Available at: <u>https://teesvalley-ca.gov.uk/lep/</u> (Accessed: 9 January ).

TVCA (undated-e) *Key Sectors*. Available at: <u>https://teesvalley-ca.gov.uk/business/key-sectors/</u> (Accessed: 4 March 2019).

TVCA (undated-f) Inward Investment Available at: <u>https://teesvalley-</u>

<u>ca.gov.uk/business/inward-investment/</u> (Accessed: 4 March 2019).

TVCA (undated-g) *Tees Valley Audit and Governance Committee Archive 2017/18*. Available at: <u>https://teesvalley-ca.gov.uk/meetings/archive-papers-and-minutes/audit-governance-committee-archive/2017-18/</u> (Accessed: 6 March 2019).

TVCA (undated-h) *Tees Valley Overview and Scrutiny Committee*. Available at: <u>https://teesvalley-ca.gov.uk/meetings/overview-scrutiny-committee/</u> (Accessed: 6 March 2019).

TVCA (undated-i) *Tees Valley Audit and Governance Committee*. Available at: <u>https://teesvalley-ca.gov.uk/meetings/audit-governance-committee/</u> (Accessed: 6 March 2019).

TVCA (undated-j) *Inspiring Our Future: Addressing long-term unemployment*. Available at: <u>https://teesvalley-ca.gov.uk/projects/education-employment-skills/addressing-long-term-unemployment/</u> (Accessed: 12 March 2019).

TVCA (undated-k) *Inspiring Our Future: Creating a Tees Valley careers and enterprise initiative*. Available at: <u>https://teesvalley-ca.gov.uk/projects/education-employment-skills/creating-a-tv-careers-enterprise-initiative-to-improve-and-extend-high-quality-careers-education/</u> (Accessed: 12 March 2019).

TVCA (undated-l) *Routes to Work DWP Pilot Project*. Available at: <u>https://teesvalley-ca.gov.uk/projects/education-employment-skills/routes-to-work-dwp-pilot-project/</u> (Accessed: 12 March 2019).

TVCA and Mayor, T.V. (2019) *Acquisition and Operation of Durham Tees Valley Airport* Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2019/01/Appendix-2-Acquisition-and-Operation-of-Durham-Tees-Valley-Airport.pdf</u> (Accessed: 18 January 2019).

TVU (2014a) *Tees Valley Unlimited Leadership Board Notes of Meeting: Stockton, 22 January 2014.* Stockton: Tees Valley Unlimited. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-22nd-January-2014.pdf (Accessed: 1 March 2017).

TVU (2014b) Tees Valley Unlimited Leadership Board Notes of Meeting: Stockton,17 February 2014. Stockton: Tees Valley Unlimited. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-17th-February-2014.pdf (Accessed: 1 March 2017).

TVU (2014c) *Tees Valley Unlimited Leadership Board Notes of Meeting: Stockton, 26 March 2014*. Stockton: Tees Valley Unlimited. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-26th-March-2014.pdf</u> (Accessed: 1 March 2017).

TVU (2014d) *Tees Valley Unlimited Leadership Board Notes of Meeting: Stockton, 30 April 2014.* Stockton: Tees Valley Unlimited. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-30th-April-2014.pdf</u> (Accessed: 1 March 2017).

TVU (2014e) *Tees Valley Unlimited Leadership Board, Notes of Meeting: Stockton, 23 July 2014.* Stockton: Tees Valley Unlimited. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-23rd-July-2014.pdf</u> (Accessed: 1 March 2017).

TVU (2014f) *Tees Valley Unlimited Leadership Board Notes of Meeting: Stockton, 22 October 2014*. Stockton: Tees Valley Unlimited. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-Minutes-22nd-October-2014.pdf (Accessed: 1 March 2017).

TVU (2014g) *Tees Valley Strategic Economic Plan*. Stockton: Tees Valley Unlimited. [Online]. Available at:

https://www.lepnetwork.net/modules/downloads/download.php?file\_name=37 (Accessed: 23 April 2017).

TVU (2015a) *Tees Valley Unlimited Leadership Board Notes of Meeting: Stockton, 28 January 2015.* Stockton: Tees Valley Unlimited. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-28th-January-2015.pdf (Accessed: 11 June 2017).

TVU (2015b) *Tees Valley Unlimited Leadership Board, Notes of Meeting: Stockton, 28 October 2015.* Stockton: Tees Valley Unlimited. [Online]. Available at:

https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-minutes-28-Oct-2015.pdf (Accessed: 1 March 2017).

TVU (2016a) *Tees Valley Unlimited Leadership Board Notes of Meeting: 27 January 2016.* Stockton: Tees Valley Unlimited. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-content/uploads/2016/04/Leadership-Board-27th-January-2016.pdf</u> (Accessed: 1 March 2017).

TVU (2016b) *Tees Valley European Structural and Investment Funds Strategy*. Stockton: Tees Valley Unlimited. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2016/10/TV-CA-ESIF-Strategy-Doc-Update-DCLG-April-16.pdf</u> (Accessed: 8 April 2019).

UK2070 (2019b) *Fairer and Stronger: Rebalaning the UK Economy (Full Report).* Sheffield: UK2070 Commission. [Online]. Available at:

https://drive.google.com/file/d/1DaPIONpLXwxS1IE2kLu3aQVkOQEmFLwB/view (Accessed: 6 June 2019).

Unboxed (2019) *Petition Map*. London: UK Parliament, [Online] Available at: <u>https://petitionmap.unboxedconsulting.com/?petition=248566</u> (Accessed: 29 September 2019).

van der Heijden, J. and Kuhlmann, J. (2016) 'Studying Incremental Institutional Change: A systematic and critical meta-analysis of the literature from 2005 to 2015', *Policy Studies Journal*, 45(3), pp. 535-554 [Online] DOI: 10.1111/psj.12191.

Waite, D. and Morgan, K. (2019) 'City Deals in the Polycentric State: The spaces and politics of metrophilia', *European Urban and Regional Studies*, 26(4), pp. 382-399 [Online] DOI: 10.1177/0969776418798678.

Walker, A. (2016) 'Heseltine hails 'incredible transformation' as Tees Valley launches industrial strategy ', *Infrastructure Intelligence*edn), 9 December 2016. [Online] Available at: <u>http://www.infrastructure-intelligence.com/article/dec-2016/heseltine-hails-%E2%80%9Cincredible-transformation%E2%80%9D-tees-valley-launches-industrial (Accessed: 2 March 2017).</u>

<u>13571220?ptnr\_rid=431380&icid=EM\_ChronicleLive\_Nletter\_News</u> (Accessed: 5 September 2017).

Walker, J. (2018a) 'Why the North of Tyne Mayor could mean Higher Council Tax Bills for People', *ChronicleLive*edn), 12 January 2018. [Online] Available at:

http://www.chroniclelive.co.uk/news/north-east-news/north-tyne-mayor-couldmean-14144151 (Accessed: 13 January 2018).

Walker, J. (2018ak) 'Northern Powerhouse Minister: 'There is a big problem with transport in the north'', *ChronicleLive*edn), 7 December 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/northern-powerhouse-minister-there-big-15517114</u> (Accessed: 7 December 2018).

Walker, J. (2018c) 'Theresa May announces £780m to improve East Coast Mail Line rail services', *ChronicleLive*edn), 23 July 2018. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/theresa-announces-780million-improve-

<u>14938677?utm\_source=chronicle\_live\_newsletter&utm\_medium=email&utm\_content</u> =EM\_ChronicleLive\_Nletter\_News\_Home\_largeteaser\_Text\_Story2&utm\_campaign=d aily\_newsletter (Accessed: 23 July 2018).

Walker, J. (2018d) 'Theresa May: 'I'll give the North East the investment, the freedom and the good Brexit it needs'', *ChronicleLive*edn), 23 July 2018. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/theresa-may-ill-give-north-</u>

14938550?utm source=chronicle live newsletter&utm medium=email&utm content =EM ChronicleLive Nletter News Home smallteaser Text Story&&utm campaign=d aily newsletter (Accessed: 23 July 2018).

Walker, J. (2018e) 'Legislation to create North of Tyne Authority and mayor goes to Parliament', *ChronicleLive*edn), 24 July 2018. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/legislation-create-north-tyneauthority-14947979 (Accessed: 25 July 2018).

Walker, J. (2019a) 'North East invited to apply for 'free port' status following claims it could create tens of thousands of jobs', *ChronicleLive*edn), 1 August 2019. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/north-east-invited-apply-free-16684517</u> (Accessed: 2 August 2019).

Walker, J. (2019e) 'Chancellor Philip Hammond provides £260m to boost Borderlands in North England and Scotland', *ChronicleLive*edn), 13 March 2019. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/chancellor-philiphammond-provides-260m-15968900</u> (Accessed: 13 March 2019). Walker, J. (2019f) 'MP joins Boris Johnson on £4,300 helicopter trip from London to Tees Valley', *ChronicleLive*edn), 28 April 2019. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/mp-joins-boris-johnson-4300-16186608 (Accessed: 28 April 2019).

Walker, J. (2019g) 'Power Up the North: Lord Heseltime launches plan to give mayors far more power', *ChronicleLive*edn), 1 July 2019. [Online] Available at:

https://www.chroniclelive.co.uk/news/north-east-news/power-up-north-lordheseltine-16515525 (Accessed: 2 July 2019).

Ward, M. (2017d) *City Deals*. London: House of Commons Library UK, P. [Online]. Available at:

https://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN07158#fullrepo rt (Accessed: 25 December 2017).

Ward, M. (2019a) *Local Growth Deals* (7120). London: House of Commons Library. [Online]. Available at:

https://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN07120 (Accessed: 18 March 2019).

Ward, M. (2019b) *Local Enterprise Partnerships* (05651). London: House of Commons Library. [Online]. Available at:

https://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN05651 (Accessed: 17 April 2019).

Ward, M. and Hardy, S. (eds.) (2013) *Where Next for Local Enterprise Partnerships?* London: The Smith Institute.

Waters, M. (2017h) *Proposed Tees Valley Housing Agreement*. Stockton: Tees Valley Combined Authority. [Online]. Available at: <u>https://teesvalley-ca.gov.uk/wp-</u>

<u>content/uploads/2017/03/Agenda-Item-10-Tees-Valley-Housing-Agreement.pdf</u> (Accessed: 12 June 2017).

Wayman, R. (2018) 'Tees Valley Mayor announced £44m pledge and 'advanced' airport talks', *Bdaily North East*edn), 14 May 2018. [Online] Available at:

https://bdaily.co.uk/articles/2018/05/14/tees-valley-mayor-announces-44m-pledgeand-advanced-airport-

talks?utm\_source=bulletin&utm\_medium=email&utm\_campaign=2018-05-15-northeast&utm\_content=readmore (Accessed: 15 May 2018).

Wayman, R. (2019a) 'Local mayor pitches on future of Teesside International Airport', *Bdaily North East*edn), 30 January 2019. [Online] Available at:

https://bdaily.co.uk/articles/2019/01/30/local-mayor-pitches-on-future-of-multimillion-pound-teesside-international-airport-

plans?utm source=bulletin&utm medium=email&utm campaign=2019-01-31-northeast&utm content=headline (Accessed: 21 January 2019).

Wayman, R. (2019b) '£260m Gateshead Quays arena in progress signs for new venue management', *Bdaily NE*edn), 9 October 2019. [Online] Available at:

https://bdaily.co.uk/articles/2019/10/09/260m-gateshead-quays-arena-andexhibition-in-progress-signs-for-new-venue-

management?utm source=bulletin&utm medium=email&utm campaign=2019-10-10-north-east&utm content=image (Accessed: 10 October 2019).

Webb, D., Raynor, R. and Ormerod, E. (2018) *Collaboration in the North East; Learning from current practice*. Newcastle: Newcastle University (Accessed: 14 November 2018). Welford, G. (2017) 'Big Meeting: the 2017 Durham Miners' Gala in pictures', *Sunderland Echo*edn), 9 July 2017. [Online] Available at:

http://www.sunderlandecho.com/our-region/county-durham/durham/big-meetingthe-2017-durham-miners-gala-in-pictures-1-8640080 (Accessed: 11 November 2017).

Whitfield, G. (2018) '50 Reasons why you should Invest in the North East', *ChronicleLive*edn), 23 March 2018. [Online] Available at:

https://www.chroniclelive.co.uk/business/business-news/50-reasons-you-shouldinvest-14445575 (Accessed: 23 March 2018).

Whitfield, G. (2019) '£600m and 10,000 jobs: the three developments that will change Newcastle city centre', *ChronicleLive*edn), 6 October 2019. [Online] Available at: <u>https://www.chroniclelive.co.uk/news/north-east-news/600m-10000-jobs-three-</u> <u>developments-17033750</u> (Accessed: 10 October 2019).

Wollmann, H. (2008) 'Reforming Local Leadership and Local Democracy: The cases of England, Sweden, Germany and France in comparative perspective', *Local Government Studies*, 34(2), pp. 279-298 [Online] DOI: 10.1080/03003930701852344.

Woods, P. (2019) *North East Local Enterprise Partnership Annual General Meeting*. Newcastle, 26 September 2019. North East Local Enterprise Partnership, (Accessed: 26 September 2019).

World Bank (2009) *World Development Report 2009: Reshaping economic geography*. Washington DC: World Bank World Bank Group. [Online]. Available at:

https://openknowledge.worldbank.org/handle/10986/5991 (Accessed: 10 February 2019).

Wright, T. (2004) 'Prospects for Parliamentary Reform', *Parliamentary Affairs*, 57(4), pp. 867-876 [Online] DOI: 10.1093/pa/gsh067.

Yin, R.K. (2003) *Case Study Research: Design and methods*. Thousand Oaks, CA.: Sage Publications.

Young, I.M. (2000) *Inclusion and Democracy*. Oxford: Oxford University Press.